The Influence of the Firm's Environmental Measures Upon Its Export Behaviour & the Level of Exporting: An Empirical Investigation

BY

HANI HAMED AL - DMOUR

BA University of Jordan, 1983 MBA University of Edinburgh, 1986

A Thesis Submitted to the University of Sheffield in Fulfilment of the Requirement for the Degree of Doctor of Philosophy

School of Management and Economic Studies

University of Sheffield

July, 1992

DEDICATION

The Work and efforts extended on this study is dedicated to my parents, my wife and daughter Rand and my son Ahmad

ACKNOWLEDGMENTS

I would first like to express my deep thanks to God (Allah) for giving me the opportunity and the ability to complete this thesis.

My sincere thank and appreciation go to my supervisors Professor Arthur Meidan, and Dr. Kirk - Smith for the effective guidance, advice and constructive criticism during the preparation of this thesis. It is true that without their directions, counselling, and friendship, the completion of the thesis would have been impossible.

My thanks are due to the Chairman of the School of Management and Economic Studies and the member of staff of the School of Management for their friendship, encouragement, and the provision of research facilities within the school.

I am very grateful for the helpful comments by Professor R. M. Loynes of the Department of Statistics and Probability, and the other staff members on the statistical techniques and their valuable comments in data analysis.

I wish to thank Mr. R. K. Richard for the computing Centre, University of Sheffield for his assistance and advice in the computing work. Thanks are also due to Mr. David Perrow of the Crooksmoor Library of Sheffield University and the rest of the libraries for their assistance in facilitating the effective use of the Library for the research purposes.

My gratitude also is due to the British Council and the University of Jordan for the sponsorship. My sincere thank is to the Minister of Industry and Trade, Dr. Zeyad Friez, the Chairman of the Amman Chamber of Industry, Mr kaludon Abu-Hassan, the Adviser of the Chamber of Industry, Mr. Ali Al-Dajni, the Chairman of the Jordan Export Association, Mr. Haleem Abu-Rahmeh for their supports and assistances during the data collection.

I want also to thank all the domain of the study population who were responded favourably to the interview.

I am deeply grateful to my wife and my daughter Rand and my son Ahmad for their moral support, patience, and encouragement during the time this work was undertaken.

Hani A1 - Dmour School of Management University of Sheffield

(iii)

The Influence of the Firm's Environmental Measures Upon Its Export Behaviour & the Level of Exporting: An Empirical Investigation

ABSTRACT

This thesis attempts to investigate the influence of the firm's internal and external environmental measures upon its export behaviour (i.e., the variation between exporters and non - exporters) and the level of exporting. In order to achieve the study's objectives, and to conduct this research in a systematic manner, a conceptual framework was developed. The conceptual framework tied together the key factors which are thought to have influenced the firm's export behaviour and the level of exporting, i.e., the internal and the external environmental dimensions.

The design of the research is guided by a number of hypotheses about the relationship between the firm's level of exporting and its internal and external environmental measures on the one hand, and about the discrimination between exporters and non - exporters according to these environmental measures on the other hand.

The required data are collected through structured - directed interviews with 387 respondents. The survey units were the manufacturing firms in Jordan, and the key single respondents approach was employed. The data are analysed by the application of a variety of statistical tests and techniques: (1) Factor analysis (Principle Components analysis); (2) Multiple regression analysis; (3) Discriminant function analysis; (4) Correlation analysis; (5) T test; (6) the Univariate F - ratio; (7) Chi - square test, and the (9) McNemar test.

This thesis is organised in 10 chapters; the first chapter discussed the research problem, objectives, and hypotheses; chapter two introduced the relevant literature review; chapter three presented the preliminary interviews; chapter four is devoted to the study's conceptual framework; chapter five and chapter six are intended to explain the research design and the research methodology; chapters seven, eight, and nine are concerned the research findings, and finally chapter ten is the conclusions.

Overall, the findings of the study support the following conclusions:

- The firm's internal environmental measures are related not only to export behaviour (i.e., the difference between exporters and non - exporters), but also to the level of exporting.
- (2) The firm's external environmental measures are also related to export behaviour and the level of exporting.
- (3) In terms of the predictive power (i.e., explanation) of the level of exporting, the firm's external environmental measures are found to be higher than its internal environmental measures.
 - (4) In respect of the predictive power of the classification of group membership (i.e., exporters vs. non - exporters), the firm's internal environmental measures are shown to be higher than its counterpart, the external environmental measures.
 - (5) The factors determining the firm's export behaviour are found to be somewhat different from the factors determining its level of exporting.
 - (6) In comparison to each environmental dimension acting alone, the integration approach of the two internal and external dimensions gives better explanation not only of the prediction of the level of exporting but also of the prediction of export behaviour. Therefore, a better understanding of export behaviour and the level of exporting requires that the firm's environmental measures to be viewed as a whole (i.e., the interaction of the internal and external dimensions) rather than being isolated fragments (i.e., only a single dimension).
 - (7) The study's findings support many of the findings of previous studies. However, some conflicting results are also reported. For example, the age of the decision - maker is not found to be an important determinant either of the level of exporting or export behaviour. These differences raise new explanation when variables are, or are not, important to export behaviour or the level of exporting.

And on the basis of the findings, the study is concluded with a number of interesting applications for the current and potential exporters, and for the public policy - makers in Jordan.

CONTENTS

	PAGE NO
The second second second as a second s	
Acknowledgment	11
Abstract	111
List of Contents	V
List of Figures	X11
List of Table	X111
CHAPTER ONE: INTRODUCTION	
and the second of the second time of specific times	2
Preface	2
1.1 The Importance of International Trade	4
for Developing Countries	
1.2 Export Expansion Versus Import	6
Substitution as Industrialization	
Polices for Developing Countries	
1.3 Statement of the Problem	8
1.4 Research Objectives	11
1.5 Research Hypotheses	14
1.6 The Significance of the Study	16
1.7 Some Background Information about Jordan	18
1.7 Dome DaekBroand Information about Soldan	
1.7.1 General Economic Features	18
1.7.2 Problems of the Jordanian Economy	19
1.7.3 External Trade Features	21
1.7.4 Jordan's Export Patterns and Structure	23
1.7.5 Characteristics of the Manufacturing Sectors	23
1.8 The Organisation of This Thesis	25
CHAPTER TWO: REVIEW OF LITERATURE	
Introduction	29
Includeetion	23
2.1 International Trade Theories	30
2.1.1 The Traditional Theory of International Trade	31
2.1.2 The Factor Abundant Theory	33
2.1.3 The Product Life Cycle theory	35
2.1.4 Limitations of the International Trade Theories	38
2.1.4 Dimitations of the international fract modifies	50
2.2 Firm and Organisational Behavioural Theories	41
2.2.1 The Theory of the Firm	41
2.2.1 The Organisation Behavioural Theories	42
2.3 Export Behaviour Studies	45
2.3.1 Studies Relating to the Firm's Internal Environment	46
(vi)	

2.3.1.1 The Initiation of Exporting 2.3.1.2 The Firm's Management Export Orientation	46 52
 2.3.1.2.1 Motivation for Export Marketing 2.3.1.2.2 Firm's Management & Export Marketing 2.3.1.2.3 Perceived Obstacles to Export Marketing 2.3.1.2.4 Export Marketing Mix Decisions 2.3.1.2.5 Firm's Demographic and Decision - Maker's Individual Characteristics 	52 59 64 69 76
2.3.2 Studies Relating to the Firm's External Environment	84
2.3.2.1 Economic and Commercial Infrastructural Environment 2.3.2.2 Political - legal Environment	88 91
2.3.3 Export Modelling Studies	95
2.4 The Common Characteristics of Export Behaviour Studies	112
Summary	113
CHAPTER THREE: THE PRELIMINARY IN DEPTH INTERVIEWS AND SOME INFORMATION ABOUT EXPORT POLICIES, PROCEDURES AND EXPORT PROMOTION INSTITUTIONS IN JORDAN	
Introduction	117
 3.1 Objectives of the Preliminary Interviews 3.2 Structure of the Preliminary Interviews 3.3 Planning for the Preliminary Interviews 3.4 Data Gathered Through the Preliminary Interviews 	118 118 119 119
3.4.1 The Role and Functions of the Export	119
3.4.2 The Export Regulations and Procedures 3.5.3 The Results of the Initial Interviews	127 130
Summary	136
CHAPTER FOUR: THE CONCEPTUAL FRAMEWORK FOR THE	
INTERNAL AND EXTERNAL DETERMINANTS OF EXPORT BEHAVIOUR	
Introduction	138
 4.1 The Nature of the Conceptual Framework 4.2 How It Works 4.3 Constructs of the Conceptual Framework 	138 139 141
4.3.1 The Firm's Export Behaviour and the Level of Exporting	141

(vii)

4.	3.2	The Co Enviro	nstructs of the Firm's Internal nmental Dimension	143
	4.3.	2.1 M	Janagerial Goals and Aspirations Level	144
	4.3.	2.2 M	anagement's Expectations	146
	4.3.	2.3 T	he Firm's Comparative Advantages	149
	4.3	2.4 T	The Firm's Commitment to Export	152
	4.5.	2.4 I	arketing Activity	132
	4.3.	2.5 T	The Socio - Demographic Characteristics	154
	4.0.		of Decisions - Makers	245
			bit Date Indit Chrant	
4	3.3	The Co	onstructs of the Firm's External	156
		Enviro	onmental Dimension	
		Statisat	wal without bard for Youling	
	4.3	.3.1	The Economic and Commercial	156
			Infrastructural Environment	
	4.3	.3.2	The Political - Legal Environment	159
		United	ista i lett	
Summa	ary			163
CH	APTER	FIVE:	RESEARCH DESIGN AND DATA COLLECTION	
Intro	oducti	on		166
5.1	Eval	uating	the Alternative	166
	Data	Collec	tion Methods	
5.2	Deci	ding or	the Most Appropriate	170
	Type	e of Que	estioning Methods	
5.3	Deci	ding or	n the Appropriate	172
	Stru	icture d	of the Interview	
5.4	Deci	ding or	the Domain of Respondents	173
5.5	Deci	ding or	1 the Appropriate	175
	Key	Respond	lent Approach	
5.6	Deci	ding or	the Appropriate	176
	Inst	rument	of Measurement	
5.7	The	Ouestic	onnaire Development Process	178
5.8	The	Intervi	iew Schedule and	182
	Rest	onse Ra	ate and the lost of the lost	2179
5.9	Prei	paring f	for Data Analysis	185
			ie Sate Factore Skat Senter He Skat - The	1000
Summa	arv			187
o crunin				
CI	APTER	R SIX:	RESEARCH METHODOLOGY	
Intro	oducti	ion		190
6.1	Clas	sificat	tion of Statistical Techniques	190
6.2	The	Statist	tical Methods Used for Research Objectives	192
6	.2.1	Factor	c Analysis	192
			trastrastanes anytransan	
	6.3	2.1.1	Methods of Extracting Initial Factor.	193
	6.3	2.1.2	Factor Analysis Input / Output	194
	6.3	2.1.3	Use fo the Factor Analysis in the Study	195

6.2.2	Multiple Regression Analysis	196
6.2.	Caution in the Use of Multiple RegressionUse of Regression Analysis in this Study	198 200
6.2.3	Simple Correlation Coefficient	200
6.2.4	Discriminant Function Analysis	202
6.2. 6.2. 6.2.	 4.1 The Applications of DFA 4.2 DFA Date Input / Output 4.3 Use of DFA in this Study 	202 202 205
6.3 The S Resea	tatistical Methods Used for Testing rch Hypotheses	205
6.3.1 6.3.2 6.3.3 6.3.4	F - Test Univariate F - Test Chi - Square Test McNemar Test	206 207 207 207
Summary		211
CHAPTER	SEVEN: THE FACTOR ANALYSIS FINDINGS	
Introduction	n	213
7.1 The F: 7.2 The In	indings of the Factor Analysis nterpretation of the Final Factor Analysis	213 214
7.2.1	The Constructs of the Firm's Internal Dimension	214
7.2.1.	1 The Main Factors That Underlie the	215
7.2.1.	2 The Main Factors That Underlie the	216
7.2.1.	3 The Main Factors That Underlie the	219
7.2.1.	4 The Main Factors That Underlie the Firm's Commitment to Export Activity	221
7.2.1.	5 The Main Factors That Underlie the Characteristics of the Decision - Makers	223
7.2.2	The Constructs of the Firm's External Dimension	225
7.2.2.	1 The Main Factors That Underlie the Firm's Legal - Political Environment	225
7.2.2.	2 The Main Factors That Underlie the Firm's Economic & Commercial Infrastructural Environment	226
	all provide toring transform but	

Summary

CHAPTER EIGHT: THE DETERMINANTS OF THE FIRM'S LEVEL OF EXPORTING

Intro	duction	236
8.1	The Bivariate Correlation: Factor Findings	237
8.2	The Interpretation of the Correlation Factor Findings	237
8.	2.1 Internal Environmental Dimension: Factor Correlations	237
8.	2.2 External Environmental Dimension: Factor Correlations	242
83	The Stepwise Multiple Regression: Dimension Findings	244
8.4	The Interpretation of the Stepwise Regression Findings	245
8.	4.1 Stepwise Multiple Regression: Internal Dimension	245
8.	4.2 Stepwise Multiple Regression: External Dimension	246
8.	4.3 Stepwise Multiple Regression: Combination	251
8.5	The Bivariate Correlation: Variables Findings	255
8.6	The Interpretation of the Correlation Variable Findings	255
Summa	ry & Conclusions	279
	APPENDIX (21) a. Deperionnative threelond Version)	
С	HAPTER NINE: THE DETERMINANTS OF EXPORT BEHAVIOUR: EXPORTERS VS. NON - EXPORTERS	
Intro	duction	283
Incro	and the legal to all the distance and the	
9.1	Testing the Research Hypotheses	284
9.1	.1 Testing the Discriminant Function /	284
	Factors: Chi - Square & Univariate F Ratio	
9.1	.2 Testing the Improvement in Predicting	289
	Group Membership; the McNemar Test	
9.2	Identification and Prediction of Export Behaviour:	291
	Interpretation of DFA	1462
9.2	.1 Comparative Analysis Between the Internal and	291
	External Dimensions in Respect of Exporting	
9.2	.2 An Integration Approach to Identify and to	294
9.2	.3 Validation of the Prediction of Export Behaviour	298
	(Exporter's and Non - Exporter's Group Membership)	
9.2	.4 The Relative Important of the Predictor Factors in Terms of Their Contributions	301
9.3	The Differences Between Exporters and	309
	Non - Exporters: Variable Findings	
9.4	The Interpretation of the Variable Findings	309
Summa	ry and Conclusions	333
	(x)	

CHAPTER TEN: SUMMARY, CONCLUSIONS AND RECOMMENDATIONS

10.1 Summary	337
10.2 Main Conclusions of the Research Findings 10.3 Research Contributions	339 350
10.3.1 Theoretical Contribution 10.3.2 Practical Contribution	350 351
10.4 Recommendations	353
10.4.1 Recommendations For Current and Potential Exporters 10.4.2 Recommendations For Public Policy - Makers	353 357
10.4 Research Limitations	365
10.5 Areas for Further Research	365
REFERENCES	367
APPENDIXES	401
APPENDIX [1]: Reliability	402
APPENDIX [2]: a. Questionnaire (English Version) b. Questionnaire (Arabic Version)	411 419
APPENDIX [3]: Comparison Between the Results of the Determinants of the Level of Exporting and the Results of the Determinants of Export Behaviour	431
APPENDIX [4]: Recommendation Letters	436
 The Student's Supervisor The Minister of Industry and Trade 	437 438
3. The Chairman of the Chamber of Industry	439
4. The Chairman of Jordan Trade Association	440
5. The Student's Sponsorship	441

List of Figures

Figure	1.1	Organisation of the Thesis	27
Figure	2.1	Weidersheim & Welch's Model	93
1.0		Factors Affecting the Pre - Export Activities	
Diama	2.2	Di die Films Beidie Export Behavioural Model	95
Figure	2.2	Reid's Export Benavioural Model	101
Figure	2.3	of a Firm's Export Behaviour	101
Figure	2.4	Kolhede's Export Orientation Model	103
Figure	2.5	Aaby & Slater's A General Model	109
Tables		For Assessing Export Performance	
		and Variables	
Figure	2.6	Yang, Leone and Alden's Model	111
Figure	3.1	The Government Export Procedures	129A
Figure	4.1	The Conceptual Framework of the Influence	140
		of the Interaction of the Firm's Internal	
		and External Environment Upon its Export	
		Behaviour and the Level of Exporting	
Figure	5.1	Research Design Process	167
Figure	5.2	The Process of Selecting the	169
		Appropriate the Appropriate Data	
		Collection Method	
Figure	5.3	Deciding on the Appropriate	171
1 10010	6.6	Communication Methods	
Figure	5.4	Questionnaire Development Process	179
Figure	9.1	All Groups Stacked Histogram (Run * 1)	294
Figure	9.2	All Groups Stacked Histogram (Run * 2)	296
Figure	9.3	All Groups Stacked Histogram (Run * 3)	296

List of Tables

Table	1.1	Some Basic Indications about Jordan's Economy	20
Table	1.2	Some Basic Indicators about the External Trade	22
Table	1.3	Composition of Export by Sector of Origin	24
Table	2.1	Export Initiation Process Studies	48
Table	2.2	Motivations For Export Marketing	54
Table	2.3	The Firm's Comparative Advantage(s)	60
Table	2.4	Management & Organisational Structure Characteristics	61
Table	2.5	Barriers to Export Marketing	65
Table	2.6	Export Marketing Mix Studies	71
Table	2.7	The Decision - Maker's Characteristics	77
Table	2.8	The Firm's Demographic Characteristics	81
Table	3 1	The Objectives of Each Stage of the	120
Table	J.1	Preliminary Interview	120
Table.	4. 0	The Flowents of Menseementle Expectations	147
Table	4.2	The Elements of Management's Expectations	147
Table	4.5	The Elements of the Firm's comparative Advantages	150
Table	4.4	The Elements of the Firm's commitment to Export	155
		Marketing Activity	1.5.5
Table	4.5	The Socio - Demographic Characteristics	155
	2.22	of Decision - Makers	100
Table	4.6	The Elements of the Economics Determinants	158
Table	4.7	The Elements of the Political - Legal Env.	161
Table	4.8	The Study's Conceptual Framework: Summary	164
Table	5.1	The Domain of the Study's Respondents	175
		According to the Type of Industrial Sector	
Table	5.2	The Distribution of the Respondents' Firms	183
		in Terms of the Type of Industrial Sector	
Table	5.3	Chi - Square Analysis of the Study's Respondents	184
		In Terms of the Industrial Sector	
Table	6.1	Fourfold Table for Use in Testing Significance	208
		of Change	
Table	6.2	The Research Hypotheses and Their	209
10010		Relevant Statistical Tests	200
Table	7 1	The Number of Factors that Underlie Fach	214
lable	1.1	Construct of the Internal and External Dimension	214
Table	7 0	The Main Fostern Underhair and External Dimension	015
lable	1.2	The Main Factors Underlying the construct	215
m.1.1	7 0	of the Managerial Aspiration Level	017
Table	1.3	The Main Factors Underlying the Construct	217
		of Management's Expectations	
Table	7.4	The Main Factors Underlying the Construct	220
		of the Firm's Comparative Advantages	
Table	7.5	The Main Factors Underlying the Construct	222
		of the Firm's Commitments to Export activity	
Table	7.6	The Main Factors Underlying the	224
		Decision - Maker's Characteristics	
Table	7.7	The Main Factors Underlying the Construct	226
		of the Political - Legal Environment	10.6
Table	7.8	The Main Factors Underlying the Construct	229
		of the Economic & Commercial Environment	
Table	7.9	Summary of the Factors Underlie the Major	222
		Constructs of the Firm's Internal Environment	200
Table	7,10	Summary of the Factors Underlie the Major	001
10010		Constructs of the Firm's External Environment	234
		SAME AND AN AND ANAL AND DALGINGI LIVIIUMPIL	

Table	8.1	The Correlation Coefficients Between Each Factor and the Level of Exporting	238
Table	8.2	A Summary Results of Stepwise Multiples Regression	244
Table	8.3	The Stepwise Regression Analysis: Internal Dimension	244
Table	8 4	The Stepwise Regression Analysis, Internal Dimension	241
Table	8 5	The Stepuice Regression Analysis, External Dimension	245
Table	0.5	The Correlation Coofficients Detrees the Venichles	252
Table	8.0	The correlation coefficients between the variables	256
		comprising Each Independent Factor	
		(Taken Separately) and the Level of Exporting	
Table	8.7	The Correlation Coefficient Between the Variables	257
		of the "Managerial Aspiration" Factor and	
		the Level of Exporting	
Table	8.8	The Correlation Coefficient Between the Variables	258
		of the "Operational Obstacles" Factor and the Level	
		of Exporting	
Table	8.9	The Correlation Coefficient Between the Variables	259
10010		of the "Perceived Risk" Factor and the Level of	
		Fynorting	
Table	0 10	The Correlation Coefficient Detween the Verichles	260
Table	0.10	of the "Broot Objectional" Best on the Variables	200
		of the "Export Objectives" Fact or and the Level	
		of Exporting	
Table	8.11	The Correlation Coefficient Between the Variables	261
		of the "Export Benefits" Factor and the Level of	
		Exporting	
Table	8.12	The Correlation Coefficient Between the Variables	262
		of the "Export Barriers" Factor and the Level of	
		Exporting	
Table	8.13	The Correlation Coefficient Between the Variables	262
		of the "Competitive Situation" Factor and the	
		Level of Exporting	
Table	8 14	The Correlation Coefficient Detween the Newighles	263
Table	0.14	of the "Mayket Competition Character" Feeter and	205
		the Level of Burgeticity Strengths Factor and	
	0.15	the Level of Exporting	~ ~ ~
Table	8.15	The Correlation Coefficient Between the Variables	264
		of the "Size of the Firm" Factor and the Level of	
		Exporting	
Table	8.16	The Correlation Coefficient Between the Variables	265
		of the "Firm's Technological Strengths" Factor and	
		the Level of Exporting	
Table	8.17	The Correlation Coefficient Between the Variables	266
YAB1m		of the "Business Experience" Factor and the Level	199
		of Exporting	
Table	8 18	The Correlation Coefficient Between the Variables	266
Table	0.10	of the "Droduct Characteristics" Factor and the	200
		Level of Executive	
1.350.2.0		Level of Exporting	
Table	8.19	The Correlation Coefficient Between the Variables	267
		of the "Commitment to Export Related Activities"	
		Factor and the Level of Exporting	
Table	8.20	The Correlation Coefficient Between the Variables	268
		of the "Willingness to Change" Factor and the	
		Level of Exporting	
Table	8.21	The Correlation Coefficient Between the Variables	260
	10.34	of the "Education" Factor and the Level	209
		of Pypayting	

Table	8.22	The Correlation Coefficient Between the Variables of the "Demographic" Factor and the Level of Exporting	270
Table	8.23	The Correlation Coefficient Between the Variables of the "Government's Export Policies and	271
Table	8.24	Assistance" Factor and the Level of Exporting The Correlation Coefficient Between the Variables of the "Awareness of Government's Export Policies" Factor and the Level of Exporting	272
Table	8.25	The Correlation Coefficient Between the Variables of the "Government's Export Facilities Related Problems" Factor and the Level of Exporting	272
Table	8.26	The Correlation Coefficient Between the Variables of the "Government's Export Procedures Related Problems" Factor and the Level of Exporting	274
Table	8.27	The Correlation Coefficient Between the Variables of the "Foreign Market Conditions" Factor and the Level of Exporting	274
Table	8.28	The Correlation Coefficient Between the Variables of the "Commercial Facilities Related Problems"	275
Table	8.29	The Correlation Coefficient Between the Variables of the "Raw Material Related Problems" Factor and the Level of Exporting	276
Table	8.30	The Correlation Coefficient Between the Variables of the "Export Costs" Factor and the Level of	277
Table	8.31	The Correlation Coefficient Between the Variables of the "Physical Distribution" Factor and the Level of Exporting	278
Table	8.32	The Correlation Coefficient Between the Variables of the "Domestic Market Conditions" Factor and the Level of Exporting	278
Table	8.33	A Summary Review of the Correlation Coefficients Between the Variables Constituting Each Independent Factor (Taken Separately and Together) and the Level of Exporting	280
Table	9.1	Univariate F Value and Its Significant Level in the DFA of the Internal Environmental Dimension: Taken Separately	287
Table	9.2	Univariate F Value and Its Significant Level in the DFA of the External Environmental Dimension: Taken Separately	288
Table	9.3	Fourfold Table of McNemar Test	290
Table	9.4	Comparative Results of Two Separate Discriminant Analysis Function	292
Table	9.5	Canonical Discriminant Function and Classification Matrix (Run * 3)	295
Table	9.6	Validation of the Discriminant Functions:	299
	9.28	Comparison of the Analysis and validation Sample	
Table	9.7	A Comparison Between the Hit Ratios of the DFA and the Jacknife Method	301
Table	9.8	Group Means and Standardized Discriminant	303

Table	9.9	A Summary Results of the T - Test of the Differences Between Exporters and Non - Exporters in Terms of Variables Which Constitute Each Independent Factor, Taken Separately	310
Tab1e	9.10	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Managerial Aspiration" Factor	311
Table	9.11	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Operational Obstacles" Factor	312
Table	9.12	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Perceived Risk" Factor	313
Table	9.13	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Export Objectives" Factor	313
Table	9.14	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Export Benefits" Factor	314
Tab1e	9.15	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Export Barriers" Factor	315
Table	9.16	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Competitive Situation" Factor	316
Table	9.17	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Market Competitive" Factor	317
Table	9.18	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Size of the Firm" Factor	317
Table	9.19	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Technological Strengths" Factor	318
Table	9.20	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Business Experience" Factor	319
Table	9.21	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Product Characteristics" Factor	320
Table	9,22	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Willingness to Change" Factor	320
Table	9.23	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Commitment to Export Belated Activities" Factor	321
Table	9.24	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Demographic" Factor	322
Table	9.25	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Education" Factor	323
Table	9.26	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Comprise the "Government's Export Assistance & Policies" Factor	324

Table	9.27	T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the	325
		"Awareness of the Government Export Assistance	
Table	9 28	T - Test Between Fynorters and Non - Fynorters	326
Table	1.20	in Terms of Variables Which Comprise the	520
		"Government's Export Facilities" Factor	
Table	9.29	T - Test Between Exporters and Non - Exporters	327
10010		in Terms of Variables Which Comprise the	
		"Government Export Procedures" Factor	
Table	9.30	T - Test Between Exporters and Non - Exporters	328
100000		in Terms of Variables Which Constitute the	
		"Foreign Markets Conditions" Factor	
Table	9.31	T - Test Between Exporters and Non - Exporters	329
		in Terms of Variables Which Constitute the	
		"Commercial Facilities Related Problems" Factor	
Table	9.32	T - Test Between Exporters and Non - Exporters	330
		in Terms of Variables Which Constitute the	
		"Raw Material Related Problems" Factor	
Table	9.33	T - Test Between Exporters and Non - Exporters	330
		in Terms of Variables Which Constitute the	
		"Export Costs" Factor	
Table	9.34	T - Test Between Exporters and Non - Exporters	331
		in Terms of Variables Which Constitute the	
		"Physical Distribution Related Problems" Factor	
Table	9.35	T - Test Between Exporters and Non - Exporters	332
		in Terms of Variables Which Constitute the	
		"Domestic Market Conditions" Factor	
Table	9.36	Profile Analysis For The Two Groups Means	334
		in Terms of the Components of the Internal and	
	S-Salarin	External Environmental Dimension	
Table	10.1	Recommendations and Suggested Actions	355
	Sec. 3	For Exporters and Potential Exporters	
Table	10.2	Recommendations and Suggested Actions	359
		For Government to Improve and Encourage	
		Export Involvement	

CHAPTER ONE

INTRODUCTION

Preface

- 1.1 The Importance of International Trade for Developing Countries
- 1.2 Export Expansion Versus Import Substitution as Industrialization Policies for Developing Countries

1.3 Statement of the Problem

- 1.4 Research Objectives
- 1.5 Research Hypotheses
- 1.6 The Significance of the Study
- 1.7 Some Background Information about Jordan
 - 1.7.1 General Economic Features
- 1.7.2 Problems of the Jordanian Economy
 - 1.7.3 External Trade Features
 - 1.7.4 Jordan's Export Patterns and Structure
 - 1.7.5 Characteristics of the Manufacturing Sectors
 - 1.8 The Organisation of This Thesis

CHAPTER ONE

INTRODUCTION

Preface

".. The economic well - being of most countries, especially the smaller ones, is affected by the international trade position. Failure to generate a sufficient volume of exportable goods and services has been a recurring source of economic crisis in many developing countries as well as in the developed ones. Experience suggests that when a country suffers from a chronic trade deficit, the remedy can not be left to Adam Smith's invisible hand. Responsibility for corrective measures must rest squarely on the shoulders of the government. If exports are to be increased with the help of conscious policies undertaken by the government or by the exporters themselves, more is needed to be known about the factors which determine exports and about how these factors interact" (Hirsch 1971).

The theme of this study is based upon the fact that the firm's involvement in export marketing operations is very important for two major reasons. First, for industrial firms, exporting often offers an attractive alternative strategy for growth and survival in an increasingly international market place - typically, it is an outlet for exploiting its comparative advantages and overcoming certain unfavourable conditions in the home market. Second, for a country, exports are known to be essential for the economic well - being of the country as well as a critical factor in the economic development process. Exports not only increase the demand for domestic production but are a major source of foreign exchange to finance the capital goods that are imported. In the long run the balance of payments have to be maintained if the nations' wealth is to improve. A better understanding of the underlying determinants of the firm's export behaviour and the level of exporting is, therefore, critical.

This study will mainly focus on those determinants (variables) which are behind the firm's export behaviour^{*} and its level of exporting. The firm's export behaviour is hypothesized to be a function of the interaction of two major groups of variables (dimensions):

(1) Variables related mainly to the firm's internal environment including its organisational and managerial characteristics.

(2) Variables related mainly to the firm's external environment including characteristics of the local and foreign environment where the firm already or expects to operate.

The aims of this study are to provide a comprehensive understanding of how these two groups of variables interact and influence the export behaviour of manufacturing firms in developing countries such as Jordan, and also, to suggest solutions to the main export marketing problems that are encountered by them.

Export behaviour is an expression used here as an indicator for the differences between exporters and non - exporters.

1.1 The Importance of International Trade for Developing Countries

"Even though international trade can not be expected to be engine of economic growth today, there are many ways by which it can contribute to the economic development of today's developing nations" (Keesing 1979).

Hibbert (1989) pointed out the following important contributions of international trade for the developing nations which can lead to the full utilization of underemployed local natural resources. Through trade, a developing country can move from an inefficient production point inside its production frontier, with unutilized resources because of inadequate internal demand to a point on its production frontier with trade. More generally, it allows a developing economy to specialize upon its relatively plentiful resources, while importing products that would be uneconomical, or impossible to produce domestically. Therefore, through international trade the country can probably obtain more of each type of product than it could produce for itself.

According to international trade theory, benefit from trade can also be obtained through economic scale, i.e., an advantage from large scale production. The extent of the division of labour and economics of scale are limited to the size of the market. Therefore, a developing country will be weakened by the high unit cost if it tries to produce a little of each product for its inadequate domestic market. Such a country stands to obtain much by importing many of the products that can be produced cheaply only on a large scale, and by building its manufacturing industries partly around exports, so that a larger scale of production can be achieved.

Furthermore, international trade is the vehicle for the transmission of new ideas, new technology and new managerial and other skills. Indeed, trade goes hand in hand with travel by businessmen and

technical specialists, through which the flow of information and technology is transferred across international boundaries. In addition to which it is an excellent anti-monopoly weapon because (when allowed to operate) it stimulates greater efficiency by domestic producers to meet foreign competition (Keesing 1979).

Learning to export and even trying to export products is valuable partly for this reason. Competition from imports, too, can probably motivate improved performance, because, in developing countries many manufacturers consist of only one or a very few enterprises, as a reflection of the small size of the domestic market. Unless subjected to foreign competition, these firms are more likely to conspire than compete and they will tend to raise prices, neglect quality and perform insufficiently in terms of technology. Import competition helps to eliminate inadequate local firms while motivating others to greater effort and higher performance standards (Hibbert 1989).

However it is worth mentioning that even though international trade is worthwhile for all these reasons, it does not mean that more trade is required in all cases. Trade policy must not only follow these advantages but must also balance them against the gains made from learning to produce goods and services that could be imported.

1.2 Export Expansion Versus Import Substitution as Industrialization Policies for Developing Countries

Since 1970, most developing countries have made a deliberate attempt to industrialize rather than continue to specialize in the production of primary commodities (food, raw material and minerals) prescribed by traditional trade for export, as theory. Industrialization was depended upon to provide; faster technological progress; the creation of jobs to overcome the serious unemployment problems faced by most developing countries; rising terms of trade and more stable export prices and earning, and relief from balance of payment difficulties (Keesing 1979).

Therefore, for development and industrialization, a developing country has to choose between import substitution, or an export oriented strategy. In order to pursue either policy within a certain developing country, an evaluation must be made in the light of the advantages and disadvantages with regard to the country's situation.

Development and industrialization through import substitution includes the following advantages: it can save foreign exchange which would be used to pay for necessary imports (Hibbert 1989), it could be easier for developing countries to protect their domestic market against foreign barriers and against foreign competition than to force developed nations to lower trade barriers against their manufactured export products.

On the one hand, substitution policy leads to insufficient industries because the domestic markets in many developing countries such as Jordan are small. It does not allow them to take advantage of the economic of scale. Furthermore, effective protection from foreign competition and the absence of domestic challenge may enable local firms to increase profit and sales mainly by increasing production

which is often obtained at the expense of quality services and consumer choice (Hibbert 1989). Finally, after the simpler manufactured imports are replaced by the domestic production, import substitution becomes more capital - intensive and, technologically advanced imports are then replaced by the domestic production.

On the other hand, industrialization and development through export expansion for Jordan as a developing country, will overcome the smallness of the domestic market and will allow benefit from economies of scale. Furthermore, production of manufactured goods for export require and stimulate efficiency throughout the economy.

(a) States at wittenades have dive a fidere to back of the second entrates of strengthings transmit transmit a or second which the second entrates (b) States are entransmitted at 11 (SVI) and at an entrates theories is sets ground to resident excelete the store of because anothings for sectors is entransmitted by the trained to solve to be anothings for sectors is entransmitted by the trained to solve the states is entrany is entransmitted by the store of the solve of the store is stored by antised by a store of the solve of the solve to be solved by the solve of the solve of the store of the solve of the store of the store is solved by a store of the solve of the solve

Jones's scores position is sufficing deeply from a decise derive is the balance of trade. In 1987 and 1988, this are closed to 01.1 without to 372 of its proper settional granters (1987), despise the higher contribution of the enternal trade, by given downstile predict (cor) which reacted 143,12 is 1987. This growth of the enternal trade and attribute more to importate them to support from here detail story is a model of because is economy take settings 1.7.3 do this Chapter). As a model of ingenting the model of alternation, and is particular.

1.3 Statement of the Problem Second Problem Second Problem Second

Export growth is not only necessary to developed countries but also to developing countries such as Jordan. Export aptitude is an important determinant of a country's economic growth. Therefore, an increase in economic growth and development is tied to an increase in all types of exports (Keesing 1979).

According to Kaynak & Gurol (1987) "less - developed countries (LDS) rely heavily on export marketing to obtain much needed foreign currency necessary for their socio - economic and technological development. Marketing domestic products in foreign markets provide less - developed countries (LDS) with a reliable source of income to purchase the capital goods they need for this development from the industrialized countries of the west".

The search for export growth and expansion is critical for the Jordanian economy. According to a recent economic report about Jordan prepared by the UINDO (1987); "manufacturing exports needed to be enhanced in order to redress problems stemming from an increasing balance of payment deficit and the stagnation of workers' remittance and the drastic fall in the price of phosphates".

Jordan's economic position is suffering deeply from a chronic deficit in its balance of trade. In 1987 and 1988, this was close to \$1.7 billion or 37% of its gross national product (GNP), despite the higher contribution of the external trade to gross domestic product (GDP) which reached 143.1% in 1987. This growth of the external trade was attributed more to imports than to exports (for more detail about the problems of Jordan's economy, see Section 1.7.2 in this Chapter).

As a means of improving its economic situation, and in particular, the balance of trade, co-operative planners of the Jordanian government have increasingly put their efforts into the expansion and

promotion of export marketing among manufacturing firms. This was stated in the country's five year plan of economic development for the period between 1986 -1990, (Jordan, Five Year Plan 1986 - 1990). It includes the following goals: (1) To increase national export on an average rate of 12.6% at current prices, (2) To increase Jordan's foreign currency holding through encouraging exports and reducing imports, and (3) To encourage the establishment of export oriented industries.

In August 1988, the government of Jordan announced a series of measures to encourage expansion of exports from Jordan (Ministry of Industry, Unpublished Paper, Government Export Promotion Measures, 1988) which include the following^{*}:

- Encourage establishment of specialised trade companies in order to export Jordanian products and give those companies similar incentives and exemptions which are given to exporters.
- Establish Trade Centres wherever possible in Arab and Foreign countries, and continue organising specialised international trade fairs.
- Simplifying customs procedures that relate to the temporary entry system and duty - drawback system.
 - 4. Prepare a feasibility study for the establishment of export credit guarantees and a government insurance agency for exports.
 - 5. Appointment of Commercial Attaches in Arab countries, EEC countries, Japan and the United States.
 - The Central Bank of Jordan to allow exporters to keep 30% of export income in hard currency.
 - 7. Reimbursement of production tax on export products.
 - 8. Acceptance of re discount of bills in the same currency.
- " Most of these measures have not yet put in action.

Despite the Government's efforts to stimulate exports from Jordan, it is recognised that there are a large number of Jordanian industrial firms which are thought to be capable of exporting, but which do not get involved in export marketing activities. Furthermore, some of those firms which are already involved in export marketing, do not do as well as they could.

This might be because these policies are formulated on the basis of macro oriented analysis and the actions recommended which they generate. In other words, the government policy makers appear to have ignored the important unit of the business's economy activity, which is the individual firm.

A better understanding of the factors that are associated with export activity at the firm level might be extremely useful for both public policy - makers and managers. For the public policy maker, the knowledge of those factors which determine the export behaviour at the firm level would assist them in formulating an effective export promotion policy. For the decision - makers of manufacturing firms, knowledge of the influence of the firms' internal characteristics on export could influence the type of changes that should be implemented within their organisations.

1.4 Research Objectives

Researchers in the past (i.e., studies at the macro level) have tended to ignore behavioural factors in explaining international trade, they have focused mainly on explaining why countries would trade in spite of the fact that countries do not trade but industrial firms do.

Product life cycle theory was an early attempt to draw analytical attention to the firm as the unit of analysis (e.g., Vernon 1966, and Vernon and Wells 1986). According to the Product Life Cycle concept many manufacturing goods such as electronic products and office machinery, undergo trade cycles.

However, most of these studies focused almost exclusively on identifying industrial and other structural and contextual variables as determinants of export activity. To date, their efforts have only been able to identify conclusively technology and product variables as important structural factors contributing to export entry into new foreign markets (Reid 1980).

In other words, these studies have failed to provide an explanation of why, at a point in time, some firms are involved in export marketing and others in the same industry are not, or why some firms are very highly involved in export marketing, while others are not. This failure stems from the tendency of conventional economic theory to ignore the behaviour aspects of the firm and the factors that govern their behaviour.

As a result, the firm's export behaviour has received significant research attention recently, particularly in developed countries (e.g., U.K, U.S.A) where the majority of export behaviour studies have been undertaken.

Despite the fact that *a few empirical studies have systematically* examined the firm's export marketing behaviour, these studies have tended to focus largely on investigating the influence of the firm's organisational and a managerial characteristics.

However, according to the organisational and the firm's behavioural theories (e.g., Cyert and March 1963), the firm's behaviour is linked inseparably to the environment in which it takes place. In other words, the firm's business behaviour is not only determined by its internal characteristics, but also through the interaction of its internal environmental characteristics and the factors that exist in its external environment.

Therefore, the main aim of this study is to understand how the interaction of the firm's internal and external characteristics can effect the firm's export behaviour and its level of exporting in developing countries such as Jordan.

The key objectives of this research are as follows:

1. To develop a theoretical framework through the integration of international trade theories, the firm and organisational behavioural theories, and export behaviour studies. This framework consists of two broad dimensions; the firm's external environment and the firm's internal environment. The major constructs of each dimension are presented in detail in Chapter Four.

2. To find out the main pattern of factors (i.e., components) that underlie each construct of both environmental dimensions (i.e., internal and external). The firm's internal dimension consists of five constructs: (1) Managerial aspirations level (2) Management's expectations (3) The firm's comparative advantages, (4) Commitment to export marketing activity, and (5) Socio - demographic characteristics of the decision makers. The firm's external dimension includes two

constructs: (1) Economic and commercial infrastructural environment, and the (2) Legal - political environment.

3. To find out the effect of the firm's internal environmental measures (together and separately) upon its level of exporting.

4. To find out the effect of the firm's external environmental measures (together and separately) upon its level of exporting.

5. To find out which environment; the external, or internal, or the interaction of both environments, can explain larger the variations of the level of exporting among manufacturing firms in Jordan.

6. To identify and profile the exporter's and the non - exporter's groups on the basis of their internal and external environmental measures.

7. To compare the firm's internal environmental measures with its counterpart (i.e., the external environmental measures) in terms of their predictive power of classification of the group membership, i.e., exporters and non - exporters.

8. To explain the variation in export behaviour (i.e., exporters vs. non - exporters) according to both internal and external environmental dimension measures.

1.5 Research Hypotheses

In the formulation of the hypotheses which guided the research, several sources were employed as follows:

(1) Previous research studies related directly or indirectly to the

phenomena under investigation were reviewed.

(2) In - depth interviews were conducted with a small number of the top executives of Jordanian manufacturing firms.

The main aim of this study was to investigate the effect of the firm's internal and external environmental measures upon its export behaviour and the level of exporting. Thus, the number of hypotheses to be tested will be very large, using the formula the maximum number of inter - relationships that can be generated:

K = N (N - 1)/2

(b) The set Where: a fill a set of the permet of the permet description of the present of the

K = The number of inter - relationships N = The number of variables

Since N = 111, the maximum number of the inter - relationships could be up to (6150). In addition to that, it was expected that the findings would lead to the formulation of new hypotheses that could be tested in future research.

Therefore, the main hypotheses of this research stated in their null form are as follows:

(1) There is no significant relationship between the two environmental measures (i.e., internal and external) and the level of exporting, taken separately.

(2) There is no significant relationship between the internal environmental measures and the level of exporting, taken together.
(3) There is no significant relationship between the external environmental measures and the level of exporting, taken together.

(4) There is no significant relationship between each independent variable (i.e., the factors underlying each construct of both internal and external environmental dimension) and the dependent variable level of exporting, taken separately.

(5) There is no significant relationship between the explanatory independent variables which comprise each factor and the dependent variable level of exporting, taken separately.

(6) There is no significant difference (i.e., variation) between the two groups (i.e., exporters and non - exporters) in terms of their internal environmental measures, taken together.

(7) There is no significant difference (i.e., variation) between the two groups (i.e., exporters and non - exporters) according to their external environmental measures, taken together.

(8) There is no significant difference (i.e., variation) between the two segments (i.e., exporters and non - exporters) in terms of their internal environmental measures, taken separately.

(9) There is no significant difference (i.e., variation) between the two segments (i.e., exporters and non - exporters) according to their external environmental measures, taken separately.

(10) There is no significant improvement in the discrimination between the two groups (i.e., exporters and non - exporters) after the addition of the external environmental measures to its counterpart, the internal environmental measures.

(11) There is no significant difference (i.e., variation) between the two segments (i.e., exporters and non - exporters) according to the explanatory independent variables comprising each factor, taken separately.

1.6 The Significance of the Study

Whilst there is a large body of literature on the topic of economic under - development and export, most of it has a macro economic focus and, a perspective of the firm and its problems. National economic policies seem to be developed mainly on the basis of macro oriented analysis and the recommendations which they generate. The success of the actual policies though is often an uncertain affair seemingly dependent on the random tides of economic and political events both at the national and international level.

In the literature, one encounters little empirical research on the exporter at the firm level in an underdeveloped economy and, there is a complete void of Jordanian firm level studies. This study would make a modest contribution by providing background knowledge of Jordan's export as a less - developed country as seen from the firm level.

The information unearthed by the study will contribute to an understanding of export behaviour in the developing countries. More important though, the study would assist and guide the public policy makers in developing countries in fashioning appropriate policies which take into account, based an empirical evidence, the problems and needs at the level of the firm. Finally, should the study enable developing countries to make some progress in their efforts to expand their export of manufactured products, the aggregate welfare of both exporting and importing nations would be enhanced.

The major potential contributions of the present study can be summarised in the following points*:

 The area researched is important to the firm as well as to the economy as whole.

^{*} For more details about the contributions and the applications of the present study's findings, see Chapter Ten.

- (2) To the best Knowledge of the researcher, this study is the first attempt undertaken in the Jordanian environment to study export behaviour at the micro level (i.e. at the level of the individual firm). Most of the exporting studies in Jordan have been carried out at the aggregate level (i.e., at the national level).
- (3) An integrated approach is developed for the factors that determine the firm's export behaviour and its level of exporting, and the establishment of propositions. This approach integrates ideas and elements from the theory of the firm (e.g., aspiration level, expectation, etc.), international trade theory (e.g., the principle of comparative advantage), the product life cycle concept of international trade; the result of pertinent empirical studies as well as the results of the preliminary interviews which had been conducted in Jordan by the researcher.
- (4) Contribution is made to the scant number of studies that look at factors within the firm as determinants of export involvement.
- (5) As shown in the literature, the body of research on export marketing at the firm's level is heavily oriented to the firm in developed countries. This research has broadened its scope to include less - developed countries (LDS).
- (6) By examining the determinants of export marketing involvement, Jordanian co - operative planners could formulate an appropriate promotion programme aimed at assisting manufacturers to become very actively involved in exporting.
- (7) The managers of manufacturing firms could also benefit from this study by learning about the factors that determine or enhance the level of their firms' exporting.

1.7 Some Background Information About Jordan

In this type of study, it is useful to provide some relevant background information about the test country. The main objective is to familiarize the reader with the country under investigation.

1.7.1 General Economic Features

Jordan can be classified as a middle - income economy with a small domestic market, as reflected by the small size of population, low per capital income and low relative consumption relative to developed countries (See Table 1.1). In 1988, the total population of Jordan was estimated to be 3.38 million inhabitants, with a population growth rate of about 4% which is among the highest growth rate of population in the world (Kanovsky 1989). Jordan's per capital income is very low in comparison to other developing countries, particularly among the Middle East countries, but recently it has witnessed a high growth rate in GNP per capital of about 5.8%. This growth rate is regarded as a remarkable development in Jordan's economy (World Bank 1987).

The total area of Jordan is about 90,000 square Klm. The greater part of the country consists of a plateau lying between 1,000 and 3,000 feet above sea level. Most of this area is desert and supports only a small nomadic population. The country is without many natural resources. It is characterized by a high degree of openness with respect to labour, goods and capital market. Although Jordan is one of the world's major exporters of phosphate, its economy in fact is service oriented, with more than 60 % of GDP being obtained from services. Manufacturing contributes only around 15% of GDP and agricultural activities contribute only 8% of GDP (EIU 1986).

1.7.2 Problems of the Jordanian Economy

Due to the heavy dependence on foreign aid, particularly from the rich Arab countries, and the steady decline of this aid, the Jordanian economy started to face serious problems which began to appear around 1988. However, the indications of those problems were apparent from the early 1980's. The main problems can be summarised as follows:

1. The high rate of unemployment: The unemployment percentage has risen from 2% in 1981 to about 15% in 1988 and probably 30% in 1990 (Kanovsky 1989). This rate may also have accelerated as a result of the large scale of repatriation of Jordanian workers employed in the Gulf States and Saudi Arabia.

2. Decline in foreign exchange reserve: The decline in Jordan's convertible foreign exchange reserve, which reached maximum level in 1986 of (\$275) Million (Business America 1987). In 1988, the government was obliged to abandon the fixed exchange rate of the Jordanian currency which lead to a serious devaluation (About 50% against the major currency).

3. The rising burden of debt repayment obligation: The debt repayment burden is likely to increase as Jordan substitutes commercial borrowing for official grants and loans from OPEC members and the United States. The external debt rose by 94.7% in the period between 1981-1985, in 1985 interest payment on the government's external debt came to nearly JD 33 Million compared with Jd 11.2 Million in 1985 (Philip 1986).

4. Steady trade deficit: Jordan has always suffered from a considerable trade deficit. The trade deficit rose up to \$ 1.8 billion in 1987 (see Table 1.2).
Table (1.1)

Some Basic Indications About Jordan's Economy

- GDP (1987):	1,686.3 JD	Million	the energy
- Population (mid 1988):	3.38 Milli	on	
- Average Annual growth rate: of population	<u>1965–1980</u> 2.6	$\frac{1980 - 1985}{3.7}$	$\frac{1985 - 1990}{3.8}$
- Labour force (1987): - GNP per capital (1985):	595,000 \$ 1,560		
- Average annual growth rate of GDP	<u>1976 –19</u> 8.5	$\frac{1980}{6.4}$	$\frac{1985 - 1990}{2.16}$
- Distribution of GDP	1970	1980	1985
(Percentage) Agriculture Industry	10.2	6.6	6.0 16.5
Manufacturi Others	ng 4.3 82.3	11.1 76.4	11.2
- Rate of inflation: <u>1982</u> <u>1</u> 7.4	$\frac{983}{5.3}$ $\frac{1984}{3.8}$ $\frac{1985}{3.6}$	$\frac{1986}{2.5} \frac{1987}{1.0} \frac{19}{1.0}$	988 <u>1989</u> <u>1990</u> 1.0
- Exchange rate: 198 (Jordanian Dinar 0.3 equivalents to \$1)	5 <u>1986</u> <u>19</u> 94 0.350 0.	<u>87</u> <u>1988</u> 344 0.560 0	<u>1989</u> <u>1990</u> 0.780 0.680
- Private consumption: <u>1</u> per capita	972 -1982 4.5%	<u> 1982 </u>	-1987 0.5%

* Sources: A Company and the test additional of the second states the second states and the second states and

Central Bank of Jordan (1989).
 Jordan, Department of Statistics (1986).
 Jordan, Five Year Plan (1986 - 1990).
 United Nations, International Trade Statistics (1987).
 UNIDO (1987).
 World Bank (1987).

1.7.3 External Trade Features

The foreign trade sector plays a vital role in Jordanian economic activity. It has special position due to the size of its contributions to the GDP, to employment and to its role in meeting the country's need for consumer and capital goods. The contributions of the total value of import and export of goods and services to the GNP was about 120% in 1987.

Despite the difference in growth of export and import in the last two decades (see Table 1.2), Jordan's economy suffers from a chronic deficit in its balance of trade. Jordanian economic expansion process is heavily dependent upon imported goods and services, i.e., the ratio of imported goods and services to gross domestic product is very high, this was calculated to be about 82.9% in 1988 (Abuzir 1988).

This huge amount of imports can not be financed only by exporting almost the entire Jordanian GDP. Instead two methods of external finances have been available: the first is the remittance of the Jordanian worker abroad and the second is the financial assistances (Amerah and Al-Hajji 1987). However, since 1982 the latter two sources, i.e., the workers' remittance abroad and the financial assistance from official sources - mainly OPEC countries have shown a declining trend. In 1980, net official transfer to Jordan amounted to \$1.31 billion, however, this figure had been halved by 1985.

Furthermore, export growth during the 1980s has generally been higher than the growth of imports. Imports growth was at an annual average rate of 3.5% during 1981 - 1985. While exporting recorded a growth rate of 6.5% during this period, export growth rate slowed after 1986 to be about 4%.

Table (1.2)

Some Basic Indicators About The External Trade

EXPORT	a analasis of anne signal consister brids	
- Total Value (1988):	JD 324.788 Million	
- Main Exports (1988): (percentage)	Food and other consumer products goods (54.1) Phosphates 929.8), Construction Material (3.4), Others (12.7).	
- Main Destination:	Iraq (18.8), Saudi Arabia (12.3), (1988) EEC (4.5) India (15.1), Romania	
harmesettasila. Phasein u	(3.9), Indonesia (3.6) Kuwait (3.9), Others (39.1).	
IMPORTS	Jorfeet, they are chosinal and read	
 Total Value (1988): JD 1,022,469 Million Principle Imports: (percentage) JD 1,022,469 Million Consumer goods (38.7), Raw material (33.4) Capital goods (23.6), Others (4.9). 		
BALANCE OF PAYMENT: 19	<u>81 1982 1983 1984 1985 1986 1987 1988 1989</u>	
(Current account – balance \$ Million)	39 -333 -391 -265 -261 -40 -352	
- External Debt (1987):	\$ 4 Billion	
 Debt Services Ratio: (percentage of export earning) 	$\frac{1985}{23.1}$ $\frac{1986}{28.7}$	
- Foreign Debt: 1/2	<u>984</u> <u>1985</u> <u>1986</u> .491 2.470 3.079	

* Sources:

1. Central Bank of Jordan (1989).

- 2. Jordan, Department of Statistics (1986)
- 3. Jordan, Five Year Plan (1986 1990).
- 4. United Nation, International Trade Statistics (1987).
- 5. UNIDO (1987).
- 6. World Bank (1987).

1.7.4 Jordan's Export Pattern and Structure

Jordan's export pattern is characterized by a high degree of concentration on specific commodities and destinations. The domestic commodities export of Jordan consists of three types; consumer goods, raw material and capital goods. Phosphate and potash make up a major portion of natural exports accounting for about 50% of the commodities export. Export of citrus and other fruits and vegetables amount to 12% while the balance was made up of industrial products such as pharmaceuticals, plastic and paints. Two product categories dominate manufactured exports from Jordan, they are chemical and metal products. Together they accounted for about 63% of the total manufacture export earnings in 1985. Food and the manufacturing of textiles are other major export earnings. Their share in manufactured exports stood at 13.5% in 1985.

1.7.5 Characteristics of the Manufacturing Sectors

Jordan may be said to possess a dualistic industrial structure. Most small and medium sized firms are in the textile and food manufacturing branches. The larger establishments with state participation are in the chemical, non - metallic, mineral and petroleum refining branches. These accounted for 46.1% of MVA (Manufacturing Value Added) in 1985 up from 30.0% in 1975 (UNIDO 1987). Most manufacturing enterprise are localized in a narrow industrial belt stretching from Amman to Zarqa in the north West of the country.

According to Abdeljaber (1986) the Jordanian industry was a fast growing economic sector in Jordan, rising from JD 40.5 million in 1974 to JD 237 Million in 1984. In 1985 - 1987, the share of industry (including mining and manufacturing) accounted for 20% of the gross

domestic product and 10 - 11% of the total employees. The share of manufacturing in total export has also risen from 46.6% in 1979 to almost 59 % in 1984. Table (1.3) shows the compositions of export by sector of origin 1981 - 1984.

Table (1.3)

Composition of Export 1981 -1984

(Percentage)

strend -		1981	1982	1983	1984	
- т	otal Exports	100.0	100.0	100.0	100.0	
- A	griculture	23.5	19.6	19.5	11.5	
- M	ining Quarry	23.2	22.8	25.7	29.3	
- M	anufacturing	53.3	57.6	54.9	59.1	
- F	ood, Beverage &	8.8	7.8	6.6	6.4	
Т	obacco					
- W	ood & Wood Products	2.9	2.1	1.6	2.9	
- T	extile	5.6	5.1	2.8	7.1	
- P	aper & Paper Products	1.8	1.3	1.1	1.2	
- C	hemicals	9.6	11.9	21.4	26.1	
- N	on - Metallic Minerals	3.4	3.6	3.4	3.6	
- B	asic Metal	1.1	1.0	0.4	0.4	
- M	etal Manufacturing	19.8	23.2	16.9	11.2	
- 0	ther Manufacturing	0.3	1.6	0.6	0.2	
A. LAND	and store swell with a weak	inter Ro	a large a	and the local days		

* Source: United Nations, <u>International Trade Statistics</u> (1984), Vol.1 New York (1987) P. 556

1.8 The Organisation of This Thesis

Chapter One provides an introduction to the thesis, starting the importance of exporting to the developing countries, the research problem, research objectives and hypotheses, and the significance of the study. It also gives some brief background information about Jordan with special emphasis on the economic dimension. Figure (1.1) gives a summary of the organisation of the thesis.

Chapter Two presents a review of international trade theories, the firm's behavioural, and organisational theories. The chapter primarily focuses on the existing literature related to export behaviour. The main findings and limitations of the previous research are presented.

Chapter Three deals with the preliminary in - depth interviews. It gives an explanation of it's main purposes, structure, and planning. It also provides some background information about government export procedures, policies and assistance.

Chapter Four presents the research framework. It gives in detail the main constructs of the study's framework.

Chapter Five deals with research design and data collection. This chapter starts with evaluating the alternative methods of data collection, and provides the basis and rationale for selecting an appropriate method. The selection of the scale of measurement, the key respondent approach, the domain of the study's population, and questionnaire development are also presented.

Chapter Six presents the research methodology. The chapter starts with a review of the alternative statistical techniques available, the assumption behind these methods and the basis for the selection of the appropriate techniques. The techniques used in this research are factor analysis, discriminant analysis, simple correlation, regression analysis, T - test, F - test, Chi - square test, and the McNemar test.

The chapter gives a description of these analyses and the justification for their use in the research.

Chapter Seven presents the research findings related to the main pattern of factors that underlie each construct of the firm's internal and external environmental dimension.

Chapter Eight deals with the research findings and interpretation related to the determinants of the firm's level of exporting.

Chapter Nine provides the research findings and interpretation concerning the determinants of the firm's export behaviour (i.e., exporters vs. non - exporters).

Chapter Ten starts with a summary review of the entire study, it presents the main conclusions of the present study with some implications for manufacturers and the Government. The research contributions in terms of theory and practice also presented. Then, two types of recommendations were discussed, i.e., recommendations for potential and current exporters and recommendations for public policy makers. After that, the limitations of the research are specified. Finally, the chapter concludes by suggesting some areas for future research.

Figure (1.1)

Organisation of the Thesis



CHAPTER TWO

REVIEW OF LITERATURE

Introduction

2.1 International Trade Theories

2.1.1 The Traditional Theory of International Trade

2.1.2 The Factor Abundant Theory

- 2.1.3 The Product Life Cycle theory
- 2.1.4 Limitations of the International Trade Theories
- 2.2 The Firm and Organisational Behavioural Theories

2.2.1 The Theory of the Firm 2.2.2 The Organisation Behavioural Theories

2.3 Export Behaviour Studies

2.3.1 Studies Relating to the Firm's Internal Environment

2.3.1.1 The Initiation of Exporting 2.3.1.2 The Firm's Management Export Orientation

2.3.1.2.1	Motivation for Export Marketing
2.3.1.2.2	Firm's Management and Export Marketing
2.3.1.2.3	Perceived Obstacles to Export Marketing
2.3.1.2.4	Export Marketing Mix Decisions
2.3.1.2.5	Firm's Demographic and Decision - Maker's Individual
	Characteristics

2.3.2 Studies Relating to the Firm's External Environment

2.3.2.1 Economic and Commercial Infrastructural Environment 2.3.2.2 Political - legal Environment

2.3.3 Export Modelling Studies

2.4 The Common Characteristics of Export Behaviour Studies

Summary

CHAPTER TWO

REVIEW OF LITERATURE

Introduction

The theme of this chapter is two fold:

- (1) To present and discuss empirical studies relating to the supply side of international marketing, by means of a content analysis of the findings of export marketing behaviour studies.
- (2) To use this literature review in order to develop a conceptual framework within which to conduct the research. This framework consists of integrated literature at the macro and micro level, i.e., international trade theories, the firm's economic and organisational behavioural theories, and empirical studies of export marketing behaviour.

2.1 International Trade Theories

The historical explanation of the determinants of the factors governing the pattern of trade can be found in traditional economic theories. In general the main purpose of international trade theories can be summarised as follows:

- 1. Why countries trade?
 - 2. What goods countries will export and import?
- 3. How gains from trade are divided between trading nations?
 - 4. How adjustments are brought about export when trade partners are distributed (Kinsey 1988).

None of these theories is generally applicable. As E1-Agraa (1983) states "there is no single theory capable of explaining international trade in all commodities at all times". Furthermore, international trade theories have a limited utility for explaining the export pattern of the individual firms (e.g., why do firms get involved in exporting), because they assume that trade will be determined by relative factor cost and by demand countries, both parameters can not be manipulated by individuals' firms unless they are monopolies (Hirsch 1971). However, they are still widely accepted as the conceptual basis for understanding and predicting events in the world of international business. Ellsworth (1964) has noted:

"Factors at the micro- level are fundamental elements in explaining the direction and structure of international trade, and their importance, therefore, should not be discounted".

Despite their limitations and their macro economic orientation, international trade theories can be useful for business management studies, and they are still an obvious starting point and continue to form the type of business thinking for many features of the global marketing environment (Kinsey 1988). In this respect Mason *et al.* (1975) state:

"A thorough understanding of the trade theory can provide the business manager with some comprehension of the large systematic factors influencing his operation in the world". Hirsch (1971) also noted:

"Trade theories help to delineate the boundaries within which individual firms operate and point to the degree of freedom which they possess and process selection".

Therefore, a better understanding of these theories, their contributions and limitations, are important for the purpose of this research as a theoretical perspective. <u>A brief presentation of the</u> theories and limitations are given below:

2.1.1 The Traditional Theory of International Trade

Traditional trade theory purports to explain why international trade exists and the pattern of this trade. Their answer to this question is based on the concept of economic advantage. The concept of economic advantage holds that "countries tend to specialize in those products in which they have an advantage, namely lower cost production" (Albaum *et al.* 1989).

In other words, the traditional approach to the explanation of the pattern of trade begins with the proposition that trade, and the pattern of trade, depends on pre - differences in (relative) price. The main question at this stage is why countries differ in their economic advantage (cost or prices). In other words, what factors determine their economic advantages. The explanation has been suggested in terms of two different situations; (1) absolute advantage and (2) comparative advantage.

The absolute advantage was put forward by Smith in 1776 (Smith 1937), and states that a country will export those commodities in which it has an "absolute advantage". The explanation of this depends on assuming: (i) a labour theory of value, i.e., the only cost of production is the number of man - hours (m/h) required per unit of

output multiplied by the wage rate (w), and (ii) that commodity price are cost - determined.

According to economic analysts, Adam Smith's explanation is insufficient to explain the international differences in price or cost (e.g., Bhagwati 1964 and Haberler 1961).

The comparative advantage was developed by the classical economist, Ricardo (1817, 1951). He theorized that nations engaged in trade because of "the special ability of a country to provide one product or service relatively more cheaply than other products or services". It is not vital that a country holds an absolute advantage over other countries.

Ricardo refined Smith's argument by explaining why price should vary between countries. Prices differ because countries producing the commodities, or products under consideration have a different comparative cost structure. Therefore, a country benefits if it produces more of the goods in which it is relatively more efficient and exports these in exchange for those in which its absolute advantage is least. In other words, international trade is determined by comparative rather than by absolute advantage.

However, Ricardo's explanation fails to explain how the actual terms of trade are determined. As Young (1963) observed:

"In his illustration, Ricardo helped to explain the conditions under which specialization and trade would be profitable ..that is, why trade take place .. but he did not explain how the actual terms of trade are determined ..that is, why just so many yards of cloth trade for so many bottles of wine".

Ricardo explains that trade occurs within the limits set by comparative cost. However, his explanation does not give any indication of how the actual rate of exchange rate (export volume) will be determined and how the benefit from trade will be distributed.

Another weakness of Ricardo's law of "comparative advantage" is that it does not consider such aspects as tariff, transport costs, different political ideologies, special interest groups and cultural barriers. However, it is useful to the extent that it helps predict what products, or what amount will be entered into with international trade (Kinsey 1988).

Despite its limitations, comparative advantage theory remains widely accepted (Kreinin 1971). Fayerweather (1969) noted:

"The comparative advantage, resources - distribution approach has persisted to this day as the basic doctrine of the field".

2.1.2 The Factor Abundant Theory

While traditional theories argue that international trade occurs as a result of differences in international production characteristics and factor productivity, due to domestic differences in natural advantage and acquired advantage, other economist analysts noted that these theories do not explain sufficiently the cause of differences in comparative advantages (Hecksher 1919 and Ohlin 1933).

The factor abundant theory, however, offers an explanation for the differences in comparative cost among trading partners (Hecksher 1919 and Ohlin 1933). According to this theory, international differences in supply conditions explain much of the international trade. Supply conditions including factor endowments, as well as productivities.

The underlying assumption in this theory is: trading partners (two countries) have the same demand conditions, use the same technology, use factors of production that are of uniform quality. This means, therefore, the productivity or efficiency of a given resource unit is similar for both trading countries.

The factor abundant theory argues that relative price levels differ among countries, because they have a different relative endowment

factor of production, and different commodities require that the factor inputs be used with differing intensities in their production (Albaum *et al.*, 1989). In short, a country will export (import) those commodities which are intensive in the use of its a abundance (scare) factor. In the discussing of Heckscher - Ohlin's model, Ford (1965) has noted that the basic model is constructed on rigid and very unrealistic assumptions and he then states:

"They are really contrary to what are observed around us, the Ohlin Heckscer premise of perfect competition, no differences in the quality of factors, no product differentiation, lack of return to scale industries, fixed production increasing function are all unrealistic. Furthermore, in connection with the function of these premises, there is scarcely any reference to the social economic forces that are undoubtedly of importance in shaping the pattern of trade, for example, the desire for economic progress and its effect on consumer demand and income is left untouched in the basic theme.

Based upon Ricardo's model, he further developed a model in connection with other factors, such as the social economic forces, and he has come up with the following conclusion:

"The basis of its trade is part of its original factor endowments, but a very important part is played by the level of development actually reached by each country."

Another deficit of the comparative advantage theory was introduced by Kindleberger (1973). He noted the assumption was that each country has the same technological possibilities of producing a given product, i.e., that the production functions are the same in both countries. However, it was found that the existence of trade was based on technological differences between countries. A similar argument was also put forward by Linder-Burenstam (1961) in which he assumed that differences of comparative advantages between nations are due to technological factor rather than those explained by Ricardo. Moreover, because the theory defines the factor endowment broadly (land, capital and labour), it has become difficult to define which

of these factors is for the purpose of its explanation (Kindleberger 1973).

The explanation of trade - flows between different countries therefore, requires an elective view. Different export products and exporting countries are subject to a variety of forces, therefore, different theoretical explanation of are required.

2.1.3 The Product Life Cycle Theory

International trade theories examine why countries trade, but this is insufficient to explain why the production of goods and services are transformed across national boundaries (Kinsey 1988). For various reasons such as rapid technological advancement and the use of multinational corporations, the international theories (based on the economic advantages of factor endowment) have failed to explain the international pattern of trade as they evolved starting in the 1960's.

The international product life cycle theory developed by Vernon (1969, 1971) and his associates-particularly Wells (1968, 1972) has become one of the leading explanations of international trade patterns. According to this theory, the process of movement in world markets can be explained in terms of trade cycle, that is, that trade exists as a result of life cycle.

The model of international trade life cycle has been developed in terms of four stages. The first stage starts when an innovator country (the United States as described by Vernon) establishes a breakthrough in the production of a manufactured item. This country initially has a lack of knowledge about this type of technology and is typically a high- income developed economy. At this stage, because of the relative small market for the product (i.e., the product is only common to a few customers) and the high technological uncertainties (degree of

risk) associated with its production and marketing, mass production is not feasible.

In the second stage, as the product becomes recognized and accepted 7 in the domestic market, the innovator country may begin to export to the nearest countries (West Europe, Canada) which have almost similar tastes and income characteristics. In the third stage, the innovator manufacturer finds that during this stage of growth and expansion, its market becomes large enough to support mass production operations and can eliminate inefficient production methods, which means that an increasing amount can be exported to foreign markets.

With the passage of time, the innovator manufacturer may find its product has become more commonplace and the costs which were associated with its production and marketing, become sufficient to expand his export operations to include other foreign markets (lessdeveloped countries). Although the innovator country's market position may be protected by legal patents, this may break down after a period of time. This is because knowledge tends to be widen in the long term. The foreign producer starts to imitate the production operations, In other words, as time passes, the innovator country begins to lose it comparative edge and its export trade cycle begins to face a declining phase. The trade cycle is complete when the production process becomes so standardized that it can be easily utilized by other countries. At this stage the innovator country may finally become the importer of the product, as its market position is lost due to foreign competition.

The main assumptions underlying the international product life cycle are: (1) The flow of information across national borders is restricted and that products undergo predictable characteristics in the production and marketing character over a period of time and this

assumption is in contrast with the traditional theory assumption, and (2) The productive process is characterised by economics of scale, that is, change over time, i.e., each country does not account for fixed proportions of expectation for buyers at different income levels given a set of international prices.

As long as the information does not flow freely across national boundaries, this leads to three important conclusions, according to Wells (1972), which are as follows: (1) Innovation of a new product and process, it is argued, is more likely to occur near a market where there is a strong demand rather than in a country with little demand, (2) A businessman is more likely to simply risk capital for the production of the new product if the demand is likely to exist in his home market than if he has to turn to a foreign market, and (3) A producer located close to a market has lower costs in transferring market knowledge into product design, than one located far from the market.

Based on the product life cycle theory, a series of studies of specific industries has provided evidence of international PLC behaviour (e.g., Wells 1968, Hirsch 1971, and Ayal 1981). For example, Ayal (1981) has attempted to test the applicability of PLC theory to Israeli export performance, and concluded that narrow range specialised products is an important determinant of such success.

However, the product life cycle has been attacked on several accounts. According to Terpstra (1978) "the overall model suffers from the recurrent difficulties of life cycle models". Thus, not all products can be expected to follow the cyclical pattern of trade. For example, some products which for a particular reason never get beyond a specific stage, and other products may only have slightly increasing return to scale, or the domestic market may never become large enough

to justify establishing manufacturing capabilities. Others also argue that trade tends to be made on the basis of product differentiation, or specialization rather than on the basis of standardisation as Cannon (1980) noted; "..as markets become more sophisticated, there is likely to be an increased adaptation of product rather than simple adaptation".

Although the model neglects the economic differences between nations (i.e., the economic forces which govern each country), because it considers the United States as a focal point for all the international trade activities. Furthermore, it provides little help to policy-makers and managers in developing countries such as Jordan. For individual firms, the product life cycle theory only suggests a certain product (i.e., high technological products) as a candidate for export but it does not provide guidance as to how the level of exporting may be improved.

2.1.4 Limitations of the International Trade Theories

Having reviewed the international trade theories and major arguments toward the factors which govern the export behaviour at the country level, the question remains to what extent are these theories able to explain export behaviour at the firms level.

It is contested here that international trade theories are unable to provide us with sufficient explanations for the following reasons:

1. Their assumptions of trade are based on two main factors; the relative factors of cost and demand. However, both parameters are insufficient to explain thoroughly the firm's export behaviour unless they are monopolies (for more detail, see Hirsch 1971).

2. There are many other factors which also need to be taken into consideration. These factors may have a major effect on the firm's

export behaviour, i.e., the firm's internal factors. However, these factors have been ignored in the explanation of international trade (see for example, Cavusgil 1976, Axinn 1985, etc.).

3. The explanations of international trade theories give us little insight into the external factors or the boundaries that firms are assumed to operate within. For example, the theory of comparative advantage fails to address socio - political and economic factors (for more detail, see Ford 1965).

4. Although the product life cycle theory was an attempt to use the firm level as the unit of analysis, to date analysts' efforts have only been confined to identifying technology and product variables that are associated with export activity (Reid 1980).

5. Despite the explanatory power of international trade theories as tools for the promotion of export, they provide little help to policymakers and managers. The international trade theories do not provide any guidance as to the conditions under which firms should be encouraged to export.

Therefore, the explanation of the pattern of trade between countries (export behaviour) is required to reach beyond the identification of the macro – economic factors. There is a need to investigate the factors which govern the pattern of trade at the micro – level.

However, with respect to this study, international trade theories are useful in two ways: 1. The theory of comparative advantage provides some theoretical support for the study of export behaviour at the firm's level (e.g., Schlegelmilch 1983). It is used here in this study's framework to represent one of the major constructs of the firm's internal environmental dimension (see chapter 4), and 2. The Product Life Cycle theory of trade, with its emphasis on domestic

demand as a stimulus to product development, also provides some theoretical background into the impact of the domestic demand on the firm's export behaviour, so that it is expected to be one of the main reasons for initiating the firm's export process.

newes in account theory in a transformation muit (presention told) teached isolate and planting these is used any to except with the similar of the activities and (insectors) is emergific against almosthem and restrained by inspect plaint methods find (e.g., stranges 1984) Conse (915 med Notative 2964). In other tolds, the fire of booker upon on a studie with it is pression against dealing with the problem of any a studie with it is pression and dealing with the fire problem of any a studie with it is pression and dealing with the problem of any a studie with it is production.

The accomptions of the classical theory of the first inserver, have been attached on ordered account. One of the major fabrics is conserved with the achiever of anony first Res Engeneral (1964) a buyle objective (server) of every first Res Engeneral (1964) expandings (1965), and herbenish (1947) agains that first way be is presente of objectives other than profit emainstation. If is, therefore, difficult to obtain a systematic prediction of the first work is acquiring antificiant includes a systematic prediction of the first such is expanding antificiant includes a systematic prediction of the first such is expanded and the classical economic taking of the first such is expanded and the classical economic taking at the first such is expanded and the classical economic taking of the first such is expanded at the classical economic taking of the first such is expanded at the classical economic taking of the first such is expanded at the classical economic taking of the first of the engentist, and the classical and there is start 1963). In approachers the conventional theory of the first in another engentist of desarrows is often the theory of the first in another engents is disagreeness where the theory of the first is the engents of the start the theory is (2) the exact to which the engents is defective and (3) approximate economic for expression the theory. 2.2 The Firm and Organisational Behavioural Theories

2.2.1 The Theory of the Firm

The theory of the firm as a subset of the macro economic field tends to concentrate on the allocation of scarce resources and the determination of income distribution (McGuire 1964). The firm is viewed in economic theory as a transformation unit (production unit), taking inputs and altering these in some way to output, with the limits of its activities set (functions) by specific market situations and restrained by its goal profit maximisation (e.g., Strauess 1944, Coase 1937 and McGuire 1964). In other words, the firm is looked upon as a single unit, or as a person dealing with the problems of marketing, pricing and production.

The assumptions of the classical theory of the firm however, have been attacked on several accounts. One of the major debates is concerned with the validity of the profit maximization as a single or a basic objective (motive) of every firm. But Hoogstrat (1964) Papandreou (1952), and Rothchild (1947) argue that firms may be in pursuit of objectives other than profit maximisation. It is, therefore, difficult to obtain a systematic prediction of the firm's behaviour from the classical economic theory of the firm such as sufficient liquidity. convenience of the security. management market position, restraining of achievement, maintenance of the competition, etc. (McGuire 1964, and Cyert & March 1963).

In approaching the conventional theory of the firm to establish their conceptual behavioural model, Cyert and March (1963) write that:

"There is disagreement about the theory of the firm in three respects: (1) what the theory is (2) the extent to which the theory is defective and (3) appropriate methods for improving the theory".

SHEFFIELD UNIVERSITY LIBRARY Several authors have argued that the classical theory of the firm has failed to view the "firm" as an organisation, i.e., this theory is confined to a few characteristics of organisation and there is nothing about complexity of organisation, conflict and control aspects (Cooper 1949, 1951, and Papandreou 1952). Cyert and March (1963) argue that the traditional theory has also failed to provide a model that applies specifically to the decision - making process within the firm, they state:

"Decision- making is treated rather vaguely in the traditional theory of the firm. Most of the discussions of decision making are concerned with normative prescribing of the steps to be followed in choosing between alternatives".

They also note that the conventional economic theory of the firm has three major characteristics which are; (1) it deals with a particular set of decisions, i.e., particular aspects of the firm, (2) it uses aggregation as a tactic in attempting to specify total market supply and demand curves, and (3) there is no attention to, or interest in, the actual process by which individual firms reach decisions.

2.2.2 Organisational Behavioural Theories

In comparison to the conventional theory of the firm, the organisational theories emphasis on the study of the firm's internal processes, i.e., what goes on in an organisation. Despite the fact that there has been much recent effort to bring some kind of order to the field, the study of the organisation is more widespread than the study of the economic activity (Cyert and March 1963). There are three major branches of organisation theory:

(1) The sociological aspect: This centres on the phenomena of bureaucracy, for example conflict, organisation life cycle etc.

- (2) The social psychologist aspect: This is built primarily on an experimental basis with the emphasis on "efficiency" and a major focus on the individual within the organisation as the relevant unit of analysis, for example motivation, leadership style, etc.
- (3) The administrative aspect: This focuses on the problems of the executive when dealing with an organisation.

However, existing organisation theory provides only a very partial basis for looking at the theory of the firm. The sociological and social psychological approach have emphasized questions that are only marginally relevant to either the objective of conventional theories of the firm, or the objective of predicting individual firm behaviour. However, the administrative approach is the only branch of organisation theory which contributes to the understanding of the behaviour of the firm, because it is mainly concerned with the internal process of decision making, i.e., how decision are made in organisations and the impact of organisational position on individual goals and perceptions (e.g., Bernard 1938).

Because of many problems and difficulties associated with the use of conventional economic and organisation theories to predict and explain the firm behaviour, Cyert and March proposed the behavioural model of the firm, which states that the firm's economic behaviour can be predicted and explained as <u>a result of interaction</u> between the internal operations of the individual firms and factors within the firm's external environment.

The behavioural theory of the firm which is proposed by Cyert & March (1963) states that it is possible to analyse the process of the firm's decision - making in terms of the variables that have an effect on organisational goals, and the variables that effect organisational expectations and organisational choice.

- 1. Organisational goals are recognized in terms of two sets of variables: (a) The dimension of the goals, i.e., why things are viewed as important and (b) The aspiration level on any particular goal dimension. It is viewed as a determinant of risk taking behaviour, i.e., the value which the individual places on the achievement of goals.
- 2. Organisational expectations are seen as the result of drawing inferences from available information, i.e., the belief which individual have that particular actions on their part lead to certain outcomes.
- 3. Organisational choice (alternative decision), they are: the past experience of the organisation and the past record of organisation slack.

The behaviour theory of the firm is adopted here as a theoretical basis for this study's framework (see Chapter 4), because it suggests that the firm's business behaviour (e.g., export behaviour, the level of exporting, etc.) is a function of the interaction of its internal operation process and factors that exist in outside its boundary (i.e., external environment). Furthermore, variables relating to the firm's internal environment, such as managerial aspirations and management's expectations are constructed in our study's conceptual framework (for more details, see Chapter 4).

In the next sections, the internal and external environmental dimension studies relating to the firm's export behaviour will be discussed respectively.

2.3 Export Behaviour Studies

In the last three decades, the export behaviour of the firm relating to the supply side of international trade theory has been a topic of interest in international business research. A substantial amount of research has been undertaken in an attempt to investigate and to test the factors that determine the export behaviour at the level of the individual firm. Some researchers have developed models to verify the export behaviour aspects. conceptual Nevertheless, the majority of these studies have failed to give a complete account of the factors underlying the export marketing behaviour, since they have tended to focus largely on the firm's internal characteristics, i.e., the firm's decision-makers' attitudes, the firm's organisational characteristics, and have given little attention to the effect of external factors, particularly government procedures and regulations concerning export.

For the purpose of this study, the content - analysis* of the findings of export marketing behavioural studies is presented under the following three major sections:

1. Studies relating to the firm's internal environment

- 2. Studies relating to the firm's external environment
- 3. Export modelling studies

* Separate evaluations of each study were not attempted because most studies tended to be concerned with a narrow aspect of the phenomenon under investigation.

2.3.1 Studies Relating to the Firm's Internal Environment

The content - analysis of this dimension is further divided and discussed under the following subsections:

- 1. The initiation of exporting
 - 2. The Firm's management export orientation
 - 2.1 Motivation for export marketing
- 2.2 Firm's management and export marketing
 - 2.3 Perceived obstacles to export marketing
 - 2.4 Export marketing mix decisions
 - 2.5 Firm's demographic and decision maker's
 - individual characteristics

2.3.1.1 The Initiation of Exporting

Studies concerned with the initiation of export activity have tended to focus on the effect of internal and external stimuli. A review of these studies reveals that firms become involved in export marketing operations because of internal stimuli, external stimuli, or a combination of both. Cavusgil (1976) states:

"There seems to be basically two types of stimuli which lead firms into exporting, initial participation begins either as result of management's own interest, or the firm "happens to receive" an external inquiry, without actively soliciting orders".

The internal and external stimuli which are thought to initiate exporting are given in Table (2.1). Literature relating to the effect of the external stimuli, such as competitive pressure, unsolicited order, government incentives, and others have yielded conflicting findings.

Whereas Tesar (1975) in his study of small and medium sized manufacturing firms in Wisconsin (USA) reported that export performance of the competing firms played an important role in inducing a firm to pursue its competitors by being involved in export marketing activities, Johnston & Czinkota (1983) and Rabino (1980) have reported that competitors pressure was not a frequent stimuli for exporting. Simpson & Kujawa (1974) in their study of small and medium sized manufacturing firms in the State of Tennessee found that the most important stimuli to initiate the firm to export was an unsolicited order from foreign customers. A similar result was also reported by Beamsih and Munro (1986). In contrast, Reid (1980) argued that the role of unsolicited export orders in inducing a firm to export was too simplistic, and could not stand alone as a sole explanatory factor. Additional factors such as adverse home market conditions and fluctuating exchange rates must also be considered. This result was also supported by Weaver & Pak (1990).

The importance of adverse home market conditions as mentioned by Reid was also confirmed by other studies (e.g., Bilkey (1978), Pavord & Bogart (1975) and Wiedersheim - Paul *et al.* (1978)). Pavord & Bogart (1975) who were more specific, concluded that the primary drive which initiated the firm's management to become involved in exporting activity was related to its management beliefs that adverse home market conditions will threaten the profitability and the eventual survival of the firm. However, the opposite has been indicated by Sinai (1970) who found a relatively strong inverse relationship between the perceived importance of participation in export trade to the firm, and the extent to which the firm felt the degree of saturation of the domestic market to be an important consideration in reaching a decision on whether to enter, or expand its export operation in foreign markets.

Studies about the effect of government export incentives as a reason for initiating exporting also produced conflicting results. While Rabino (1980) concluded that government incentive was one of the major stimuli inducing small firms to become involved in exporting activity, however, this was not supported by Simpson & Kujawa (1974).

Table (2.1)

Export Initiation Process Studies

Source of Initiation	Discussed By Author(s)	
1. External Stimuli		
- Unsolicited order	Perkett 1963; Snavely <i>et al.</i> 1964; Sinai 1970; Daniels & Goyburo 1976; Simpson & Kujawa 1974 ; Pavord and & Bogart 1975; Tesar 1975; Bilkey & Tesar 1977; Wiedersheim-Paul <i>et al.</i> 1978; Bilkey 1978; Reid (1980, 1983); Johnston & Czinkota 1982; Joynt 1982 Cavusgil 1984; Kaynak and Kothari 1984; Malekzadeh & Rabino 1986; Beamish and Munro 1986; Sullivan & Bauerschmidt 1990; Barker & Kaynak 1992; Weaver & Pak 1990.	
– Saturated home market	Pavord & Bogart 1975; Bilkey 1978; Wiedersheim - Paul <i>et al.</i> 1978; Reid 1980; Kaynak <i>et al.</i> 1987; Kaynak and Kothari 1984.	
- Competitive pressure	Tesar 1975; Rabino 1980; Joynt 1982 Kolhede 1984; Karafakioglu 1986; Johnston & Czinkota 1982; Cavusgil 1982.	
- Government incentives	McDougall & Stening 1975; Souza <i>et al</i> 1983; Sullivan & Bauerschmidt (1988, 1990); Welch & Wiedersheim-Paul 1979; Christensen <i>et al.</i> 1987; Karafakioglu 1986; Simpson & Kujawa 1974.	
 Better opportunities available in foreign markets 	Kaynak & Kothari 1984; Wiedersheim-Paul <i>et al.</i> 1978.	
2. Internal Stimuli		
- Top management orientation	Tookey 1964; Pinney 1970; Cunningham & Spigel 1971; Simpson 1973; Roux 1979; Wiedersheim-Paul <i>et al.</i> 1978; Khan 1978; Tesar 1977; Langston 1976; Simmonds & Smith 1968; Holzumller & Kasper 1990.	
 Manager assessment of the firm's competitive advantages 	Cavusgil 1976; Khan 1978; Daniels and Goyburo 1976; Kacker 1975; McGuinness 1976; Johnston and Czinkota 1982; Kolhede 1984; Schlegelmilch 1983.	

Karafakioglu (1986) and Sullivan & Bauerschmidt (1988, 1990) in their separate studies. Christensen *et al.* (1987) found that government incentives work more as a secondary stimulus, or reinforcement to the company's decision rather than as a primary motivation for aggressive exporters.

In discussing the effect of the firm's internal stimuli, studies found that a member of the firm's top management (decision - makers) who is interested in, and enthusiastic about exporting is the most important factor. However, it is disputed what factors determine the interest of the top management. While Simpson (1973) postulated that management's diffused impression of the attractiveness of exporting as an abstract ideal influence whether the firm's management takes the initiative in exporting, Wiedershiem - Paul *et al.* (1978) argued that it is the degree of the firm's international orientation that will determine management initiative in pursuing exporting opportunities.

A study by Cunningham & Spigel (1971) suggested that the international orientation of the firm's management is determined by the firm's management background and traditions, they state:

"The motivation of dynamic exporting, lies rather, in the recognition, that growth and long - term improvement in profitability are only achievable through the adoption of an international marketing outlook.."

Similar arguments are suggested by Perlmutter (1969), Pinney (1968), Wiedersheim - Paul *et al.* (1978), and Simmonds & Smith (1968). A study by Langston (1976) indicated that for United States firms foreign attitudes of top management correlate with whether or not they had studied a foreign language at school, whether they had lived abroad sufficiently long enough to have experienced cultural shock, and whether that foreign experience was attractive. A study by Pinney

(1971) showed that younger managers tend to be more internationally minded than older ones.

Another determinant of management's drive in initiating exporting is its confidence in the firm's competitive advantage as measured in terms of the management's perception of whether the firm has technology, marketing, financial, or price advantage (e.g., Cavusgil 1976, Tesar 1975, Snavely *et al.* 1964, Johnston & Czinkota 1982, Schlegelmilch 1983, and Kolhede 1984), whether the firm possesses exclusive information about a foreign market or customer, whether the firm has a potential product, and whether the firm has an efficient distribution network (e.g., Tesar 1975, and Cavusgil 1984).

Based upon the firm's management orientation, or outlook to export marketing, and the type of stimuli needed for initiation in exporting, Tesar (1975) identified three main types of firms: (a) Regionally oriented firms, i.e., firms dependent on the local market, and prejudiced against exporting (b) Opportunist oriented firms, i.e., these firms which are not actively seeking export markets but will not forego opportunities to export. These firms are described as passive exporters and, (c) Strategic oriented firms, these firms are operated by managers interested in promoting the business and their products. These managers are constantly looking for new opportunities are described as active including exporting and or aggressive exporters.

Having reviewed the major findings on the firm's export initiation process (see Table 2.1), the question arises as to which of the various stimuli is the most important for firms to become involved in exporting?. In response to such questions, studies in developed countries reveal that the firm's internal change - agents are the most important stimuli initiating the firm's export process. For

example, Cavusgil & Nevin (1981) in their study of the export marketing behaviour of manufacturing firms in the United States, concluded that; "the real barriers to the firm's involvement in exporting are internal rather than external to the firm". A similar view was also confirmed in Rabino's study (1980), in which he asked the exporters whether they would increase their export involvement without government assistance, he found that thirty six out of forty six surveyed firms have indicated that they did not need government support. However, he states:

"The argument was that as long as a product is unique, price is competitive, or highly acceptable on export market demand, it will increase anyway ,..., finally an observation was made that after investing such effort in deciding to initiate exporting, the commitment was firm regardless of government assistances".

The inference was also supported by Kolhede's study (1984). She concluded that the top management's perceptions of government export programme assistances was insignificant upon the export orientation of exporting and non - exporting of small manufacturing firms in the United States. However, it might be argued that the external stimuli, particularly government assistances, are more likely to initiate the firm's export process in developing countries, because it might be difficult for firms in these countries to obtain foreign export orders without government assistances. The intervention of the government here might be due to the lack of financial resources and know - how which many manufacturing firms suffer from. As Kaynak (1982) notes:

".. small scale private traders and producers in LDS will never be in a position to obtain by their own means sufficient and accurate information on conditions. This should be done by government".

2.3.1.2 The Firm's Management Export Orientation

Export orientation has been defined by Abdel - Malek (1974) as "the degree to which management appears willing and feels able to mobilize resources to initiate or expand the firm's involvement in the export market". Kolhede (1984) observed that changes to increase or decrease the firm's export marketing basically depends on export orientation.

She stated "a positive export orientation will cause the manager to decide to increase his firm's export marketing: a negative export orientation will result in the firm's resistance to export expansion".

Therefore, the top management's attitudes or their export orientation is expected to have major influence on the level of exporting and export behaviour. The degree of the firm's export orientation is based on the level of management's perception or attitudes toward environmental forces, export barriers, motivations, and their firms' capabilities and comparative advantages.

The content - analysis of the findings of the management export orientation studies are presented under the following subheadings:

- 1. Motivation for export marketing
- 2. Firm's Management and export marketing
- 3. Perceived obstacles to export marketing
- 4. Export marketing mix decisions
- 5. Firm's demographic and decision maker's individual characteristics

2.3.1.2.1 Motivation For Export Marketing

Several studies have tackled the motivation and the initiation of exporting simultaneously (e.g., Schlegelmilch 1983, Johnston & Czinkota 1982). Bilkey (1978) has argued that even though they are often related, it is very useful to maintain a distinction between them, because some firms have been pushed into exporting by an external change - agent, or simply by taking advantage of export opportunities as they rise without evident objectives in mind, while other firms were truly motivated and deliberately planned to become involved in export marketing activities as an alternative market strategy.

The literature relating to the motivation factors produced conflicting findings (see Table 2.2) as discussed here. The role of profit and its contributions to the firm as export motives are offered by several studies. Johnston & Czinkota (1982) found that the proactive motivation-profit advantage ranked first among all possible motivation factors for exporting. Ogram (1982), Wiedershiem - Paul et al. (1978), Weaver & Pak (1990) and Barker and Kaynak (1992) in separate studies reported a similar result. Cavusgil & Nevin (1981), Jatusripitak (1984), Schlegelmilch (1983), and Bourantas & Halikias (1991) found that management favourable expectations of, or their belief and value toward the contributions of exporting to their firms are the most important motivators for their firms to consider export marketing as an alternative business strategy. However, Hirsch (1971) in his study of 497 Danish, Dutch & Israel manufacturers could not establish such a relationship between the firm's likelihood to export and the export profit expectation. A similar result was also reported by Axinn (1985, 1988).

Bilkey (1978) hypothesized that the differences between these findings could be explained by variations in the experience of firms under investigation in respective of studies. He states:

"The percentage of sales exported by experienced firms tends to vary according to the perceived profitability of exporting but the initiation of experimental exporting seems to relate to non-profit considerations".

However, Bilkey (1982) later contradicts this hypothesis in findings which not only indicate a low correlation between the perceived

Motivations For Export Marketing

Motive Variables	Discussed By Author(s)
- Contribution to a firm's overall profit.	Sinai 1970; Simpson 1973; Hirsch 1971; Alexandrides 1971; Cavusgil 1976; Rabino 1980; Johnston & Czinkota 1982 Kolhede 1984; Axinn (1985, 1988); Weaver & Pak 1990; Barker & kaynak 1992; Kaynak 1992.
- Alleviation of a firm's excess capacity.	Sinai 1970; Kaynak and Kothari 1984 Pavord and Bogart 1975; Bilkey and Tesar 1977; Rabino 1980; Brooks & Rosson 1982; Ogram 1982; Johnston & Czinkota 1982; Kaynak 1992.
- Confidence in a firm's competitive advantage(s),	Snavely et al. 1964; Simmonds & Smith 1968; Cooper et al. 1970; Bilkey 1978 Brooks and Rosson 1982; Johnston & Czinkota 1982; Reid 1983; Cavusgil and Nevin 1981; Schlegelmilch 1983; Kolhede 1984; Karafakioglu 1986.
- Managerial aspiration for greater profit, sales growth, stability.	Simmonds & Smith 1968; Hirsch & Lev 1971; Cavusgil 1976; Hunt <i>et al.</i> 1967; Bilkey 1978; Rabino 1980; Czinkota & Johnston 1983; Kolhede 1984; Cavusgil & Nevin 1981; Cavusgil & Naor 1987; Schlegelmilch 1986; Ursic & Czinkota 1983; Weaver & Pak 1990; Moon & Lee 1990; Karafakioglu & Harcar 1990.
 To improve its market position (prestige). 	Cooper <i>et al</i> . 1970; Sinai 1970; Karafakioglu 1986; Rabino 1980.
- To offset adverse home market conditions.	Hunt <i>et al.</i> 1967; Cooper <i>et al.</i> 1970; Sinai 1970; Reid 1982; Sullivan and Bauerschmidt 1990.
 To avoid increased competition in home markets. 	Alexandrides 1971; Sinai 1970; Abdel-Malek 1978; Johnston & Czinkota 1982; Kolhede 1984.
- Management's belief about the benefits to be gained from the foreign trade.	Langston 1976; Bilkey & Tesar 1977; Roy & Simpson 1980; Reid 1980; Withey 1980; Bradley & Keogh 1981; Cavusgil 1984; Wiedersheim - Paul <i>et al.</i> 1978; Jatusripitak 1984; Kolhede 1984; Axinn (1985 1988); Sullivan and Bauerschmidt 1990.

- Management's expectations	Brasch 1981; Hirsch & Adar 1974
profitability and	Simpson & Kujawa 1974; Bilkey 1978
riskiness.	Czinkota & Johnston 1983; Cavusgil
an and information of the state	and Nevin 1981; Edmunds & Khoury
	1986; Schlegelmilch 1986; Bourantas
and there experies derived when the	& Halikias 1991.
relative profitability of exporting and the percentage of total sales exported ($\mathbb{R}^2 = .012$) but also a negative correlation between the perceived profitability of exporting and export experience (Partial correlation = - .361). The later result is remarkable in that at the same time export experience correlates positively with the percentage of sales exported (Partial correlation = +.416). This relationship can either be rationalised in that experienced exporters perceive less risk to exporting and are therefore, more willing to accept limited profits in order to increase export sales, or that exporting is indeed less profitable than new exporters tend to perceive.

In a review of literature related to the firms motives for exporting, Cavusgil (1976) notes:

"The studies fail to provide answers to two important questions, one is the relative importance of profit objectives and growth objectives, and the second is whether the passive exporter had any clear - cut objectives in the initial stage of export involvement other than short run objectives".

Conflicting results were also produced about the effect of the firm's excess production capacity as a reason for the firm to become involved in exporting. Brooks & Rosson (1982) concluded that a major motivation among Canadian manufacturing was the management's view that exporting provides a means of reducing (alleviating) a firm's excess production capacity. A similar finding was also put forward by Daniels & Goyburo (1976), Ogram (1982), and Kaynak (1992) in their separate studies. In contrast, the research subjects who participated in the studies conducted by Rabino (1980) and Johnston & Czinkota (1982) did not regard the utilization of excess production capacity as a major reason for small firms to become involved in exporting. They concluded that tax incentives were found to be a major drive for small firms to become involved in export marketing activities rather than the alleviation of excess production capacity.

Rabino (1980) and Abdel-Malek (1974), in separate studies, have examined the notion that the firm's management are motivated to peruse exporting by the prospect that international marketing involvement enhances a firm's reputation. Rabino (1980) refers to this exporting motivation as "Keeping up with the competition" and explains it as follows:

"Some industries are characterized by a high propensity to export. Hence some manufacturers are forced to export and maintain some level of exposure in international trade shows because their dealers or customers expect them to develop an international image".

However, no support was found in studies conducted by Johnston & Czinkota (1982) and Karafakioglu (1986).

The firms' desire for sales stability is considered as a motive factor for exporting. Wiedersheim - Paul *et al.* (1978) in a survey of Australian exporting revealed that the desire for sales stability with growth was the most frequently mentioned reason for exporting. They indicated that:

"... the avoidance of undue instability in its sales performances is related back to a basic goal of the firm's, i.e. security and survival. The more unpredictable the firm perceives variations in its sales performance to be, the more concerned it will be to find other sources of sales and growth. In order to insulate it from potential disturbances. By raising the risk and uncertainty of present operations, instability reduces the perceived risk and uncertainty of creating and marketing a new product, or selling internationally both of those forms of diversifications provided an insulating function ensuring that a firm while suffering a loss in one market, is less likely to experience a loss in all. Thus where the basic security of the firm is threatened by market fluctuations, a powerful reason for export may be developed".

A similar vein was also put forward by Hirsch (1971), Barker and Kaynak (1992) and Rabino (1980). In this respect Rabino notes:

"The diversification and market product is particularly important from a business strategy stand-point, involvement in export markets facilitating the introduction of the firm's products in a number of market, some of these products may be in a strong position relative to competitors and others may be in a weak position. Thus, the firms performance does not depend solely on one market and a few product lines but on the relatively greater significance when these markets do not confirm to the same explicable values as that of the US economy. Thus a down turn in the US economy could coincide with repaid growth of some export markets. This growth can effect the decline in sales in the domestic market".

Barker and Kaynak (1992) reported that search for sales stability through market diversification was found to be one of the most important factors initiating exporters.

However, researchers of the product life cycle theory, Wells (1972) and Vernon & Wells (1986) have argued that the international trade pattern is more a result of a "four - phase cycle" rather than a result of firms seeking to diversify their markets, i.e., the firms move into world markets as a result of a trade cycle. The main problem of the Wells & Vernon approach, is that it regards the United States as the originating point of all export marketing activities as discussed previously (see Section 2.1.3). Furthermore, their approach does not take into consideration the variance between countries, industries and firms.

Several studies, in particular, Cavusgil (1976), Bilkey (1978), Cavusgil and Nevin (1981) and Moon & Lee (1990) have highlighted managerial aspirations for profit, growth and market security as an indication of the extent of the firm's export involvement. Their findings suggested that the high aspiration for both profit and growth are related to exporting whereas low aspiration is related to non exporting. Higher aspiration for market security and growth have also been associated with non - exporting (e.g., Cavusgil and Naor 1987; Karafakioglu & Harcar 1990). Therefore, research suggests a positive relationship between the desire for profit and the level exporting while the desire for stability may be expected to be negatively associated with the level exporting.

2.3.1.2.2 Firm's Management and Export Marketing

The following studies have examined the effect of a firm's management characteristics such as the quality of management and organisation structure on its export behaviour. Doyle & Schommer (1976) found a tendency for non - exporting firms to believe that some one outside the firm should be responsible for proving that exporting would be successful for them. Simpson (1973) in a study of 50 exporting and 70 non - exporting firms supported the notion that the quality of management was the main reason behind reluctance to export. Bilkey and Tesar (1977) reported that the quality of management may vary directly with whether or not a firm initiates exporting.

In differentiating between exporting and non - exporting firms, Reith & Ryan (1981) reported that the exporting firm's management felt more confidence about handling export - related problems than did non - exporting firms. Furthermore, Cavusgil, Bilkey & Tesar (1977), Cavusgil (1984), Layton & Dunphy (1970), Schlegelmilch (1986), Christensen *et al.* (1987), Malekzadeh (1982) and Bourantas & Halikias (1991) have found that the management of exporting firms tended to be more strategic planning oriented and have a higher commitment to control, and that they allocated resources more systematically to export activities relative to non - exporting firms.

Other analysts have investigated the impact of management assessment of the firm's competitive advantage for exporting upon the firm's level of involvement in exporting. These advantages as described by Cavusgil (1976) are derived from the nature of the firm's product, market, technological orientation, research, and such things as technically superior products and technological level (Table 2.3).

The firm's possession of such advantages will influence its willingness to become involved in export marketing.

Table (2.3)

The Firm's Comparative Advantage(s)

Advantage(s)	Discussed By Authors(s)
- Product advantage (specific design feature; high quality product etc)	Khan 1978; McGuinness 1978; Bilkey 1978; Cunningham & Spigel 1971; Mayer & Flynn 1973; Wiedersheim - Paul <i>et al.</i> 1978; Cavusgil 1976; Daniels & Goyburo 1976; Beamish and Munro 1987; Kirpalani and Macintosh 1980; Kolhede 1984; Schlegelmilch 1986; Cavusgil & Naor 1987; Sriram <i>et al.</i> 1989; Louter <i>et al.</i> 1991; Da Rocha <i>et al.</i> 1990; Moon & Lee 1990; Bourantas & Halikias 1991.
- Level of technology	Hirsch 1971; Cavusgil 1976; McGuinness 1978; Reid 1983; Daniels and Goyburo 1976; Bilkey & Tesar 1977; Czinkota & Johnston 1981; Axinn 1985; Burton & Schlegelmilch 1987; Ogram 1982; Schlegelmilch 1986; Snavely <i>et al.</i> 1964; Cooper and Kleinschmidt 1985.
- Distribution advantage	Tookey 1964; Cunningham and Spigel 1971; McDougall & Stening 1975; Philpot 1975; Kizilbash and Maile 1977; Kleinschmidt 1982; Cavusgil and Naor 1987; Schlegelmilch 1986; Louter <i>et al.</i> 1991.
- Competitive price	Tesar 1977; Kirpalani and Macintosh 1980; Cavusgil & Naor 1987.
- Proximity to the foreign market	Welch & Wiedersheim - Paul 1978; Tesar 1977; Bilkey 1978; Schlegelmilch 1983; Sullivan & Bauerschmidt 1990; Barker & Kaynak 1992.
- Market area (geographic scope)	Wiedersheim-Paul <i>et al.</i> ; 1978; Snavely et al. 1964; Schlegelmilch 1983; BETRO 1977; Diamantopoulos and Inglis 1988; Johanson & Vahlne 1977; Axinn 1988.
- Sales volume	Tesar 1977; Cavusgil 1976; Kolhede 1984 Kleinschmidt 1982.
- Excess capacity	Cunningham & Spigel 1971; Simpson & kujawa 1974; Daniels & Goyburo 1976; Wiedersheim-Paul <i>et al</i> . 1978.
- Financial advantage	Abdel-Malek 1978; Bilkey & Tesar 1977; Cavusgil & Naor 1987; Malekzadeh and Nahavandi 1985; Cavusgil & Nevin 1981;

Table (2.4)

Characteristics	Discussed By Authors(s) Doyle & Schommer 1976; Simpson 1973 Bilkey & Tesar 1977; Reith & Ryan 1981; Brooks & Rosson 1982.	
- Quality of management		
- Management capability	Tesar (1975, 1977); Kolhede 1984; Simpson & Kujawa 1974; Schlegelmilch 1983.	
- Level of management	Daniels & Goyburo 1976.	
- Delegation of responsibility	Christensen <i>et al.</i> 1987; Snavely <i>et al.</i> 1964; Layton & Dunphy 1970; Schlegelmilch 1983; Daniels and	
a.e. 987 5.	Goyburo 1976.	
- Planning & control	Meidan 1974; Walters 1985; Daniels and Goyburo 1976; Layton and	
rodies frond that the masages	Dunphy 1970; Fenwick & Amine 1979 Malekzadeh 1982; Schlegelmilch 1983	
apobility and the risk sensiti	Tesar & Terleton 1982; Snavely <i>et al.</i> 1964; Christensen <i>et al.</i> 1987;	
bich the firs is willing to	Walters & Samiee 1990; Bourantas & Halikias 1991.	

Management & Organisational Structure Characteristics

Bilkey & Tesar (1977) found that management's perceptions of the firm's competitive advantages play a major role in the exploration of exporting. However, export studies produce conflicting results in respect of the many sources of the firm's comparative advantages. For example, the findings with regard to the technological level as source of the firm's comparative advantages have been mixed. While several studies showed a positive relationship between the level of technology and the firm's propensity to export (Cavusgil 1976, Cavusgil & Nevin 1981, Czinkota & Johnston 1981, Reid 1987, Bilkey & Tesar 1977, Schlegelmilch 1983, and Daniels & Goyburo 1976), others have not established any relationship (Axinn 1985, 1988 and Cavusgil & Naor 1987).

Tesar (1975), Kolhede (1984) and Bilkey & Tesar (1977) in separate studies found that the management's perception of the firm's export capability and the risk involved in exporting may effect the level to which the firm is willing to get involved in exporting activities. While the firm may tend to become involved for the reasons of market stability, the indications are that there is an element of risk-taking which is likely to determine the firm's export marketing propensity. Roux (1979) established a link between the manager international orientation and risk. In addition, from a survey of managers in 120 manufacturing firms, Simpson & Kujawa (1974) concluded that firms which developed into exporters were less fearful of the risk involved in doing so than the firms which did not become exporters. Furthermore, portfolio theory suggest that an exporting firm probably faces less total market risk than a non - exporting firm, because of its market diversification (Hirsch & Lev 1971). Hirsch (1971) in a study of Danish, Dutch and Israel firms concluded that foreign entry is more hazardous than domestic selling. Tesar (1975) found that the

"light exporters" in his sample perceived more risk from exporting than "heavy exporters".

In contradiction to the above findings, Brooks & Rosson (1982) in their study of Canadian manufacturers concluded that the exporters (managers) do not perceive their quality of management as being significantly stronger than non - exporters (managers). Bilkey (1978) has attributed the contradictory results of these studies to the measurements which have been used for this purpose which are highlighted as follows: (1) In the peer evaluation of the efficient way in which firms best perform product planning, advertising, research and sales administration function. For example, Perkett (1963) in his study of British Columbia manufacturing firms found that exporting firms tended to be evaluated more highly than nonexporting firms, (2) The evaluation of the quality of management as rated by their own management; for example, Doyle & Schommer (1976) reported that exporters tended to evaluate their management as more aggressive than did the non - exporters (3) Comparing managers (a) attitude and activities, & (b) the firm's function and organisational structure with accepted good management practice (e.g., Tookey 1964, Cunningham & Spigel 1971, Wiedersheim-Paul et al. 1978, Tesar 1975, and Bilkey & Tesar 1977).

With regard to the firm's organisational structure characteristics, (see Table 2.4) several researchers have found that exporting firms relative to non - exporting firms are more likely to delegate more responsibility (Christensen *et al.* 1987, Schlegelmilch 1983, Daniels and Goyburo 1976, Snavely *et al.* 1964, and Layton & Dunphy 1970), having a higher commitment to planning and control (Malekzedeh 1982, Schlegelmilch 1983, Cavusgil 1976 and Beamish & Munro 1986, 1987), have a wider geographic scope (Wiedersheim-Paul *et al.* 1978, Snavely

et al. 1964 and Axinn 1988), are more likely to have more than one plant (Snavely *et al.* 1964), have a higher utilization of production capacity (Ogram 1982), and are more likely to have a formal quality control (Da Rocha *et al.* 1990).

Daniels & Goyburo (1976) found that the amount of organisational control is significantly higher and more decentralised in exporting firms than non - exporting firms. Meidan (1979) suggests that the form of control must be suitable for export market and depends largely on the organisational structure of the firm. However, Tesar & Terleton (1982) found that even companies classified as aggressive exporters tended not to plan their exporting activity. Therefore, Reid (1982) stresses that the function specialisation does not correlate with export behaviour.

2.3.1.2.3 Perceived Obstacles to Export Marketing

Researchers have attempted to identify various obstacles which inhibit exporting by manufacturing firms, the rationale being that a government could stimulate exporting by removing these barriers. Several studies have tended to compare between exporters and non exporters with regard to their managers' perceptions of the various obstacles to exporting. However, only some of researchers have tried to answer the question whether such perceptions would influence the firm's level of exporting (e.g., Weekly & Bardi (1975), Axinn (1985, 1988) and Burton & Schlegelmilch (1987)).

The results are conflicting, while some studies have found that non - exporting firms perceived significantly more serious obstacles than exporting firms (Bilkey 1970, Simpson 1973, Alexandrides 1971, Rao & Weinrauch 1974, Tesar 1975, Bilkey & Tesar 1977, Schlegelmilch 1983, Kolhede 1984, Sharkey *et al.* 1989, Keng & Jiuan 1989, Tseng & Yu 1991

Table (2.5)

Barriers to Export Marketing

Discussed by Author(s)
970; Groke & Kreidle 1967; 1971; Alexandrides 1971; & Bardi 1975; Pavord & Bogart einrauch 1973; Tesar 1975;
1 1976; Ogram 1982; Perkett abino 1980; Brooks & Rosson xinn 1988; Yang 1988; Kaynak Yeaver & Pak 1990: Korth 1991;
Yu 1991; Verhoeven 1988; & Walters 1990a; Yang <i>et al.</i> Well <i>et al.</i> 1992; Keng & Jiuan
1968; Perkett 1963; Rabino zinkota & Johnston 1983; & Bardi 1975; Tesar 1975; & Bogart 1975.
rides 1971; Groke & Kreidle gram 1982; Rabino 1980; Tarleton 1982; Kaynak & 1984; Axinn 1988; Bilkey Colhede 1984.
1963; Weekly & Bardi 1975; & Tesar 1977; Daniels and 1976; Czinkota & Johnston
dmunds & Khoury 1986; Pavord t 1975; Groke & kreidle 1967; & Wilkinson 1986; Brooks & 1982; Bauershmidt <i>et al.</i> 1985 rides 1971; Axinn 1985, 1988; <i>al.</i> 1992; Huu-Phuong & Soo- 989; Samiee & Walters 1990a; ren 1988.
rides 1971; Weinrauch 1973; & Bardi 1975; Pavord & Bogart roke & Kredile, 1967; Cheong 1986; Christensen <i>et al.</i> arafakioglu 1986; Pinney 1968 & Wilkinson 1986; Barker & 1992; Weaver & Pak 1990; <i>et al.</i> 1990; Tseng & Yu 1991;

 Cost & availability of of transport

- Low product quality, high production cost
 & unmarketable product
- Government restriction & regulations

 Lack of finance for exporting & export incentives

- Foreign competition
 & Foreign trade
 restrictions
- Marketing problems
 - Maintaining control over middleman
 - 2. Identification of potential markets
 - 3. Packaging adaptation
 - 4. Difficulty in meeting product specification
 - 5. Export transaction
 - 6. Promoting products overseas
 - 7. Provision of follow up services
 - 8. Pricing for foreign markets

Pinney 1968; Perkett 1963; Groke & Kreidle 1967; McDougall & Stening 1975; Kaynak & Kothari 1984, Weekly & Bardi 1975; Yang *et al.* 1992, Barker & Kaynak 1992, Kaynak 1992; Gripsrud 1990; Verhoeven 1988; Samiee & Walters 1990a; Huu-Phuong & Soo-Jiuan 1989.

Tesar 1975; Karafakioglu 1986; Chreitensen *et al.* 1987; Daniels & Goyburo 1976; Keng & Jiuan 1989.

Bilkey 1978; Ogram 1982; Rabino 1980; Alexandrides 1971; Groke and Kreidle 1967; Pavord & Bogart 1975 Weekly & Bardi 1975; Kolhede 1984; Kaynak & Kothari 1984; Weaver & Pak 1990; Huu-Phuong & Soo-Jiuan 1989; Kaynak 1992; Sullivan & Bauerschmidt 1989.

Bilkey 1978; Bilkey & Tesar 1977; Ogram 1982; Rabino 1980; Yaprak 1985; Alexandrides 1971; Kedia and Chhokar 1985; Axinn 1985: Groke and Kredile 1967; Barker & kaynak 1992 Sullivan & Bauerschmidt 1989; Hook & Czinkota 1989; Samiee & Walters 1990a; Korth 1991; Keng & Jiuan 1989 Kathawala *et al.* 1989.

Rabino 1980; Alexandrides 1971; Dichtl *et al.* 1984; Ogram 1982; Groke and kredile 1967; Kedia and Chhokar 1985; Kolhede 1984; Kaynak & Kothari 1984; Hook & Czinkota 1989; Sullivan and Bauerschmidt 1989; Weaver & Pak 1990.

Weekly and Bardi 1975; Czinkota and Johnston 1983; Kedia & Chhokar 1985; Pavord & Bogart 1975; Ogram 1982; Weinrauch & Rao 1974; Keng & Jiuan 1989; Hook & Czinkota (1988, 1989); Kathawala *et al.* 1989; Howard and Borgia 1990. and Barker & Kaynak 1992), others found either no significant relationship, or an inverse relationship that non-exporters perceived fewer obstacles than did exporting firms (Doyle & Schommer 1976 and Bilkey 1978). Kolhede (1984), in her study of export orientation of small manufacturing firms in the United States, reported that exporters' assessment of the obstacles related to exporting are insignificant determinants of their degree of export orientation. Kolhede notes:

"It appears that although the non - exporter might cite the various obstacles surveyed in research as justifiable reasons for his refusal to export. Such reasons could be offered merely as rationalization or excuse and not truly underlies his negative export orientation".

Bilkey (1978) concluded that these seemingly contradictory findings can be explained by differences in the export development of the firms in the various studies. Furthermore, non - exporters (which Bilkey defined as firms in export stage one or two, see section 2.3.3) have no basis for knowing their barriers to exporting, and they, therefore, tend to list fewer obstacles to exporting than do exporting firms, whereas it could be different to those firms which have explored the feasibility of exporting (those firms in stage three).

This suggests that the means of overcoming such obstacles, from a policy viewpoint, will need to vary according to a firm's level of exporting experience. Bilkey (1978) then concluded that the government promotion programme which aims to remove these obstacles should be directed at the experienced exporter rather than at the non exporter. However, it can be argued that Bilkey's explanation is insufficient, because it does not necessarily follow that two firms at the same stage of export development will perceive similar problems. Other factors such as management skills, organisation resources, marketing orientation should also be included.

Howard & Borgia (1990), Samiee & Walters (1990a) and Barker & Kaynak (1992) indicated that there are some significant differences in the pattern of problems encountered by small and large firms. Barker and Kaynak (1992) reported that the lack of trained personnel for export operation was a more critical problem for firms with less than ten employees while it was not perceived as an impediment by those respondents who had more than fifty employees. Samiee & Walters's study (1990a) revealed that while larger firms are more concerned with foreign market information, the small firms focus is more on operational problem such as export finance. However, Verhoeven (1988) reported that the most serious problem encountered by small-medium sized enterprises in Netherlands was the lack of knowledge about foreign markets.

Within the various classes of barriers (Table 2.5), it was found that the most important problems inhibiting firms in developed countries are price intensive competition in foreign markets, and a lack of knowledge about foreign market opportunities. In a comparative analysis of the export problems encountered by Japanese & German manufacturing firms, Dichtl *et al.* (1986) concluded that foreign competition, pricing, and personal expertise ranked as the most important export problems in both nations. Also Alexandrides (1971), in his study of the Georgia firms in the United States, concluded that foreign competition was significantly the most important problem encountered by these firms, while government restriction was found to be insignificant in inhibiting exporting. A similar result was also reported by Sullivan and Bauerschmidt (1989).

Furthermore, Korth (1991) reported that a lack of knowledge of foreign opportunities was considered one of the major reasons for the reluctance of American firms to enter international markets. Hook &

Czinkota (1988, 1989) also indicated that the difficulty of locating foreign markets was perceived as the most important problem by the current and potential exporters of Hawaiian firms.

Therefore, it can be concluded that such problems are mainly related to the firm's internal environmental characteristics (e.g., the comparative advantage) rather than to its external environmental forces. Although these problems cannot be underestimated to have influenced manufacturing firms, which are operating in developing countries, it can, however, be argued that the firm's external barriers are expected to be highly rated by the decision-maker, i.e., problems such as government restrictions, inadequate infrastructure, and poor transportation are expected to be the most ranked problems perceived by Jordanian firms.

2.3.1.2.4 Export Marketing Mix Decisions

There is a conspicuous lack of research studies examining the export marketing mix decisions within the context of the firm's export behaviour. Although a few studies have been undertaken to investigate the association between the firm's export behaviour and their export marketing decisions (elements), they have shown relatively definitive findings.

Functional or marketing advantages have been found to differentiate between export prone and non - export prone firms. For example, Layton and Dunphy (1970) found that exporting firms are more likely promotion awareness, use lower price premiums and consider after sales more important than non - exporting firms. In a similar vein reported by Schlegelmilch (1983), it was found that exporters are more confident in their marketing skills than non - exporters and they also perceive themselves as employing marketing techniques more frequently.

Kaynak and Kothari in their study (1984) found that managing of the marketing mix is a major determinant in the firm's commitment for international marketing.

Therefore, the degree of the firm's needs to adapt or manage its marketing mix strategies to foreign market requirements is expected to have influenced the firm's probability to export. However, Kolhede (1984) reported that the degree of management willingness to modify export marketing elements has a negative association with the export orientation of smaller (exporting and non - exporting) manufacturing firms in the United States.

The findings related to export marketing mix decisions are discussed below:

(1) Product Policy

Product policy is commonly considered to be the most vital element of marketing mix decisions, regardless of whether the firms operate in the domestic market, or are involved in foreign markets operations. In this respect Terpstra (1978) notes:

"International product policy is the cornerstone around which all other international marketing activities must be designed".

The most important decisions concerning the product policy is whether, and to what extent products should be modified (adapted) to meet foreign consumer requirements. The product design for the foreign market is very important in winning export business, as Thomas (1969) notes:

"The key success in product planning is to adapt the product line to the tastes and economic characteristics of a particular foreign market."

The following studies produce conflicting findings with regard to product adaptation policies. Studies by Weinrauch (1973) and Sinai (1970) reported that the physical product modification was not

Table (2.6)

Export Marketing Mix Studies

Elements	Discussed by Author(s)	
- Product policies (adaptation vs. standardization)	Weinrauch & Rao 1974; Mayer & Flynn 1973; Meidan 1974; Schelgelmilch 1983; Cavusgil 1984; Christensen <i>et</i> <i>al.</i> 1987; Edmunds & Khoury 1986; Khan 1975; Cooper & Kleinschmidt 1985; Diamantopoulos and Inglis 1988; Sullivan & Baurschmidt 1990; Buatsi 1986; Seifert and Ford 1989; Ortiz- Buonafina 1990; Walters & Samiee 1990 Samiee & Walters 1990a; Koh 1991.	
- Distribution policies (distribution channel)	Meidan 1974; Khan, 1975; Cavusgil 1976; Pinney 1970; Brady & Bearden 1979; Seely & Iglarsh 1981; Rosson & Ford 1980; Kirpalani & Macintosh 1980; Hunt <i>et al.</i> 1967; O'Rourke 1985; Diamantopoulos & Inglis 1988; Madsen 1989; Fenwick & Amine 1979.	
- Promotion policies	Tookey 1964; Weinrauch & Rao, 1974 Meidan 1974; McFarlane 1978; Garnier 1982; Cavusgil 1983; Brasch 1978; Kirpalani & Macintosh 1980; Schlegelmilch 1983.	
- Prices policies	Muir 1977; McDougall & Stening 1975; Pircey 1983; Beamish & Munro 1987; Khan 1975; Cavusgil 1983; Meidan 1974 Schlegemlich 1983; Malekzadeh 1982; Christensen <i>et al.</i> 1987; Seifert & Ford 1989; Buatsi 1986; Samules <i>et al.</i> 1992; Koh 1991.	
- Other marketing functions	Martation colletes.	
1. Export market research	Meidan 1974; McDougall & Stening 1975 Khan 1978; Tookey 1964; Philpot 1975; Schlegelmilch 1986; Madsen 1989.	
2. Use of external information	Daniels & Goyburo 1976; Mayer and Flynn 1973; Philpot 1975; Reid 1983; Cavusgil 1983.	
3. Export marketing dep.	Hunt <i>et al.</i> 1968; Hirsch 1971; Meidan 1974; Philpot 1975; Fenwick & Amine 1979; McDougall & Stening 1975; Bourantas & Halikias 1991.	

regarded as significantly important among the sample they examined. Christensen *et al.* (1987) also reported that successful exporting was more likely to be the manufacture of standard products (products that were sold in the domestic market, not designed and manufactured only to fill foreign customer specification). Seifert and Ford's study (1989) indicated that neither the size of the firm nor its experience in exporting would influence its product policies. He concluded that firms do not believe that product modifications are necessary. Koh's (1991) study revealed a similar result.

However, Meidan (1974) concluded that firms which are export marketing oriented tend to specialize their products for certain overseas markets. Cooper and Kleinschmidt (1985) & Walters and Samiee (1990) in separate studies found that adapting products for specific markets was correlated with a high growth of export sales and profitability. Samiee & Walters (1990a) reported that the differences between regular exporters and sporadic exporters are more likely to be due to the suitability and adaptation ability of products for international market and export marketing programmes. A similar result was reported by Ortiz-Buonafina (1990). His study indicated that the larger the firm is, the higher likelihood that the firm would be engaged in product and market adaptation policies.

These contradictory results can be attributed to many factors, such as: (1) either that export operations are insignificant to these firms under investigation, as their aim might be only to dispose of surplus products which are not sold in the domestic market, or (2) it might relate to the type of the firms in the countries under investigation (i.e., the nature of their products), because industrial products require a somewhat different product policy in comparison with consumer products.

(2) Pricing Policy

Export policy plays an important role in connection with other elements of marketing. In discussing the pricing methods used by exporting firms Cavusgil (1976) notes:

"Export prices seem to be determined largely by cost consideration rather than demand analysis. Most firms build up the final price via works direct cost and overhead and profit margin".

In a somewhat different statement to Cavusgil's, Khan (1978) found that although 35% of the firms base their prices on cost, a higher percentage, i.e., 45% price their products as high as the market can tolerate.

The decision of adapting pricing policies to foreign markets is very likely to effect the firm's export involvement. However, studies in this respect produced contradictory results. Weinrauch & Rao (1974) in their study of Arkansas (USA) export manufacturing firms, to determine the extent to which marketing variables would have to be modified for exporting, found that pricing and credit policies required the least modification in exporting. Samules *et al.* (1992) reported that smaller firms are not willing to alter their export pricing policies. Seifert & Ford's study (1989) revealed that exporters appear to use standardised pricing (i.e., the export prices were the same as the domestic prices).

However, Schlegelmilch (1983) in discriminating between exporting and non - exporting firms with regard to the pricing policies such as differentiation prices, reported that exporters are more inclined to differentiate the price by region than non - exporters. Kirpalani & Macintosh (1980) in their study also concluded that pricing was significantly associated with the firm's export performance. A similar result was also reported by Koh (1991).

(3) Promotion Policies

Traditionally, promotion policy is divided into advertising, sales promotion, and public relations. With regard to these methods and their degree of importance in export marketing, the export studies revealed mixed findings as we discuss below. While Brasch (1981) confirmed the hypothesis that more profitable firms are more prone to show an interest in international trade shows (this statement has to be treated carefully as it is based on a very small database, and reflect the view of the firms rather than factual statements on their profitably). Cavusgil (1984) and Weinrauch & Rao (1974) in separate studies found that promotion was not at all important. In this respect Cavusgil states:

"promotion appears to be the least significant element of the marketing mix among the firms surveyed"

The justification for this statement was provided by Sinai's study (1970) in which he found that a large proportion of firms surveyed leave promotion to their agent, it is conceivable that firms which indirectly export underestimate the impact of promotion. The influence of advertising on the export ratio is hardly discussed in the literature; one little contribution stems from NEDO (1977) that "an expected association between export sales and the advertising budget allocation to the export market is strongly confirmed", a similar vein was also put forward by Schlegelmilch (1986), in which he reported that exporters perceive advertising techniques to be more important than non - exporters and assessed that successful exporters are more used to staff promotion and trade fairs than less successful exporters.

(4) Distribution Policies

Although there is a consensus in the literature with regard to the importance of distribution policy, the findings of the value of it's specific distribution policy are conflicting. However, it is not surprising, since the selection of distribution policy is determined for the firm by a number of factors, such as nature of product, resources availability, experiences of the firm, scope and complexity of distribution functions, availability of inadequate distributors in foreign markets, etc. Therefore, no distribution system can be right or wrong, because it always has to be related to the specific situation of a firm.

In the selection of the distribution policy and its impact on the level of exporting, Pinney (1971) observed that the use of a direct distribution method seems to become more desirable with an increase in foreign trade involvement. In this respect Cavusgil (1976) notes:

"When foreign sales account for only a marginal proportion of the firm's total business, firms tend to use an indirect method of distribution, once the volume of export sales reach a certain level, and the foreign market becomes more important, the use of direct distribution becomes increasingly desirable".

The BETRO'S report (BETRO 1977) points out that in the firms surveyed, direct selling was significantly associated with the higher export ratio. Meidan (1974) observed that the most effective channel of distribution marketing oriented firms is selling directly through the overseas importer, while the export oriented firms consider the most effective channel of distribution to be selling through an exclusive agent. Koh's (1991) study suggested that exporting directly to the final end-users is the most profitable channel for industrial goods. However, Tookey (1964) highlighted the point that most firms did not restrict themselves to one market for selling. A little

support was provided by Cavusgil and Naor (1987) to Tookey's statement. However, Kirpalani and Macintosh (1980) and Madsen (1989) in separate studies concluded that distribution did not have any significant relationship with sales volume.

It can be concluded that the degree to which the firm needs to adapt or the willingness of commitment to its marketing mix policies to foreign market environment conditions is expected to have influenced its level of exporting. In developing countries it can be assumed that the higher the degree that the firm needs to adapt its business operations (i.e., marketing mix policies) to meet the foreign market conditions, the lower its level of exporting will be. It might be that firms in developing countries tend to be less marketing orientated than firms in developed countries. As Cateora & Hess (1987) noted: "Marketing orientation is neglected in developing countries". Also, Kaynak (1982) states:

"in the prevailing atmosphere of staff international competition, it is extremely difficult for a developing country to secure an overseas order unless it makes a planned and systematic effort in the direction of product adaptation. No developing country can think of pursuing a policy of product expansion as that done by some multinational corporations".

2.3.1.2.5 Firm's Demographic and Decision - Maker's Individual Characteristics.

Several studies have investigated the effect of the managers and the firm's demographic characteristics upon export behaviour and the level of exporting. The findings of these studies are discussed below:

(1) Decision - Maker's Profile and Export Marketing

There is a broad agreement in the literature on export behaviour that the decision - makers' attitudes and characteristics are critical not only with regard to export initiation, but also significantly to the level of exporting (e.g., Cavusgil 1976 and Schlegelmilch 1983).

Table (2.7)

The Decision - Maker's Characteristics

Characteristics	Discussed By Author(s)
- Level of education	Layton & Dunphy 1970; Mayer & Flynn 1973; Daniels & Goyburo 1976; Reith &
	Ryan 1981; McConnell 1979; Simpson & Kujawa 1974; Cavusgil and Naor 1987;
	Schlegelmilch 1983; Axinn 1985; Dichtl et al. 1984; Holzmuller & Kasper 1990;
	Moon & Lee 1990; Keng & Jiuan 1989; Koh 1991.
	sta Danta da Parda dalata a Rojana Sara
- Age	Layton and Dunphy 1970; Pinney 1968;
	Reid 1983; McConnell 1979; Dichtl et al
	1984; Cheong & Chong 1988; Cavusgil &
	Naor 1987; Moon & Lee 1990; Karafakioglu & Harcar 1990
	Adialakiobid a halcal 1990.
- Proficiency in	Simpson 1973: Langston & Teas 1976:
foreign language	Reid 1981: Cavusgil and Naor 1987;
	Schlegelmilch 1983; Dichtl et al. (1984
	& 1990); Cavusgil & Naor 1987;
	Karafakioglu & Harcar 1990.
	or (1.76) and Primer (1903) (spected the
 Experience 	Daniels & Goyburo 1976; Mayer & Flynn
	1973; Reid 1981; Garnier 1982; 1974
	Bilkey 1978; Wiedersheim - Paul et al.
	1978; Dichtl <i>et al.</i> 1984; Khan 1975
	et al 1990. Moon & Lee 1990.
	Karafakioglu & Harcar 1990.

Various analysts have examined the effect of the decision - maker's characteristics such as the level of education, age, proficiency in foreign languages, and experience on the adoption of export activity. However, while some studies show some consistent findings with regard to the relationship between export propensity and the level of education of decision - makers (Mintzberg 1973, Layton & Dunphy 1970, Simpson & Kujawa 1974, Wiedersheim - Paul *et al.* 1978, Langston & Teas 1976, Reid 1983, Mayer & Flynn 1973, Keng & Jiuan 1989, Holzmuller & Kasper 1990, 1991 and Koh 1991), other did not establish such a relationship (Ogram 1982, Garnier 1974, Daniels & Goyburo 1976, Cavusgil & Naor 1987, Axinn 1985, Reith & Ryan 1981, and Moon and Lee 1990). Dichtl *et al.* (1984) reported that the level of education produces very little discriminating power with regard to export experienced and non - experienced managers.

Studies about the decision-maker's age also produced conflicting findings. While Layton & Dunphy (1970) and Pinney (1968) reported that younger managers are more likely to engage their firms in exporting than older ones, Cheong & Chong (1988) in their study of the furniture industry and the electronic industry reported that managers of furniture exporting firms were older than their counterparts in non exporting firms. Moon & Lee's study (1990) showed that the manager's personal characteristics (i.e., age, education and experience) were not related to export stage development. Bourantas & Halikias's study (1991) also produced a similar result. Their study indicated that the decision-maker's characteristics (i.e., education, knowledge of language and experience) were not found to be important in discriminating between systematic exporting firms and non - systematic exporting firms.

Other mixed results are also revealed regarding the effect of the decision manager's language skills on the level of exporting. While Schlegelmilch (1986), Khan (1978), Langston (1976), Tesar (1975), Mayer & Flynn (1973) and Karafakioglu & Harcar (1990) show a positive relationship, Simmonds and Smith (1968), Reid (1980), Daniels & Goyburo (1976) and Cavusgil & Naor (1987) have attached either little or no relationship to such skills.

In the area of the decision-maker's experience studies also revealed contradictory findings. While Langston & Teas (1976), Axinn (1985), Karafakioglu & Harcar 1990 and Da Rocha *et al.* (1990) found that the amount of such experience would bear a significant positive relationship with the degree of the manager's orientation and the firm's export behaviour, Ogram (1982) did not find experience to be strongly underlying the determinants of export behaviour among small and medium sized exporting and non - exporting firms of Georgia.

(2) Firm's Demographic Characteristics and Export Marketing

Not only the decision - maker's characteristics have been studied but also the firm's demographic characteristics, such as size, ownership, and experience. The firm's size structure has received great attention in the research literature. The findings of studies about the relationship between the firm's size and its propensity to export have been mixed and inconclusive. While some studies found no relationship (Snavely *et al.* 1964, Doyle & Schommer 1976, Bilkey & Tesar 1977, Johnston & Czinkota 1982, Kirpalani & Macintosh 1980, Axinn 1985 & 1988, Keng & Jiuan 1989, Ali & Swiercz 1991, Samiee & Walters 1990a & 1991, Bourantas & Halikias 1991 and Liouville 1992), others have revealed a positive relationship (Tookey 1964, Simmonds and Smith 1968, Reid 1980, Karafakioglu & Harcar 1990, and Ortiz-

Buonafina 1990). Tookey (1964) found that larger firms tend to be more successful in exporting. Simmonds and Smith (1968) reported that firms with less than 300 employees demonstrate a lack of interest in exporting, and Reid (1980, 1983 & 1985) concluded that the size of a firm predominantly effects the export entry into new foreign markets. However, Gripsrud's study (1990) indicated that based upon the experience of the firm, smaller firms tended to have a more positive attitude toward exporting than the larger ones. He explained that his finding might be related to the different aspirations and goals of management of the firms under investigation.

Louter *et al.* (1991) indicated that the export marketing practice and strategies would be influenced by the size of the firm. His study revealed that medium - sized exporters approach exporting with a more systematic approach and segmented their markets more often than small firms. Culpan (1989) also concluded that firms demonstrate behavioural differentials in export performance according to their sizes. However, Czinkota and Johnston (1983, 1985) indicated that the size of the firm does not make a difference in exporting practice.

Bilkey (1978) suggests a possible inter - correlation of the firm size with the quality of management and process, that size could vary directly with a firm's propensity to export. In another study in 1982, he found that firms which had obtained their own initial export orders were much larger than firms where the initial export order was unsolicited.

Sharrard (1975) postulated that company size to a great extent dictates the size of business opportunities that can be undertaken. For example, small firms find that the risk is too great and capital demands too high to undertake very large research projects.

Table (2.8)

The	Firm's	Demographic	Characteristics
-----	--------	-------------	-----------------

Characteristics	Discussed By Author(s)
Size	Porkett 1962, Snovely et al. 1964.
- 2126	Tookey 1964; Simmonds and Smith 1968;
	Hirsch 1971; Hirsch & Adar 1974; Abdel- Malek 1974; Daniels & Goyburo 1976;
	Layton & Dunphy 1970; Khan (1975, 1978) Withey 1980; McGuinness 1978; Cavusgil
	1976; 1984; Bilkey & Tesar 1977; Doyle and Schommer 1976; O'Rouke 1985;
	Schlegelmilch 1983; Liouville 1992; Gripsrud 1990: Samiee & Walters 1991 &
	1990b; Louter <i>et al.</i> 1991; Culpan 1989; Axinn (1985, 1988); Diamantopoulos and
	Inglis 1988; Ortiz-Buonafina 1990;
	Moon & Lee 1990; Ali & Swiercz 1991;
	Bourantas & Halikias 1991;
	Karafakioglu & Harcar (1990).
Firenience	Pilkey 1079, McDeucell & Steping 1975,
- Experience	Ursic & Czinkota 1984; Kirpalani and Macintosh 1980; Daniels & Goyburo
	1976; Ogram 1982; Reid 1982; Tseng & Yu 1991; Louter <i>et al.</i> 1991; Samiee &
	& Walters 1991; Gripsrud 1990; Da Rocha et al. 1990: Moon & Lee 1990: Keng &
	Jiuan 1989; Ali & Swiercz 1991.
- Ownership	Daniels & Guyburo 1976; Abdel-Malek 1974; McDougall & Stening 1975.

ent existent (Artain 1976, Quentur 1974, Mysic & densitie 1983 Synce (1982, and Densitie & Goodenne (1976), Mirgainer & Herrintoph (1983) constitut a monthly efformation age of the first and its equal constitut they found that new First tended in the artains conconstitut the even structs through the poster. The estents conequitable of first electrony the base of the second struct reported by Topic & Grants (1993) that the poster, first endored arger performance of chair ente and have a some formation attribute arger performance of chair ente and have a some formation attribute const the second of their ente and have a some formation attribute arger performance of their enter and have a some formation attribute const exporting these alfers of the rest formation is attribute attribute Withey (1980) reported that the more employees, the greater the firm's probability of engaging in exporting, even among small firms with less than 200 employees. Further support for the positive relationship was provided by Hirsch & Adar (1974), Khan (1978), and Schlegelmilch (1983).

Studies by Hirsch (1971) and Cavusgil (1976) attempted to reconcile these findings by suggesting that a correlation between size and export propensity occurs only within a certain range of a firm's size, which excludes only very small firms (where no correlation between exporting and size is observed). Cavusgil (1984) later argued that firm size is an unreliable indicator of export activity, he states:

"Size might be seen as an approximation for associated advantages of scale including bargaining power and the capacity to hire better trained managers than suggesting a comparative advantage based on lower costs and subsequently lower price".

In respect of the firm's age which is expressed in terms of the firm's length of time in business or exporting, findings are also mixed. While some studies reported that the firms' export activity is positively related to the age of a firm (Snavely *et al.* 1964, Alexandrides 1971, Wiedersheim - Paul *et al.* 1978, McDougall & Stening 1975 and Ali & Swiercz 1991), other studies concluded that firms tend to be more interested in a foreign market at an earlier stage of their existence (Brasch 1978, Garnier 1974, Ursic & Czinkota 1983, Ogram 1982, and Daniels & Goyburo 1976). Kirpalani & Macintosh (1980) reported a negative association between age of the firm and its export activity, they found that new firms tended to be more anxious to export than seek growth through the market. In a similar vein reported by Ursic & Czinkota (1983) that the younger firms export a larger percentage of their sale and have a more favourable attitude about exporting than older ones. Tseng & Yu (1991) also reported that

exporters have been in business for a significantly shorter period of time than non - exporters. However, Samiee & Walters's study (1991) indicated that the regular and sporadic exporters are not different in terms of the number of years in business. Keng & Jiuan's study (1989) does not support the contention that younger firms are more likely to be interested in exporting, either.

Cavusgil & Naor (1987) highlighted an inverse relationship between the firm's age and export profitability they stated:

"Although the difference is not significant, it is conceivable that non - exporters are more optimistic about export profitability due to their lack of experience and unfamiliarity with their activity, for example many associated costs of export are not well known to the non- exporter".

Stuides about the effect of the ownership factor on the firm's export behaviour are also produced mixed results. While Abdel-Malek (1974) found little or no significant differences between the firms domestically owned and the firms foreign owned in terms of their export orientation, Daniels & Goyburo (1976) were more clear in their conclusion in which they reported that the term ownership should be interpreted in the light of the two kinds of ownership; the first is divided according to the domestic legal structure which includes, for example, individual, cooperative, cooperation and state, and the second is divided according to which the firm belongs either to domestic owned or to overseas owned. Regarding the former they found no significant relationship, but it was significant with regard to the second division. They reported that foreign owned firms are more likely to have an interest in exporting rather than those owned domestically. A similar vein was reported by Layton and Dunphy (1970) who concluded that exporters are more likely to be firms owned overseas.

2.3.2 Studies Relating to the Firm's External Environment

The concept of environment with its components and relevant dimensions has not been well specified in the literature (Lawrence & Lorsch 1967 and Thompson 1967). One of the shortcomings of much of the theoretical and empirical research on organisational environment has been the failure clearly to conceptualize organisational environment, or the elements comprising it (Duncan 1972).

However, various authors have implied that the businesses environment is composed of physical, cultural and technological components (Homans 1950), and that the organisational environment can be broken into external and internal environments (Duncan 1972). While the internal environment is viewed as consisting of all the relevant physical and nominal factors within the boundaries of the organisation that influence the decision - making behaviour of individuals in the organisation, the external environment is viewed as consisting of those relevant physical and non - physical factors outside the boundaries of the organisation or specific decision unit, that are into consideration. The multifacetd nature taken directly of constituent groups in the external organisational environment has been presumed by a number of writers (e.g., Dill 1958 and Estafen 1971) to include; competitors, suppliers, consumers, stockholders, governments, regulatory agencies and the labour market.

The organisational behaviour literature reveals that the firm's organisation structure is largely directed and shaped by its environment. However the degree of the firm's adaptation to its environment is based upon its management's perception of these environmental forces. In other words, the firm's behaviour and objectives are very likely to be determined through the interaction of

its internal operations with factors existing in its external environment. In this respect Yasai-Ardekani (1986) writes:

"Management's perception of the magnitudes and direction of social, political & economic forces influences organisational structure, such forces should manifest themselves as constraints, threats and opportunities from the organisational perspective".

Aldrich's concept of environment capability states that organisation seeks out environments that permit organisational growth and stability (Aldrich 1979). Such growth and stability may allow the organisation to generate slack resources (Cyert & March 1963) which can in turn provide a buffer for the organisation during periods of relative scarcity. Aldrich clearly articulated the view that the resources required for organisational survival are the most relevant in defining organisational environment. He stated that:

"Environment affects organisation through the process of making available or withholding resources and organisational firms can be ranked in terms of their efficiency in obtaining resources".

In the business - policy literature, Douglas (1975) demonstrated that the interaction between a firm and its environment is related to its performance. This finding was supported by the hypothesis that the performance of the firm is primarily determined by the net result of all managerial actions, or interaction within the firm which is reflected by some transactions across the environmental interface. The structure - behaviour - performance model of industrial economists also indicates that the firm's environment, influences its behaviour and the directions of its business strategies (e.g., Koh 1991).

In discussing the effect of the environmental factors on the marketing structure, business studies produce conflicting results. Kaynak & Hudanah (1987) and Kaynak (1982) argue that the marketing structure like all other forms of human behaviour is linked

inseparably to the environment in which it takes place, variables such as socio economic traditions, culture, population, government, suppliers, technology and level of education shape marketing practice to a notable extent. Kaynak (1982) in his empirical study concluded that the environmental factors impact a country's marketing structure through their impact on institution operation, he stated:

"The environment contains operating conditions that limit the scope of the marketing institution activities and effect their organisations structure, factors such as income level, culture attribute, consumer characteristics and the structure of the channel of distribution have a bearing on the market potential and thus the scale of marketing institution operation".

However, Douglas (1971), Dougals & Wind (1972) & Douglas & Dubois (1977) argue that the effect of the environmental factors upon the marketing system is indirect and complex in nature, and that it cannot be explained solely in terms of environmental development. Dougals notes;

"Individual firms may respond in different ways and in different degree to environmental conditions and here no consistent pattern of response emerges at the national level. Such environmental change as these just noted are in the large part beyond the control of the marketing system. Generally, the system is able only to adjust and adapt these piece of the total environment to which are the same for all establishment in a given area, it is true, however, that firms and establishment seek to exert as much as possible over the government in which they function".

In her empirical study of the impact of the environmental forces on the marketing structure, Douglas (1971) concluded that the environmental factors effect marketing systems through their impact on individual firms. However, her study does not support the development theory that the marketing structure is closely related to the environment, she states:

"Despite the comprehensive scope of the survey, there was little evidence to support the theory that the development of the marketing structure is closely parallels to that of social environment and cultural environment".

According to the management's strategic studies, it is essential to the firm's behaviour to match its organisational capabilities and its environment. The management should make a greater effort to adapt to its external environmental conditions since it is critical for the firm's behaviour in pursuing its basic objectives. However, studies that focus on the impact of the external environmental characteristics on the marketing behaviour are still few and fragmented and most of them have only been conducted at the macro level.

Goodnow & Hansz (1972) have investigated the effect of the environmental forces on overseas market strategies. Although their study is conducted at the macro level, it is supported by the "Banting-Litvak hypothesis" that the extent of investment as well as the degree of control exercised by the parent firm over the distribution channel in other countries is related to external environment and that the extent of investment and degree of control decline as the environment becomes less favourable. Another similar study was also presumed (Maclayton, et al. 1980) that a firm's decision to enter foreign marketing is largely based on environmental forces, the higher the barriers to invest in marketing in foreign market, the higher the risk the firm will encounter. In the case of the higher risk, the firm has to think deeply in analysing the environmental force. However, their study has tended to concentrate with only a univariate measure of the environment such as gross national product, where there are many environmental factors needed to be considered such as political - legal factors.

Studies that are related directly to investigating the extent of the influence of the external environmental measures upon the export behaviour and the level of exporting are few in nature and fragmented.

Therefore, much of the work in this section is derived from the contribution of international economic and management studies.

For the purpose of the present study, the external environmental forces are suggested under two interrelated constructs: (1) Economic and commercial infrastructural environment, and the (2) Political legal environment.

2.3.2.1 Economic and Commercial Infrastructural Environment

It is believed that an assessment of economic environmental factors is important to the firm's business success either in the domestic, or foreign market (e.g., Keegan 1984). In other words, the economic environment of a country tends to have a great impact on the firm's activities and the ways in which things are done. The main components of the market can be captured by considering variables such as those relative to the population (e.g., population growth, income distributions, consumption patterns) and its various characteristics, such as level of technological development, commercial infrastructure, physical endowment, geographical features of the environment, trade pattern, foreign involvement in the economy, etc..

In discussing the effect of domestic market conditions on the international trade, international business studies concluded that the marketer should be very sensitive to economic perspectives in the domestic market. It was put in this way;

"In appraising domestic economic development, short run foreign trade and internal economic movements can no longer be ignored even when we are looking at the world's largest and most self - sufficient economy" (McCracken 1973).

Indeed, firms react to changing domestic and international economic environment and can be expected to shift their relative emphasis in promoting domestic versus foreign trade. Kindelberger (1962) notes:

"Capacity to transfer is capacity to react to change originating at home or abroad by adapting the structure of foreign trade to the new situation in an economic fashion".

Orientation towards overseas markets is somewhat related to the domestic economic environment (Cooper *et al.* 1970 & Rao *et al.* 1990). An empirical study on the subject by Rao (1977) indicated that during slack conditions in the United States, foreign markets provide a realistic alternative for maintaining business. Based on his survey of 104 firms in Arkansas, Rao concluded:

"While there are wide variations as to the nature and extent of the impact of the recent economic recession, more than one third of the reporting firms were adversely effected. During the recession world period, about 40 % of respondents have intensified their overall export marketing efforts. Some of the export facilitating developments such as the U.S dollar devaluation, exporters and tax incentives expand their exports. During the reactionary period more than one third of the reporting firms have intensified their export marketing efforts by arranging deliveries according to the needs of their overseas customers, appointing more overseas agents and visiting overseas markets more frequently. More than one third of the respondents have reported that their export sales growth has helped them to cushion the slack in domestic sales resulting from the recessionary conditions."

An article in Business International (1974) notes a typical response of firms as they deal with production shortages. These responses include: (1) Allocation of production to the maintenance of domestic markets at the expense of more profitable exports, (2) Simultaneous cultivation of domestic markets and long established foreign accounts, and (3) Forsaking domestic customers in favour of more profitable export business.

The firm's export behaviour is also very likely to be determined by the many commercial factors in its local environment. The main ones are: local tax structure, transportation facilities, availability and access to finance, rate of employment, interest rate, country's trade position, level of economic development, exchange rate, inflation, the availability of resources (raw material) and intensity of competition. The technological development taking place within the environment in which the firm operates is also expected to have an effect on the firm's export behaviour and its level of exporting. According to Hirsch (1971) know how is not created within the firm, it may be purchased, the level to which their purchase relies upon the level of technological development and adaptation taking place within the environment in which the firm operates. This adaptation of technology will influence the methods of production used, works techniques and skills of the labour force through training and education courses. A study by Daniels & Robles (1982, 1985) indicates that there is a significant relationship between the adoption (choice) of technology and export commitment of manufacturing firms.

The physical characteristics of individual markets in terms of distance, topography, climate and natural resources is expected to have influenced the international marketing decision to enter a market and the possible adjustment needed in the marketing mix. In this context distance relates to the physical separation of the marketer from potential customers. In many cases physical distance is accompanied by psychological distance which manifests itself in cultural differences as stated by Johanson and Vahlne (1977) and Dichtl *et al.* (1984).

According to Czinkota & Ronkainen (1988) the availability and quality of an infrastructure is of critical importance to the expansion of export and in evaluating marketing operations abroad. Every international marketer will depend heavily on the services provided by the local market for transportation, communication and energy, as well as an organisation participating in the facilitating functions of marketing communication distribution, information and financing. Transportation network by land, waterway or air are

essential for physical distribution. Communication is as important as transportation, the ability of a firm to communicate with entities both outside and within the market can be estimated by using indicators of the communication infrastructure, telephone, broadcast media and print media in use, data on the availability of commercial infrastructure, bank, advertising agencies, distributors to assess new markets. It was indicated by Cateora & Hess (1987) that;

"The quality of infrastructure directly effects a country's economic growth potential and the ability of an enterprise to engage effectively in business".

Therefore, commercial infrastructure is very likely to effect the level of exporting. However, inadequate commercial infrastructure is expected to, or very likely to have a negative influence on the firms' export involvement, particularly in developing countries where firms are unable to, or not in the position to obtain, or provide export facilities without government assistance.

According to Cateora & Hess (1987); "the less developed a country is, the less adequate the infrastructure is for conducting business". Kaynak & Samli (1984) also stated:

"In developing countries, the marketing infrastructure is under developed. Commercial media, financial facilities, research agencies, postal and other communications and physical distributions are generally under represented, creating constraints on the organisation".

2.3.2.2 Political - Legal Environment

In fact, a country's political system is a mirror of its economy structure and situation. Therefore, it is difficult if not impossible to separate the political system from the whole economic environment. The political - legal environment of international marketing includes any national, or international factors which can effect its operations
or its decision - making (Terpstra 1988). The elements of this environment are the:

- Role of the government in the economy
- Economic and political ideology
- Interrelated relationship
- Business government relative in general

The political forces tend to have a great impact on many aspects of business and economic behaviour. It tends to determine the degree of freedom that organisations are obliged to follow in pursuing their business's goals. In other words, the government's role can effect the firm's organisational behaviour (structure, direction, resource allocation, etc.) through a number of processes of legislation and regulations that firms are obliged to conduct.

Much of the literature in economic and international business has been discussed the forms and the degree of the effect of government intervention in economic and business operations. However, it was reported by several studies that the extent of the government's role differs among countries and even in the same country, but it appears to be larger in developing countries than in developed countries as noted by Poynter (1982):

"....the extent of the government's role varies from country to country, even in the same country, from industry to industry, while government intervention in the form of financial support and trade protection is sought by firms, most government intervention is costly. Corporate reaction to undesirable forms of government intervention has varied. Some firms aggressively attempt to control such behaviour directly while others treat intervention as a fact of life over which they have little, if any, control".

The manner of government intervention in export marketing takes two forms, while it acts as barrier to export marketing activities through a number of restrictions such as tariffs, quotas, legislation and procedure requirements such as licence requirement, it acts as a promoter through its action on the areas of subsidies, tax incentives, provision of information other types of assistance, and free trade zones. In this respect Kaynak (1982) notes:

both direct and indirect government levels of "Various intervention in the distribution system of LDS have been recorded. Direct intervention includes activity such as publicising controlled prices, the operation of government owned retail outlets. Indirect intervention includes economic legislation, licensing dealers and the support and encouragement of non - profit organisations, the consumer cooperations"

"There are many rules and regulations that hinder efficient marketing in less developed countries.For example distribution in most LDS are burdened by outdated laws that restrict or limit the assortment of goals they can handle" (Kaynak 1982).

In his study of export marketing practices of the UK organisations in developing countries Michell (1979) confirmed the relative significance of the government political factor. The three most popular types of export constraints mentioned by respondents were: political instability, internal currency restrictions and difficult legal problems. Rojemta (1986) reported a similar result in his study of constraints that were encountered by American industrial firms in Thailand, he found that government restrictions and inadequate infrastructure were among the most important constraints rated by the respondents.

Terpstra (1988) notes that the political instability of the country and rapid change create a climate of uncertainties unfavourable for business. Political stability has been found to be the most crucial variable that companies weigh when considering going overseas (Schollhammer & Nigh 1984). Czinkota & Ronkainen (1990) views that the effect of politics on international marketing is determined by both the bilateral political relationship between home and host countries and the multinational agreement government relationship among groupings of countries.

The policies and promotional activities and programmes developed by government organisations and agencies to promote exporting are increasingly important factors in the international environment. However, studies in developed countries indicated that promotional and financial assistances provided by governments are viewed as insignificant to increase the firm's involvement in export operations (Kolhede 1984 and Wiedersheim-Paul *et al.* (1978). The export incentive was found to be only important to initiate the export process of small firms (Rabino 1980). It can be argued that export firms in developing countries very likely need to rely on the government's assistance in order to enhance their ability and to increase their export involvement in increasingly competitive international markets.

(ariosa)isa retiit(a lagaf taa Jacaro) tiaaca (Jacaro) (aroo) (rieg waam ordi) (latia radiaretiin() la bardo a eriecharo ((reesaringga wit) rie (aro) (latia) (reesaringga wit) retigan (stroro) (a c

the stars segme and so estimate a market of the stars the stars of the stars of the stars of the stars of the stars and the stars and the stars of t

laise a north a na britteria i forma in the second a lass - northerappet (1) -ard brand a britteria for the second second satisfies a britteria transdiscission (these of states to be second and states and so a transat a coster of "second at restor" to around the second of atoms

2.3.3 Export Modelling Studies

Export models are concerned with the identification of the relevant variables, the specification of the interdependencies among these variables at a given time, and the specification of the dynamics between the variables (Bilkey 1978). A brief summary of each model, including advantages and limitations, is given as below:

(1) Etgar & McConnell's Model

Etgar & McConnel (1976) formulated a static causal and effect theoretical model of export behaviour in the form of an equation:

$$\mathbf{A} = \mathbf{Q} (\mathbf{E}, \mathbf{I}, \mathbf{C})$$

Where:

A = represents a vector of export related behavioural decision
 E = represents a group of internal and external environmental
 factors (location of markets, technological, institutional factors,
 behavioural forces, economic forces, and legal political influences)

I = represents a group of information stimuli (from mass media, personal contacts, and previous experiences)

C = represents the information processing complex (including technological choice constructs.

However, the relationship among variables on the right side of the equation either within groups or between groups were neither indicated nor tested empirically. According to Bilkey (1978) their model yield inference is that it harmonizes with the observable behaviour.

(2) Wiedersheim - Paul & Welch's Model

They proposed a conceptual framework focusing on a firm's preexport activities before commencement of export. The model basically suggests that different types of "Triggering cues" in relation to

decisions about pre - export behaviour - attention evoking - factors are exposed to the decision maker.

The type and amount of alternatives evoked are dependent not only on the individual decision - maker but also on the environment in which he is acting, i.e., the environment of the firm and the firm itself. Four factors are thought likely to have an impact on the preexport behaviour of the firm:

- 1. Goals of the firm and their degree of realisation
- 2. Type of product line
- 3. History of the firm
- 4. The degree of extra regional expansion.

Figure (2.1)

WIEDERSHEIM & WELCH'S MODEL

Factors Effecting the Pre - Export Activities of the Firms*



^{*} Source: Wiedersheim - Paul, F., H. Olson & L. Welch, "Pre - Export Activity: The First Step in Internationalisation", <u>Journal of</u> <u>International Business Studies</u>, (Spring - Summer 1978), pp. 47 - 58.

Specifically, the attention evoking factors - those influences that cause the firm to consider the adoption of export as a possibility, are divided into internal and external stimuli. Pre - export activities were studies from the perspective of the information form the firm surveyed, these activities being described on the basis of:

- 1. Willingness to start exporting
- 2. Information transaction activity
- 3. Information collection activities

Among the dimensions, firms were classified into three groups: active, passive and domestic. Empirical testing of various aspects of the models operations were conducted in Australian small manufacturing firms. However, the data was generated from a small sample size, which is insufficient to generalize the findings. Furthermore, it should be noted that the model still needs to be tested empirically and the specific types of functional relationships among the variables in the model have also not been identified. The authors, as they recognize, this model represents a framework for the developing for more partial models of qualitative and quantitative types which are open to operationalization.

With regard to the present study, this model seems to be one of the most comprehensive models, since it includes all the concerned factors that could possibly effect the firm's export behaviour. However, these factors were advanced in broader categories.

(3) Reid's Model

Reid (1980) developed a theoretical framework to explain the firm's decision to export. His model focused on the role of individual factors (characteristics). Export behaviour related variables were classified into three groups: (1) Contextual factors such as the industry environment and the foreign market environment.

Figure (2.2)

REID'S EXPORT BEHAVIOURAL MODEL

The Role of the Individual in Export Decision Making

Contextual Factors



* Source: Reid, S. "A Behavioural Approach to Export Decision-Making" American Marketing Association Proceedings, (1980), PP. 265 - 268. (2) Individual mediating factors which consisted of two levels of variables: the first group consisted of the decision - maker's past experiences, his objective knowledge which can be taught and more easily acquired and experimental knowledge that could be gained only from experiences which made it possible to perceive concert opportunities; the second group consisted of the decision - maker's export orientation, his motivation to export expectation, and (3) The decision - maker's intention to export which served as dependent variables. The general relationship amongst these groups of variables are shown in Figure (2.2).

According to the model, the decision - maker's characteristics may act on several levels. First the decision - maker is exposed to all the information stimuli from the firm's and the foreign market environment. The information is processed and influenced by his experience and knowledge, in turn, this information processing effects the decision - maker's export organisation, motivation to export, and his expectation of export effects. These effected mediating variables, finally influence the decision - maker's intention to initiate exporting. However, the specific type of relationship among the variables in the model are not indicated and the model is still not empirically tested.

(4) Bilkey & Tesar's Model

Bilkey & Tesar (1977) formulated a theoretical model by using Roger's theory of Diffusion of Innovations (Rogers 1961). The model involves the following export stages:

- The firm is unwilling to export; it would not even fill a unsolicited export order, because of apathy, dislike of foreign activities, busy doing things ..., etc.
- 2. The firm fills unsolicited export order but does not explore the feasibility of export.

- The firm explores the feasibility of exporting (this stage may be omitted by the receipt of unsolicited export orders).
- 4. The firm exports experimentally to one or a few markets.
- 5. The firm is an experienced exporter to those markets.
- 6. The firm explores possibilities of exporting to additional market and so on.

The questionnaire was mailed to a sample of (423) small and medium sized firm in Wisconsin, and the analytical methodology employed involved treating each stage of the export development process as the dependent variables in a multiple regression equation - Coefficients differing at each stage to allow for experience gained in the preceding stage:

$$\mathbf{A} = \mathbf{a} + \mathbf{b}\mathbf{E} - \mathbf{c}\mathbf{I} + \mathbf{d}\mathbf{F} = \mathbf{e}\mathbf{M}$$

Where:

- A = The firm's export activity for the stage in question.
- E = Management's expectation regarding the benefit of exporting after it has been developed.
- I = The inhibitors (mainly serious infrastructure and institutional obstacles) that management perceive to initiating exporting.
- F = The facilitators (unsolicited order, information, subsidies, infrastructure and institution aid, etc.) management perceive to initiate exporting.
- M = The quality and dynamism of the firm's management plus the firm's organisational characteristics that effect exporting.

The specific findings in this regard are as follows; the major correlation of export movement from stage I to stages 2 and 3, were whether management planned for exporting (pertains to factor M) and management's impression of the firm's differential advantage (factor F), no relationship was found with the management's expectation as to what exporting would contribute to the firm's profit's, growth etc, (factor E), nor with management's perception of inhibitors to exporting, (factor I). Movement from an earlier stage to stage 4 (experimental exporting) correlate directly with whether the firm received an unsolicited initial export order (factor F), and directly with the quality of the firm's management; and to a small extent, directly with the firm's size, (factor M). Again, there was no relationship with the management's expectation as to what exporting would contribute to its firm, nor with management's perception of export inhibitors. Movement from stage 4 to stage 5 as well as the extent to what stage 5 was pursued (measured by the percentage of the firm's total sales derived from exporting), correlated directly with the management's perception of gains from exporting, inversely with the number of perceived inhibitors to exporting, and inversely with the quality of the firm's management.

The basic conclusion that Bilkey & Tesar derived from their study was summarized as follows: (1) the export development process of the firms tended to proceed in stages, (2) considerations that influence a firm's progression from one stage to another tend to differ by stage, (3) within the size range of the firms studied, size was relatively unimportant when account was taken of the quality and dynamism.

(5) Czinkota and Johnston's Model

In discussing Bilkey and Tesar's model, Czinkota & Johnston (1981) note: "Although this study was successful in differentiating among firms, the criteria according to which firms are classified were quite general in nature". Based upon Bilkey & Tesar's Model, they further developed the process of the firm to encompass eight stages instead of six stages as discussed by Bilkey and Tesar, which are as follows:

Stage (1): The completely uninterested firms (the unwilling firms)
Stage (2): The partially interested firms
Stage (3): The exploring firms
Stage (4): The experimental exporters
Stage (5): The disappointed exporters
Stage (6): The temporary exporters
Stage (7): The experienced small exporters
Stage (8): The experienced large exporters

Ouestionnaires from 237 U.S manufacturing firms (three industries) were segmented according to the above stages, and analysis of variances (ANOVA) were employed to compare the mean group score for the group configuration of such segmentation approach. The findings show that change in exporting attitudes, motivation, and perceived exporting obstacles are associated with the growing length of exporting experiences, although, the defined categories did not statistically identify the precise pattern of the development process. Moreover, the criteria which was used to classify the firms into these above stages were composed in a more rigourous fashion by using a quantitative cut - off value for relative and absolute export sales volume, length of export experience, type of countries exported to, and manpower committed to exporting. In addition, the stages were slightly altered to include only six (all the firms except the firms in stage 5 and stage 6).

Specifically, although Bilkey and Tesar's model and Czinkota & Johnston's model could be used as a basis for a development promotion programme, they do not provide evidence as to whether the firms in each stage will react similarly to a certain stimuli*; whether the factors that determine firms in each stage are similar among the firms surveyed, i.e., are there any specific relationships among the variables which are discussed, and whether the firm has to pass through each stage in order to be fully involved in export marketing activity. Furthermore, both models assume that exporting begins by accident and then commitment to exporting develops later. This situation is fairly common in developed countries where these two models were tested. However, in developing countries such as Jordan,

^{*} Recent studies in exporting suggest that firms seldom go through a concrete, stepwise process of export development or internationalisation (e.g., Sullivan & Bauerchmidt 1990).

it can be argued that the government's export assistances (financial and non - financial incentives) are important for the exporters of manufactured products to develop their commitment to export first and then develop their export sales. In other words, the process of export development may begin either with export commitment, or with export sales.

(6) Cavusgil & Nevin's Model

Cavusgil and Nevin (1981) have developed an empirically tested static path model of export marketing behaviour, utilizing (473) manufacturing firms in Wisconsin as research subjects. As can be seen in Figure (2.3), this model consists of internal determinants of the firm's export marketing behaviour. These determinants are further divided into background and intervening independent variables.

Cavusgil & Nevin distinguished between background variables which are divided into two groups of differential firm advantage and the strength of managerial aspiration of business goals. Intervening variables which represented the present situation include two groups: management's expectation about the firm's commitment to export marketing. Specifically, differential advantage and managerial aspiration are thought to promote management's expectation regarding the effect of exporting on the firm's goals and indicators of the resources allocation. Intervening variables hold firm's the distinction of being more subject to change in the short term than background variables, i.e., the former are more dynamic as far as determining export behaviour. Through multiple classification analysis (MCA), they reduced 19 internal determinants to 10 independent variables. A summary discussion of these follows. The differential firm advantage (one of the Background group) include: Firm size (as

Figure (2.3)



CAVUSGIL AND NEVIN'S STATIC - PATH MODEL OF A FIRM'S EXPORT BEHAVIOUR*

measured by sales volume), the technological intensiveness of the firm's industry, the firm's possession of a unique product and the firm's proximity to foreign markets. As discussed previously, although these variables are not sufficient by themselves to initiate export marketing, these unique advantages are attention evokers (Wiedersheim - Paul *et al.* 1978), i.e., "they are important in preparing the firm and in providing initial motivation for management".

The other group of "Background determinants" i.e., the strength of managerial aspiration for business goals, include the independent variables of managerial aspiration for growth and managerial aspiration for security markets. According to this analysis a positive relationship exists between export marketing behaviour and the decision maker preferences for these business goals.

^{*} Source : Cavusgil, S. & J. Nevin, "Internal Determinants of Export Marketing Behaviour: An Empirical Investigation", <u>Journal of Marketing</u> <u>Research</u> (Feb.) 1981, P. 115.

One of the two groups of intervening determinants, i.e. expectation of management regarding the impact of exporting on firms business goals consists of the independent variable of management's expectation about the effect of exporting on the firm' growth. The second group of intervening determinants relate to the management's level of commitment to export marketing and consists of whether or not management is involved in market planning; systematically explores the feasibility of exporting and developed a policy toward exports.

According to Cavusgil & Nevin, the firm's willingness to allocate adequate resource to exporting as measured by the above variable is critical in determining the extent of a firm's export marketing behaviour.

Based on these findings, these analysts have come up with a conclusion relating to the means by which (USA) business export marketing activity may be stimulated. They state:

"The result seems to support the contention that the reluctance of firms to export may be largely attributable to top management's lack of determination to exports and, the firm's top management should be able to stress the importance of growth as an organisational objective as well as the potential positive impact of exporting on growth and then committing resources to engage in market planning systematically exploring the possibilities of exporting and developing an export policy.

In the discussing of Cavusgil & Nevin's model, it was noticed that the number of internal determinants of exporting behaviour are very limited, i.e., only a few numbers of the firm's internal characteristics have been tested. Also, the relationships among variables called "background variables "have not been identified.

As far as their major hypothesis is justified and explained, export marketing behaviour of the manufacturing firms can be explained largely by the firm's internal characteristics. Our argument is in opposition to their conclusion that the export marketing behaviour can be better explained through the interaction of the firm's internal characteristics and forces that exist on its external environment rather than based on either of them individually.

(7) Kolhede's Export Orientation Model

Kolhede (1984) developed a model by using Roethlisberger's theory, the major promise of her study is based on the fact that the causal elements of attitudes embodied in "Roethlisbergers's theory (1941)" are; underlying determinants of small business, employees response to change in the workplace environment can be modified for use as tools for understanding both the export orientation of the small business manager as the resulting response of small companies, to the change involved in the notion of their expansion export involvement.

The particular change examined is the notation of export undertaking by small firms. The manager's response to such a proposed change based on his export orientation. A positive export orientation (+) will cause the manager to decide to increase his firm's export marketing, a negative (-) export orientation will result in the firm's resistance to export expansion.

The questionnaire was mailed to a sample of (403) small manufacturing US firms. Analysis of variances, factor analysis and multiple regression analysis techniques were employed to compare the export orientation in both types of small firms, exporting and non exporting.

The result indicated that by using a multiple regression coefficient of determination (R square) which was about 44% of the variation in the dependent variables and which was explained by the seven factors discussed by the author, i.e., the factors which were assumed to determine the small firm's export orientation (Figure 2.4)

Figure (2.4)*

KOLHEDE'S EXPORT ORIENTATION MODEL

PROPOSED CHANGE RESPONSE

MANAGER'S OR DECISION - MAKER'S

Expand Company Export Export Involvement Increase/ No Increase In Company

Export Marketing.

MANAGER'S EXPORT ORIENTATION (+/-)

ATTITUDES TOWARD ATTI TUDINAL ENVIRONMENTAL FORCES PERSONAL'S BACKGROUND VARIABLES

SOCIAL AT WORK

Assessment (+/-):

Inducement for company exporting Benefit to the firm of export marketing involvement Company resources essential for exporting Company's competitive advantage for exporting Willingness to modify management marketing mix for exporting Involvement Personal enjoyment of Obstacles to company's export involvement Effectiveness of Federal government export programme

Assessment (+/-):

Foreign travel experience Foreign language proficiency Formal education Work experience Group affilitations outside the company Assessment of social importance of US small business exporting challenge of exporting

Export orientation (+/-) of fellow Company members

* Note: +/- denotes positive / negative

Source: Kolhede, E., "The Underlying Determinants of Export * Orientation of Small Manufacturing Firms", PH.D Dissertation, Golden State University (1984).

Although the results of her study were found to be in conflict with many previous works as discussed through the literature, the study can be considered as an additional new approach to export marketing behaviour. Furthermore, her study was mainly based on examining the attitudes of the decision - maker rather than examining the factual data which was behind each firm when it becomes involved in exporting, either small or large firm, or both.

(8) Aaby & Slater's Strategic Export Model

Aaby & Slater (1989) developed a strategic export framework based on a review of empirical literature between 1978 to 1988. They argue that current export knowledge should be synthesised at two broad level: (1) the external environmental level, and (2) the firm business strategy and functional level. The environmental level includes macro - economic, social, political and cultural aspects which influence export management, behaviour and performance. The business strategy level includes assessment of key business policies and capabilities within the firm's control.

This framework has only focused on a review of literature that related to managerially controllable factors. They include; firm characteristics, firm competencies, and firm strategy (see figure 2.5). The relationship among these groups of variables has been discussed through an extensive review of studies during the 1978-88 period.

According to Aaby & slater "although there is no clear cut formula for developing a successful programme, there appears to be sufficient evidence to draw a number of general conclusions: First company size by itself is not an important factor unless it is associated with other aspects such as financial strength or variables related to





* Source: Aaby, N.E. & S.F. Slater, "Management Influences on Export Performance: A Review of the Empirical Literature 1978-88" International Marketing Review, Vol. 6. No. 4, (1989). economic of scale. Second, in companies where management is firmly committed to export, export performance tends to be higher. Third, firms that have better management systems and plan export activities well are more successful than those that done. Fourth, export experience is important, and firms that have experience are likely to do better than that are just starting".

(9) Yang, Y.S., R.P. Leone & D.L. Alden's Model

Yang *et al.* (1992) have developed an empirically tested model. It suggests that the manager's export intention is an indicator of the firm's future export behaviour and is related to a firm's market expansion ability and the manager's perception of external (e.g., general environmental) and internal (e.g., firm-specific) barriers to exporting (see figure 2.6).

Following Fishbein ans Ajzen (1975), they proposed the model as:

 $EB \sim EI = (MEA, BE)$

Where:

EB = The actual export behaviour EI = The export intention MEA = Market expansion ability of the firm BE = Barriers to exporting

The tilde between EB and EI suggests a time interval between export intention and actual export behaviour. The relationship among export intention, market expansion ability and perceived barriers to exporting were examined by using 345 non - exporters of manufacturers firms in Texas (U.S.A).

The specific findings in this regard are as follows: (1) The export intention of non - exporters is positively related to market expansion ability and perceived export barriers. (2) The probability of the firm's future export activity is largely related to the market expansion ability variables than those for the perceived barriers to exporting. This suggests that firm's prior domestic expansion experience and resources may be more important to identify its potential export activity. However, no significant relationship was found between market expansion ability and perceived barriers to exporting.

Although this model was limited to the experience of one geographical area in the U.S.A (i.e., Texas), it might be useful in the prediction and identification of the characteristics of the potential exporters in other developed countries.

Figure (2.6)

Yang, Leone and Alden's model*



^{*} Source: Yang, Y.S., R.P. Leone & D.L. Alden, "A Market Expansion Ability Approach to Idnetify Potential Exporters", <u>Journal of</u> Marketing, Vol. 56, (January 1992), PP. 84 - 96.

2.4 The Common Characteristics of Export Behaviour Studies

<u>To the best knowledge of the researcher</u>, previous studies in export marketing behaviour, have in common the following characteristics:

Only some of these studies have developed a conceptual framework.
In many studies, research has been confined to only some aspects, therefore, other important aspects have been given little attention in the specific analysis making the findings less relevant, for example, the effect of some external environmental measures such as those related to the government export procedures and regulations.
The majority of these studies are confined to the experience of

developed countries, such as the U.K, USA, Germany, etc.

- Empirical evidence produced mixed findings regarding many aspects as discussed through the literature, for example, the impact of the firm's parameters (i.e., size, age, etc.) on the firm's export behaviour and its level of exporting.

- Some of these investigations were conducted in isolation, without benefit from the experience of findings from other studies.

- In many of these studies, practical implications of research findings are only stated in general terms.

- Little attempts might be made to report the reliability of the scale of measurement for the data collection.

Therefore, the aim of this study is to overcome the above limitations of the previous research and to improve the understanding of export behaviour at the firm level.

In this chapter, a review of international trade theories, firm and organisational behavioural theories were presented firstly as a basis theoretical background for the purposes of the study. A discussion of empirical studies in export marketing behaviour then followed in an attempt to identify the variables which are most likely to have an impact on the firm's export behaviour and its level of exporting.

The traditional international trade theories were introduced, as it provides a basic explanation for the global factors that determine the pattern of trade among countries. However, the international trade theories explanation which is based on the doctrine of competitive advantages was found to be insufficient, or failed to provide an answer for why cost or price differs among trade partners, because cost or price parameters can only be determined at the micro level, i.e., the individual firm's level.

Product life cycle concept, which has been recently developed, holds that the moves of a firm or country towards international marketing can be explained in terms of trade cycle. However, the product life concept has been attacked on several accounts. While it provides a sufficient explanation for certain types of products particularly those highly technological products which are largely innovated and produced in advanced countries, it might be insufficient for traditional products which are mass produced in developing countries. However, both international trade theories are useful for this study, since they provide some theoretical support for some variables that could be examined at the firm's level.

Firms and organisational behavioural theories were presented and discussed simultaneously as theoretical perspectives for the purpose

of the study. The firm's behaviour is viewed as a result of an interaction between its internal operations and the factors that exist in its external environment. Therefore, it is possible to analyse the process of decision making within individual firms on the basis of the factors influencing the organisation. Administration theory which is one of the branches of the organisational behavioural theories is also discussed, since it is of major concern with the process of decision making within the organisation with emphasis on the role of decision makers.

Export marketing behaviour literature which relates to the supply side of the international trade theory are comprehensively reviewed and presented under three sections: (1) Studies relating to the firm's internal environment, (2) Studies relating to the firm's external environment, and (3) Export marketing behaviour modelling studies

The studies that are concerned with the firm's internal export marketing are further presented under five headings: (1) The exporting initiation (2) Firm export orientation (3) Management characteristics and export marketing (4) Export marketing mix decisions (5) Firm and decision - maker's individual characteristics.

The studies that are related to the firm's external environment, however, are few in nature and fragmented, therefore, to a greater extent, much of the work in this section is drawn from the contribution of the economics and organisational theories. For the purposes of this study, the constructs of the firm's external environment are presented and discussed under the following headings: (1) Economic and commercial infrastructural environment, and (2) The political - legal environment.

With regard to export marketing behaviour modelling studies, nine export modelling studies were presented and reviewed. However, only

five of them were tested empirically. The review of export models suggest that forces within the firm and its management may be centred around export behaviour of the nine models, all included some characteristics of the firm, the characteristics of the management, and important factors included aspects of the environment, whether local or foreign environment. The conceptual model which will be presented in Chapter Five, borrows some of the concepts and ideas of these export models, in addition to the ideas that derived from the organisational behaviour and international trade theories.

CHAPTER THREE

THE PRELIMINARY IN DEPTH INTERVIEWS AND SOME INFORMATION ABOUT EXPORT POLICIES, PROCEDURES AND EXPORT PROMOTION INSTITUTIONS IN JORDAN

Introduction

3.1 Objectives of the Preliminary Interviews

3.2 Structure of the Preliminary Interviews

3.3 Planning for the Preliminary Interviews

3.4 Data Gathered Through the Preliminary Interviews

- 3.4.1 The Role and Functions of the Export Promotion Institutions in Jordan
- 3.4.2 The Export Regulations and Procedures

3.4.3 The Results of the Initial Interviews

Summary

CHAPTER THREE

THE PRELIMINARY IN DEPTH INTERVIEWS AND SOME INFORMATION ABOUT EXPORT POLICIES, PROCEDURES AND EXPORT PROMOTION INSTITUTIONS IN JORDAN

Introduction

As indicated in the previous Chapter, most of the aspects of the firm's external environmental dimension have been neglected, or have received little attention in export behaviour studies.

Therefore, the researcher conducted several interviews with relevant persons in both private and public institutions, as well as with a number of initial respondents of manufacturing firms in order to develop some measures of the firm's external environment, in particular, elements of the political - legal environment, and to formulate specific hypotheses that were relevant to the export situation in Jordan.

In this Chapter, the main purpose of the preliminary interview, its structure and results are presented. Furthermore, some background information about the export policies, regulations, and procedures is also given.

The preliminary interview constants and from a subsection with the tocontinuous of the secondarierity firms, as well as with the probability of both public and private institutions that undertake various endow stated dather to format. The otherwise of these interviews were own contentored interviews. The otherwise cashed was attempted at the content of the present study.

3.1 Objectives of the Preliminary Interviews

The objectives of the preliminary interviews can be divided into two types: (1) general objectives, and (2) specific objectives.

The general objectives of the preliminary interviews are: (1) to achieve more insight into the study problem under investigation, (2) to formulate specific hypotheses in respect of the objectives of the study, (3) to explore new areas which are more relevant to the export marketing behaviour of the manufacturing firms in Jordan, (4) to be able to prepare and design an appropriate questionnaire that will handle the export marketing behaviour in Jordan.

The specific objectives of the preliminary interviews are: (1) to acquire more information about the external environmental characteristics of the manufacturing firms in Jordan; in particular, the export policies, regulations, procedures, the types of export assistances and services that are offered by the public and private institutions, and the major obstacles that are encountered by manufacturing firms from the point of view of the authorities, (2) to shed light onto the export situation in Jordan from the point of view of the decision - makers of manufacturing firms with regard to the above issues.

3.2 Structure of the Preliminary Interviews

The preliminary interview consisted of free discussion with the top executives of five manufacturing firms, as well as with the authorities of both public and private institutions that undertake various export related issues in Jordan. The general form of these interviews were openstructured interviews. No statistical analysis was attempted at this stage of the present study.

3.3 Planning for the Preliminary Interviews

The in - depth interviews were conducted in two stages. At the first stage, the authorities in both public and private institutions were contacted by telephone and their cooperation was sought. The prior arrangements for the meeting were well scheduled in terms of the name of the person to be interviewed, and the time and place of the meeting.

At the second stage, the top executives of six manufacturing firms were telephoned and their cooperation was sought. Of the six, five agreed to a meeting. The interviews were held separately at the organisation's premises. The main objectives of each stage are given in Table (3.1).

3.4 Data Gathered Through the Preliminary Interviews

The data collected through the preliminary interviews are presented under the following headings:

- (1) The role and functions of the export promotion institutions
- (2) The government's export procedures and regulations
- (3) The results of the initial interviews

3.4.1 The Role and Functions of the Export Promotion Institutions in Jordan

There are several bodies who undertake the various export related tasks in Jordan. Public bodies include the Ministry of Industry and Trade, the Ministry of Finance and Customs, the Central Bank of Jordan and others. The private institutions consist of the Chamber of Industry, the Commercial Banks, and The Jordan Trade Association. The role and the function of each institution in respect of exporting is given in detail according to the type of institution, i.e., public or private, as follows:

Table (3.1)

The Objectives of Each Stage of the Preliminary Interview

e of Interview	19.19.282	Objectives
Several authorities in five organisations were interviewed.	(1)	To get more specific information about the role of each organisation
Ministry of Industry		in respect of exporting.
Trade Ministry of Finance	(2)	To acquire more inform- ation about the types of
Customs Central Bank of Jordan		export policies and assistances that are offered b
Chamber of Industry Trade Corporation Centre		each organisation.
	(3)	To identify the main aspect of the government export procedures and
		regulations.
	(4)	To get the point of view
		respect of the export situation in Jordan.
		export barriers, and any suggestions to improve th export level.
Conduct free discussion with five executives of	(1)	To acquire their point of view in respect of the
manufacturing firms.		stances that are offer by each organisation.
	(0)	(ing convertigh and in
	(2)	hindrances that might be encountered by exporters
		& potential exporters.
	(3)	To obtain their point of
		VIEW IN TESDELL III THE
	Several authorities in five organisations were interviewed. Ministry of Industry Trade Ministry of Finance Customs Central Bank of Jordan Chamber of Industry Trade Corporation Centre	Several authorities (1) in five organisations (1) were interviewed. (1) Ministry of Industry (2) Ministry of Finance (2) Customs (2) Central Bank of Jordan (1) Chamber of Industry (1) Trade (2) Ministry of Finance (3) (3) (4) Conduct free discussion (1) with five executives of (1) manufacturing firms. (2)

3.4.1.1 Public Sector Institutions

(1) The Ministry of Industry and Trade

The Ministry consists of several departments. Of these, three are tied to export activities. These departments are: the Commercial department, Standards and Specifications department, and Economic Co - operation and Export Promotion department.

The main functions of the Commercial department in respect of export are: (1) issuing export licenses, and (2) making sure that the export licence requirement is fulfilled by the relevant person. The export regulations and export procedures are presented in section 3.4.2 of this Chapter.

Export products need to be checked and approved by the Standards and Specifications department, in particular, those products which are exported to countries where the government holds trade agreements with. The main function of the Standards and Specifications department is to make sure that: (1) the export product is produced according to the trade agreement conditions (2) the export product meets the safety and specification standards, and (3) issuing a Quality Certificate.

The role of the Economic Co-operation and Export Promotion Department can be summarized as follows: (1) preparing commercial and trade agreements (2) carrying out studies to improve and amend the agreement, (3) conducting research and studies in relation to the activities of the Arab Economic Unity and the Arab Economic and Social Councils. (4) preparing studies concerning exhibitions and fairs, marketing Jordanian products abroad, and supervising Jordanian commercial attaches, (5) participating in exhibitions and international trade fairs, and (6) preparing a list of potential Jordanian export commodities. containing information, and addresses names and of Jordanian manufacturers.

According to the export regulations, the manufacturer has to get approval of this department. The reason for this procedure is to ascertain whether the exporter is qualified to be exempted from export license fees. The exporter can be exempted from the fees if he is exporting within the trade agreement, otherwise, he has to pay a certain amount of money (1).

(2) Jordan Commercial Trade Centres Corporation

The Trade Centres Corporation was established in 1973 by decreed Law No. 21 of Year (1972), with the objectives of establishing Trade Centres in the Arab countries according to the articles of agreement and protocols ratified between the government of Jordan and other Arab nations, in order to promote and encourage market and export Jordanian agricultural and industrial products to concerned partner countries.

The functions and duties of the Trade Centres Corporation with regard to export include the following: (1) to act as a commercial agent or representative for Jordanian firms in the areas of industry, services and trade, and (2) to encourage the marketing of Jordanian products through advertising activities, trade propaganda, participation in international fairs and organizing bilateral exhibitions.

Currently, the Corporation has only got three trade centres in operation. One in Iraq and the others in Egypt and Sudan.

(3) The Ministry of Finance and Customs

The Ministry of Finance and Customs administers the various trade tariffs, taxes and custom procedures in Jordan. The Ministry also approves the Certificate of Origin for the exported products.

Unpublished information from the Ministry of Industry and Trade. A Visit on 28 July 1990.

The tariff system in Jordan shows tremendous import protection and plays a significant part in the contributions to local government revenue. For exports, however, there are no export taxes as such, nevertheless, all materials (if they are not available in the domestic market) imported by manufacturers for the purpose of exporting are exempt from custom duties. There are two systems operating for the purpose of achieving this objective. First, the temporary entry scheme, under which raw materials are imported, manufactured and exported in a period of less than six months, are exempt from customs duties, and second, is the duty - drawback system, under which customs duties are reimbursed to manufacturers after export takes place⁽²⁾.

According to the Jordanian Custom and Excise Law No. 1 of (1962): (1) the exported goods should be inspected by the customs and selected in accordance with custom procedures, (2) the imported goods under the Law should be exported within a six months period which must not be exceeded unless the Authority advises otherwise⁽³⁾.

(4) The Central Bank of Jordan

The Central Bank of Jordan's impact on the export sector is manifested most significantly in the monitoring of the following:

- 1. The Jordan Dinar Exchange Rate
- 2. The financial arrangement of Jordan's trade agreement with other countries.
 - The bilateral agreements with Iraq concerning barter trade and payments facilities.

⁽²⁾ An interview with the Head of the Temporary Entry Department, Mr. Mahmoued Alatwi, on 30 July 1990.

⁽³⁾ Ministry of Finance and Customs: Jordan's Customs and Excise Law No. 1, 1962 PP. 1 -20.

4. The commercial bank loans rediscount facility to exporters.

5. The interest rate on the loans to exporters.

The financial arrangement of trade agreements include the Letters of Credit and payment orders for imports, in addition to the Letters of Credit for exports. In an effort to promote exporting the Central Bank has extended the short term re - discount and advancement facility of the Letters of Credit for export from six to nine months. Moreover, the interest rate on the advancement of the Central Bank to exporters was reduced from 6% to 5%.

(5) The Free Zones Corporation

The Free Zones Corporation was established in 1973 and came into operation in 1974. The function of the Corporation is to establish, manage and develop Free Zones, in addition to utilizing them in order to serve the national economy, promote international trade exchange, exchange transit trade and encourage export oriented industries. Currently, there are three trade zones corporations in operation. One in Agaba, and the others in Zarka and Queen Alia International Airport.

The incentives offered to industrial and commercial projects established in the Free Zones Corporations includes tax holidays for twelve years, exemptions from import fees, custom duties and other taxes, in addition to free transfer of capital and profits outside Jordan⁽⁴⁾.

⁽⁴⁾ The Free Zones Corporation Law and Regulations and Instructions: Jordan, 1980.

3.4.1.2 Private Sector Institutions

The private sector institutions, which are concerned with export in Jordan, are mainly the Chambers of Industry, the Jordan Trade Association, and the Commercial Banks.

(1) The Chambers of Industry and Trade

The Law of the Chambers of Industry and Trade was promulgated in 1949 and was amended in 1961. The functions of the Chambers of Industry and Trade include the following:

- To collect, tabulate, and publish industrial and trade data, and to make this data available to the interested parties (i.e., the members of the Chambers).
 - 2. To participate in the preparation of the laws and regulations that are related to the economy as a whole, and to industrial and commercial activities in particular.
- 3. To organize economic conventions and seminars inside and outside Jordan.
- To act as an intermediary between manufacturers and traders in case of conflict or disagreement⁽⁵⁾.

In respect of exporting, the main function of the Chamber of industry, is to stamp export invoices and to issue a Certificate of Origin as a prerequisite for the export licenses. Other functions include the provision of information about potential foreign markets.

⁽⁵⁾ Jordan Legislation Encyclopedia, Volume 18, The Function of the Chamber of Industry and Trade, No. 14 of Year 1949, and Law No. 21 of Year 1961.

(2) The Commercial Banks and Other Financial Institutions

The financial sector in Jordan is made up of a number of institutions providing a variety of debt and equity financing facilities. Lending facilities are provided by 18 commercial banks, 6 specialised banks, 13 finance companies, 2 investment banks, 33 insurance companies and others. Although Jordan is considered to have a reasonably developed financial sector, short term financing for export and equity participation is relatively underdeveloped. In other words, export financing is rarely available and access to bank credit facilities is highly restricted. This might be attributed to two factors: (1) the high risk associated with the financing of export activities, and (2) the inability of many manufacturing firms to meet the commercial banks' requirements.⁽⁶⁾

(3) The Jordan Trade Association

The Jordan Trade Association is a non - profit making organisation. It was established in late 1987 with the aim of expanding and promoting export products and services. The main functions of the Jordan Trade Association include:

1. The collection of information about export markets, and the potential opportunities available in these markets. In particular, the information about: (1) the suitable channel of distribution for products and services to be exported to foreign markets, (2) the international trade law and system, (3) the trade law of the countries where the products or services are to be exported to, (4) the standard specifications required for products and services in foreign markets and, (5) the prices of the alternative products in international markets.

 ⁽⁶⁾ Unpublished information from the Central Bank of Jordan, A visit on 3 August 1990.

2. Undertaking directly, or with the cooperation of the local and foreign consultancy firms, with regard to the collection and exchange of important information, to promote the products of the private sectors outside Jordan.

3. Conduct economic studies for the purpose of promoting Jordanian exports. These studies and their structure should be approved by the general assembly of the Association.

4. Prepare and distribute special bulletins about the activities of the Trade Association among their members, hold specialized conferences and seminars with regard to export activities, and participate in the international trade fairs⁽⁷⁾.

3.4.2 The Export Regulations and Procedures

The main purpose behind of the Government export regulations and procedures is: (1) to control the trade transaction in terms of the quantity of export and the destination of goods (administrative purpose), (2) to prevent smuggling out of any necessary products, in particular those products which are subsidised by the Government for the purpose of domestic consumption, (3) to make sure that the product is exported to a country that it is allowed to be exported to (political reason), (4) to prevent the smuggling out of the value of export products (currency restriction), (5) to make sure that the exporter is qualified for exemptions of certain fees, and (6) to make sure that the exporter meets other Government requirements, such as income tax payment.⁽⁸⁾

In accordance with export regulation No.66 of (1974) the exportation of all goods outside Jordan are subject to export licenses. Export

⁽⁷⁾ Unpublished information from the Jordan Trade Association, A visit on 29 July 1990.

⁽⁸⁾ Unpublished information form the Minister of Industry and Trade. A visit on 2 August 1990.
licences are obtainable from the Ministry of Industry and Trade for a revenue stamp of 2 JD to be affixed on the application for the licenses. However, these export regulations have not ruled out the possibility of increasing or imposing new fees for revenue purposes, and it may be necessary sometimes, to limit the exportation of certain items. The duration of the license is three months, and it is not transferable to another person or firm.⁽⁹⁾

The following procedures must be followed by the manufacturer who wants to export (see Figure 3.1).

- Step (1): An export invoice should be prepared to provide detailed information about the items that are to be exported. This invoice should be submitted to the Chamber of Industry to be certified and, accordingly, to get a Certificate of origin in quadruplicate, a copy of which is kept in the Chamber.
- Step (2): The Certificate of Origin and an invoice should be stamped by the Ministry of Industry and Trade who will, in turn, issue an Export Permission in five copies. In this Ministry, the Export Permission should be approved by the Head of the Commercial department, and then by the Director of the Standard and Specification department who, in turn, issues a Certificate of Inspection (i.e., a Quality Certificate).
- Step (3): Before it is to be finally approved by the Director of Economic Co- operation and Promotion department in the Ministry of Industry and Trade, it must be checked and approved by the Director of the Jordan Commercial Trade Centre. The purpose of this procedure is to ascertain whether the exporter qualifies to be exempted form the export license fees.

⁽⁹⁾ Ministry of Industry and Trade: Export Regulations No. 66 of 1979, Trade Publication, Amman).

Figure (3.2): The Export Procedures in Jordan



Source: Developed by the researcher through the preliminary interviews
* The time required to finish the above procedures is estimated between 2
to 5 days.

- Step (4): The four documents (i.e., the Certificate of Origin, the Export Permission, the Certificate of Inspection, and the Export Invoice) should be stamped and approved by the Ministry of Foreign Affairs.
- Step (5): The duly certified copies of all documents must be certified by the embassy of the destination country.
- Step (6): According to the regulations of the Central Bank of Jordan, the exporters must also obtain a bank guarantee for securing payment by the export customers. However, this is not a requirement for Jordanian exporters selling by means of a Letter of Credit. Under the current system of the Central Bank, the exporter must post a guarantee for the amount of 10% of the export value of the goods that the exporters' customers will pay for the goods.
- Step (7): Upon the submission of the said four certified documents to the Ministry of Finance and Customs, an Export Declaration is issued. The Customs keeps copies of all documents. Copies of the Export Declaration will be despatched to the Departments of Statistics and Income Tax. On the border, or at the Port of Agaba the exported merchandise is checked and inspected by the customs, in accordance with the accompanied documents.

For obtaining and certificating all the mentioned export documents, the manufacturer has to pay fees. Also, to complete all the above procedures, the manufacturer needs at least three to five days (this estimation is based on our observation). This is not only because of the large amount of paperwork, but also because of the distances between the locations of the six institutions which are concerned with the export procedures (see Figure 3.1).

3.4.3 The Results of the Initial Interviews

In the previous section, information about several bodies of institutions who undertake export related issues, export procedures and regulations was presented. In this section, comments on the above by the chief executives of five manufacturing firms are summarized under the following subheadings:⁽¹⁰⁾

(1) The Government's Export Procedures

It was noted that there are a number of procedures and regulations that Jordanian manufacturers are obliged to follow in order to export their products, otherwise, they would not be allowed to sell their products to any foreign market. These procedures were presented in the previous section, they include mainly; obtaining a Certificate of Origin, export license, and Certificate of Inspection and others. These procedures were considered to be superfluous and time - consuming. One of the interviewees indicated that "the costs of those documents (export paper work) are not the problem, rather it is the length of time which one needs to complete those procedures, since there are a lot of signatures involved and many requirements need to be followed" (Interviewee B).

Another one said (Interviewee E) that: "it is much easier for me to import rather than to export" and then he added (Interviewee C) that "I wonder, how a new firm or those firms with little experience could be involved and encouraged in export activity if the Government intervention was so very high". Another executive confirmed (Interviewee A) that "if I want my firm to be involved in export marketing and to be competitive in

¹⁰ Taking into consideration the confidentiality of details with regard to the interviewees (i.e., names, companies, etc.) can not be enclosed in the thesis. However, the researcher posses documentation in this respect.

^{*} Interviewees are labelled here by A, B, C, D, and E.

international marketing, the Jordanian Government should simplify all these superfluous export procedures". He explained that the delivery time is very critical in capturing opportunities in foreign markets.

(2) Financial Services and Credit Facilities

Whilst Jordan can be considered to have a reasonably well developed financial system, the conservative nature of the system tends to hinder Jordanian firms from entering international markets due to:

- The scarcity of bank financing on competitive terms for foreign account receivables, i.e., post export finance.
 - The limited facilities for financing, required for particular inputs, i.e., pre - export finance.

Although it might be possible that exporting can be done through a confirmed Letter of Credit, which is opened in the country of the imports and confirmed by a local Jordanian bank, which in turn shifts the risk of non - payment from the exporter to his bank: "it does not always allow the exporter to offer competitive financial terms to his customers" (Interviewee D).

According to those executives, it is difficult to get access to the banking system, because they rely heavily on collateral and personal guarantees. One said "I wonder how small firms or a new firm can finance its export activity in the light of the tough banking system requirements" (Interviewee A).

Furthermore, the export credit guarantee and insurance against non payment risks system is not available in Jordan, and so many export transactions are done through a Letter of Credit, which in turn: "hampers the firm's ability to be more secure about their export transactions with foreign customers" (Interviewee C). The Central Bank of Jordan recently offered a special short term rediscount facility for commercial loans to exporters (under this procedure, exporters apply to commercial banks for loans at a lower interest rate "5% instead of 6%" and the Central Bank covers the difference). However, not many firms can benefit from this facility, because they are unable to meet the other conditions of the commercial bank. This may be due to the small size of manufacturing firms, as well as to the high risk associated with financing export activities (comments from Interviewees A and C).

(3) Duty Rebates Policy

Two of the executives who were interviewed (A and E) were not aware, or were unfamiliar with the Duty - Drawback system. Others who knew about it were not satisfied, because not only were the instructions of the drawback system not clear, but also the need to keep records and files about every transaction was time - consuming. Also, the refund sometimes takes time to be made.

(4) Tax Concessions

One of the executives said (Interviewee C) that "the Government of Jordan used tax exemptions as an instrument for industrial development rather than as an incentive for export oriented firms". In discussing the tax income policy, it appeared that tax incentives were provided for all firms, whether they did or did not export. In accordance with the Encouragement of Investment Law, profit of approved economics projects are exempted from income and social taxes for a period of 6 - 12 years (Jordan Five Year Plan, 1986 - 1990). However, the Encouragement of Investment Law only concentrated on tax exemptions, and ignored the other incentives such as reduction of production cost. One executive's opinion (Interviewee B) is that: "although the government claims that profits generated from export are exempt from tax, a high percentage of export for marketing, advertising, travel and other expenses connected with export is disallowed by the income tax authorities, and more often the decision about tax exemptions is arbitrary".

According to one executive's comment: "the income tax rebate policy does not really help our industry very much because one has to wait for a whole year before one realizes how much allowance he will get and how the allowances are to be distributed on the product policy" (Interviewee D).

(5) Trade Centres and Commercial Attaches

Currently, the Government of Jordan has got only three Trade Centres in Iraq, Sudan, and Egypt and four Commercial Attaches in Syria, Iraq, Sudan, and Egypt. However, there were no Commercial - Attaches or representatives for Jordan in European countries. One of the executives said "the lack of Trade Centres and Commercial Attaches in some foreign markets that we wish to export to make us unable to enter them" (Interviewee A).

(6) Provision of Information about Foreign Markets

Several bodies of both private and public institutions such as the Chamber of Industry, the Trade Centres and the Trade Association provide information about overseas markets. In discussing the importance of such information, it appeared that essential information is rarely available about the markets which some firms wish to export to, and sometimes the information available is out of date (Interviewees A, C, and D).

(7) Organising of International Trade Fairs and Exhibitions

The Jordan Trade Centres Corporation, in addition to other institutions, organise and participate in several international trade fairs and exhibitions with aims of expansion and promotion of export products of Jordanian manufacturing firms. In discussing the importance of such services, it appeared that these services are important if they are well organised in terms of time, place, and method of payment (Interviewees C & B). However, according to one executive, he said that: "I have been waiting for more than six months since the last exhibition^{*} and I have not received any payment for the goods which were sold at that exhibition" (Interviewee D).

(8) Exchange Rate Policies

In late 1988, the Government adopted a new policy for exchange rates as a result of the economic situation. This policy, which floated the Jordanian currency against the hard currency, was intended to enhance exporting. According to the executives, the devaluation of the Jordanian dinar would encourage firms to export more, but the Central Bank regulation concerning the currency exchange rate was not flexible and could be described as restrictive.

(9) The Central Bank Guarantee Requirements

A bank guarantee is needed as proof that the exported product is received by the foreign customer, and to make sure that money is not being smuggled out of the country. However, the high cost of the bank guarantee, the short period of the guarantee, and the uncertainty involved, do not encourage firms to export (comments from Interviewees A and E).

^{*} This exhibition was organised by the Tunisia government in 1989.

3.4.4 A Summary of the Results of the Initial Interviews

The major complaints of the five managers in respect of the government's export policies, regulations and assistances are summarised in the following points:

- The complexity of the export license requirements (the length of time and the number of the departments involved).
- The complexity of the customs clearance process.
- Inadequate tax incentives.
- Inadequate information about export markets that firms wished to export to.
- The length of the inspection process.
- Difficulty of the duty drawback system.
- The Central Bank regulations concerning the bank guarantee condition.
- Lack of export credit guarantee.
- Lack of Commercial Attaches in the foreign market that firms wished to export to.
- The high cost of paperwork which include: (1) export licence (2) bank guarantee (3) Certificate of Origin and (4) Certificate of Inspection.
- Lack of knowledge or awareness in respect of some government's export assistances, such as the duty rebate policies.
- Difficulty of access to the banking system.

The above complaints are included in our study's conceptual framework (see chapter 4) to represent some elements of the political - legal environment.

Summary

The purpose of this chapter is two - fold: (1) to present the main purpose of the preliminary in - depth

interviews and its structures, and (2) to provide some background information about the export promotion institutions, export policies, regulations and procedures in Jordan. Also, comments on the Government's export procedures, policies and assistances by the chief executives of the five manufacturing firms, are presented.

The variables which were generated from the results of the preliminary interviews will be integrated in the conceptual framework of this study to represent some elements of the political - legal environment.

In the next chapter, the main constructs of this conceptual framework will be elaborated.

CHAPTER FOUR

THE CONCEPTUAL FRAMEWORK FOR THE INTERNAL AND EXTERNAL DETERMINANTS OF EXPORT BEHAVIOUR

Introduction

4.1	The Nature of	the	Conceptua1	Framework			
4.2	How It Works						
4.3	Constructs of	the	Conceptual	Framework			

4.3.1 The Firm's Export behaviour and the Level of Exporting 4.3.2 The Constructs of the Firm's Internal Environmental Dimension

4.3.2.1	Managerial Goals and Aspirations Level
4.3.2.2	Management's Expectations
4.3.2.3	The Firm's Comparative Advantages
4.3.2.4	The Firm's Commitment to Export Marketing Activity
4.3.2.5	The Socio - Demographic Characteristics of Decision -
	Makers

4.3.3 The Constructs of the Firm's External Environmental Dimension

4.3.3.1	The Economic	and Commercial	Infrastructural				
	Environment						
			and the second se				

4.3.3.2 The Political - Legal Environment

Summary

CHAPTER FOUR

THE CONCEPTUAL FRAMEWORK FOR THE INTERNAL AND EXTERNAL DETERMINANTS OF EXPORT BEHAVIOUR

Introduction

The intention of this chapter is to discuss the conceptual framework for this study, the main dimensions of the conceptual framework, and the expected relationships among these dimensions.

4.1 The Nature of the Conceptual Framework

In Chapter Two, empirical studies on export behaviour, as well as the relevant theoretical literature on both international trade theories and organisational behaviour theories, were reviewed and integrated to develop a conceptual framework to guide this study.

The framework has tied together the factors (i.e., constructs) which are postulated to determine the firm's export behaviour and its level of exporting (see Figure 4.1). These factors are mainly derived form two broad dimensions, i.e., the firm's internal and external environment. According to the organisational and behavioural theories, the firm's business behaviour (e.g., Cyert and March 1963) is linked inseparably to the environment in which it takes place. Therefore, the firm's export behaviour (or the level of exporting) is thought to be a function of the interaction of the constructs of both environmental dimensions.

However, the effect of the external environmental dimension has been given little attention in previous export studies. In other words, the effect of the internal environmental dimension was the main focus of the previous studies in this field (see Chapter Two). The study's framework developed here is based partially on the Cavusgil and Nevin's model (1981). Their model was composed of four categories of the internal determinants of the firm's export activity, they are; (1) managerial aspirations (2) management's expectations (3) the firm's comparative advantages, and (4) commitment to export marketing activity (see section 2.3.2 of Chapter Two).

Based upon the studies of organisations and international business, Cavusgil and Nevin's model is modified here to include other determinants of the export activity (i.e., the firm's external environmental measures). These determinants are proposed under two major constructs; (1) the economic and commercial infrastructure, and (2) the political - legal environment.

4.2 How It Works

The conceptual framework developed here suggests that the firm's export behaviour and its level of exporting is thought to be a function of the interaction of the constructs of the internal and external environment (see figure 4.1). For example, the extent of the firm's commitment to export marketing activity, i.e., the willingness of the management to allocate resources to export marketing functions is assumed to be a function of the level of the managers' perceptions of their firms' internal and external environmental characteristics (e.g., Wiedersheim – Paul *et al.* 1978, and Reid 1981). These perceptions interact with the managers' characteristics (i.e., value system, experience, goals and aspirations) which result in an overall impression of the desirability of exporting. In turn, the level of desirability of exporting (i.e., the extent of the management's expectations in respect of the impact of the achievement business) is also thought to be a function of the manager's perception of:

Figure (4.1)

The Conceptual Framework of the Influence of the Interaction of the Firm's Internal and External Environment Upon its Export Behaviour and the Level of Exporting



* This model was partially developed by Cavusgil and Nevin (1981), but it is modified by the researcher to include the influences of the firm's external environment.

(1) the firm's export capabilities, i.e., the level of confidence of management with their firm's comparative advantages (e.g., Johnston and Czinkota 1982 and Schlegelmilch 1986), (2) the firm's domestic conditions, and (3) the foreign market conditions. In this context, the favourable perceptions of the foreign market conditions (e.g., the high expectation of better opportunities in the foreign market), coupled with the confidence of management with their firm's capabilities to export, the higher it is likely that their firm' will be committed to export marketing activity, which should result in a higher level of exporting. However, if the domestic market conditions are perceived to be unfavourable at the same time (e.g., the poor marketing commercial infrastructure in the domestic market), the firm's level of exporting might be decreased to the same degree.

4.3 Constructs of the Conceptual Framework

The constructs of each dimension are presented below and a $\underline{brief}^{\approx}$ reference is given about the studies which were concerned with them. Furthermore, the expected relationships among these dimensions are clearly defined and discussed throughout the presentation of each construct.

4.3.1 The Firm's Export behaviour and the Level of Exporting

There are two indicators used separately to achieve the study's objectives: the level of exporting and export behaviour.

The level of exporting (i.e., the level of the firm's export involvement) is used to indicate the extent to which a firm depends upon export marketing activity for its business goals. In other words,

^{*} A full details about these studies is given in Chapter Two.

the extent to which a firm has adopted and implemented export marketing activity as an alternative business strategy.

Export studies, such as Axinn (1985, 1988), Kolhede (1984) and Schlegelmilch (1983, 1986) indicated that there was no agreement on how best the level exporting can be measured. However, the most common measure which has been used as an indicator of the level of exporting is the export sales level as a percentage of the firm's overall sales volume. This measure is adopted here for the following reasons: (1) it is readily available, (2) it is a direct expression of the overall importance of export to the firm, (3) from a national economic planners' view - point, the total export volume, which is aggregate of the export level of industry, is of primary concern, (4) it facilitates the comparison of the findings of the present study with past research in export.

However, the use of this measure is not without disadvantages such as the time lapse between the use of this measure and the time of the data collection. In other words, the use of this measure may not be representative of the level of exporting. Nevertheless, the researcher believes that time is not a problem, because any major changes in the firm's environment takes a long time to filter through.

The firm's export behaviour is used here as an indicator for the differences between exporters and non - exporters on the basis of their environmental measures (internal and external). In other words, why are some firms involved in exporting and others are not.

Although this measure (i.e., export behaviour) is used in dichotomous categories * (i.e., exporters vs. non - exporters), it is adopted here for the following reasons:

^{*} This measure was used by many studies in this field, such as Cavusgil and Naor 1987, Daniels and Goyburo 1976, Tesar 1977, and Layton and Dunphy 1970.

(1) To facilitate the comparison between the results of using it and the results of using the other measure (i.e., the level of exporting). This comparison will help the public policy - makers to formulate appropriate export promotional programmes for enhancing the level of exporting and encouraging other firms to become involved in exporting.

(2) Understanding the differences between exporters and non exporters is of primary concern not only to the government, but also to the potential exporters who wish to enter foreign markets. This comparison will help them to identify the types of changes which should be implemented within their organisations in order to become involved in exporting.

(3) To facilitate the comparison between the findings of this study and the findings of previous works in this field (i.e., export behaviour).

Same stokazatoja - 1990.) - Mr. 1996. - 198. - 1981. - 1981. - 2008.

4.3.2 The Constructs of the Firm's Internal Environmental Dimension In the literature of the theory of a firm, the firm's internal dimension is described as a function of the management's goals, policies, abilities and position with past and present activities (Cyert and March 1963). In the empirical export marketing behaviour studies, Cavusgil & Nevin (1981) have advanced four broad categories for the firm's internal determinants. They include: (1) managerial goals and aspirations, (2) management's expectations, (3) the firm's comparative advantages, and (4) commitments to export marketing activity.

These above constructs, to some extent, were found to be significantly related to the export behaviour (see section 2.3.2 of Chapter Two). However, the types and the directions of relationships

among these constructs of internal determinants have not been well specified in previous studies. In particular, Cavusgil and Nevin (1981), in their export marketing behaviour model, did not define the types of relationships among the variables called 'background', i.e., the firm's differential advantages and its level of managerial aspirations. Therefore, the aim of the present study is to find out empirically the impact of the interaction of these constructs upon the firm's export behaviour, as well as its level of exporting.

In this study, the socio - demographic characteristics of decision - makers have been added as a fifth construct for the internal dimension, because the firm's desire to export is assumed to be a function of the manager's value system, experience, and aspirations level, etc. (see for example, Reid 1981).

4.3.2.1 Managerial Goals and Aspirations Level

The concept "aspiration" is introduced to decision - making theory by psychologists; and it is widely discussed in the theory of the firm (Cyert & March 1963). This behavioural concept of aspirations is defined as a determinant of risk - taking behaviour (McGuire 1964). Siegel (1957), who based his discussions of aspiration level on the work of Kurt - Lewin, viewed aspirations as "a person's utility of the goals in question". Another behavioural scientist emphasised the aspirations level as "the value which the individual places upon the achievement of the goals rather than upon its utility" (Atkinson 1957).

Regardless of variations in definitions, the implications of this construct (i.e., managerial aspiration) put forward by Cavusgil (1976) were that decision - maker's preferences for basic business goals such as profit, growth, or the importance he places upon the achievement of

each goal is a direct determinant of his decision - making behaviour. Empirical studies supported this by reporting a strong relationship between export behaviour and the level of managerial aspirations to such goals growth, profit and security (see section 2.3.1.2.1 and Table 2.2 in Chapter Two). Therefore, the strength of importance that is placed upon the achievement of the basic business goals, such as growth, security, and profit, is more likely to explain the variations of the firm's export behaviour. For example, Cavusgil and Nevin (1981) found a strong relationship between the firm's export behaviour and the managerial keenness in pursuing basic business goals, such as growth and profit. However, previous empirical export studies did not investigate the type and the direction of the relationship between the managerial aspiration level and other constructs developed here in respect of the firm's export behaviour and its level of exporting.

The managerial goals and aspiration level are constructed and measured here in terms of the importance that the decision - maker places upon the achievements of each of the following exporting goals:

- (1) Improving the firm's market position
 - (2) Increasing the firm's overall profit
 - (3) Securing steady growth for the firm's product (stability)
 - (4) Reducing the cost of production
- (5) Increasing the sales volume of the firm's exporting products
 - (6) Developing the firm's market strategies.

4.3.2.2 Management's Expectations

The effect of management's expectations on business behaviour has been recognised and emphasized by many researchers in macro - economic literature, as well as in the theory of the firm. There have been several attempts to measure empirically the significance of the impact of the management's expectations (e.g., Bilkey 1978, & Cavusgil and Nevin 1981). In the theory of the firm, Cyert & March (1963) have included the "theory of expectations" as one of four major elements of the behavioural model of the firm.

In export marketing behaviour, management's expectation is viewed as the decision - makers subjective evaluations of the profitability, riskiness, and costliness of exporting (Cavusgil 1976), and is based upon the decision maker's own or other firms' experiences, as well as their perceptions of the potential impact of the constantly changing environment of international markets (e.g., Bilkey & Tesar 1977 and Cavusgil & Nevin 1981).

Several researchers have reported a strong relationship between the management's expectations and the firm's export behaviour (for more details, see section 2.3.1.2.1 of Chapter 2). But they did not specify any relationship between the construct of management's expectations and other constructs of the present study's model in respect of the level of exporting.

Therefore, the aims of this study are: (1) to investigate the effect of the interaction of the management's expectations level and the other constructs of this study's model upon the firm's export behaviour (exporters vs. non - exporters) and its level of exporting, (2) to find out whether the existing relationship between the level of exporting and the management's expectations can be extended to the firms investigated here.

Based on the review of export marketing literature, the elements of management's expectations level can be constructed in terms of the motives for, and the benefits of exporting, the perception of riskiness, and the perception of export operational obstacles (i.e., these obstacles which are mainly related to the firm's inability to undertake export marketing activities). These elements and the relevant variables can be seen in Table (4.2). However, it is worth

Table (4.2)

The Elements of Management's Expectations

Elements	Influencing Variables
1. Motives for, and the ben of exporting to the firm	 efits Overcoming instability demand on the local market. The expectations of management with regard to the effect of exporting upon the firm's annual sales (growth). Preventing dependency on the local market. Overcoming competition on the local market. Using the firm's excess production capacity. Overcoming the small size of the domestic market*. The expectations of management with regard to the effect of exporting on the firm's profits. The expectations of top management with regard to the effect of exporting on the firm's security (stability). Benefiting from the government's export assistance. Keeping pace with local competitors we wish to export to. Better opportunities for our products in the foreign market.
2. Perception of riskiness	 Exchange rate risks* Transportation risks Production risks Collection on foreign credit sales Political risks or instability

^{*} Sources: These variables have been selected from several empirical studies on export behaviour (see sections 2.3.1.2.3 and 2.3.1.2.2 of Chapter Two). However, while some of them had been modified, others* were developed by the researcher in order to suit the reality of the export situation in Jordan.

Table (4.2) continued

 Perception of export obstacles 	
-	Difficulty in obtaining sales representative.
13.3.3 The First's Conserva-	Difficulty of pricing the product in the foreign market.
The term "Gospara" ive 4	Difficulty of maintaining control over foreign middlemen.
lated to the opecial	Inability to provide follow up services to overseas customers.
initian as other time-	Inability to provide credit facility
everal constructors have -	Difficulty in collecting money on foreign sales.
naran artatica that tanan-	Insufficient knowledge of communicating with overseas' customers.
maineren (ist marm data-	Difficulty of locating the demand for our products.
and Tables 2.3 and 7.7. Cha-	Lack of personnel expertise for handling export tasks.
the derived (ros the to-	Inability to establish distribution systems in foreign markets.
- Linka (technika) -	Size of production*.

mentioning that some of the variables of the management's expectation have been modified, and others have been developed by the researcher in order to suit the reality of the export situation in Jordan.

4.3.2.3 The Firm's Comparative Advantages

The term "Comparative Advantage" or "Differential Advantage" is related to the special characteristics that a firm possesses in comparison to other firms. In export marketing behaviour studies, several researchers have sought to explore the set of a firm's characteristics that seemed to facilitate a firm's export marketing involvement (for more details, see sections 2.3.1.2.2 and 2.3.1.2.5 and Tables 2.3 and 2.7, Chapter 2). These advantages (characteristics) are derived from the nature of the firm's capability, product features, technological orientation, managerial experience and knowledge, in addition to such things such as competitively priced products, technically superior production methods, marketing techniques and proximity to the market (e.g., Tesar 1977, Cavusgil 1976, Kolhede 1984, and Schlegelmilch 1983, etc.).

The effect of a firm's differential advantages were found to be significantly related to the firm's export behaviour (e.g., Cavusgil and Naor 1987). It has been indicated in previous studies that a firm's possession of such unique advantages, and the extent of the decision - maker's confidence in them would contribute to their willingness to consider export marketing as an alternative possible strategy. In other words, the uniqueness of advantage may serve as "initiation- evokers" for firms that are experimenting with exporting (Wiedersheim - Paul *et al.* 1978). However, they are insignificant by themselves (i.e., taken separately) to initiate exporting. For example, empirical findings on the relevance of a firm's size as a

Table (4.3)

The Elements of the Firm's Comparative Advantages*

Elements	Influencing Variables
1. Firm's demographi	ic
1.1 Size	
	- Number of employees
	- Size of the capital assets
	- Total gross sales volume
1.2 Experience	
	- Number of years in business
	- Number of years in exporting
2. Technological str	rengths
	- The superior product equipment
	- The superior production methods
3. Product character	ristics
	- Product uniqueness
	- Product quality
	- Product competitive price
4. Market competitiv	re
	- The ability to modify the product
	- Ability to offer special discounts*
	- Proximity to foreign market
	- Ability to hire highly skilled labour
	- Ability to modify organisational system
	- Strength of management
	- Reputation*
	- Firm's toobnological standard lovel

^{*} Sources: These variables were borrowed from several export studies (see section 2.3.2.7 and Table 2.3 of Chapter Two). Other* variables were developed by the researcher.

source of differential advantage for export behaviour have been mixed, because the larger size firm allows for a greater availability of financial, production and managerial resources (Cavusgil & Nevin 1981). Therefore, the relationship which appears is not between size and export behaviour, but between the various advantages which are associated with the larger size firm and export behaviour. For more details about the findings of the impact the other sources of competitive advantages, see section 2.3.1.2.2 of Chapter Two.

The aims of this study are: (1) to investigate the impact of the interaction of the firm's comparative advantages and other constructs of this study's model developed here upon the firm's export behaviour and the level of exporting (2) to explore whether the existing relationship between the firm's level of exporting and the elements of the firm's comparative advantages can be extended to our study's investigation. These elements and the relevant variables that are related to the firm's comparative advantages are constructed in terms of four sources of comparative advantages, they are; (1) firm's demographic (2) product characteristics (3) technological strengths and (4) market competitive strengths (see Table 4.3). Being aware that some of the variables of these elements have been modified by the researcher in order to suit the current export situation in Jordan.

4.3.2.4 The Firm's Commitment to Export Marketing Activity

The term "commitment" is viewed as the extent of the firm's allocation of resources, as well as the willingness of the management to adapt their organisational and marketing policies to foreign market conditions (e.g., Kolhede 1984). The effect of the "management's commitment" upon the export behaviour is recognised and emphasized by many researchers. They assert that a favourable orientation, keen

interest, willingness to allocate resources, and to adapt organisational and marketing mix policies to meet foreign market conditions are important determinants of export marketing involvement among firms (e.g., Pinney 1970; Khan 1975; Schlegelmilch 1983, 1986, Kolhede 1984, Cavusgil & Nevin 1981; Kaynak & Kothari 1987; Sullivan & Bauerschmidt 1990; Aaby & Slater 1989; Walters & Samiee 1990; and Gemunden 1991).

Involvement in export marketing requires that a firm devotes human and financial resources as well as management willingness to carry out tasks that are new to the firm, and for building the infrastructure of export marketing. Foreign market potential needs to be assessed, policies regarding export marketing need to be formulated, production for foreign markets need to be planned, new staff need to be hired and trained, and so on.

Although the commitment of top management, in terms of the extent of resources allocated for exporting, was found to be significant in determining the firm's export behaviour, there is still an argument as to whether such a relationship can be confirmed in the present study's investigation. It is believed that the degree of the firm's commitment to export marketing activities is not widespread among the firms operating in developing countries such as Jordan.

As far as the degree of the commitment is concerned, there is a need to find out the effect of the interaction of the management's commitment to export activity and other constructs of this study's model upon the firm's export behaviour and the level of exporting. The relevant variables that are related to the firm's commitment to export marketing activity are presented in Table (4.5).

Table (4.4)

The Elements of the Firm's Commitment to Export Marketing Activity

Elements	Influencing Variables
1. Willingness to change	. Romerger, there are insightlearn b
hemselves faction arguinter-	Willingness to modify product policies to foreign markets. Willingness to modify organisational
brie cherneteristics (for -	and control system. Willingness to change pricing policies
aula 2.7, Chapter 3). This-	to foreign markets. Willingness to modify distribution policies to foreign markets.
2. Export marketing activit	ies
Therefore, 21 15 crows	Participating in international trade fairs.
	Constantly seeking export opportunities* Planning systematically to explore foreign markets.
agent in the dira's erport-	Promotion of products in foreign markets Having a formal policy for handling
an des intel of experting 10	several export activities. Using export research.
This study ales to (1-	Sending sales representative to foreign markets*.
anora the first level ->	Carrying out a systematic research of possibilities in exporting.
asserbate theree attends a	icons (ik Taras invostigates), as to the

there there is my relationship Appress the decipies - values

^{*} Sources: These variables were derived form several empirical studies of export, see section 2.3.1.2.4 and Table 2.4 of Chapter Two. Other variables* were developed by the researber.

4.3.2.5 The Socio - Demographic Characteristics of Decision - Makers

In export marketing behaviour studies, the decision - maker's socio - demographic characteristics such as age, experience, education level, and proficiency in foreign languages, are found to be critical not only to export initiation, but are also significantly related to the firm's export behaviour. However, these are insignificant by themselves (acting separately) to initiate exporting. Empirical export marketing studies produced conflicting findings with regard to each of these characteristics (for more details, see section 2.3.1.2.5 and Table 2.7, Chapter 2). This might be related to the fact that these characteristics were investigated separately (i.e., as a constant factor) rather than as concomitant variables (i.e., as a group).

Therefore, it is expected that these characteristics of the decision - makers taken together are more likely to have a larger impact on the firm's export behaviour (exporters vs. non - exporters) and the level of exporting than taken separately.

This study aims to find out whether there is any relationship between the firm's level of exporting and the decision - maker's demographic characteristics among the firms investigated, and to find out whether there is any relationship between the decision - makers' characteristics (taken together) and other constructs of this study's model upon the firm's export behaviour (exporters vs. non-exporters).

Table (4.5)

The Socio - Demographic Characteristics of Decision - Makers

Elements	Influencing Variables
1. Demographic Characteristic	Successive accessive the tradition of the
	- Age
2. Education Background	
IT'S The adjustments and cable	- Level of education
	- Proficiency in English
	language skills
3. Experience	
White state to wanted with	- Number of years in business

Sources: These variables were derived form several empirical export studies, see section 2.7 of Chapter Two. 4.3.3 The Constructs of the Firm's External Environmental Dimension

As we indicated in Chapter Two empirical export studies with regard to the effectiveness of the external environment upon the firm's export behaviour and its level of exporting are still few and fragmented. Therefore, much of work here is based to a large extent on the firm's organisational and economic studies. The firm's external environmental dimension is proposed under two major constructs:

(1) The economic and commercial infrastructural environment

(2) The political and legal environment.

The purpose of this study is to explore empirically whether there is any relationship between the firm's external environmental measures and its level of exporting. Also, to find out the effect of the interaction of the internal environmental measures and the external environmental measures upon the firm's export behaviour (exporters vs. non - exporters) and its level of exporting.

4.3.3.1 The Economic and Commercial Infrastructural Environment

Based upon a review of the economic literature, the economic environment of a country tends to have a greater impact on the firm's activities and the practices which are required to be done (e.g., Keegan 1984). Therefore, one could expect that the firm's export behaviour can be viewed as a result of the economic conditions that exist in its environment.

It was found by studies of international business that orientation towards the foreign market is somewhat related to the domestic market conditions (Rao *et al.* 1990). In other words, the domestic market conditions may serve as external "initiation - evokers" that cause the firm to move or to consider the foreign markets as a possible alternative strategy. These domestic market conditions include: the

size of the local market in terms of population and income per capital, pattern of demand and consumptions, degree of competition, availability & quality of commercial infrastructure, availability & quality of resources (raw material), the technological development, in addition to other economic indicators such as inflation rate, exchange rate, interest rate and so on (e.g., Cateora and Hess 1987, and Terpstra 1988).

On the other hand, these economic conditions may act as a hindrance for firms wishing to be involved in export marketing. For example, while the small size of the domestic market in terms of population or the income per captia may cause the firm to move into other markets, the inadequate commercial infrastructure of the local market may hinder its ability to be competitive in foreign markets, which in turn affects its level of exporting.

Based on the consequences of the preliminary interviews which were undertaken with the initial respondents of this study survey (see Chapter 3), the domestic market conditions of Jordan, where the study has been conducted, are expected to have a negative effect on the firm's export behaviour. Despite the fact that Jordan has a small domestic market, in terms of population and income per capital, which might encourage firms to become involved in exporting (see section 1.7.2 and Table 1.1, Chapter One), the inadequate commercial infrastructure, the low quality and shortage of raw material, and other economic indicators, such as inflation rate, exchange rate are very likely to effect their level of exporting negatively.

Therefore, this study aims to explore the effect of the domestic market conditions upon the firm's export behaviour (i.e., exporters vs. non - exporters) and the level of exporting. Furthermore, the study aims to find out whether there is any relationship between the

Table (4.6)

The Elements of the Economic and commercial infrastructural Environment

Elements	Influencing Variables
1. The Domestic Market Co	onditions
inginar, and its level of	According may also take two population
1.1 The Commercial Infra	astructure
	- Availability of transportation
	- Availability of information
	- Availability of warehouse facilities
	- Ease of access to financial institution
	- Availability of communications
	- Availability of export credit*
	insurance
	 Availability of effective trade* companies
	- The cost of warehouse*
	- The cost of highly skilled labour
	- The cost of communication
	- The cost of transportation
1.2 Raw material resource	es
the annual see, the person	- Availability of raw material*
	- The quality of raw material*
	- The cost of raw material*
1.3 Other domestic market	teign weiker conditions, and similar the
indicators	- The demostic market demand
	- Degree of competition in the
	domestic market
	- The interest rate*
2 Foreign Market Conditio	ons
	- Degree of competition in the
	foreign markets
	- Geographical distance
	- Government's trade restrictions
	- The political stability*
	- Business practice in foreign markets

* Sources: These variables have been selected from several empirical studies on export behaviour (see section 2.3.2.1 of Chapter 2). However, while some of them had been modified, others* were developed by the researcher in order to suit the reality of the export situation in Jordan. domestic market conditions and the other constructs of this study model which developed here upon the firm's export behaviour and the level of exporting. The variables related to the domestic market conditions are presented in Table (4.6).

The effect of foreign market conditions on the firm's export behaviour, and its level of exporting may also take two forms: as a motivator or as a barrier. For example, while the attractiveness of the conditions of foreign markets may serve as a motivator for firms to be involved in, such as the availability of high demand or expected good marketing opportunities, the restrictions imposed on dealing with these markets may act as a hindrance for those firms wishing to operate there. In this study, it is expected that the foreign market conditions to have a greater effect on the firm's export behaviour and its level of exporting.

To summarize, the purpose of this study is to find out the extent to which the firm's export behaviour, and its level of exporting can be explained by the foreign market conditions, and whether there is a positive or negative relationship between the foreign market conditions and other constructs of this model upon the level of exporting and export behaviour. The variables related to foreign market conditions are presented in Table (4.6).

4.3.3.2 The Political - Legal Environment

It has been stated that a country's economic structure and development is a reflection of its political - legal environment characteristics (Keegan 1984). The international business literature as well as the economic literature assessed that the political forces of a given country tend to have a greater impact upon many aspects of business and economic behaviour. It tends to determine the extent of

freedom in which firms are committed to conduct their business in their pursuance of business goals. Therefore, the degree of government intervention on the firm's activities is very likely to determine their business behaviour.

In export marketing behaviour, the Government interventions can take two forms, as a hindrance or promoter. Whilst the Government can act as a hindrance through a number of restrictions, such as tariff, quota, legislation, procedure requirements, such as licence and inspections, and so on, it can act as promoter through its actions in the area of tax incentives, trade agreements, free trade zones, provision of information and so on.

As far as the effect of the Government's policies and procedures with regard to export is concerned, the Government of Jordan has adopted several export policies recently in order to enhance and to facilitate the level of exporting. Policies which resulted from this preliminary study interviews are; trade agreement policies, credit discount programme, devaluation of the Jordanian currency against foreign currencies, the temporary – entry system, the income – tax rebate policies, free – trade zones facilities, the duty – drawback system, the provision of information and provision of managerial services (e.g., trade fairs) through private institutions, such as the Chamber of Industry and the Trade Centre.

This study will attempt to find out if there is any relationship between the firm's level of exporting and the degree of importance attached to the Government's export policies among firms. It will also try to investigate whether there is a significant difference between exporters and non - exporters in terms of their evaluations of these Government's export measures.

Table (4.7)

The Elements of the Political - legal Environment *

Elements	Influencing Variables
1. The government's export	
assistances	
	- The trade agreement policies
	- The export credit discount
	- The duty - drawback system
	- Provision of information
	- Managerial services
	- Free trade zone facilities
	- Devaluation of the JD
	- The Temporary Entry system
	- The government tax policy
2. Knowledge or awareness	
of the government's export	and should adve should be been and
assistances policies	
	- Free Trade Zones facilities
	- The duty - drawback facilities
	- The export credit discount
	- The managerial services
	such as trade fair services
	- The temporary entry system
	- The Chamber of Industry services
3. The government's export	
procedure and assistances	
related problems	
	- The customs export clearance process
and the Appendix Lar 170	- The export license requirements
	- The inspection process
	- The cost of export documents
	- The duty - drawback instructions
	- The high import duties imposed on
	- The bank guarantee requirement
	- Ine balk guarantee requirement
	- The lack of Commercial Attaches
The Had not clother to	ters in the extentionship between it

^{*}Sources: All of these variables have been developed by the researcher through the preliminary interviews, see Chapter Three.

It is expected that the degree of the decision - maker's awareness and knowledge of these polices and assistances could result from the Government's attention in export promotion. Therefore, this study also aims to find out whether there is any relationship between the firm's export behaviour and the degree of the decision - maker's awareness and knowledge of these export policies.

On the other hand, the Government's policies and procedures with regard to export marketing are not without criticism, The results of the preliminary interviews with the initial respondents of this study showed that there were some complaints about them (for more detail see Chapter 3). The major complaints about them were: the complexity of the customs export clearance process, the complexity of the export licence requirements, the length of the inspection process for exported products, the complexity of currency exchange procedures, the high cost of export documents, difficulty in the duty – drawback system, the high duties imposed on the spare parts needed for production equipment, inadequate information about foreign markets, inadequate government tax incentives, and the lack of export credit guarantees system against non – payment risks.

This study will attempt to explore the extent to which these problems regarding the Government's policies and procedures are determining the firm's export behaviour and the level of exporting, and to find out whether there is any relationship between this construct (i.e., political – legal environment) and the other constructs of this model upon the firm's export behaviour and the level of exporting. The variables relating to this construct are presented in Table (4.7).
Summary

In this chapter, the conceptual framework for this study has been established through the integration of the factors that are assumed to govern the firm's export behaviour (exporters vs. non - exporters) and its level of exporting. Two broad dimensions are constructed to be related to the firm's export behaviour and its level of exporting, i.e., internal and external environment. The constructs of each dimension and the expected relationships among their constructs were discussed.

Table (4.8) represents a summary of the expected relationships investigated in this research. The generalised relationship stipulates that the level of exporting as expressed in terms of the export sales ratio is a function of the interaction of the internal variables and the external variables. The relationship is also applied to the firm's export behaviour (exporters vs. non - exporters). The combinations of these relationships represent the present study's framework, they are:

- The level of exporting / export behaviour is a function of the internal variables.
- (2) The level of exporting / export behaviour is a function of the external variables.
- (3) The level of exporting / export behaviour is a function of the interaction of the internal and external variables.

Table (4.8)

The Study's conceptual Framework Relationships: Summary



CHAPTER FIVE

RESEARCH DESIGN AND DATA COLLECTION

Introduction

5.1	Evaluating the Alternative Data Collection Methods
5.2	Deciding on the Most Appropriate Type of Questioning Methods
5.3	Deciding on the Appropriate Structure of the Interview
5.4	Deciding on the Domain of Respondents
5.5	Deciding on the Appropriate Key Respondent Approach
5.6	Deciding on the Appropriate Instrument of Measurement
5.7	The Questionnaire Development Process
5.8	The Interview Schedule and Response Rate
5.9	Preparing for Data Analysis

Summary

CHAPTER FIVE

RESEARCH DESIGN AND DATA COLLECTION

Introduction

The intention of this chapter is to explain the selection of an appropriate research design and data collection methods. This chapter represents the main constructs of the research and is divided into seven sections (see figure 5.1).

5.1 Evaluating the Alternative Data Collection Methods

The data required for this study is categorised into two main types: secondary and primary data.

Secondary data is defined as "data already collected and published for purposes other than the specific research needs at hand" (Kinnear & Taylor 1991). In addition to the saving in time and cost over primary data, secondary data also has other advantages, which include:

- (1) Helping to understand the problem under investigation
- (2) Suggesting improved methods to tackle the problem
- (3) Providing comparative data by which primary data can be interpreted and evaluated more meaningfully (Aaker & Day 1986).

The secondary data used in this research were the Government Census and other publications giving the external trade size, trade patterns, distribution and the structure of the industrial sectors which were selected for this study. These data were used to gain insight into the export position in Jordan, and to assist in identifying those sectors which show high export potential.

The other secondary data used in this study related to the existing literature concerned with the research problem. The purpose of using those sources of information was to have a better understanding of the

Figure (5.1)

Research Design Process



problem, and to determine the required data and the suitable method for data collection.

However, in spite of its importance, and the necessity for using it, as mentioned above, it was found that secondary data was not sufficient to solve the research problem. This was due to the major limitations of the secondary data, which related to finding the most suitable and the accurate data.

As mentioned in the first chapter, the problem investigated in this study has not been undertaken before in Jordan, or in any of the Middle Eastern countries in a similar economic position. Marketing research in general, and export marketing research, in particular, are very rare in Jordan. There is a lack of recorded data. El-Ansary (1985) points out: "Marketing is the most neglected function in the enterprise of Middle Eastern Countries". Also, Drucker (1958) noted: "In less developed countries, marketing and marketing channels are often neglected". Amerah & Samdi (1979) noted: "Marketing in Jordan has been neglected, unfavourably evaluated & that its role in economic development and in promoting exports has not been appreciated yet".

Therefore, and in spite of the cost in money and time, there was no alternative but to conduct a field study to collect the primary data required for this study.

Primary data can be obtained through experimentation, analogies and respondents (Kinnear and Taylor 1991). The experimentation and analogous methods were unsuitable because the limits of time and budget. Thus, the respondents method was chosen.

Parasuraman (1986) classified the primary data collection methods into two broad categories: questioning and observation. In the questioning approach respondents play an active role, while in the observing approach respondents do not directly interact, or

Figure (5.2)



The Process of Selecting the Appropriate Type of Data and the Appropriate Data Collection Method communicate with the researcher. The communication (questioning) approach was considered more appropriate method for this study because of time limitations and also the numbers of and types of variables that needed to be measured.

A key strength of the observation method is that it is more likely to provide more accurate data, since distortions deriving from respondents will be much lower than in studies employing questioning methods. Questioning methods are also less convenient for respondents, and require the need to secure their co- operation (Parasuraman 1986). However, Kinnear and Taylor (1991) suggest that these limitations can be controlled by properly designing the data collection instruments.

5.2 Deciding on the Most Appropriate Type of Questioning Methods

There are three main questioning methods: personal interview, telephone interview, and postal interview.

The telephone interview method was eliminated from the outset, because it had to be shorter than other two alternative methods, and the level of co-operation of the respondents was likely to be lower than free to free contact.

In this research, the personal interview questionnaire was preferred to the postal interview alternative, although it was more expensive, and probably more biased, because of the influence of the interviewer.

There are seven major reasons for this decision:

(1) The postal survey questionnaire would almost certainly remain unanswered by the firms in Jordan, due to a general lack of acceptance and familiarity with this means of gathering information, combined with a cultural reluctance to give written information to an unknown person.



Deciding on the Appropriate Communication Methods



(2) The personal interview questionnaire permitted us to ask a relatively larger number of questions. Clarification and follow up remarks were also possible to supplement the knowledge gathered.

(3) One of the purposes of this questionnaire was to obtain information coded by some firms "confidential" or classified, such as annual sales, export ratio, number of countries exported to, and so forth. This type of information is less likely to be secured by the postal questionnaire.

(4) The personal interview questionnaire allows better clarification of the meaning of terms or misunderstanding.

(5) The appointments were pre - arranged by telephone calls directly with the persons concerned. Thus, by personal interview, we derived a relatively higher percentage of response, about 73.6% of the population surveyed.

(6) Identity of the respondent could be ascertained, and we could also obtain general information about the respondent.

(7) Furthermore, it was expected that people would be more forthcoming in face to face interviews. In fact, in the preliminary interviews, the interviewees indicated that were more comfortable with personal contact more than indirect approaches.

The disadvantage of a personal interview using direct questionnaires are relatively limited. The most important disadvantage is the danger of personal bias. This potential disadvantage was limited by presenting the respondent with the questionnaire and asking him to complete it by himself (with the interviewer present).

5.3 Deciding on the Appropriate Structure of the Interview

There are two broad types of interviews: structured and unstructured, or standardised and unstandardised (Kerlinger 1986).

Structure was defined by Churchill (1987) as the degree of standardisation imposed on the questionnaire. Directness is the amount of knowledge about the purpose of a study communicated to a respondent.

The structured - direct technique is used in this research. This technique necessitates the questions being presented with exactly the same wording, and in exactly the same order to all respondents. The reason for standardisation is to ensure that all the respondents are replying to the same questions (Churchill 1987). The major advantage of using the standardised - direct interview, is that it is simple to administer and easy to tabulate and analyse.

5.4 Deciding on the Domain of Respondents

As discussed in the previous section, it was decided that a field study was necessary for the current study's objectives. Because this study is mainly concerned with the investigation of the export behaviour of manufacturing firms, firms engaging in any business outside this industrial category were excluded from the study, such as agriculture products or services.

The survey units in this study were the individual manufacturing firms of exporters and potential exporters. The decision of the choice of the individual firm as a focus for study derived from the nature and the objectives of the study. In other words, the investigation was conducted at the micro level.

Furthermore, the choice of the individual manufacturing firm as a unit for the field study was supported by the theory of the firm. For example, Cyert and March (1963) suggested that;

"the individual firm is a unit which coordinated and undertook critical aspects of economic activity".

The identification of the individual manufacturing firms in the country (Jordan) was done by obtaining names of all firms, as well as their addresses, from a variety of private and public sources, such as the Chamber of Industry, the Trade Centre, the Ministry of Commerce, and the Central Bank of Jordan. These sources were reviewed in order to identify the type of industrial sector, and the range of the number of firms in each sector.

Restrictions of time and financial resources made the inclusion of all firms impossible. It was decided to choose certain types of manufacturing sectors for the current study. On the basis of the study's objectives, and the data analysis techniques to be employed, two conditions were considered in the selection of the type and number of these sectors. These conditions were: (1) the industrial sector must exhibit a high export potential and contains a substantial number of exporting firms, and (2) their relative importance to Jordan's economy must be shown to be high. The relative importance of each sector is measured in terms of its contribution to the GNP (Gross National Product), see Chapter (1).

Based upon the two above conditions, six major industry sectors were selected for the field of this study. They were: Engineering, Clothing, Supply, Plastic and Leather, Chemical and Chemist, and Construction.

The total population of the firms in all the selected industry sectors constituted 526 firms which represented about 70% of all the manufacturing firms in the industry sectors in Jordan. The distribution of firms in the selected industry sectors is presented in Table (5.1).

Table (5.1)

Name of Sector	Number of Firms	Percentage	
Engineering Industries	94	17.9	
Clothing Industries	159	30.3	
Chemical & Chemist Ind.	46	08.7	
Construction Industries	78	14.8	
Plastic & Leather Ind.	59	11.2	
Supply Industries	90	17.1	
Total	526	100.0	

The Domain of the Study's Respondents According to the Type of Industrial Sector

5.5 Deciding on the Appropriate Key Respondent Approach

The major sources of data were the individuals to whom the self administered questionnaire was subsequently directed. Their selection a very important issue. Campbell (1955) suggested that the was informant would not be chosen for statistical representativeness, instead he would be chosen because he possessed special qualities. The informant should occupy a role that makes him more knowledgeable, regarding the issues under the study, and more capable of "speaking the language of the researcher". Pennings (1979) supported the use a single key informant where most of the informants occupy top executive, or ownership positions. He argued that managers at the higher level of management and owner managers were the key figures in dealing with the external environment and were suitably qualified to speak for the firm.

However, these views have come under criticism (Philips 1980, 1981 and Seidler 1974). The criticism has been that the single informant is not capable of providing reliable data. Using a single or a few informants can lead to unreliable data. Although there is still some argument regarding the particular reliability of the key informant, it is essential for this study that the target respondent should be the director of the firm. The reason is that the type of information sought makes it mandatory that the respondent be not only in a position of being the firm's policy maker, whose decision will have a strong influence on the direction the firm will pursue, but also must occupy a position that makes him knowledgeable regarding foreign trade involvement, as it relates to the firm's overall business operations.

Furthermore, the relatively small size of Jordan's manufacturing firms (in terms of number of employees and capital assets), as well as the consuming time and cost associated with the use of the multiple informant approach made it is essential to rely upon a single informant for collecting data for this study. In addition, previous works in exporting supported this notion by claiming the top executive to be the key informant in this type of study (e.g., Cavusgil 1976 and Olson & Wiedersheim - Paul 1978).

As a result, an effort was made to access the person at the higher level of management of the individual firm, i.e., the chief executive or the director.

5.6 Deciding on the Appropriate Instrument of Measurement

Measurement is defined as "the rules for assigning of numbers to objects in such a way as to represent quantities of attribute" (Churchill 1987). There are four general levels of measurement: nominal, ordinal, interval and ratio. However, the selection of the appropriate level of measurement is difficult. This arises mainly from disagreement over the statistics that can legitimately be used at the different levels of measurement (Kerlinger 1986). Churchill (1987)

suggested that the empirical evidence indicated that:

"None of the scaling devices is superior in all instances. Each one has its place nor is there one single optimum number of scale positions or single optimum conditions for other measured characteristics. The nature of the problem, the characteristics of the respondent and the planned mode of administration will and should affect the choice as to which technique should be used in a particular instance and what features the scale should possess".

In the first part of this study's questionnaire (from Q1 to Q101), the rating 7 point scale was used. The justification for using this type of scale was as follows: (1) It is relatively easy to construct and administer, and (2) Subjects generally find it easy to respond because the response categories allow sufficient expression of intensity of feeling (Churchill 1987).

Furthermore, the selection of the rating 7 point scale is based on the fact that empirical studies have suggested that scales with three or more points can, and do provide a valid measure (Jacoby and Matell 1971, Bendig 1954, Boote 1981, and Burns and Harrison 1979). Also, in discussing the validity and reliability of different scales, Churchill and Peter (1984) concluded that the reliability of different scales would be increased as the number of items used in the final scale as well as the number of the scale points increased.

Boote (1981) suggests restricting scales to five and seven points. On the one hand, any fewer than five points would reduce the scale's ability to discriminate, since the respondent would be less able to express refined gradations. On the other hand, more than a seven point scale would be less than the optimum, because of the limited increase in information gathered. Lehmann & Hulbert (1972) commented:

".. increasing the number of scale points reduces the rounding error as benefit, but may also increase cost of administration, non - respondent bias and respondent fatigue, since averaging tends to reduce the rounding error. When scale points aim to be averaged the cost of increasing the number of scale points will usually out-weight the benefit".

The nominal scale was employed in parts 2 and 3 of this questionnaire, which covered the socio - demographic characteristics of the decision - maker, as well as the firm's parameters. Though this scale is simplest amongst those available, it is appropriate for such category data (e.g., the type of industry, type of education, etc.). The questions in the nominal scale cannot be used for normal arithmetic calculating, adding, subtracting, multiplying or dividing.

5.7 The Questionnaire Development Process

The questionnaire development process used here was suggested by Churchill (1987), see figure (5.4). Steps (1) and steps (2) were presented in the previous section.

Step (3) determines the content of the individual question: The content of the questionnaire depends upon the type of data required to be collected, data collection methods, and the ultimate use of the results. Since this study is concerned with the impact of the firm's environment (external and internal) upon its export behaviour, and the level of its exporting, it is necessary that the main constructs of both external and internal environments be covered in the questionnaire.

Each respondent was asked about each of the variables which constitute each construct. These variables were drawn from the literature review, as well as from a series of preliminary interviews. All unnecessary or confusing questions were either altered or modified during the pre - test stage.

Step (4) determines the form of response to each question: Several forms of responses were suggested in the marketing research literature including the open - ended question, the multichotomous question, the dichotomous question and the scale. The open - ended questions were



* Source: Churchill (1987) p. 272.

eliminated due to the use of a structured - direct questionnaire.

Step (5) deciding on question wording: Since the rating 7 point scale was selected, the meanings of the questions were stated clearly and directly in simple language.

Step (6) deciding on question sequences: Several strategies are suggested to tackle the question sequences (see Churchill 1987), such as using simple interesting opening questions or using the funnel approach design, branching questions with care and placing classification information last.

As far as possible these strategies were used, for example starting the first part of the questionnaire with questions considered to be easiest and leaving the other questions which might be considered difficult or sensitive to the last. The general information about both the firm and the decision - maker (classification questions) were placed in the last part of the questionnaire.

Step (7) determine physical characteristics: The questionnaire was typed and revised several times by professional people before it was copied and distributed. When the questionnaire appeared to have a satisfactory appearance, it was copied and then pre - tested. All the comments and suggestions received, whether relating to the appearance of the questionnaire, or the wording of some parts, were considered when preparing the final copy. The only complaint which could not be rectified was about the length of the questionnaire. It was a very lengthy questionnaire, but all the variables covered were important for the study, and it was not possible to omit any of them. However, the response rate 73%, in comparison to similar studies, can be considered a satisfying indicator that the questionnaire was manageable.

Step (8) re - examination and revision of the questionnaire: Churchill (1987) suggested that each question should be reviewed to ensure that the question was not confusing or ambiguous, potentially offensive to the respondent, leading or bias inducing, and that it was easy to answer. The questionnaire was pre- tested at three different stages: Areas to be covered by the pre - test were:

- 1. The content validity of the questionnaire
- 2. The ease of understanding of the contents
- 3. The willingness and ability of executives to respond to the questions.

We indicated in the previous section that the contents of the study's questionnaire was based on an review of related literature on export behaviour, as well as on the results of preliminary interviews conducted with a number of authorities (in both public and private institutions). and a number of initial respondents in the manufacturing firms in Jordan. In the first stage, the first copy of the questionnaire which had been developed, designed and translated reviewed by researcher's supervisors. The English. was into questionnaire was then redesigned in the light of their suggestions and comments. At the second stage of the pre - test, staff members Department of Business at the University of Jordan. the of knowledgeable in international business and in questionnaire design reviewed the questionnaire and commented on its clarity and relevance.

After incorporating their comments in a revised questionnaire, stage three of the pre - test was carried out with few respondents firms. The presidents of 10 manufacturing firms were contacted and their cooperation solicited. Of the ten firms contacted, 8 managers were able and willing to participate in the interview. The validity of the questionnaire content was investigated through open - ended interviews. The content of the questionnaire was discussed. This

procedure allowed the researcher to check for possible misunderstandings, and also permitted assessment of willingness and ability to respond to the questions. As a result of this stage the questionnaire was re - edited for the final stage.

5.8 The Interview Schedule and Response Rate

The interview process for collection of data for this research was undertaken over five months. The response rate was 73.6% of the all firms surveyed (see Table 5.2), because interviews were planned as follows:

(1) An exploratory telephone call was made to each of the potential respondent firms included in the domain of the study's population, in order to identify the key respondent, the exact address of the respondent's firm (the place where the interview took place), and ask for their possible cooperation in order to make an appointment for the interview. The researcher was also able to visit many firms to ask for cooperation and to make an appointment for an interview. This was done with many of the respondents' firms, because their locations were easy to reach.

Having identified the name of the potential respondent's firms who had responded favourably to the interview, as well as the time and the place, the field study (the interviews) were then started. Each interview was held separately, and was carried out at the informant's place of business.

(2) Five recommendation letters were attached with each questionnaire. They were addressed generally to the President of each respondent's firm, in order to gain their full cooperation and assistance. These letters were signed by the sponsor of the researcher (i.e., the University of Jordan), the Minister of Industry and

Commerce, the Chairman of the Chamber of Industry, the Head Officer of the Export Association, and the researcher's supervisors (Appendix 4).

(3) The average time for each interview was approximately one hour, and after each interview was finished, appreciation of their cooperation was conveyed to them orally.

Table (5.2)

The Distribution of the Respondents' Firms in Terms of the Type of Industrial Sector

Type of Sector	Number of Firms No. %		
Engineering	75	19.4	
Clothing	105	27.1	
Chemical & Chemist	40	10.3	
Construction	59	15.2	
Plastic & Leather	44	11.4	
Supply	64	16.5	
Total	387	100.0	

As a result of this effort, the number of firms which were actually co - operative, and participated in the study was 387. This total represents 73.6% of all the domain of the firms selected. The distribution of the co - operating firms in terms of the industrial sectors which they belong to is given in Table (5.2).

One hundred and thirty nine did not respond representing about 27% of all firms approached. The reasons for their non - participation in this study are explained as follows:

(1) 23 firms were closed down at the time of the data collection.

(2) 51 firms refused totally to participate in this study, or to give any sort of information. This could be due to the fact that the collection of data was carried out at the same time that the Gulf crisis was taking place which made them unwilling to take part in this study. It is worth mentioning that about 20% of the exported products of Jordan were directed to Iraq (the country which was one of the main participants of the Gulf Crisis).

(3) 12 firms belonged to more than one sector (i.e., 7 firms belonged to 2 sectors, 3 firms belonged to 3 sectors, and 2 firms belonged to 4 sectors, which make a total <u>19</u> firms not included in this study). In this case, these firms were assigned to the major sector to which they belonged, in order to avoid a multiple response from the same firm.

(4) 19 firms did not participate in this study because their directors were out of Jordan at the time of the data collection.

(5) 27 of the respondents' firms failed to finish the interview completely (8 of them hastily completed the questionnaire, 7 of them completed half of the questionnaire, and the rest gave their apologies for their inability to continue the interview).

Table (5.3)

Type of Sector	Observed Frequencies (Fo)	Expected Frequencies (Fe)	$(Fo - Fe)^2$	(Fo - Fe) ² Fe
Engineering	75	69.16	34.11	0.493
Chemical	40	33.84	38.00	1.137
Supply	64	66.22	05.00	0.073
Clothing	105	117.00	144.00	1.230
Construction	59	57.40	2.56	0.005
Plastic & Leather	44	43.40	0.36	0.008
Total	387	note and Tayl	do 1591)	2.977

Chi - Square Analysis of the Study's Respondents In Terms of Industrial Sector

Industry sector total

— x 387

* Expected frequencies (Fe) = -

```
526
```

- * Calculated chi square value = 2.977
- * Degree of freedom = 5
- * Level of significance < 0.05
- * Tabled value of Chi square = 11.07

The chi - square test (goodness of fit) was employed to measure whether there was a significant difference between the distribution of the respondents' firms and the distribution of the non - respondents' firms in terms of the industry sector which each firm belongs to.

As can be seen in Table (5.3), the calculated chi - square value was not significant. Therefore, it can be concluded that each sector had a similar response rate.

5.9 Preparing for Data Analysis

Before starting the data analysis, it was necessary to undertake the preliminary steps of editing, coding, and tabulating the data.

Editing: The term editing, as used in marketing research, refers to the process of examining completed questionnaires and taking whatever corrective action is needed to ensure that the data is of a high quality (Parasuraman 1986). Editing is often done in two stages: the field edit and the central - office edit. The field edit is a preliminary edit, designed to detect the most obvious omission and inaccuracies in that edit. In this research, effort has been made to keep the data accurate.

Office editing: This involved a more complete and exacting scrutiny and correction of the completed return questionnaire (Churchill 1987). There are five areas in which the editing function should be concerned, they are: legibility, completeness, consistency, accuracy and response classification (Kinnear and Taylor 1991).

In this study, most of the editing work was done by the researcher himself. All questionnaires were inspected, to ensure that they were properly filled in, and that no significant omissions were allowed. The partially filled in questionnaire in which any questions were left unanswered omitted from analysis. However, if the omitted questions

were few, the questionnaire was allowed to be used in the final analysis and unanswered questions were assigned a missing value. Also, if the questionnaire appeared to be hastily filled in (for example, assigning number 7 for all the variables, then the questionnaire was eliminated from the final study).

Coding and entering the data: Coding means translating answers into both class membership and into a symbolic representation of this membership, usually a column and position designation on a punch card used for machine tabulation (occasionally coding is used in manual tabulation, but this is more of a type of shorthand than of a truly symbolic code) (Ferber 1974).

In this research, the coding was done manually. There was little difficulty in coding the questionnaire, since most of the questions were to be rated on a scale of seven points (Q1 to Q101), but the other questions, i.e., the firm's parameters as well as the socio demographic characteristics of the decision makers were categorically measured. Each edited and coded question was transferred to a Coding Sheet. Every completed and edited coding sheet was sent directly to the Computing Department to be entered into the computer and copied onto computer diskettes on a Mainframe.

This chapter explains the various processes of selecting and designing the research instrument, as well as the data collection procedures. Although secondary data sources are always important, they are not always sufficient to resolve the research problem. A review of the main data collection methods, including observations, experiments and questioning have been presented.

Questioning was the method chosen for this research, and justification for using this method rather than an alternative method was made. The advantages and disadvantages of the different methods were also discussed. For time and cost reasons the questioning method was the most appropriate method for this research. However, the questioning method was not free from drawbacks. These drawbacks and the measure taken to reduce their impact were presented in the chapter. The various methods of questioning techniques (direct interview, telephone interview and postal interview were also discussed). Direct interview selected as the best method for this research.

Several forms of the questionnaire were looked at including: direct non - structured, indirect non - structured, direct structured and indirect structured. On the basis of relative advantages and limitations, it seemed that the direct - structured form was the most appropriate form for this research.

The domain of the study population was also been explained, and the decision to select appropriate industrial sectors for the field of the present study justified. The selection of the appropriate key respondent approach to be contacted within the firms was discussed and

"the single informant approach" method was found to be most appropriate.

Several forms of measures were also explored. The reasons for selecting the 7 point scale, nominal scale and ratio scale for various parts of the questionnaire were explained. The questionnaire development stage was explained and the various stages suggested by the marketing research literature for developing a sound questionnaire were followed. The pre - test stage of the questionnaire proved to be useful in eliminating, adding and re - wording some of the questions, and finally, the stage of data preparation for final analysis was explained. These processes included: editing, coding, transferring the coded data to the coding sheet and data entry.

CHAPTER SIX

RESEARCH METHODOLOGY

Introduction

6.1 Classification of Statistical Techniques

6.2 The Statistical Methods Used for Research Objectives

6.2.1 Factor Analysis

6.2.1.1 Methods of Extracting Initial Factor 6.2.1.2 Factor Analysis Input / Output 6.2.1.3 Use of the Factor Analysis in the Study

6.2.2 Multiple Regression Analysis

6.2.2.1 Caution in the Use of Multiple Regression 6.2.2.2 Use of Regression Analysis in this Study

6.2.3 Simple Correlation Coefficient

6.2.4 Discriminant Function Analysis

6.2.4.1 The Applications of DFA 6.2.4.2 DFA Date Input / Output 6.2.4.3 Use of DFA in this Study

6.3 The Statistical Methods Used for Testing Research Hypotheses

6.3.1 T - Test 6.3.2 Univariate F Test 6.3.3 Chi - Square Test 6.3.4 McNemar Test

Summary

CHAPTER SIX

RESEARCH METHODOLOGY

Introduction

In the previous Chapter, the process of data collection, design of the questionnaire, and data preparation for the final stage of analysis were fully presented. The aim of this chapter is to provide a brief explanation of the statistical analysis techniques used in achieving the research objectives and hypothesis testing.

6.1 Classification of Statistical Techniques

Marketing research literature suggests different methods for data analysis which can be classified into three techniques according to the type of data and number of variables (e.g., Churchill 1987). Namely, Univariate, Bivariate, and Multivariate. The univariate technique is used if there is a single measurement of each of the Kth sample objects, or if there are several measurements of each of the Kth observations, but each variable is to be analyzed in isolation. The central tendency measures (mean, median, mode) and the measures of dispersion (standard deviation, relative and absolute frequencies), as well as the T - test, F - test, and chi - square test are among the suggested techniques which can be used.

The bivariate analysis technique allows the researcher to examine the interaction between variables taken two at a time. For example, the investigation of the relationship among pairs of variables. Suggested bivariate techniques are: linear correlation coefficient, rank correlation coefficient, contingency coefficient lambda, t- test

on regression coefficient, Mann - Whitney U - test, Kolmogorov -Smirnov test, chi - square test and others (Kinnear & Taylor 1991).

The multivariate analysis technique is concerned with the investigation of interaction among a set of variables. The multivariate techniques can be classified as either dependent or independent. The dependent methods imply that one or more variables are specified as being predicted by a set of independent variables, while the independent method implies that there is no variable selected as being a dependent variable.

The dependent method might include: analysis of variance (ANOVA), analysis of variance and covariance (ANCOVA), multiple regression, automatic interaction detection (AID), multiple classification analysis (MCA), and discriminant analysis (DFA). The independent methods might include: cluster analysis, factor analysis, latent structure analysis and non - metric multidimensional scaling.

The decision was made in the research to use a combination of the above data analysis techniques. From the multivariate techniques, factor analysis, regression analysis, and discriminant analysis were employed.

The following bases were used for selecting these statistical techniques. According to Churchill (1987) the selection of the appropriate technique depends on:

- 1. The type of data (nominal, ordinal, interval and ratio)
 - 2. The research design (dependency of the observation, number of
 - observations per object, number of groups being analyzed)
 - 3. The assumptions underlying the test statistics.

This research study focuses on the investigation of the effect of the firm's internal and external environmental measures upon its level of exporting and export behaviour. The external and internal environmental variables were measured on 7 point scale which was

assumed to have an interval property. This necessitated the use of various statistical techniques suitable for each level.

6.2 The Statistical Methods Used for Research Objectives

6.2.1 Factor Analysis

Factor analysis is an interdependence multivariate technique. It can be defined as a procedure that takes a large number of variables or objects and searches to see whether they have a small number of factors in common which accounts for their inter - correlation (Kinnear & Taylor 1991).

The common factor analysis assumes that each variable is a function of the same set of underlying common factors plus a factor unique to that variable. However, each variable has a different set of weigh associated with the factor analysis (Kim & Muller 1978).

In applying factor analysis one is interested in examining the strength of the overall association among variables in terms of a smaller set of linear composites of the original variables that preserve most of the information in the full data (Aaker & Day 1986). In other words, the factor analysis procedure involves finding a way of linearly transforming the original variables into a new smaller set of independent factors, which when multiplied together in a special manner will produce the original correlation matrix as closely as possible.

Factor analysis can be applied for two major functions. One function, is to identify underlying constructs in the data (Aaker & Day 1986) by deriving dimensions in the data which combine each group of similar variables under specific termed factors. A second function of factor analysis is simply to reduce a large number of variables to a more manageable set (Brown 1980). The smaller set of variables

express that which is common among the original variables. Generally speaking, factor analysis can be useful to the analyst in three ways (Wells & Sheth 1971). Firstly, it can point out the latent factors or dimensions that determine the relationship among a set of observed or manifest values. Secondly, factor analysis can be helpful in pointing out the relationship among observed values that were there all the time but not easy to see. Thirdly, factor analysis is useful when things need to be grouped.

6.2.1.1 Methods of Extracting Initial Factor

The main objective of the extraction step in exploring factor analysis is to determine the minimum number of common factors that would satisfactorily produce the correlation among the observed variables (Kim & Muller 1978).

In this research, the principal component analysis is employed. According to Hair *et al.* (1987), this method is an appropriate one when the objective is to reduce a large number of variables to a smaller set of uncorrelated variables for subsequent use in a regression or other prediction techniques, and also when the researcher has prior knowledge suggesting that unique and error variance represents a relatively small proportion of the total variance.

In using factor analysis, the researcher must in one way specify the number of factors to be considered, since we normally begin an analysis without knowing how many factors, or which factors underlie a set of manifest variables. Jackson (1983) stresses the importance for the investigator not to leave out any important factors. If this occurs, the results will be basically worthless. On the other hand, if the researcher instructs the programme for many factors more than the important ones, those factors will appear on the programme output but contribute little to the explanatory power of the factor model.

In fact, carrying the analysis too far has penalties, it is wasteful of computer time as well as obscuring the meaning of the findings. An exact quantitative method for determining the number of factors to rotate has not been developed, therefore, two rules of thumbs are simultaneously used here for this purpose:

(1) Interpretability - by this method the smaller factors are retained only if they have sufficient substantial meaning to be interpreted (see for example, Nunnaly 1967).

(2) Eigenvalue - by this criterion the analysis is limited to the number of factors with an eigenvalue^{*} greater than one (Corsuch 1973). The rationale for this approach is that any individual factor should account greater for at least the variance of a single variable if it is to be retained for interpretation (Hair *et al.* 1987).

6.2.1.2 Factor Analysis Input / Output

The input of factor analysis is usually a set of variable values for each individual or object in the sample. In this present research, the input is a set of attributes that relate to each construct alone. In other words, the variables which express each construct of the firm's environmental dimensions were used as inputs for factor analysis. Factor analysis uses a derived matrix of correlation, the components of which provide a measure of similarity between variables. Factor analysis has value only when correlation amongst a subset of variables really exists. The higher these inter correlation are, the better defined are the resulting factor dimensions. The most important

^{*} Eigenvalue: The column sum of square for a factor; also referred to as the Latent root. It represents the amount of variance accounted for by a factor (Hair *et al.* 1987).

outputs are; factor loading, the factor scores and variance explained percentages. Each of the original variables has a factor loading on each factor. The factor loading is the correlation between the factors and the variables. These are used to interpret the factors^{*}. Furthermore, the nearer to one the factor loading is, the stronger the association between the variable and the factor (Crawford and Lomas 1980). Normally, factor loadings are crystallized by using a rotation procedure, the most commonly used is the varimax orthogonal rotation which attempts to produce some high loading and some near zero loading on each factor. The varimax orthogonal rotation method is preferred when the objective is to utilize the factors results in a subsequent statistical analysis (Hair *et al.* 1987). This because the factors are orthogonal (uncorrelated) and therefore eliminate the collinearity.

The interpretability of factors is facilitated when individual factor loading is high or low (Cattell 1978). Aaker (1973) also reminds us that while it attempts to maximize the number of factor / variable correlations that are either high or low, it also minimizes the number of factors with which a variable is correlated.

6.2.1.3 Use of the Factor Analysis in the Study

Factor analysis was used in this study for the following objectives:

(1) To find out the main patterns of factors that underlie each construct of the internal and external environmental dimensions.
(2) To use the output of the factor analysis as an intermediate step for further analysis by regression and discriminant analysis.

^{*} It is decided that the cutoff point for the factor loadings should not be less than .30. The rational for this those variables which load above or equal .30 on any factor are considered significant (Hair *et a1.* 1987).

(3) To overcome the potential problem of intercorrelation among independent variables, i.e., the multicollinearity problems.

6.2.2 Multiple Regression Analysis

Multiple regression is a multivariate statistical technique through which one can analyse the relationship between a dependent or criterion variable, and a set of independent or predictor variables. Multiple regression can be viewed either as a descriptive technique by which the linear dependence of one variable on an other is summarized and decomposed, or as an inferential tool by which the relationship is the population evaluated from the examination of sample data (SPSS 1990).

Multiple regression analysis attempts to determine the functional relationship between a single metric dependence variable (criterion) and a number of independent (explanatory) variables (Jain et al. 1982). Multiple regression is the appropriate method of analysis when the researcher has a single dependent variable which is presumed to be a function of other independent variables. Usually, the dependent variable is predicted or explained by a group of independent variables. Aaker & Day (1986) have suggested two different concepts of independent variable on the basis of the study goal. Firstly, the independent variable (explanatory) sometimes are called the predictor variable when prediction is the goal. They help to predict the value of dependent variable (criterion). Secondly, they are called explanatory variables because they explain variation in the dependent variable. When constructing the model, the analyst must include all relevant variables. If an important variable is omitted, the power of the model is reduced.

The construct of a regression model usually starts with the specification of the dependent variable and the independent variable or variable.

 $Y = a + b_1 X_1 + b_2 X_2 + \dots + b_n X_n$

Where:

Y = the estimated value of the criterion variable

a = constant derived from the analysis

 b_1 = coefficient associated with the predictor variable

 X_1^{T} = predictor variable that influence the criterion variable (Tull & Hawkins 1990).

In multiple regression analysis the relationship is assumed to be linear and additive. However, these are important assumptions. Linearity is the assumption that for each independent variable, the amount of change in the mean value of the dependent variable associate with a unit increase in the independent variable "holding all other independent variables constant" is regardless of the level of independent variable. While, additivity is the assumption that for each independent variable, the amount of change is the expected value of the dependent variable associated with a unit increase in the independent variable "holding all other independent variables constant" is the same regardless of values of the other independent variables in the regression equation (Berry and Feldman 1985).

When regression analysis is used to gain understanding of the other relationship between variables, the primary question is "which of the independent variable has the greatest influence upon the dependent variable?". This can be answered by obtaining the partial regression coefficients; the beta coefficient, which measures the degree of association between each independent variable (when all the independent variables are expressed in the standardized form) and the dependent variable. Since the Beta coefficients can be compared with each other in order to evaluate the relative effect of predictor variables, the larger the Beta coefficient, the stronger the impact of that variable upon the criterion variable. In addition, the Beta weight enables the analyst to see how well a set of explanatory variables explain the criterion variable, and to determine the most influential explanatory variables.

The simple R^2 (the coefficient of multiple determination) through which one can measure the proportion of the variation in the dependent variable, tends to overestimate the population value of R^2 . Therefore, adjusted R^2 attempts to correct the optimistic bias of the simple R^2 .

Adjusted R^2 does not necessarily increase as additional variables are added to an equation and is the preferred measure of goodness of fit because it is not subject to the inflationary bias of unadjusted R^2 .

In Summary, Multiple regression is often used to gain an understanding of the relationship between variable by:

- (1) Finding a function or formula by which one can estimate the value of the criterion variable from the predictor variable (Green & Tull 1978).
- (2) Determining which of the independent variables has the greatest influence upon the dependent variable (Kinnear and Taylor 1991).

6.2.2.1 Caution in the Use of Multiple Regression

The use of the multiple regression analysis is not without its problems. One of the most common problems in applied regression analysis is the multicollinearity. Multicollinearity refers to the situation in which some, or all of the independent variables are very highly correlated. In other words, when independent variables are related to each other and not truly independent of each other, multicollinearity is said to exist. Such correlation between the
explanatory variables in the regression equation makes the identification of structural relationships difficult or impossible. Berry and Feldman (1985) distinguished between two forms of multicollinearity. First, is perfect collinearity in which some independent variables regressed against the other independent variables in the model yield of an R^2 of precisely 1.00. This arises from very small data sets (i.e., small samples). The second is less extreme multicollinearity in which the independent variables in a regression equation are intercorrelated but not perfectly. The study of multicollinearity in data analysis evolves around two major problems: (1) how it can be deleted, and (2) what can be done about it. These problems are particular to marketing research where one often faces the dilemma of needing a number of variables to achieve accuracy of explanatory variables (Green & Tull 1978)

Multicollinearity can be dealt with by different approaches. Tull and Hawkins (1990) and Hair *et al.* (1984) suggested several ways for dealing with such situations. First, it can be ignored, particularly when multicollinearity may be prominent in only a subset of the explanatory variables and when this subset does not account for a large proportion of the variance in the data. The second approach, is to omit one or more of the highly correlated predictor variables. This one is recommended when two variables are clearly measuring the same thing. Thirdly, the correlated variables can be combined or otherwise transformed to produce unrelated variables. Finally, the correlated explanatory variables can be summarized in a set of explanatory factors using factor analysis. Furthermore, Kinnear and Taylor (1991) add that another way to avoid multicollinearity is by increasing the sample size.

In this research, the use of the principal components analysis technique was the only possible way to overcome the potential problem of multicollinearity.

6.2.2.2 Use of Regression Analysis in this Study

The regression analysis techniques (stepwise regression method) was preferred here, since it fulfils the requirements of the study's objectives (3, 4, and 5). The primary purposes behind using this technique are:

- (1) To find out statistically whether there is a significant relationship between the two set of environmental dimension measures (internal and external) and the dependent variable level of exporting, taken separately or together.
- (2) To discover whether the addition of the external environmental dimension measures to the internal environmental dimension measures would produce a better explanation for the dependent variable level of exporting.
 - (3) To conclude whether these explanatory variables (taken together) are strongly relevant to the level of exporting.
 - (4) To determine the most important independent variables explaining the variation of the dependent variable.

6.2.3 Simple Correlation Coefficient

The Pearson correlation coefficient is an appropriate statistical method for measuring the degree of association between variables that are interval or ratio scaled (Parasuraman 1986). The correlation coefficient is the standard measure of the linear relationship between two variables and has the following properties:

(i) It is pure number and independent of the units of measurement.

(ii) Its absolute value varies between zero when the variables have no linear relationship, and one, when each variable is perfectly predicted by the other. The absolute value thus gives the degree of relationship.

(iii) Its sign indicates the direction of the relationship. A positive sign indicated a tendency for high values of one of the variables to occur with high values of the other variable. A negative sign indicates a tendency for the high value of one variable to be associated with low value of the other. Reversing the direction of measurement of one of the variables will produce a coefficient of the same absolute value but of the opposite sign. A coefficient of equal value but opposite sign (e.g., .50 or -.50) thus indicates an equally strong linear relationship but in the opposite direction (Cohen & Cohen 1975).

The primary purposes behind the use of this technique in the present study are:

- (1) To identify the strength and the direction of the relationship between each explanatory independent variable (taken as a factor or as a specific variable) and the dependent variable level of exporting.
- (2) To find out whether there is a statistical significant association between the independent variables and the level of level of exporting, taken together or separately.

6.2.4 Discriminant Function Analysis

Discriminant analysis is a multivariate technique whose end purpose generally is to provide a procedure for classifying individual observation into one of a set of groups or population (Jackson 1983). Simply stated, the primary objective of DA is to predict an entity's

likelihood of belonging to a particular class or group bases on several predictor variables (Sheth 1971).

Classification is achieved through a series of classification functions. Fisher (1936) was the first to suggest that classification should be based on a linear combination of discriminating variables. Fisher proposed using a linear combination which maximised group differences whilst minimising within the groups (Klecka 1980). The linear discriminant function can be expressed as follows:

$Z = W_1 V_1 + W_2 V_2 + W_3 V_3 \cdots W_n V_n$

Where :

Z = the discriminant score
W = the discriminant weights
V = the predictor variables

DFA combines those predictor variables that contribute to the discrimination of the "a prior" groupings. The discriminant weights (W) are assigned according to the discriminating power of the predictor variables (Hair *et al.* 1979). Once the predictor variables are selected and the discriminate weights are assigned, they are multiplied together and added as seen above, the sum of which is referred to as the discriminant score of Z. Each individual in the analysis is then classified according to where its z score is in relation to the single "cutting score", which is the Z value used to classify an individual into a group. Those individuals whose Z score are greater than the "cutting" score are classified in group (1), while those with a Z score less than the "cutting" score are placed in group (2).

6.2.4.1 The Applications of DFA

The use of DFA procedure in marketing research has proved most beneficial for the following purposes:

- (1) developing a predictive model to classify individuals into distinguishing groups;
- (2) detecting relationships between predictor variable and group membership;
- (3) "profiling" characteristics of groups which are most dominant in terms of discrimination, and
- (4) identifying the most important variables which differentiate best among groups (Crask et al. 1977).

6.2.4.2 DFA Date Input / Output

Discriminant analysis is mostly used to classify and to make predictions in situations where the criterion variable is in categorical form (e.g., exporters vs. non - exporters) and the predictor variable appears in a metric form (interval or ratio).

By using the SPSSX discriminant function analysis of two groups various statistics can be obtained. The key DFA output can be summarized as follows:

(1) Standardised discriminant function coefficients: These coefficients reflect the relative ability of each predictor variable to discriminate (discrimination power) between groups where the other predictors are held constant (Aaker and Day 1986). The absolute magnitude of the standardized discrimination function coefficient (which are similar to the B weight in multiple regression analysis) is used as an indication of the relative importance of a predictor variable (Tabachnick and Fidell 1983). The larger the discriminate coefficients, the more important the variable as a discrimination.

(2) Eigenvalue and canonical correlation: While eigenvalue indicates the discriminate power of the discriminant function, canonical correlation provides the degree of association between discriminant function scores and group membership (Tabachnick and Fidel1 1983).

(3) Statistical significance: All DFA programmes provide automatically the significant level of the discriminant function which has been developed. A chi - Square value with its degree of freedom and its level of significance is available in the SPSSX discriminate programmes. Moreover, the SPSSX programme of DFA automatically discontinue at or beyond the 0.05 level (Hair *et al.* 1987). Univariate F - ratio for each predictor variable is also provided, but only with the stepwise method which is employed in this research.

(4) Classification or confusion matrix: This matrix helps visualise exactly how accurate the discriminate function "predicted" group memberships. It provides sufficient information for classification of the individual into their appropriate groups. The most important factor to be considered in the classification matrix is the overall predictive accuracy of the discriminant function.

(5) Group means: The group means is considered very useful in interpreting how a predictor variable discriminates between groups. For example, if predictor variables were found to discriminate between group (a) and group (b), it would be worthwhile to compare the mean of this predictor variable for group (a) with its counterpart mean for group (b).

(6) All - groups histograms of the discriminate scores: The discriminant function scores for each group (1 and 2) are plotted in a histogram. The purpose of this histogram is to show how much two or more groups overlap and to examine the distribution of the discriminant score.

6.2.4.3 Use of DFA in this Study

Multiple discriminate analysis was considered to be the most appropriate statistical technique for accomplishing the study's objectives (6, 7 and 8), as it fulfils the following requirements:

- (1) The ability to determine whether or not a statistically significant difference exists between the two groups (i.e., exporters and non - exporters) in terms of their environmental measures.
- (2) The ability to predict group membership of firms on the basis of the internal and external environmental measures, taken together and separately.
- (3) The ability to identify the degree of association between exporters and non - exporters (i.e., export behaviour) and their environmental measures, (i.e., internal and external).
- (4) The ability to identify those independent variables which account for most of the differences between groups.
- (5) To discover whether the addition of the external environmental measure to the internal environmental measure might improve the prediction of group membership.

6.3 The Statistical Methods Used for Testing Research Hypotheses

There are alternative statistical tests available for any given research design, and it is necessary to use some rationale for selecting among them. In hypothesis testing, we must state the hypothesized value of population parameters before we begin sampling. The assumption we wish to test is the null hypothesis "HO".

A statistical test is good if it has a small probability of rejecting (HO) when it is true, but has greater probability of rejecting (HO) when it is false. If our sample results fail to support

the null hypothesis, we must conclude that something else is true. In other word, in applying a statistical test, the researcher must choose between accepting or rejecting the null hypothesis (HO). If (HO) is rejected, then he tends to use this as evidence in favour of (HI) (Siegel 1956).

Siegel (1956) suggests that there are two major considerations in choosing a statistical test. Firstly, the researcher must consider the manner in which the sample was drawn and the nature of its population. Secondly, he must consider the kind of scale of measurement (i.e., nominal, ordinal, interval or ratio) which was employed in the definition of the variables involved in the study. Luck and Rubin (1987) added another consideration which must be taken into account when deciding on the appropriate statistical test, such as, (1) how many samples are involved in the problem? "one, two or many (k) samples" (2) are the samples independent or related to each other.

In this study, four different statistical test representing parametric and non - parametric statistical methods were used to test the research hypotheses (i.e., F - test, T - test, Chi - square test, and McNemar test). Table (6.1) illustrates the research hypotheses and the relevant tests.

6.3.1 T - Test

The T - Test is a parametric statistical test. It is employed for testing hypotheses (HO1, HO4, HO5, and HO11). This statistical test is provided by the stepwise regression analysis computer programme. It is also used to measure the significance of the relationships between each independent variable (the output of principle component analysis), and the level of exporting. Furthermore, the T- test was employed to test the significant difference between the two groups

means (i.e., exporters and non - exporters) in terms of variables constituting each factor (the factors which underlie each major construct of the firm's internal and external environmental dimensions), taken separately.

6.3.2 Univariate F-Test

The SPSSX statistical package provided the result of the F-test with the results of some of the statistical techniques (e.g., DFA discriminant function and RA regression analysis). In this research the F - test was used to test the significance of regression equations, and to measure the degree of significance of the discriminating power for each predictor variable in the analysis, taken separately.

6.3.3 The Chi - Square Test

Chi - square test is employed for testing research hypotheses HO6 and HO7 (Table 6.2). The command DISCRIMINAT, in the SPSSX computer programme, routinely prints the chi - square value, the degree of freedom, and the significant level.

6.3.4 The McNemar Test

The McNemar test was used for testing research hypothesis number (H010). We are seeking the significant improvement (or changes) in the classification, and variation between the two groups in the analysis before and after the addition of the external environmental measure to the internal environmental measure. Therefore, the McNemar test for the significance of change is appropriate (Cohen and Holliday 1982).

To test the significance of any observed change by this method, one can set up a fourfold table of frequencies to represent the first and second sets of response from the same individuals. The general features of such a table are pointed out in Table (6.1), in which (+) and (-) are used to signify different responses. In the McNemar test, the research is often interested only in two cells which show change between the first and second treatment. The sampling distribution associated with this test is Chi - square distribution (Siegel 1956).

Table (6.1)

Fourfold Table for Use in Testing the Significance of Change



Table (6.2)

The Research Hypotheses and Their Relevant Statistical Tests

Hypotheses	Statistical Test
HO(1): There is no significant relationship between the two environmental measure (i.e., internal and external) and the criterion variable level of exporting taken separately.	p T - Test es e s,
HO(2): There is no significant relationship between the external environmental measures and the dependent variable level of exporting, taken together.	F - Test
HO(3): There is no significant relationship between the internal environmental measures and the dependent variable level of exporting, taken together.	F - Test
HO(4): There is no significant relationship between each independent factor (the (the output of the principal componen analysis) and the criterion variable level of exporting, taken separately	T - Test ats
HO(5): There is no significant relationship between the explanatory independent variables which comprise each factor and the dependent variable level of exporting, taken separately.	T - Test
HO(6): There is no significant difference between the two groups (i.e., export and non - exporters) according to the internal environmental measure, taken together.	Chi - square & F - Test eir n
HO(7): There is no significant difference between the two groups (i.e., export and non - exporters) according to the external environmental measure, taken together.	Chi - square & F - Test eir n
HO(8): There is no significant different (variation) between the two segment (i.e., exporters and non- exporters) terms of their internal environment measures, taken separately.	ce F – Test ts in tal

continue Table (6.2)

Hypoth	heses the second begins three and the	Statistical Test
но(9):	There is no significant difference (variation) between the two segments (i.e., exporters and non - exporters) in according to their external environmental measures, taken separately.	F – Test
НО(10):	There is no significant improvement in the discrimination between the two groups (i.e., exporters and non - exporters), after the addition of the external environmental measures to its counterpart the internal environmental measures.	McNemar Test
но(11):	There is no significant differentiation between the two segments (i.e., exporters and non - exporters) according to the independent variables which comprise each factor, taken separately.	T - Test

Summary

Based on the research objectives and hypotheses, several statistical techniques were preferred to analyse the data and to achieve the research objectives, in addition to testing the research hypotheses

The statistical techniques chosen varied from the univariate, the bivaraite and the multivariate, depending on the type of data and the number of variables. The univariate statistical methods used in this research was the chi - square test. The bivariate techniques used were the T - test, F - test, the Pearson correlation coefficients, and the McNemar test. With regard to the multivariate techniques, the following were employed: the factor analysis, the multiple regression analysis, and the discriminant function analysis.

This Chapter includes a brief description of the alternative statistical techniques which had been used in this study, the basis for choosing the appropriate statistical techniques, and the reasons for using each technique in this research.

CHAPTER SEVEN

THE FACTOR ANALYSIS FINDINGS (THE MAIN PATTERN OF FACTORS THAT UNDERLIE EACH CONSTRUCT OF BOTH ENVIRONMENTAL DIMENSIONS)

Introduction

- 7.1 The Findings of the Factor Analysis
- 7.2 The Interpretation of the Final Factor Analysis
- 7.2.1 The Constructs of the Firm's Internal Environmental Dimension
 - 7.2.1.1 The Main Factors That Underlie the Managerial Goals and Aspiration Level
 - 7.2.1.2 The Main Factors That Underlie the Management's Expectations
 - 7.2.1.3 The Main Factors That Underlie the Firm's Comparative Advantages
 - 7.2.1.4 The Main Factors That Underlie the Firm's Commitment to Export Activity
 - 7.2.1.5 The Main Factors That Underlie the Characteristics of the Decision Makers
 - 7.2.2 The Constructs of the Firm's External Environmental Dimension
 - 7.2.2.1 The Main Factors That Underlie the Firm's Legal -Political Environment
 - 7.2.2.2 The Main Factors That Underlie the Firm's Economic and Commercial Infrastructural Environment

Summary

CHAPTER SEVEN

THE FACTOR ANALYSIS FINDINGS (THE MAIN PATTERN OF FACTORS THAT UNDERLIE EACH CONSTRUCT OF BOTH ENVIRONMENTAL DIMENSIONS)

Introduction

This chapter presents the main findings of the principle component analysis. The main purpose behind the use of this techniques here is to reduce the large number of variables that underlie each construct of both dimensions (i.e., internal and external environments), into orthogonal indices, for further analysis by the regression analysis and discriminant analysis.

Furthermore, by employing the principal component analysis techniques, it may be possible to explore the patterns of factors that underlie each major construct. It was considered an appropriate method to overcome the potential problems of multicollenearity among the variables that pertain to each construct (e.g., Tull and Hawkins 1990).

7.1 The Findings of the Factor Analysis

The results of the principal components analysis indicate that 26 factors can be extracted from the seven major constructs of both environmental dimensions (internal and external). Sixteen 16 factors are derived from the five constructs of the internal environmental dimension and ten factors are extracted from the two constructs of the external environmental dimension.

Table (7.1) presents the number of factors underlying each construct of both dimensions.

Table (7.1)

Name of	Name of	Number of	Number of
Dimension	Construct	Variables	Factors
1. Internal	 Managerial aspirations Management's expectations Comparative advantages Commitment to export The decision - maker's characteristics. 	6 27 16 12 4	1 6 5 2 2
2. External	 Economic & commercial	22	6
	infrastructural envir. Political-legal envir.	24	4
Total		111	26

The number of factors that underlie each construct of the internal and external dimension

7.2 The Interpretation of the Final Factor Analysis

The main patterns of factors underlying each construct of the internal and external dimension and their interpretations^{*} are presented under the following sections:

7.2.1 The Constructs of the Firm's Internal Environmental Dimension

The firm's internal dimension consists of five major constructs (for more details see Chapter 4). These constructs are: (1) Managerial aspirations level, (2) Management's expectations, (3) Comparative advantages, (4) Commitment to export marketing activity, and the (5) Socio - demographic characteristics of the decision - maker. The interpretations of the results of the principal components analysis are presented for each of these constructs, as follows:

^{*} The varimax rotation version with Kaiser normalization was used to produce more interpretable factors. The eignvalue (> 1) criteria was used in order to determine the number of factors.

7.2.1.1 The Main Factors That Underlie the Managerial Goals and Aspiration Level

The managerial aspiration level is one of the major constructs of the firm's internal environmental dimension. It is used to measure the importance of export goals to the firm. The results of the principal component analysis (Table 7.2) indicate that only one factor can be extracted from this construct. This factor is composed of six managerial aspiration attributes; improvement of the firm's market position, increasing the firm's overall profit, securing steady growth, developing market strategies, reducing the cost of production and increasing the sales volume of the exported product. Therefore, this factor, which accounts for 54.5% of the total variance, could be reasonably labelled as the "managerial aspiration" factor.

Table (7.2)

	The	Main	Fac	ctors Under	lying	
The	Construct	of	the	Managerial	Aspiration	Leve1

Variable	Attributes Factor Lo	ading	Communality
ayours ri	Factor (1) "Manageria	al aspirati	ons"
V31	Improving the firm's market position	.65798	.43294
V32	Increasing the overall profit	.83745	.70132
V33	Securing steady growth	.45634	.20824
V34	Developing market strategies	.78564	.61722
V35	Reducing the cost of production	.82464	.68002
V36	Increasing the sales volume	.79333	.62937

Variance accounted for:

Factor (1) Eigenvalue^{*} 3.26912 Percentage of Variance 54.5

* Communality: The amount of variance an original variable shares with all other variables included in the analysis.

* Eigenvalue: It represents the amount of variance accounted for by a factor.

7.2.1.2 The Main Factors That Underlie the Management's Expectations

Six factors were extracted from 27 variables that pertain to this component (Table 7.3). These factors together explain 53.1% of the total variance. The first factor is composed of eight variables, namely; difficulty in recruiting sales representatives abroad, difficulty in pricing our products abroad, difficulty in locating potential markets, insufficient knowledge of communicating with overseas customers, lack of personal expertise in exporting, size of production, inability to establish a distribution system and the difficulty of maintaining control over foreign middlemen. These variables together represent exporting operational obstacles. This factor, which accounts for 20.6% of the variance, is labelled as a "export operational obstacles" factor.

The second factor can be identified as a "perceived risk" factor, since it is composed of the five attributes of risk. Those attributes are; risk of production, political risk, risk of transportation, foreign currency exchange risk, and the risk of collecting money (non - payment risk). This factor accounts for 10.8% of the variance.

The third factor consists of five variables, namely; making greater profit, using the firm's excess capacity, increasing the firm's annual sales, increasing the growth of the firm, and increasing the security of the firm. This factor, which explains 7.6% of the variance, could be reasonably called the "export objectives" factor.

The fourth factor can be identified as an "export benefit" factor. It includes four variables; overcoming instability demand, better opportunities in foreign markets, preventing dependency in the domestic market, and benefiting from government export assistance. This factor accounts for 5.2% of the variance.

Table (7.3)

The	Main	Factors	Under	lying	the	Construct
	of	Manageme	ent's	Expect	tati	ons

Variable	Attributes F	actor Loading	Communality
	Factor (1) "Op	perational obsta	cles"
V67	Difficulty of obtaining	.72582	.55485
	sales representatives		
V68	Difficulty of pricing of	.70502	.55571
Sec. 1	products abroad		
V65	Lack of knowledge in	.69625	.54659
	communicating		
V70	Difficulty of locating	.67671	.51700
	potential markets		
V66	Lack of Personal expertise	.67640	.50711
V75	Size of production	.65718	.55504
V69	Inability to establish a	.63049	.46921
	distribution system		
V71	Difficulty of maintaining	.60966	.57846
	control over foreign middl	lemen	
	Factor (2) "H	Perceived risk"	
103	Risk of production	82143	70847
V98	Political risk	80504	.67131
V98	Risk of transportation	77948	.66160
V78	Foreign exchange risk	.74013	.59280
V70	Risk of collecting money	.59997	.40115
***	Rabit of correcting money		
	Factor (3) "H	Export objective	s"
W27	Making greater profit	72570	55066
V21 V26	Heing the firm's excess of	macity 67311	51661
V20 V20	Increasing the annual sale	62112	.52371
1/28	Increasing the security	.54622	47797
V20 V21	Increasing the growth	.51989	35933
V21	Increasing the Browth	.51505	
	Factor (4)	'Export benefits	
	Factor (4)	hapore benerres	
V16	Overcoming instability dem	nand .70896	.55942
V29	Having better opportunitie	es in .58169	.53471
	the foreign markets		
V20	Benefiting from the govern	ment .50393	.32239
	export assistances		
V22	preventing dependency in	.47680	.31179
	the domestic market		

Variable	Attributes Fa	ctor Loading	Communality
	Factor (5)	ort harriers"	
	Factor (5)	Joit Dalliels	
V73	Inability to provide follow services	vup .74918	.58358
V74	Inability to provide credit facilities	.64660	.58933
V72	Difficulty of collecting mo	oney .47556	.42228
	Factor (6) "Con	petitive situa	tion"
V23	Overcoming competition in t domestic market	the .77506	.66195
V19	Keeping peace with local competitors	.75127	.59152

Variance accounted for:

Factor	Eigenvalue	Percentage of Variance	e Cumulative percentage
1	5.57495	20.6	20.6
2	2.91159	10.8	31.4
3	2.04169	7.6	39.0
4	1.39512	5.2	44.2
5	1.27684	4.7	48.9
6	1.12637	4.2	53.1

The fifth factor consists of three variables. These variables go together to form an "export barriers" factor. This factor includes the following attributes; the firm's ability to provide follow up services, the firm's ability to provide credit facilities and the difficulty of collecting money on export sales. This factor accounts for 4.7% of the variance.

Finally, the sixth factor is composed of two variables, namely, overcoming competition in the domestic market and keeping pace with local competitors. This factor which explains only 4.2% of the variance might be identified as the "competitive situation" factor. It represents the benefits which come from the competitive advantage in the domestic market as a result of exporting.

7.2.1.3 The Main Factors That Underlie the Comparative Advantages

The results of the factor analysis showed that the 16 advantages attributes can be clustered into five significant factors. The combination of these factors account for 62.6% of the total variance (Table 7.4).

Factor 1, which accounts for 27.6% of the variance, is composed of five comparative attributes. Those attributes are; ability of the firm to hire trained persons, ability of the firm to offer discount prices to overseas customers, strength of the firm's management, proximity of the firm to the markets, and the firm's technological standard level. These variables appear to go together to produce the level of the manager's confidence in their competitive strengths in the export markets. Thus, this Factor can be designated as the "competitive strengths" factor.

Factor 2 can be identified as a "size of the firm" factor. This factor consists of three variables which are designed to measure the

Table (7.4)

The Main Factors Underlying the Construct of the Firm's Comparative Advantages

Variable	Attributes Fa	actor Loading	Communality
	Factor (1) "Compet	titive strengt	hs"
V54	Ability to hire trained pers	sons .75289	.58164
V53	Ability to offer price disco	ounts .68213	.61068
V24	Strength of Management	.60388	.59154
V18	Proximity to foreign market	.59088	.55975
V49	Firm's technological standar	rd .52492	.44200
	level		
	Factor (2) "Size o	of the firm"	
V104	Number of employees	.8445	.74778
V106	Total of the capital assets	.8435	7.75555
V108	Total sales volume	.8004	.68640
	- 7.18 months in making		
	Factor (3) "Techno	ological stren	gths"
V50	The superior product equipme	ent .8445	.74778
V51	The superior production met	thod .8389	6.73430
V55	The ability to modify produc	et .5006	.50233
	policies		
	Factor (4) "Busine	ess experience	u googees of th
V103	Number of years in the busin	ness .8682	3 .78440
V107	Number of years in exporting	.7931	2.77135
	Factor (5) "Produc	ct characteris	tics"
	win Parlite that Underlie the		
V52	Product quality	.7102	.56959
V25	Product competitive price	.6717	.51663
V17	Uniqueness of product	.4653	.43628
VI/	ourdeness or broader	• 4033	.43020

Variance accounted for:

Factor	Eigenvalue	Percentage of Va	riances Cumulative percentage
1	4.41075	27.6	27.6
2	2.15383	13.5	41.0
3	1.27473	8.0	49.0
4	1.16543	7.3	56.3
5	1.01594	6.3	62.6

size. These variables are; number of employees, total of capital assets that are held by the firm and the total sales volume. This factor, which accounts for 13.5% of the variance, may be labelled the "size of the firm" factor.

Factor 3 can be labelled as the "technological strengths" factor. It is composed of three attributes, they are; superiority of the firm's production equipment, superiority of the firm's production method, and the firm's ability to modify its products policies. This factor explains 8.0% of the variance.

Factor 4 consists of two types of business experience; the number of years in which each firm has been in business, and the number of years in which each firm has been exporting. This factor, which accounts for 7.3%, could be reasonably called the "business experience" factor.

Finally, factor 5 can be identified as the "product characteristics" factor, since it is composed of three product attributes. Those attributes are; product quality, uniqueness of the product, and the product competitive price. This factor explains 6.3% of the variance.

7.2.1.4 The Main Factors that Underlie the Firm's Commitment to Export Activity

The examination of Table (7.5) reveals that the 12 commitment variables can be combined into 2 significant factors. The combination of those factors accounts for 56.0% of the total variance. These two factors are known as the management's commitment to export marketing activities, and the management's willingness to change, or modify export marketing policies.

The first factor is composed of the following attributes; to promote our product to overseas customers, to plan to get into foreign

Table (7.5)

The Main Factors Underlying the Construct of the Firm's Commitments to Export activity

Variable	Attributes Factor	Loading	Communality
atitur.	Factor (1) "Commitmen related a	nt to exp activitie	ort s"
V41	We promote our products to overseas customers	.76365	.65095
V40	We participate in international trade fairs	.71468	.54528
V42	We plan to get into foreign marke	t .70870	.62094
V43	We send sales representatives to foreign markets	.69825	.57849
V44	We have a formal policy for handling export activities	.64790	.43372
V38	We consistently seek export opportunities	.63883	.57693
V39	We carry out a systematic research of possibilities in exporting	h .63795	.45806
V37	We use export research	.38442	.26528
al de The	Factor (2) "Willing	ness to c	hange"
V45	Willingness to modify organisation and control systems	n .85682	.78252
V47	Willingness to change pricing policies to foreign markets	.76207	.65492
V46	Willingness to adapt product policies to foreign markets	.75477	.59246
V48	Willingness to modify distribution policies to foreign markets	n .67511	.56461

Variance accounted for:

Factor	Eigenvalue	Percentage of	Variances	Cumulative	percentage
1	5.50982	45.7		45.7	
2	1.26367	10.2		56.0	

markets, to send sales representatives to foreign markets, to participate in international trade fairs, to carry out a systematic research to explore the possibilities in exporting, to have a formal policy for handling various export marketing activities, and to use export research for our marketing decisions. These eight variables constitute the level of "commitment to export related activities" factor. This factor alone accounts for 45.7% of the variance.

Factor 2 reflects the management's willingness to change or modify their policies to meet the foreign market conditions. It is composed of four attributes; the willingness to modify their organisational and control system, to adapt their product policies, to change their pricing policies, and to modify their distribution policies. This factor, which accounts for 10.2% of the variance, can reasonably be designated as the "willingness to change" factor.

7.2.1.5 The Main Factors That Underlie the Characteristics of the Decision - Makers

Two significant factors were extracted which accounted for 74.5% of the total variance (Table 7.6). The first factor can be reasonably identified as the "education factor". It is composed of two attributes; proficiency in the English language, and level of education that each manager completes. This factor explains alone accounts for 44.7% of the variance.

The second factor consists of two types of demographic characteristics; age and experience. This factor, which accounts for 29.8% of the variance, can be identified as the "demographic characteristics" factor.

Table (7.6)

The Main Factors Underlying the Decision - Maker's Characteristics

Variable	Attributes	Factor Load	ing	Communality
	Factor (1)	"Educat:	ion"	
V120	Proficiency in a foreign	language	88883	.80766
V117	Level of education	ebe Fice's i	.86839	.76294
	Factor (2)	"Demogra	aphic"	acconstances o
V115	Experience	in combined .	.83713	.74975
V114	Age		70527	.65894

Variance accounted for:

Factor	Eigenvalue	Percentage of	Variances	Cumulative	percentage
1	1.85909	44.7		44.7	
2	1.46888	29.8		74.5	

7.2.2 The Constructs of the Firm's External Environmental Dimension

The firm's external environmental dimension was outlined under two major constructs, namely; (1) Economic and commercial infrastructural environment and (2) Political - legal environment. The purpose of this section is to discuss the pattern of factors underlying each of these constructs.

7.2.2.1 The Main Factors That Underlie the Firm's Legal - Political Environment

The examination of Table (7.7) indicates that the 24 attributes of the legal - political construct can be combined into 4 significant factors. The combinations of those factors explain 51.7% of the total variance.

The first factor consists of nine types of export policies and assistance attributes. They include; duty - drawback system policy, temporary - entry system, trade agreement policy, managerial services provided by the Trade Centre (e.g., organising international trade fairs), foreign markets information services provided by the Chamber of Industry, the government's income tax rebate policy, the export discount programme offered by the Central Bank of Jordan, devaluation of the Jordanian Dinar, and facilities provided in the area of Free Trade Zones. These nine attributes go together to form the "government's export policies and assistance" factor. This factor accounts for 23.5% of the variance.

The second factor, which explains 14.2% of the variance, can be reasonably identified as the "awareness of the government's export policies and assistance" factor. It is composed of six attributes, they are; knowledge of the temporary entry system, awareness of the managerial services provided by the Trade Centre, awareness of information services provided by the Chamber of Industry, awareness of

Table (7.7)

The Main Factors Underlying the Construct of the Political - Legal Environment

Variable	Attributes	Factor Loa	ding	Communality
	Factor (1) "(Government's policies and	export assist	ance"
173	Duty - drauback system		76000	64060
V3 179	Temporary - entry system		72819	.64060
VO V1	Government's trade agreen	nent	.72621	.58107
VI	policies	acti c	.12021	.50107
V2	Export credit discount pr offered by the Central Ba	cogramme ank	.66653	.56565
V5	Managerial services prov by the Trade Centre	ided	.63868	.58189
V4	Information services prov by the Chamber of Indust	vided ry	.62857	.59734
V9	Government's income tax policy	rebate	.62436	.48408
V6	Free Trade Zones facilit:	ies	.61116	.41957
V7	Devaluation of The Jorda	anian Dinar	.50254	.43660
	Factor (2) "An ex	wareness of t kport policie	he gov s & as	vernment's ssistance"
V13	Awareness of the services by the Trade Centre	s provided	.73575	.58147
V12	Awareness of the export of	credit prog.	.72227	.64062
V14	Knowledge of the tempora: entry system	ry -	.72220	.68508
V11	Awareness of the duty - o entry system	lrawback	.72131	.67664
V15	Awareness of the services by the Chamber of Indust:	s provided ry	.64932	.62471
V10	Knowledge about the faci in the area of Trade Zone	lities es.	.64345	.44337
	Factor (3) "G	overnment exp	ort fa	acilities
	r	elated proble	ms"	
V63	High duties imposed on spreauired for production	pare parts	.76560	.51319
V62	Lack of commercial - Atta	aches	.69325	.55849
V61	Complexity of the duty - system	drawback	.6107	.51319
V59	Inadequate tax incentives	S	.63119	.55406
V60	The high cost of export	documents	.61818	.49071
v80	The Central bank regulat: in respect to the bank - requirements	ions guarantee	.51663	.31647

Continue Table (7.6)

Variable	Attributes	Factor Loading	Com	munality
	Factor (4) "(Government expo related problems	rt proc s"	edure
V57	The export licence proced	lures .:	79828	.69765
V56	The customs export cleara	inces .	74183	.59576
V58	The export inspection pro	ocedures .	57065	.57853

Variance accounted for:

Factor	Eigenvalue	Percentage	of variance	Cumulative percentage
1	5.41193	23.5		23.5
2	3.27364	14.2		37.8
3	2.08242	9.1		46.8
4.	1.11254	4.8		51.7

a second factor explains & 8% of the variance.

the export credit programme, awareness of the duty - drawback system facilities and the knowledge of the Trade Zone facilities.

Factor 3 and factor 4 are the government's export regulations, procedures and facilities related problems. The third factor consists of those types of regulations and assistances which go together to form the "government export facilities related problems" factor. This includes, high duties imposed on the spare parts, complexity of the duty - drawback system, inadequate tax incentives, high cost of export documents, lack of Commercial Attaches in the foreign markets to be exported to, and the bank - guarantee requirement. The fourth factor deals with export licence procedures, customs export clearance processes and the export product inspection procedures. These types of procedures could be identified as the "government export procedures related problems". The third factor explains 9.1% of the variance and the fourth factor explains 4.8% of the variance.

7.2.2.2 The Main Factors That Underlie the Firm's Economic and Commercial Infrastructural Environment

The findings of the factor analysis reveal that 6 significant factors accounting for 62.1% of the total variance can be extracted from the 22 economic - environment attributes (Table 7.8). These six factors may be called; the "foreign markets conditions", the "commercial facilities related problems", the "raw materials related problems, the "export costs", "physical distribution related problems" and the "domestic market conditions" factors. The attributes and weighting for each factor are as follows:

Factor 1, which accounts for 22.8% of the variance, is composed of five foreign market attributes. These attributes are; intensity of competition, foreign government restrictions, business practices in foreign markets, political instability and geographical distance. It

Table (7.8)

The Main Factors Underlying the Construct of the Economic and Commercial Infrastructural Environment

Variable	Attributes Factor	r Loading	Communality
	Factor (1) "Foreig	n markets co	onditions"
	The famian concomment	02025	70/07
V96	restrictions	.03935	.12421
105	Intensity of competition	82899	71015
1/98	Political instability	81909	69300
V90	Business practice in	.80735	.67763
V 54	the foreign markets	.00100	.07705
V97	Geographical distance	.76942	.61390
151	ocoBraphicor arboance		
	.76621 5.8		
	Factor (2) "Comme	rcial facili	ties"
V86	Lack of effective trade companies	.82528	.74448
V64	Lack of information	.81090	.69194
V79	Lack of export credit guarante	ee .72101	.54061
	against non - payment risk.		
V76	Difficulty of access to	.58798	.51061
	banking facilities		
	Factor (3) "Raw ma	aterial prob)lems"
V85	Difficulty of obtaining raw	.83297	.73114
	material from foreign markets	0 00050	71970
V84	Low quality of faw material is	.82055	./10/2
	Chertage of ray material	20010	60134
V83	shortage of faw material	.00510	.03134
V92	High price of raw material	.36005	.37973
	Factor (4) "Expor	t costs"	
V90	Cost of communication	.82440	.69907
V89	Cost of labour skills	.78351	.63978
V91	Cost of warehouses	.65258	.52264
V101	Interest rate	.55907	.47390
	Factor (5) "Physic	cal distribu	ition"
		000055	
V81	Shipping facilities	.82957	.72750
V82	Warehouse facilities	.71905	.65011
V87	Freight transportation	.40028	.34735

Continue Table (7.7)

Variable	Attributes	Factor L	oading	Commun	ality
Laped pro	Factor (6)	"Domestic ma	arket co	nditions"	
V99 V100	Intense competition High domestic demand		.75436	.630 .568	36 02

Variance accounted for:

Factor	Eigenvalue	Percentage o	of Variance	Cumulative	percentage
1	5.01812	22.8		22.8	
2	3.05058	13.9		36.7	
3	1.76621	8.0		44.7	
4	1.65225	7.5		52.2	
5	1.44972	5.2		57.4	
6	1.60268	4.7		62.1	

is clear these variables merge together to form the "foreign markets conditions" factor.

Factor 2 is likewise identified as the "commercial facilities related problems" factor. It is composed of four commercial attributes. These attributes are as follows; lack of information, lack of effective trade companies, difficulty of access to banking facilities, and lack of export credit guarantee against non - payment risk. This factor accounts for 13.9% of the variance.

Factor 3 can be reasonably identified as dealing with the "raw material related problems" factor. It is composed of four raw material variables. These are; difficulty in obtaining raw material from foreign markets, low quality of raw materials in the domestic market, lack of raw materials needed in the domestic market, and the high price of raw materials. This factor explains 8.0% of the variance.

Factor 4 consists of four types of commercial costs; cost of communication, labour skills, cost of obtaining warehouse facilities, and interest rate on banking facilities. This factor, which explains 7.5% of the variance, could be reasonably called the "export costs" factor.

Factor 5 may be labelled as a "physical distribution related problems" factor, since it is composed of variables that mainly relate to the physical distribution characteristics. These variables include; inadequate shipping and warehouse facilities, and the cost of obtaining transportation facilities. This factor accounts for 5.2% of the variance.

The final factor is composed of two attributes; the high demand for products and intense competition in the domestic market. This factor, which accounts for 4.7% of the variance, can be identified as the "domestic market conditions" factor.

Summary

The principal component analysis techniques were performed here for the following purposes:

(1) To explore the main pattern of factors that underlie each construct of both dimensions of the firm's internal and external environment.

(2) To reduce the large number of variables of each construct into orthogonal indices which can be used (the output of the principal component analysis) as an intermediate step (input) for further analysis by the regression and discriminant analysis techniques.

(3) The principal component analysis was considered an appropriate method to overcome the potential problems of intercorrelation among the variables.

The findings of the principal component analysis revealed that 26 factors could be extracted from the seven major constructs of the firm's environmental dimensions (internal and external). Sixteen factors were extracted from the five major constructs of the firm's internal environmental dimension and ten factors were extracted from the two major constructs of the firm's external environmental dimension.

A summary of these factors, with accounting variance and eigenvalues, are presented in Table (7.9) and Table (7.10) for the firm's internal environmental dimension and the firm's external environmental dimension respectively.

Table (7.9)

Summary of the Factors That Underlie the Major Constructs of the Firm's Internal Environmental Dimension

The Name o	f Construct	Eigenvalue	Percentage of Variance
1. The Manag	erial Aspiration Level	3.26912	54.5
Factor (1)	"Managerial Aspirations"	3.31239	55.2
2. The Manag	ement's Expectations	15.32661	53.1
Factor (1)	"Operational obstacles"	5.57495	20.6
Factor (2)	"Perceived risk"	2.91159	10.8
Factor (3)	"Export objectives"	2.04169	7.6
Factor (4)	"Export benefits"	1.39512	5.2
Factor (5)	"Export barriers"	1.27689	4.7
Factor (6)	"Competitive situation"	1.12637	4.2
3. The Firm'	s Comparative Advantages	10.02068	62.6
Factor (1)	"Competitive strengths"	4.41075	27.6
Factor (2)	"Size of the firm"	2.15383	13.5
Factor (3)	"Technological strengths"	1.27473	8.0
Factor (4)	"Business experience"	1.16543	7.3
Factor (5)	"Product characteristics"	1.01594	6.3
4. Commitmen	ts to Export Marketing	6.71602	56.0
Factor (1)	"Commitment to export related activities"	5.48993	45.7
Factor (2)	"Willingness to change"	1.22609	10.2
5. The Deci	sion- Maker's Characteristics	2.97829	74.5
Factor (1)	"Education"	1.78588	44.7
Factor (2)	"Demographic"	1.19241	29.8

Table (7.10)

Summary of the Factors That Underlie the Major Constructs of the Firm's External Environmental Dimension

The Name of Construct	Eigenvalue	Percentage of Variance
1. The Political - Legal Environment	11.87993	51.7
Factor (1) "Government's export policies and assistance"	5.41193	23.5
Factor (2) "Awareness of the government export policies & assistance	3.27304	14.2
Factor (3) "Government export facilities related problems"	2.08242	9.1
Factor (4) "Government export procedures related problems"	1.11254	4.8
2. The Economic & Commercial Environment	13.65860	62.1
Factor (1) "Foreign markets conditions"	5.01812	22.8
Factor (2) "Commercial facilities"	3.05058	13.9
Factor (3) "Raw material problems"	1.76621	8.0
Factor (4) "Export costs"	1.65225	7.5
Factor (5) "Physical distribution"	1.14972	5.2
Factor (6) "The domestic market conditions"	1.03172	4.7
CHAPTER EIGHT

THE DETERMINANTS OF THE FIRM'S LEVEL OF EXPORTING

Introduction

- 8.1 The Bivariate Correlation: Factor Findings
- 8.2 The Interpretation of the Correlation Factor Findings

8.2.1 Internal Environmental Dimension: Factors Correlations 8.2.2 External Environmental Dimension: Factors Correlations

- 8.3 The Stepwise Multiple Regression: Dimension Findings
- 8.4 The Interpretation of the Stepwise Multiple Regression Findings
 - 8.4.1 Stepwise Multiple Regression: Internal Dimension
 - 8.4.2 Stepwise Multiple Regression: External Dimension
 - 8.4.3 Stepwise Multiple Regression: Combination
- 8.5 The Bivariate Correlation: Variable Findings
- 8.6 The Interpretation of the Correlation Variable Findings

Summary & Conclusions

CHAPTER EIGHT

THE DETERMINANTS OF THE FIRM'S LEVEL OF EXPORTING

Introduction

The 26 factors and associated variables identified in Chapter Seven will be analysed in terms of their relationships with (direction and strength), and ability to predict the level of exporting^{*} (defined here as a dependent variable), and thus to test the following hypotheses (HO1, HO2, HO3 HO4, and HO5, see Chapter 5).

The statistical analysis techniques used are the Pearson's Correlation and the stepwise multiple regression. Factors and variables will be analysed and discussed respectively in this chapter.

^{*} The level of exporting is measured in terms of the export volume as a percentage of the firm's total sales, see Chapter 4.

8.1 The Bivariate Correlation: Factor Findings

The Pearson correlation is used to test the following null hypothesis:

HO(1): There is no significant relationship between each independent

factor and the level of exporting, taken separately.

Table (8.2) presents the bivariate correlation^{*} coefficients between each specific factor and the criterion variable level of exporting. Of 26 independent factors, 18 factors are shown to have a significant association with the level of exporting. Of the 18 significant factors, 5 factors are related to the firm's external environmental dimension, and 13 factors are related to the firm's internal environmental dimension.

Furthermore, it was found that out of 26 factors, 10 factors are negatively associated with the dependent variable. However, only 6 of them are significantly related to the level of exporting.

8.2 The Interpretation of the Correlation Factor Findings

The possible explanations of the above findings are presented under the following two subsections:

8.2.1 Internal Environmental Dimension: Factor Correlations

The correlation analysis indicates that, out of the 16 factors of the internal environmental dimension, only 13 factors are significantly associated with the level of exporting. The possible interpretations of these 13 significant factors are given below:

^{*} A list of the 26 factors is provided in Tables (7.9) and (7.10), Chapter 7.

^{*} The T - Test of significant was used to determine the "goodness - of fit) for the correlations coefficients. The SPSSX Package produces the result of this test routinely.

Table (8.1)

The Correlation Coefficients Between Each Factor and the Level of Exporting

The Independent Factor	R devision e
1. Internal Environmental Dimension	
01 Managerial aspirations	.3607**
02 Operational obstacles	3363**
03. Perceived risk	1379**
04. Export objectives	.2629**
05. Export benefits	.1218*
06. Export barriers	.0735
07. Competitive situation	.0329
08. Competitive strengths	.3273**
09. Size of the firm	.3642**
10. Technological strengths	.0766
11. Business experience	.2719**
12. Product characteristics	.2144**
13. Commitment to export	.3711**
related activities	s that the dright
14. Willingness to change	.5173**
15. Education	.1173*
16. Demographic	.4422**
whe level of the expertise wall be	s disk ensity is
2. External Environmental Dimension	in an an an an an an an an
17 Covernment export	207644
17. Government export	.2570***
assistance and policies	440630
To, Awareness of the	.203000
government export	and the state of the
10 Coversment expert	122144
facilities related problems	1331~~
20 Concernment overset	0162
20. Government export	0103
Procedures	605044
21. Foreign market conditions	0000
22. Commercial facilities	0898
25. Kaw material problems	0007
24. COSTS TACLOF	08/2
25. Physical distribution	0419
20. Domestic market conditions	1255**

Note: Figures in the Table above show the Pearson correlation coefficient. Level of significance is shown by ** = 0.01 level, * = 0.05 level. In an examination of Table (8.1), it appears that the "willingness to change" factor exhibits a high correlation coefficient compared to the other factors which make up the firm's internal environmental dimension. This might imply that the management willingness to change the export marketing policies to meet foreign markets conditions should be important for those firms which want to export more.

Another important factor which shows a high correlation coefficient with the level of exporting is the "demographic". This result might indicate that this factor is regarded as being an important determinant of the level of exporting.

The factor of the "commitment to export related activities" is shown to be directly related to a high level of exporting. From the marketing point of view, this might indicate that the higher the firm's commitment to carry out several export marketing activities, the higher the level of its exporting will be. This result is also supported by previous works such as Cavusgil (1976), Cavusgil & Nevin (1981), and Aaby & Slater (1989).

The "managerial aspirations" factor is shown to have a direct relationship with the level of exporting. This might indicate that the higher the importance attached to export goals, the higher the level of exporting will be. This result appears to agree with the findings of the previous studies, such as Cavusgil (1976), Cavusgil & Nevin (1981), Kolhede (1984), and Schlegelmilch and Crook (1986).

Among the groups of factors which were extracted from the construct of the "Firm's Comparative Advantage" is the "size of the firm" factor. It is found to be directly related to the level of exporting. This might indicate that the larger the size of the firm is, the higher its level of exporting will be. While this result appears to be endorsed by some of the previous works (e.g., Tookey 1964, Reid 1980,

1983 & 1985, Ortize-Buonafina 1990 and Culpan 1989), it is found to be in conflict with other studies, such as Kirpalani & Macintosh (1980), Czinkota & Johnston (1983, 1985), Axinn (1988), and Diamantopoulos & Inglis (1985, 1988). The possible explanation of this conflict might be attributed to two factors; the first factor might be that the size of the firm here is tested at the aggregate level, i.e., a factor, and the other possible factor is that the "size of the firm" as a factor might be critical to those firms who want to improve their level of exporting. This might be due to the small size of manufacturing firms in Jordan (as a developing country) in comparison to the manufacturing firms in developed countries where most of the previous empirical studies were undertaken.

Another factor which is directly associated with the level of exporting is the "market competitive strengths" factor. This might indicate that the higher confidence of the management with their firms' market competitive advantages, the higher the level of their firms' exporting will be. This result corroborates the findings of the previous studies, such as Schlegelmilch (1983) and Kolhede (1984).

The "business experience" factor is shown to have a direct association with the level of exporting. This might indicate that the wider the experience of the firm, the higher the level of its exporting will be. It is also strongly supported by many export studies, (e.g., Ursic & Czinkota 1984 and Aaby & Slater 1989).

The "product characteristics" factor is shown to be directly related to the level of exporting. This might indicate that the strong of the firm's product advantages, the higher its level of exporting will be. It is found to be in agreement with many previous works, such as Khan (1978), Beamish and Munro (1987) and McGuinness (1978).

The "operational obstacles" factor is found to be negatively associated with the level of exporting. This might indicate that the higher the perceptions of the export operational obstacles, the lower the level of exporting will be. Also, the "perceived risk" factor is found to be negatively related to the level of exporting. The result might indicate that the higher the perception of risks, the lower the level of exporting will be. This result was supported by the findings of the other studies (e.g., Bilkey 1978 and Schlegelmilch 1986).

The "export objectives" factor is found to be positively related to the dependent variable level of exporting. This might indicate that the higher the expectation of management with regard to the export objectives, the higher their firms' level of exporting will be. It is in agreement with many export studies, such as Czinkota and Johnston (1983), Cavusgil and Nevin (1981) and Axinn (1985, 1988).

The "export benefits" factor is also found to be positively correlated with the level of exporting. This might indicate that the higher the perceptions, or the importance attached to the export benefits, the higher the level of exporting will be.

The last factor (i.e., "education") is shown to have a direct association with the level of exporting. This might indicate the higher the level of the manager's education, the higher the level of their firms' exporting will be. It appears to be in agreement with previous works, such as Schlegelmilch (1983) and Burton & Schlegelmilch (1987).

The lack of significance for the other factors, (i.e., "export barriers", "competitive situations", and "technological strengths") might be attributed to the fact that they may be more important in determining the firm's export behaviour rather than its level of exporting.

8.2.2 External Environmental Dimension: Factor Correlations

The interpretation of the five significant factors of the external environmental dimension is presented and discussed in terms of the their level of the correlations and their level of significance (see Table 8.1), as follows:

The "foreign market conditions" factor stands out as the highest factor correlated with the level of exporting. This might indicate that the higher the perception or the importance attached to the foreign market conditions, the lower the level of exporting will be. It is not surprising that this factor is found to be crucial to the level of exporting, because if the foreign market conditions are unfavourably perceived, the level of exporting is expected to be at its lowest. Of course, this result makes sense, because during the data collection, Jordanian manufacturing firms were experiencing tough foreign market conditions as a result of the Gulf crisis which took place in August 1990. Iraq as the main participant of this conflict, was the most important destination market for Jordanian export products (see Chapter 1).

The "government's export policies and assistance" factor is shown to be positively associated with the level of exporting. This might indicate that the higher the importance attached to the government's export policies and assistance, the higher the level of exporting will be. This might be due to the fact that these types of assistances are strongly needed among the firms surveyed, in particular those firms which are not in a position or able to export more, without government's export assistance.

Another factor which is positively correlated with the level of exporting is the "awareness of the government's export policies and assistance". This result might indicate that the higher the awareness

of the government's export policies and assistance, the higher the level of exporting will be. It also might indicate how much the awareness or knowledge of those export policies and assistance are important for firms which intend to export more.

The "government's export facilities related problems" factor is shown to be negatively associated with the level of exporting. This might indicate that the higher the perception of this factor, the lower the level of exporting will be. Also, it might imply that the inadequacy of the government's export facilities or the difficulty associated with obtaining such facilities is an important determinant to the level of exporting. Again this factor emphasizes how the government's export facilities are important for firms which want to export more.

The "domestic market conditions" factor is found to be negatively associated with the level of exporting. This result might indicate that the higher the perception of the "domestic market conditions" factor, the lower the level of exporting will be. It might be due to the fact that the domestic market conditions inhibit the firms from getting involved in a higher level of exporting. For example, the high domestic demand might discourage firms to increase their level of exporting beyond a certain point (e.g., Rao *et al.* 1990).

Among the constructs of the external environmental dimension, 5 independent factors are found to be not importantly related to the level of exporting. They are; "government export procedures related problems", "commercial facilities", "raw material related problems", "export costs", and "physical distribution". The lack of significant relationships between those factors and the level of exporting might be due to the fact they are important in determining the firm's decision to become involved in exporting.

8.3 The Stepwise Multiple Regression: Dimension Findings

The stepwise multiple regression analysis technique is performed here in order to examine the following three null hypotheses:

- HO(2): There is no significant relationship between the internal environmental measures, (i.e., 16 factors^{*}) and the level of exporting, taken together.
- HO(3): There is no significant relationship between the external environmental measures, (i.e., 10 factors^{*}) and the level of exporting, taken together.
- HO(4): There is no significant relationship between the two environmental measures (i.e., 26 factors) and the level of exporting, taken together.

Table (8.2) summarizes the results of the stepwise multiple regression analysis, with the F - ratio test, for the three above hypotheses. The results indicate that each of these hypotheses (i.e., HO2, HO3, and HO4) is correlated significantly with the level of exporting at .0000 level of significant. Accordingly, it may be concluded that there is a significant relationship between each dimension (i.e., internal, external, and combined) and the level of exporting.

Table (8.2)

Hypotheses	Dimension	Multiple R	R. Square	Adjusted R Square	DF	F - Sign.
HO(2) HO(3)	Internal External	.72370 .73165	.52375	.49454 .52289	16 10	.0000
HO(4)	Combined	.81175	.66871	.64542	26	.0000

A Summary Results of the Stepwise Multiples Regression

* A list of the 16 internal factors is provided in Tables (7.9)
 * A list of the 10 external factors is provided in Tables (7.10)

8.4 The Interpretation of the Stepwise Multiple Regression Findings

According to the stepwise multiple regression method, the variable which correlated highly with the dependent variable level of exporting is expected to enter into the regression equation. The F value at .00 level of significance will be used to determine the "goodness of fit" for the regression equation. The F value is the ratio of explained to unexplained variance accounted for by the regression equation. When the total variance accounted for is low, interpretation of the individual beta coefficient has little meaning (SPSS 1990). Therefore, when the adjusted (R) square is around .10 or above and the F value of the regression equation reaches to 0.05 level of significance the individual beta weight is explained.

Also, in this study the severity or degree of multicollinearity is tested by examining the relative size of the pairwise correlation coefficient between the explanatory independent factors. An examination of the correlation matrix indicates that the correlation for each coefficient is less than about (.50). Therefore, it is possible to interpret the results since the multicollinearity is not severe.

The results of the stepwise regression analysis are presented and discussed here under the following subsections:

8.4.1 Stepwise Multiple Regression: Internal Dimension

The results of the stepwise regression analysis indicate that the firm's internal environmental dimension (i.e., all 16 factors of the internal dimension, taken together) is significantly related to the level of exporting. This result supports the findings of the previous studies, in particular those reported by Cavusgil and Nevin (1981), Kolhede (1984), Axinn (1988), Gomez-Mejia (1988) & Gemunden (1991).

The findings also indicate that, out of those 16 explanatory independent factors, only 3 factors are included in the regression equation. These 3 factors in terms of their order of importance are; "willingness to change", "commitment to export related activities", and "size of the firm" (see Table 8.3).

The adjusted R square for these three factors is .4945. This indicates that about 50% of the variation of the level of exporting can be explained by these three factors.

The "willingness to change" factor is shown to be the most important factor that related to the level of exporting. The Beta coefficient for this factor is .5872. This might imply that the management willingness to change the export marketing policies to meet foreign markets conditions is necessary for firms which want improve their level of exporting.

The "commitment to export related activities" is the next most important factor that is highly associated with the level of exporting. The Beta coefficient indicates that this factor on its own accounted for .3565 of the variation of the level of exporting. It might imply that the commitment to export marketing activities is important for firms which want to increase their level of exporting.

The last most important factor included in the regression equation is the "size of the firm". The Beta correlation coefficient for this factor is .1595. This might indicate that this factor (i.e., the "size of the firm") is critical for firms which intend to improve their level of exporting.

Table (8.3)

The Stepwise Regression Analysis: Internal Dimension

 Multiple R
 .72370

 R Square
 .52375

 Adjusted R Square
 .49454

 Significant F = .0000

Factors Included In the Regression Equation

Factors	Step	Mult.	Rsq	Adjrsq	Sign. 1	Beta In
Willingness to change Commitment to export	1 2	.5872 .6868	.5872 .4717	.3411 .4689	.000	.5872 .3562
The size of the firm	3	.7032	.4845	.4905	.000	.1595

8.4.2 Stepwise Multiple Regression: External Dimension

The results of the stepwise multiple regression analysis indicate that the firm's external environmental dimension, (i.e., all 10 factors of the external dimension; taken together) is importantly related to the level of exporting.

To the best knowledge of the researcher, supporting evidence for the effect of the external dimension (*taken together*) upon the level of exporting might not be established in the previous studies^{*}.

The findings indicate that, out of these 10 explanatory independent factors, only 5 important factors are included in the regression equation. The adjusted R square of these factors is .5234.

<u>In comparing this result with the previous one</u>, i.e., the result of internal dimension (see section 8.2.1), it may be concluded that the combination of the factors of the external dimension (52%) produces a slightly better explanation of the variances of the level of exporting than the combination of the factors of the internal dimension (50%). This might indicate that the factors existing outside the firm's environmental boundary are more critical to its level of exporting than those ones related to its internal environmental process.

As shown in Table (8.4), these five most important factors included in the regression equation are discussed according to their order of importance as follows:

The "foreign markets conditions" factor is ranked as the most important when compared to the other factors of the external environmental dimension. This might indicate that the foreign market conditions as perceived by managers are crucial to a high level of

^{*} The variables or the constructs which comprise the external environmental dimension were only examined separately in the previous studies.

Table (8.4)

The Stepwise Regression Analysis: External Dimension

Multiple R	.73165	
R Square	.53531	
Adjusted R Square	.52289	Significant $F = .0000$

Factors Included In the Regression Equation

Factors	Step	Mult.	Rsq	Adjrsq	Sign. Beta in	i
The foreign market	1	.6881	.4734	.4721	.0006881	
The government's expor	t 2	.7046	.4938	.4936	.000 .1555	
The awareness of the government's export	3	.7163	.5130	.5092	.000 .1314	
policies and assistance The domestic market	es 4	.7245	.5249	.5199	.0001097	
The export costs	5	.7284	.5305	.5243	.0000980	

exporting. In other words, the higher the importance attached to the "foreign markets conditions" factor, the lower the level of exporting will be. Unfavourable perceptions of the foreign market conditions might discourage or inhibit firms from exporting more.

Another most important factor is the "government's export policies and assistance". The Beta coefficient indicates that there is a positive relationship between this factor and the dependent variable. This might indicate that these export assistances and policies are required for firms which want to be at a high level of exporting.

The "awareness of the government's export policies and assistances" factor is ranked as the third most important factor associated with the level of exporting. This might indicate, on the one hand, how much those export policies and assistances are important to the level of exporting, and on the other hand, it might indicate how much the government pays attention to promoting and encouraging firms to become involved in export marketing.

Another most important factor included in the regression equation is the "domestic market conditions". This might indicate that the "domestic market conditions" factor is critical for firms which want to increase their level of exporting. In fact, the domestic market conditions might discourage or inhibit firms from increasing their level of exporting beyond certain conditions (e.g., Keesing 1979 and Rao et al. 1977, 1990).

Finally, the last most important factor is the "export costs". The result indicates that there is a negative relationship between this factor and the dependent variable. This might indicate that the higher the perception of the costs of obtaining commercial facilities for exporting purposes, the lower the level of exporting will be.

8.4.3 Stepwise Multiple Regression: Combination

This approach is expected to provide evidence of the determinants of the level of exporting when compared with the solution for each dimension, (i.e., each one acts alone). More of the predictor factors are expected to enter in the regression equation.

The results of the stepwise multiple regression indicate that the combination of the two dimensions (i.e., all 26 factors, acting together) is significantly associated with the level of exporting.

The findings also indicate that, out of the 26 factors, only 7 factors are included in the regression equation. The adjusted R square for those 7 factors together is .6443, i.e., about 64% of the variation of the level of exporting is explained by them (Table 8.5).

Those 7 most important factors included in the regression equation are in terms of their order of importance; "foreign markets conditions", "management's willingness to change", "commitment to export related activities", "size of the firm", "domestic market conditions", "business experience", and the "government's export facilities related problems".

Four factors related to the internal dimension, and three factors related to the external dimension. The 4 internal dimension factors are also shown to be importantly related to the level of exporting when they taken separately. However, only 1 factor, (i.e., business experience) is not regarded as important when all the factors of the firm's internal dimension act alone.

The three factors of the external dimension are also found to be importantly related to the dependent variable when they act separately. However, only two of them are shown to be important when all of the factors of the firm's external environmental dimension act alone. These two factors are: "foreign markets conditions" and the

Table (8.5)

The Stepwise Regression Analysis: Combination

Multiple	R		.81775
R Square			.66871
Adjusted	R	Square	.64452

Significant F = .0000

Factors Included In the Regression Equation

Factors	Step	Mult.	Rsq	Adjrsq	Sign. Beta In	
Foreign market	1	.6881	.4754	.4740	.0006895	2
Willingness to change	2	.7435	.5527	.5504	.000 .3207	
Commitment to export related activities	3	.7844	.6153	.6122	.0002583	
Size of the firm	4	.7964	.6342	.6304	.000 .1457	
Domestic market	5	.8015	.6425	.6377	.0000917	
Business experience	6	.8041	.6466	.6410	.000 .0694	
Government's export facilities related problems	7	.8067	.6508	.6443	.000 .0682	

the "domestic market conditions".

The Beta coefficient indicates that the "foreign market conditions" factor is ranked as the most important in comparison to other factors included in the regression equation. This might indicate how much this factor is important to the level of exporting. The adjusted R square for this factor alone is .4740. It also indicates that about 47% of the variations in the dependent variable level of exporting can be explained only by this factor.

The "willingness to change" factor is the next most important factor that is directly associated with the level of exporting. The Beta correlation coefficient for this factor is .3207. This indicates that there is a moderate relationship between this independent factor and the level of exporting. The adjusted R square change for this factor is .0773 which indicates that this factor can add (by itself) about 8% of the explanation to the total of the adjusted R square.

Another most important factor included in the regression equation is the "commitment to export related activities". The Beta correlation coefficient of this factor is .2583. This might indicate that the higher the degree of the firm's commitment to undertake export marketing activities, the higher the level of exporting might be.

The "size of the firm" is ranked as the fourth most important factor related to the level of exporting. The Beta correlation coefficient of this factor is .1457. This might indicate that the size of the firm is crucial to a high level of exporting, because the larger firms are more capable of or in a better position to increase their level of exporting than the smaller manufacturing ones.

The "domestic market conditions" factor is ranked as the fifth most important factor associated with the level of exporting. The Beta

correlation coefficient indicated that there is a negative relationship between this factor and the dependent variable.

The result of the stepwise regression at Step 6 also indicates that the "business experience" factor might be regarded as an important factor related to a high level of exporting. Finally, the "government export facilities related problems" factor is ranked as the seventh most important factor associated with the level of exporting. This result might indicate that the worse the perceptions of these problems associated with obtaining of the government's export facilities, the lower the level of exporting might be.

In comparing this solution with the other two solutions presented in the previous sections, i.e., the results of each dimension acting alone (sections 8.2.1 and 8.2.2, Chapter 8), it may be concluded that the combination of the two environmental dimensions would give a better explanation (predictive power) of the variation of the level of exporting than either dimension acting individually. Indeed, not only because the number of determinants (7 factors) is increased, but also the rate of explanation which they account for is increased from 49% (internal dimension) and 52% (external dimension) to about 64%.

This conclusion implies that a better understanding of the determinants of the firm's level of exporting requires that the two environmental dimensions as a whole should be viewed (i.e., internal and external acting together) rather than only viewing each dimension acting individually. However, previous studies have mainly focused on the firm's internal environmental characteristics.

8.5 The Bivariate Correlation: Variable Findings

The Pearson correlation is used again to test the following hypothesis:

HO(5): There is no significant relationship between the variables

which comprise each factor and the dependent variable level of exporting; taken separately.

This major hypothesis can be further divided into 26 hypotheses according to the number of the factors included in the analysis. A summary result for each of these hypotheses is given in Table (8.6).

The results indicate that, out of the 26 factors, only 18 factors are found to be significant in terms of all their associated variables, 2 factors are shown to be insignificant in terms of all their associated variables, and 6 factors are found to be mixed, i.e., some of their associated variables are found to be significant and others not.

8.6 The Interpretation of the Correlation Variable Findings

The main purpose of this analysis is: (1) to find out which variable is highly correlated with the level of exporting, i.e., the most important variable in comparison to the other variables of each factor, (2) to find out whether the result is in agreement or disagreement with previous works in this field, and (3) to conclude whether the results of the attributes of each factor taken together or separately are similar with respect to the level of exporting.

Results of the variables analysis are presented and discussed in terms of each factor to which they belong, as follows:

Hypotheses	The Independent Factors	No. of Variables	Taken Separately
F(HO)01	Managerial aspirations	6	All signif.
F(HO)02	Operational obstacles	8	All signif.
F(HO)03	Perceived risk	5	All signif.
F(HO)04	Export objectives	5	All signif.
F(HO)05	Export benefits	4	2 signif.
F(HO)06	Export barriers	3	All insignif.
F(HO)07	Competitive situation	2	All insignif.
F(HO)08	Competitive strengths	5	All signif.
F(HO)09	Size of the firm	3	All signif.
F(HO)10	Technological strengths	3	All signif.
F(HO)11	Business experience	2	All signif.
F(HO)12	Product characteristics	3	All signif.
F(HO)13	Commitment to export	8	All signif.
F(HO)14	Willingness to change	4	All signif.
F(HO)15	Education	2	All signif.
F(HO)16	Demographic	2	1 signif.
F(HO)17	Government's export	9	8 signif.
meredier trut	assistances	antrassi at	
F(HO)18	Awareness of the	6	All signif.
F(HO)19	Government export	6	3 signif.
F(HO)20	Government export	3	1 signif.
F(HO)21	Foreign market conditions	5	All signif.
F(HO)22	Commercial facilities	4	All signif.
F(HO)23	Raw material problems	4	All insignif.
F(HO)24	Export costs	4	2 signif.
F(HO)25	Physical distribution	3	All signif.
F(HO)26	Domestic market conditions	2	All signif.

The Correlation Coefficients Between the Variables Which Comprise Each Independent Factor (Taken Separately) and the Level of Exporting

Significance level = 0.05 and 0.01

(1) The "Managerial Aspirations" Factor: Variable Correlations

The "managerial aspiration" factor is composed of six variables (Table 8.7). The result indicates that each of these independent variables is related to a high level of exporting. An examination of the direction of the relationship between each variable and dependent variable also indicates all but one are positively correlated with the dependent variable. This variable which is negatively associated with the dependent variable is found to be securing steady growth for firms. it might indicate that the need for securing steady growth is an important goal for firms with a lower level of exporting. This result is confirmed by other studies, (e.g., Bilkey 1978, Cavusgil 1976. Axinn 1985, 1988, and Moon & Lee 1990).

In comparison to other attributes of this factor, reducing the cost of production is shown to be the most important attribute related to the level of exporting. This might indicate that the higher the aspirations of management to reduce the cost of production, the higher the level of exporting will be. Therefore, it may be concluded that the "managerial aspirations" factor attributes (taken together, or separately) are important as determinants of the level of exporting, in particular, the managerial aspiration to reduce the cost of production attribute.

Table (8.7)

The Correlation Coefficient Between the Variables of the "Managerial Aspiration" Factor and the Level of Exporting

No.	Variables	R
V31	Improving the firm's market position	.2331**
V32	Increasing the overall profit	.3635**
V33	Securing steady growth	3594**
V34	Developing market strategies	.3667**
V35	Reducing the cost of production	.3878**
V36	Increasing the sales volume	.3478**

(2) The "Operational Obstacles" Factor: Variable Correlations

The "operational obstacles" factor consists of 8 variables (Table 8.8). The result indicates that each of these variables is shown to have a negative association with the level of exporting. Among the classes of obstacles, lack of export expertise is found to be the most important related to the dependent variable. This implies that lack of personal expertise or know - how is an important determinant of a high level of exporting.

The attributes of the "operational obstacles" factor (taken together) are also found to be importantly related to the level of exporting. The possible interpretation of this might be that the higher the perception of operational export obstacles, the lower the level of exporting will be.

There is a strong support for this result. Indeed, it was revealed by many of the studies, that the perception of export obstacles will adversely determine the firm's level of exporting, (e.g., Czinkota & Johnston (1983) & Bilkey (1978)).

Table (8.8)

The Correlation Coefficient Between the Variables of the "Operational Obstacles" Factor and the Level of Exporting

No.	Variables	R
V67	Difficulty of obtaining sales representatives	2219**
V68	Difficulty of pricing products	1920**
V65	Lack of knowledge in communicating	2766**
V70	Difficulty of locating potential demand	2706**
V66	Lack of personal expertise	2829**
V75	Size of production	2321**
V69	Inability to establish distribution systems	2924**
V71	Difficulty of maintaining control	2348**

** = 0.01 significance level * = 0.05 significance level

(3) The "Perceived Risk" Factor: Variable Correlations

The "perceived risk" factor is composed of five types of risks (Table 8.9). The result shows that each of these independent variables is related to the level of exporting. Also, the result of the "perceived risk" acting as a factor is found to be importantly related to the level of exporting. This result is in line with many export studies, such as Bilkey (1978), Schlegelmilch (1986) and Ortiz-Buonafina (1990).

Therefore, it might be concluded that the "perceived risk" attributes (taken together, or separately) are important determinants of the level of exporting, in particular, the political risk attribute.

Table (8.9)

The Correlation Coefficient Between the Variables of the "Perceived Risk" Factor and the Level of Exporting

No.	Variables	R
V93	The high risk of production	1711**
V98	The high political risk	7086**
V88	The high risk of transportation	1773**
V78	The high foreign exchange risk	1099*
V77	The high risk of collecting money	1779**

** = 0.01 significance level * = 0.05 significance level

(4) 'The "Export Objectives" Factor: Variable Correlations

The "export objectives" factor is composed of 5 independent variables. An examination of Table (8.10) indicates that each of these variables is shown to have a relationship with the level of exporting. When compared to other variables of this factor, the increasing the security of the firm's investment attribute is the only one that was found to be adversely associated with the level of exporting. The result also indicates that management's expectation concerning profit is the most important attribute that correlated with the dependent variable. It appears that managers view exporting primarily as an alternative means for making greater profit. This result is found to be in conflict with Cavusgil's study. Cavusgil (1976) reported that while growth was the main alternative concern for the firm, profit was found to be the second. A possible explanation of this could be related to the different originations of the firms which were studied.

Also, the attributes of this factor as taken together are found to be importantly related to a high level of exporting. Therefore, it may be concluded that the attributes of this factor are important as determinants of the level of exporting, in particular, the management's expectation of making greater profit attribute.

Table (8.10)

The Correlation Coefficient Between the Variables of the "Export Objectives" Factor and the Level of Exporting

No.	Variables	R
V27	Making greater profit	.3449**
V26	Using the firm's excess capacity	.2304**
V30	Increasing the firm's annual sales	.2391**
V28	Increasing the security of the firm's	1179*
V21	Increasing the growth of the firm	.2173**

** = 0.01 significance level * = 0.05 significance level

(5) 'The "Export Benefits" Factor: Variable Correlations

An examination of Table (8.11) indicates that, out of 4 variables which constitute the "export benefits" factor, only two variables are found to be associated with the level of exporting. These two variables are; overcoming instability demand and benefiting from the government export assistance. This result might indicate that these two variables are the most important benefits related to a high level of exporting. The lack of significance level for the other two variables, i.e., preventing dependency in the domestic market and having better opportunities in the foreign market, could be more important in determining whether or not the firm would become involved in export marketing.

However, these attributes acting as a factor are regarded as important to a high level of exporting. This might indicate that the attributes of this factor (taken together) are more important to a high level of exporting than when taken individually. The possible explanation might be attributed to either of two reasons; the use of the aggregate measure (i.e., factor) may produce more power than depending on each attribute acting individually, or perhaps the different types of the industrial firms under investigation. Some of these types might regard these attributes taken together as being critical to their high level of exporting.

Table (8.11)

'The Correlation Coefficient Between the Variables of the "Export Benefits" Factor and the Level of Exporting

No.	Variables	R
V16	Overcoming instability demand	.1988**
V29	Better opportunities in the foreign market	.0851
V20	Benefiting from the government's export assistance	.1719**
V22	Preventing dependency in the domestic market	.0851

** = 0.01 significance level * = 0.05 significance level

(6) The "Export Barriers" Factor: Variable Correlations

The "export barriers" factor consists of three independent variables. An examination of Table (8.12) indicates that each of these independent variables is not shown to be an important to the level of

exporting. The result for these variables acting as a factor are also found to be not important to the dependent variable. This might indicate that these attributes of this factor are not regarded as important determinants of the level of exporting whether taken together, or separately. The possible explanation of the lack of significance might be attributed to the fact that these types of export barriers have more influence on the firm's decision to export.

Table (8.12)

The Correlation Coefficient Between the Variables of the "Export Barriers" Factor and the Level of Exporting

No.	Variables	R
V73	Inability to provide follow up services	0418
V74	Inability to provide credit facilities	0929
V72	Difficulty of collecting money	0840

** = 0.01 significance level * = 0.05 significance level

(7) The "Competitive Situation" Factor: Variable Correlations

The "competitive situation" factor consists of two independent variables; keeping pace with local competitors and overcoming competition in the domestic market. An examination of Table (8.13) reveals that each of them is shown to be not important to the level of exporting. These attributes acting as a factor are also found to be not important to the dependent variable.

Table (8.13)

The Correlation Coefficient Between the Variables of the "Competitive Situation" Factor and the Level of Exporting

No.	Variables	R
V23 V19	Overcoming competition in the domestic market Keeping peace with local competitors	.0300

Therefore, it may be concluded that the attributes of this factor are not important determinants of the level of exporting (taken together, or separately). The possible explanation is that the competitive situations in the domestic market might be critical to the firm to become involved in exporting rather than to increase its level of exporting.

(8) The "Market Competitive Strengths" Factor: Variable Correlations

The "market competitive strengths" factor is composed of six independent variables (Table 8.14). The result indicates that each of these variables is shown to have a direct relationship with the level of exporting. In comparison to the other variables of this factor, the strength of management attribute is the most important related to the dependent variable.

Also, the findings in respect of these attributes acting as a factor are regarded to be important to a high level of exporting. Therefore, it might be concluded that these attributes (taken together, or separately) are important determinants of the level of exporting, in particular, the strength of management skills attribute.

Table (8.14)

The Correlation Coefficient Between the Variables of the "Market Competitive Strengths" Factor and the Level of Exporting

No.	Variables	R
V54	Ability to hire trained persons for the	.2703**
V53	Ability to offer discount prices to overseas customers	.2824**
V24	The strength of management	.3543**
V18	The proximity to foreign market	.1045*
V49	The firm's technological standard level	·2343**

** = 0.01 significance level * = 0.05 significance level

(9) The "Size of the Firm" Factor: Variable Correlations

The "size of the firm" factor consists of three variables; number of employees, total of capital assets, and sales volume. The result indicates that each of them is shown to be important to a high level of exporting. An examination of Table (8.15) reveals that the total capital assets, which each firm holds, is the highest one that directly correlated with the level of exporting. This might indicate that the size of the capital assets that are held by each firm would be critical to its high level of exporting.

Also, these attributes as acting together are found to be directly related to the level of exporting. Therefore, it might be concluded that the attributes of this factor are important as determinants of the level of exporting (taken together, or separately), in particular, the amount of capital assets held by each firm.

Table (8.15)

The Correlation Coefficient Between the Variables of the "Size of the Firm" Factor and the Level of Exporting

No.	Variables	R
V104	Number of employees	.3552**
V106	Total of the capital assets	.4336**
V108	Total sales volume	.3885**

** = 0.01 significance level * = 0.05 significance level

(10) The "Firm's Technological Strengths" Factor: Variable Correlations

The "firm's technological strengths" factor consists of three variables (Table 8.16). The results reveal that each of these variables is found to be directly associated with the level of exporting. However, these attributes as acting together are found to be irrelevant to the dependent variable. This might indicate that these attributes (taken individually) are more important to the level of exporting than when they are taken together, (i.e., as a factor). There appears to be a contradiction in the result. It might be attributed partially to the fact that most of the firms which were surveyed did not produce a high technological product, or it could be related to the general low level of technological development in the Jordanian environment. However, this result still needs further investigation.

Table (8.16)

The Correlation Coefficient Between the Variables of the "Firm's Technological Strengths" Factor and the Level of Exporting

No.	Variables	R
V50	The superior product equipment	.1074*
V51	The superior production methods	.1431**
V55	The higher ability to modify the product	.2332**

** = 0.01 significance level * 0.05 = significance level

(11) The "Business Experience" Factor: Variable Correlations

As can be seen in Table (8.17), the "business experience" factor consists of two types of experience; number of years in business (age) and number of years in exporting. The result indicates that each of these variables is directly associated with the level of exporting. The number of years exporting is the most important type of experience that highly correlated with the dependent variable.

Also, these attributes of this factor taken together are found to be importantly related to the level of exporting. This result is supported strongly by many studies (e.g., Bilkey 1978, McDougall & Stening 1975, Ursic & Czinkota 1984 and Da Rocha *et al.* 1990). Therefore, it can be concluded that the wider the business experience of the firm, the higher the level of exporting will be expected.

Table (8.17)

The Correlation Coefficient Between the Variables of the "Business Experience" Factor and the Level of Exporting

No.	Variables	R
V103	The number of years in business	.1361**
V107	The number of years exporting	.5202**

** = 0.01 significance level * 0.05 = significance level

(12) The "Product Characteristics" Factor: Variable Correlations

The result indicates that each of the three attributes which constitute this factor (Table 8.18) is importantly related to the dependent variable level of exporting. When compared to the other attributes of this factor, the uniqueness of product attribute is found to be the most important that highly associated with the level of exporting.

The attributes of the "product characteristics" factor (taken together) are also regarded as important to the level of exporting. This result is strongly supported by many export studies, such as Kolhede (1984), Schlegelmilch (1986), and Da Rocha *et al.* 1990 . Therefore, it might be concluded that the attributes of the "product characteristics" factor (taken together, or separately) are important as determinants of the level of exporting, in particular, the uniqueness of a product attribute.

Table (8.18)

The Correlation Coefficient Between the Variables of the "Product Characteristics" Factor and the Level of Exporting

No.	Variables	R
V52	Product quality	.1045*
V25	Product competitive price	.2805** 3180**

(13) The "Commitment to Export Related Activities" Factor: Variable Correlations

The "commitment to export related activities" factor consists of 8 independent variables (Table 8.19). The results reveal that each of these variables is importantly related to the level of exporting. When compared to the other variables, the commitment to seek export opportunities attribute is the most important related to the dependent variable.

These variables acting as a factor are also considered as important to a high level of exporting. This result appears to agree with the findings of the previous export studies, such as those reported by Cavusgil and Nevin (1981), Kolhede (1984) and Walters and Samiee (1990). Therefore, it might be concluded that the attributes of this factor (taken together or separately) are important as determinants of the level of exporting, in particular, the commitment to seek export opportunities in foreign markets.

Table (8.19)

The Correlation Coefficient Between the Variables of the "Commitment to Export Related Activities" Factor and the Level of Exporting

No.	Variables	R
V41	We promote our products to overseas customers	.3956**
V40	We participate in international trade fairs	.3595**
V42	We plan to get into foreign markets	.3919**
V43	We send sales representatives to foreign market	.4071**
V44	We have a formal policy	.2894**
V38	We consistently seek export opportunities	.4795**
V39	We carry out a systematic research into the possibilities of exporting	.3837**
V37	We use export marketing research	.4142**

(14) The "Willingness to Change" Factor: Variable Correlations

The "willingness to change" factor is composed of four independent variables (Table 8.20). The result indicates that each of these variables is shown to have a direct association with the level of exporting. In comparison to other attributes of this factor, the management's willingness to change pricing policies attribute is the most important related to the level of exporting. This might indicate that the top management of the firms who are more willing to change pricing policies, are more likely to be those firms at the higher level of exporting.

These attributes acting as a factor are also found to be importantly related to the dependent variable. This result supports previous studies, such as Ortiz-Buonafina (1990), Seifert & Ford (1989), and Cooper & Kleinschmidt (1985). Therefore, it may be concluded that the attributes of this factor (taken separately, or together) are important as determinants of the level of exporting.

Table (8.20)

The Correlation Coefficient Between the Variables of the "Willingness to Change" Factor and the Level of Exporting

No.	Variables	R
V45	Willingness to modify organisational and control system	.3962**
V47	Willingness to change pricing policies	.5646**
V46	Willingness to adapt product policies	.2893**
V48	Willingness to modify distribution system	.5239**

** = 0.01 significance level * 0.05 = significance level

(15) The "Education" Factor: Variable Correlations

The "education" factor consists of two types of education; the level of education that each manager completes and proficiency in a foreign language. An examination of the correlation coefficients results in Table (8.21) indicate that each of these is importantly related to a high level of exporting. When compared to other variables in the same category, proficiency in a foreign language is the most important attribute related to the level of exporting. This might indicate that knowledge of a foreign language is necessary for firms which want to improve their level of exporting. However, previous studies produced mixed results with regard to the relationship between the decision - maker's foreign language skills and the level of exporting. While Schlegelmilch (1986), Khan (1978) and Langston (1976) showed a positive relationship, others such as Simmonds & Smith (1968), Reid (1980) and Moon & Lee (1990) attached little, or no relationship to such skills.

Table (8.21)

The Correlation Coefficient Between the Variables of the "Education" Factor and the Level of Exporting

No.	Variables	R
V120	Proficiency in a foreign language	.4738**
V117	Level of education	.3081**

** = 0.01 significance level * = 0.05 significance level

(16) The "Demographic" Factor: Variable Correlations

The "demographic" factor is composed of two types; age and experience. An examination of Table (8.22) indicates that the manager's experience is the only attribute importantly related to the level of exporting. This might indicate that the wider the experience of the manager in business, the higher the level of exporting is probably. Previous studies reported a strong relationship between the manager's foreign experience and the level of exporting, such as Schlegelmilch (1986), Axinn (1988) and Da Rocha *et al.* (1990). The lack of significance for the age attribute might be related to the fact that it might be more important to the firm's decision to become involved in exporting. However, these attributes acting as a factor are found to be importantly related to the dependent variable. This might be due to either the use of the aggregate measure here which may produce more effect on the dependent variable than each variable acting individually, or perhaps the different demographic characteristics of the decision - makers in the firms under investigation.

Table (8.22)

The Correlation Coefficient Between the Variables of the "Demographic" Factor and the Level of Exporting

No.	Variables	R
V115	Experience	.1523**
V114	Age	0028

** = 0.01 significance level * = 0.05 significance level

(17) The "Government's Export Policies and Assistance" Factor: Variable Correlations

As can be seen in Table (8.23), the "government's export policies and assistance" factor consists of nine variables. The result indicates that 8 of them are correlated with the level of exporting. Among these significant eight variables, the devaluation of the JD and the trade agreement policies are the most important variables associated with the dependent variable. The insignificant variable is found to be the Free Trade Zone facilities. The lack of significance of this variable might indicate that it is either not important as determinant of the level of exporting, or it could be important, but it has not been considered yet, due to the lack of awareness of such a facility among the firms surveyed.
However, the attributes of this factor (taken together) are regarded as important to a high level of exporting. This might indicate that these attributes when taken together are more important as determinants of the level of exporting than when they are acting separately. This might be attributed to either the use of the aggregate measure here (i.e., factor), or perhaps the different types of the industrial firms under investigation. Some of these types might consider these attributes taken together as being critical to their high level of exporting.

Table (8.23)

The Correlation Coefficient Between the Variables of the "Government's Export Policies and Assistance" Factor and the Level of Exporting

No.	Variables	R
V1	Government's trade agreement policy	.2850**
V2	Export credit discount programme	.2224**
V 3	Duty - drawback facilities	.2701**
V4	Foreign market information	.2112**
V5	Managerial services	.1976**
V6	Trade Zones facilities	.0680
V7	Devaluation of JD	.3039**
V8	Temporary - entry facilities	.2859**
V9	The income tax rebate policy	.2024**

** = 0.01 significance level * = 0.05 significance level

(18) 'The "Awareness of the Government's Export Policies and Assistances" Factor: Variable Correlations

The "awareness of the government's export policies and assistances" factor consists of 6 variables. An examination of Table (8.24) indicates that each of these variable is shown to have a direct relationship with the level of exporting.

Also, these attributes acting as a factor are found to be important to the level of exporting. Therefore, it may be concluded that the attributes of this factor whether taken separately or together are important as determinants of a high level of exporting.

Table (8.24)

No.	Variables	R
V10	Knowledge of Free Trade Zone facilities	.1365**
V11	Awareness of the duty- drawback facilities	.3146**
V12	Awareness of the export credit programme	.2690**
V13	Awareness of the managerial services	.2694**
V14	Knowledge of temporary entry facilities	.2313**
V15	Awareness of the export services	.1899**

The Correlation Coefficient Between the Variables of the "Awareness of Government's Export Policies" Factor and the Level of Exporting

(19) The "Government's Export Facilities Related Problems" Factor: Variable Correlations

The "government's export facilities related problems" factor consists of six variables (Table 8. 25). The results indicate that only three variables of them are shown to be related to the level of exporting; the high cost of export documents, lack of commercial attaches, and the bank - guarantee requirement. This might indicate that these variables are the most important that related to a high level of exporting when compared to the other variables of this factor (i.e., the inadequate tax incentive, complexity of the duty - drawback system, and the high duties).

Table (8.25)

The Correlation Coefficient Between the Variables of the "Government's Export Facilities Related Problems" Factor and the Level of Exporting

No.	Variables	R
V59	Inadequate tax incentives	0024
V60	High cost of export documents	1471**
V61	Complexity of the duty drawback system	1273
V62	Lack of Commercial - Attaches	1086*
V63	High duties imposed on spare parts	0747
V80	The bank - guarantees requirement	1256*

However, the attributes of this factor as taken together are found to be related to the level of exporting. The possible explanation of this result might be attributed to either of the two reasons; the use of the aggregate measure (i.e., factor) may produce more effect on the level of exporting than depending on each attribute acting separately, or perhaps some of the different types of the industrial firms under investigation might consider these attributes taken together as being critical to their high level of exporting.

(20) The "Government's Export Procedures Related Problems" Factor: Variable Correlations

The "government's export procedures related problems" factor consists of three types of procedures. Namely; customs' export clearances procedures, export licence procedures, and the export inspection procedure. An examination of Table (8.26) indicates that the export inspection procedures is the only variable related to the level of exporting. The lack of significance for the other two variables might indicate that they are more important in determining the firm's decision of whether or not to become involved in export marketing.

These attributes acting as a factor are also found to be not related to the level of exporting. This result might indicate that these attributes whether taken together, or separately are not regarded as important determinants of the level of exporting, except for the export inspection procedures. This might be attributed to the fact that the inspection process is considered to be one of the main problems encountered by manufacturing firms in Jordan. It was concluded through the preliminary interviews (see chapter 3) that the

time and the routine procedures associated with the export inspection

process is long and complicated.

Table (8.26)

The Correlation Coefficient Between the Variables of the "Government's Export Procedures Related Problems" Factor and the Level of Exporting

stoms export clearance procedures	0274 0683
	stoms export clearance procedures port license procedure port inspection process

** = 0.01 significance level * = 0.05 significance level

(21) The "Foreign Markets Conditions" Factor: Variable Correlations

The "foreign markets conditions" factor includes 5 independent variables (Table 8.27). The results indicate that each of these variables is shown to be related to the level of exporting. When compared to the other variables of this factor, political instability is the most important attribute related to the level of exporting.

These attributes acting as a factor are also found to be related to the level of exporting. Therefore, it may be concluded that the attributes of this factor (taken together or separately) are important as determinants of the level of exporting.

Table (8.27)

The Correlation Coefficient Between the Variables of the "Foreign Markets Conditions" Factor and the Level of Exporting

No.	Variables	R
V94	Difficulty of business practice	1628**
V95	The intensity of competition	6235**
V96	Foreign government restrictions	5461**
V97	Geographical distances	4898**
V98	Political instability	7086**

** = 0.01 significance level * = 0.05 significance level

(22) The "Commercial Facilities Related Problems" Factor: Variable Correlations

The "commercial facilities related problems" factor consists of 4 variables (8.28). These independent variables are; lack of information, lack of export credit guarantees, difficulty of access to banking facilities and lack of effective trade companies. The results indicate that each of these variables is related to the dependent variable. When compared to the other variables of this factor, the lack of export credit guarantee system against non - payment risk is the most important attribute related to the level of exporting.

However, the attributes of this factor (taken together) are found to be irrelevant to the level of exporting. This might reveal that these attributes are more relevant to the dependent variable when they are acting separately than together as a factor. It might be due to the different types of the industrial firms under investigation, some of which might consider only some facilities as being critical to their high level of exporting. The lack of export credit guarantee system against non - payment risk might discourage firms to increase their level of exporting beyond a certain point. This result is recommended for further investigation.

Table (8.28)

The Correlation Coefficient Between the Variables of the "Commercial Facilities Related Problems" Factor and the Level of Exporting

No.	Variables	R
V64	Lack of information	2091**
V76	Lack of export credit guarantee against non - payment risk	2402**
V79	Difficulty of obtaining banking facilities	1370**
V86	Lack of effective trade companies	1474**

(23) The "Raw Material Related Problems" Factor: Variable Correlations

The "raw material related problems" factor is composed of 4 variables, Table (8.29). The result indicates that each of these variables is not related to the level of exporting. Also, the result of these attributes acting as a factor is found to be insignificant.

Therefore, it may be concluded that these attributes are not considered important as determinants of the level exporting whether acting as a factor or separately. The possible explanation of this might be attributed to the fact that these attributes are more important in determining whether the firm would become involved in exporting rather than the level of exporting.

Table (8.29)

'The Correlation Coefficient Between the Variables of the "Raw Material Related Problems" Factor and the Level of Exporting

No.	Variables	R
V83	The Difficulty of obtaining raw material	0522
V84	The low quality of raw material	0023
v85	The shortage of raw material	0439
V92	The high price of raw material	0939

** = 0.01 significance level * 0.05 = significance level

(24) The "Export Costs" Factor: Variable Correlations

The "export costs" factor consists of 4 types of costs (Table 8.30). These four types are; cost of communication, cost of labour skills, warehouse facilities, and interest rate. The result indicates that only two variables are related to the dependent variable. These two variables are; cost of labour skills and high interest rates. This might indicate that these two variables are more important to the high level of exporting when compared to the other variables of this factor, i.e., the cost of communication and the cost of warehouse facilities.

However, these attributes acting as a factor are found to be not related to the level of exporting. This might due be to the different types of the industrial firms under investigation. Some of these types might regard only some attributes as being critical to their high level of exporting.

Table (8.30)

The Correlation Coefficient Findings Between the Variables of the "Export Costs" Factor and the Level of Exporting

No.	Variables	R
V89	High cost of communication	0571
V90	High cost of labour skills	1469**
V91	High cost of obtaining warehouse facilities	0936
V101	High interest rate	1869**

** 0.01 significance level * 0.05 significance level

(25) The "Physical Distribution" Factor: Variable Correlations

The "physical distribution" factor is composed of six variables (Table 8.31). The result indicates that each of these variables is importantly related to the level of exporting variable. However, these variables acting as a factor is found to be irrelevant to the dependent variable. This might be attributed to the different types of the industrial firms under investigation, some of which might consider only some attributes of this factor as being important to their high level of exporting.

Therefore, it may be concluded that these variables are more important to the dependent variable level of exporting when they are taken separately than when they are taken together.

Table (8.31)

'The Correlation Coefficient Between the Variables of the "Physical Distribution" Factor and the Level of Exporting

No.	Variables	R
V81	Difficulty of obtaining shipping facilities	1340**
V82	Difficulty of obtaining warehouses facilities	1099**
V87	High freight of transportation	1895**

** 0.01 significance level * 0.05 significance level

(26) The "Domestic Market Conditions" Factor: Variable Correlations

The "domestic market conditions" factor consists of two variables; intense competition and the high domestic demand. An examination of Table (8.32) shows that each of them is related to the level of exporting.

exporting.

Also, these two variables acting as a factor are found to be importantly related to the dependent variable level of exporting. Therefore, it may be concluded that these variables (taken separately, or together) are important as determinants of the level of exporting.

Table (8.32)

The Correlation Coefficient Between the Variables of the "Domestic Market Conditions" Factor and the Level of Exporting

No.	Variables	R
V99	Intense competition	6926**
V100	The high domestic demand	6620**

** = 0.01 significance level * = 0.05 significance level

Summary and Conclusions

This chapter presents the main findings related to the study's objectives and testing relevant hypotheses (HO1, HO2, HO3 HO4 and HO5) by using the Pearson correlation coefficient and the stepwise multiple regression. The findings can be summarized as follows:

(a) Of 26 explanatory independent factors, 18 factors were each found to be significantly associated with the level of exporting. These factors were: (1) "Foreign market conditions related problems", (2) "Willingness to change" (3) "Demographic" (4) "Commitment to export related activities", (5) "Size of the firm", (6) "Managerial aspirations", (7) "Operational obstacles" (8) "Market competitive strengths", (9) "Government's export policies and assistances", (10) "Business experience", (11) "Export objectives" (12) "Awareness of government's export policies and assistance" (13) "Product characteristics" (14) "Perceived risk", (15) "Government's export facilities related problems", (16) the "Domestic market conditions related problems", (17) "Education", and (18) "Export Benefits".

(b) The firm's internal environmental dimension acting alone (i.e., all 16 factors, acting together) was found to be significantly associated with a high level of exporting. The most important factors of the regression equation were: (1) "Willingness to change", (2) "Commitment to export related activities", and (3) "Size of the firm". The adjusted R square result indicated that about 49% of the variation on the dependent variable level of exporting could be explained by the above significant factors, acting together.

(c) The firm's external environmental dimension acting alone (i.e., all 10 factors, acting together) was found to be significantly related to a high level of exporting. The most important factors of

A Summary Review of the Correlation Coefficients Between the Variables Constituting Each Independent Factor (Taken Separately and Together) and the Level of Exporting

The Independent Factors	No. of Variables	Taken Together	Taken Separately
01. Managerial aspirations	6	Signif.	All signif.
02. Operational obstacles	8	Signif.	All signif.
03. Perceived risk	5	Signif.	All signif.
04. Export objectives	5	Signif.	All signif.
05. Export benefits	4	Signif.	2 signif.
06. Export barriers	3	Insignif.	All insignif.
07. Competitive situation	2	Insignif.	All insignif.
08. Competitive strengths	5	Signif.	All signif.
09. Size of the firm	3	Signif.	All signif.
10. Technological strengths	3	Insignif.	All signif.
11. Business experience	2	Signif.	All signif.
12. Product characteristics	3	Signif.	All signif.
13. Commitment to export	8	Signif.	All signif.
14. Willingness to change	4	Signif.	All signif.
15. Education	2	Signif.	All signif.
16. Demographic	2	Signif.	1 signif.
17. Government's export assistances and policies	9	Signif.	8 signif.
18. Awareness of the government export	6	Signif.	All signif.
19. Government export facilities	6	Signif.	3 signif.
20. Government export procedures	3	Insignif.	1 signif.
21. Foreign market conditions	5	Signif.	All signif.
22. Commercial facilities	4	Insignif.	All signif.
23. Raw material problems	4	Insignif.	All insignif.
24. Export costs	4	Insignif.	2 signif.
25. Physical distribution	3	Insignif.	All signif.
26. Domestic market conditions	2	Signif.	All signif.

Significance Levels = 0.01 & 0.05

regression equation were: (1) "Foreign markets conditions", (2) "Government's export policies and assistances", (3) "Awareness of the government's export policies", (4) "Export costs" and (5) the Domestic market conditions". The adjusted R square result indicated that about 52% of the variation of the level of exporting could be explained by those five significant factors.

(d) Comparing between (b) and (c), it may be concluded that the variation of the level of exporting is more explained by factors outside the boundary of the firm, i.e., the firm's external environmental dimension than its internal environmental dimension.

(e) The combination of the two dimensions, (i.e., all 26 factors, acting together) was found to be significantly related to a high level of exporting. There were 7 factors included here in the regression equation: (1) "Foreign markets conditions" (2) "Willingness to change" (3) "Commitment to export related activities" (4) "Size of the firm" (5) "Domestic market conditions" (6) "Business experience", and the (7) "Government export facilities related problems". The adjusted R square indicated that about 64% of the variations on the dependent variable could be explained by the above factors.

(f) The integration approach of the internal and external dimensions not only comes up with a better explanation of the variation of the dependent variable level of exporting, but also comes up with more number of determinants in comparison to each dimension acting alone.

(i) The results of the relationship between the variables constituting each independent factor and the level of exporting were presented and discussed. A summary review of these findings is given in Table (8.33).

CHAPTER NINE

THE DETERMINANTS OF EXPORT BEHAVIOUR: EXPORTERS VS. NON EXPORTERS

Introduction

- 9.1 Testing the Research Hypotheses (HO6 To HO10)
- 9.1.1 Testing the Discriminant Function / Factors: Chi Square and Univariate F Ratio
- 9.1.2 Testing the Improvement in Predicting Group Membership: The McNemar Test
- 9.2 Identification and Prediction of Export Behaviour: Interpretation of DFA
- 9.2.1 Comparative Analysis Between the Internal and External Dimensions in Respect of Export Behaviour
 - 9.2.2 An Integration Approach to Identify and to Predict Exporter Status
- 9.2.3 Validation of the Prediction of Export Behaviour (Exporter's and Non Exporter's Group Membership)
 - 9.2.4 The Relative Important of the Predictor Factors in Terms of Their Contributions
- 9.3 The Differences Between Exporters and Non Exporters: Variable Findings

9.4 The Interpretation of the Variable Findings

Summary and Conclusions

CHAPTER NINE

THE DETERMINANTS OF THE EXPORT BEHAVIOUR: EXPORTERS VS. NON - EXPORTERS

Introduction

In Chapter Eight the main findings pertaining to the determinants of the level of exporting (the percentage of export volume) were analysed and discussed by using the Pearson correlations coefficients and the stepwise multiple regression analysis.

In this Chapter, the 26 factors^{*} and associated variables identified in Chapter 7 will be analyzed again for the following purposes:

- To find out whether or not a significant difference exists between the exporters and non - exporters in terms of the internal and external environmental measures, (i.e., 26 factors).

- To predict group membership of the export behaviour on the basis of these 26 factors.

- To identify the degree of association between the export behaviour (i.e., exporters and non - exporters) and the internal environmental measures, (i.e., 16 factors).

- To identify the degree of association between the export behaviour (i.e., exporters and non - exporters) and the external environmental measures, (i.e., 10 factors).

- To discover whether the addition of the external environmental measures (i.e., 10 factors) to the internal environmental measures (i.e., 16 factors) might improve the prediction of the group membership, (i.e., classification).

- To find out whether the two groups (i.e., exporters and non - exporters) are different in terms of the variables comprising each factor.

The statistical analysis techniques used are; the discriminant analysis functions, Chi - Square, F - test, McNemar test, and the T - test. Factors and variables will be analyzed and discussed respectively.

* A list of the factors is given in Tables (7.9) and (7.10), Chapter 7

9.1 Testing the Research Hypotheses (i.e., HO5 to HO10)

9.1.1 Testing the Significant of Discriminant Function / Factors: Chi - Square and Univariate F Ratio

Before attempting to interpret the output of DFA, it was thought that it would be better to check on its statistical significance. A statistically significant function means that there is a meaningful differentiation of the groups on the discriminant score (Churchill, 1983). For testing the solution of DFA in this study, two key statistics were used: Chi - square test and univariate F ratio.

The chi - square test was employed to determine the significance of the discriminant function for each dimension (i.e., internal dimension, external dimension, and both), or otherwise the distinction between exporter's group and non - exporter's group in terms of their internal and external environmental dimensions. For testing the significance of each discriminator (or predictive factor), the univariate F statistics was used as well.

The following tested hypotheses are those pertaining to the significance of the discriminant function of the components of the internal environmental dimension (16 factors, taken together), and the components of the external environmental dimension (10 factors, taken together).

The null hypotheses were stated as follows:

- HO(6): There is no significant difference between the two groups (i.e., exporters and non - exporters) in terms of their internal environmental measures, taken together.
- HO(7): There is no significant differentiation between the two groups (i.e., exporters and non - exporters) on the basis of their external environmental measures, taken together.

For hypothesis HO(6), the computed chi - square is (443.54) with 13 degrees of freedom. The obtained chi - square value exceeds the critical value (34.54) at .000 (or far beyond .001 level) Table (9.4). The decision, therefore, is to reject the null hypothesis HO(6), and conclude that the discriminant function of the components of the internal environmental dimension is statistically significant. Stated somewhat differently, the internal environmental measures (i.e., "managerial aspirations", "willingness to change", "business experience", "export benefits", "market competitive strengths", "technological strengths", "demographic", "product characteristics", "size of the firm", "export objectives") do discriminate between exporters and the non - exporters.

With regard to hypothesis HO(7), the computed value of chi- square is (214.82) with 7 degree of freedom (see the output of DFA in Table 9.4). The obtained chi - square value exceeds its critical value (20.98) with .000 (or too far beyond .001 level of significance). Therefore, the decision is to reject the null hypothesis HO(7), and conclude that the discriminant function of the components of the external environmental dimension is statistically significant, i.e., the "foreign market conditions", "awareness of the government's export policies and assistance", the "government's export policies and assistance", "commercial facilities related problems", "export costs", and "raw material related problems" do distinguish the exporter's group from non - exporter's group.

In addition to the chi - square test, the univariate F statistics test was employed to determine the significance level of each predictor independent factor included in each discriminant function, the null hypotheses were stated as below:

- HO(8): There is no significant differentiation (i.e., variation) between the two groups (i.e., exporters and non - exporters) in terms of their internal environmental dimension measures (mentioned above), taken separately.
- HO(9): There is no significant determination (i.e., variation) between the two groups (i.e., exporters and non - exporters) in terms of their external environmental dimension measures, taken separately.

Tables 9.1 and 9.2 summarize the computed value of F ratio and its significance level for each predictor factor included in the two discriminant functions of the internal and external environment, respectively. A closer look at the computed univariate F value in Table (9.1) indicates that "managerial aspirations", "willingness to change", "commitment to export related activities", "business experience", "market competitive strengths", "perceived risk", "technological strengths", "export objectives", "operational obstacles", and "demographic" factors are found to be significant beyond .001 or .01 level. However, the "education", "export barriers", "competitive situations", "product characteristics", and "export benefits" factors are found to be insignificant.

Table (9.2) shows that, out of 10 external environmental factors included in the analysis, 8 are found to be significant beyond .001 or 0.05. Only 2 external factors are found to be insignificant. Therefore, one should reject the null hypothesis HO(9) with respect to those 8 significant factors, but accept HO(9) regarding the others. Each of these 8 external factors (i.e., the "government's export policies and assistance", "awareness of the government's export policies and assistance", the "government's export facilities related problems", "government export procedures", "foreign market conditions"

Table (9.1)

Univariate F Value and Its Significant Level in the DFA of the Internal Environmental Dimension: Taken Separately

Internal - Environmental Independent Factors	Univariate F Value	Sign. Level
El Openvettent - etchica paireste	20.00	
1. Managerial aspirations	269.40	0.0000
2. Operational obstacles	38.83	0.0000
3. Perceived risk	14.43	0.0002
4. Export objectives	88.67	0.0000
5. Export benefits	3.36	0.0675
6. Export barriers	0.28	0.5923
7. Competitive situation	2.05	0.1529
8. Competitive strengths	79.28	0.0000
9. Size of the firm	3.72	0.0545
10. Technological strengths	7.64	0.0060
11. Business experience	95.56	0.0000
12. Product characteristics	1.96	0.1622
13. Commitment to export	31.80	0.0000
related activities	8.2	
14. Willingness to change export marketing policies	228,20	0.0000
15. Education	2.30	0.1304
16. Demographic	101.80	0.0000

Univariate F Value and Its Significant Level in the DFA of the External Environmental Dimension: Taken Separately

External - Environmental Independent Factors	Univariate F Value	Sign. Level
1. Government's export policies and assistance	70.2	0.0000
2. Awareness of the government export assistance	70.6	0.0000
 Government's export facilities related problems 	4.6	0.0326
4. Government export procedures	6.0	0.0147
5. Foreign market conditions	77.5	0.0000
6. Commercial facilities related problems	21.8	0.0000
7. Raw material related problems	4.2	0.0416
8. Export Costs	8.0	0.0048
9. Physical distribution related problems	0.5	0.4706
10. The domestic market conditions	0.2	0.6705

"commercial facilities related problems", "raw material related problems", and the "export costs") distinguish the exporter's group from the non - exporter's group.

9.1.2 Testing the Improvement in Predicting Group Membership

The McNemar test for the significance of change was used to determine the significant improvement in the classification of group's membership (i.e., exporter's group and non - exporter's group) after adding all the 10 external environmental factors to the other 16 factors of the internal environment in the DFA model. It was hypothesized that:

HO(10): There is no significant improvement in the discrimination (i.e., variation) between the two groups (i.e., exporters and non - exporters) after the addition of external environmental measures (i.e., 10 factors) to the internal environmental measures (i.e., 16 factors) in the prediction model of DFA.

Based upon the classification results obtained from the first (internal dimension) and second (external dimension) runs of DFA, individual cases are tabulated as to whether they are correctly or incorrectly classified in the early discriminant function run (i.e., internal environmental measures only) and the later discriminant function run (i.e., external environmental measures combined with internal environmental measures) in Table (9.3).

Referring to Table (9.3), those individuals who have the same results in both run (cell A & D) should be ignored since they show no change (or improvement). Cell B includes those individuals who were incorrectly classified in the early run and correctly classified in the later run, and cell C includes individuals who were correctly classified in the early run and incorrectly classified in the later.

Fourfold Table of McNemar Test

Early run of DFA (Internal dimension only)

		Correct	Incorrect	
Later run of DFA After the addition of the external dimension)	Correct	356 (A)	16 (B)	
	Incorrect	5 (C)	8 (D)	

Therefore, the chi - square distributions for change is

$$X^2 = \frac{(/B - C/ - 1)^2}{B + C}$$

Naturally the null hypothesis of no improvement would only be rejected, if more individuals became correctly classified after the addition of external predictors (i.e., B > C). Therefore, if B > C and the computed value of X^2 is greater than its critical value within 1 degree of freedom at .05 level of significance, we reject the HO(9) and conclude that the addition of a predictor in the later run of DFA (i.e., external environmental measures) has improved the solution.

By applying the above formula, the

$$x^{2} = \frac{(/16 - 5/-1)^{2}}{21}$$

= 4.76

The obtained X^2 value of (4.76) with 1 degree of freedom exceeds its critical value (3.84) at .05 significance level. As a result, the decision is to reject the null hypothesis, and to conclude that there was significant change or improvement in the classification of group membership after adding the 10 independent factors of the external environment to the 16 independent factors of the internal environment.

- 9.2 Identification and Prediction of Export Behaviour: Interpretation of DFA
 - 9.2.1 Comparative Analysis Between the Internal and External Dimensions in Respect of Export Behaviour

Various attempts have been made in the literature to describe and predict export behaviour, (e.g., Cavusgil and Naor 1987, Schlegelmilch 1983, Burton and Schlegelmilch 1987 and Barker & Kaynak 1992, etc.). These studies employed internal environmental variables as predictors of export behaviour. However, external environmental variables have received little attention in classification of exporters from non exporters.

Therefore, one of the aims of the present study is to classify and predict the groups' membership (i.e., exporters and non - exporters) on the basis of those two environmental dimensions (i.e., 16 internal factors and 10 external factors, taken separately) and to compare between those two dimensions in terms of their predictive power.

In order to accomplish this objective, the 16 independent factors of the internal environmental dimension and the 10 independent factors of the external environmental dimension were submitted to the stepwise DFA computer programme in SPSSX. Two stepwise runs of DFA were performed on the factors of internal and external environments respectively. Table (9.4) contains the summary results of the two separate discriminant functions of internal and external factors.

The eigenvalue (2.35 and 0.75) and associated correlation coefficient (0.83 and 0.65) in Table (9.4) denote the relative degree of relationship between each type of the two dimensions (i.e., internal and external) and export behaviour (i.e., exporters vs. and non - exporters). The internal dimension has a higher degree of association with export behaviour than the external dimension (Table 8.4). Moreover, the low Wilks' lambda (0.30) and the high value of chi

- square (443.54) associated with the first discriminant function indicate that the internal dimension and export behaviour are highly related.

In addition, the classification results obtained from each discriminant function helps to visualize exactly how accurate each discriminant function was in predicting exporter's and non- exporter's group membership. By looking at Table (9.4), it can be seen that the first discriminant function (internal dimension) has correctly classified (95.84%) of the respondents into two discriminant groups, whereas the second discriminant function (external dimension) is correctly classified (83.38%).

Table (9.4)

Comparative Results of Two Separate Discriminant Analysis Function

Results	Run * 1 Internal Dimension	Run * 2 External Dimension
Eigenvalue Canonical correlation	2.24811 0.8319426	0.76133 0.6574558
Wilk's lambda Chi - squared	0.3078714 443.54	0.5677519
Degree of freedom Level of significance	13 0.0000	7 0.0000
Cases correctly classified	95.84%	83.38%
Perfectly accurate classification	17.84	etseessa 5.38 meessemeese

One way to evaluate the classification from a discriminant function is to compare it to the classification which one would expect by chance (Hair *et al.* 1987). The question is, however, how much better than chance should the accuracy of classification be? One practical approach recommended by Hair *et al.* (1987) suggests that in order to be useful the classification accuracy must be at least 25.0% greater than chance. The probabilities associated with chance, in each discriminant function in the research, are given by the following formula:

$$C = P^2 + (1 - P)^2$$

Where:

C = chance P = proportion of cases in group 1 1 - p = proportion of cases in group 2

By applying, the above formula (in any of the two discriminant function) then :

$$c = \left(\frac{272}{387}\right)^2 + \left(1 - \frac{272}{387}\right)^2$$

= 53%

Based upon the Hair *et al.* approach, this means that a perfectly acceptable classification level would be at least 25% greater than 53% of the achieved by chance 66.25%. In the discriminant function of the internal dimension, the classification accuracy level is 29.57% greater (i.e., 95.84% - 66.25% = 29.57%). Whilst the perfectly acceptable level is 17.13% greater (i.e., 83.38% - 66.25% = 17.13%) in the discriminant function of the external dimension.

Further comparative results between internal and external dimensions can be obtained from the all groups' histogram of the two separate discriminant analysis functions. Figures (9.1) and (9.3) give a clear picture of the differentiation between the two groups according to their internal and external dimensions respectively. Group G1 (exporters) and group G2 (non - exporters) are well separated on the basis of internal dimension (see Figure 9.1), but they are less separate on the basis of the external dimension (see Figure 9.2).

This comparative analysis between the external and internal environmental dimensions in relation to export behaviour supports the

^{*} The survey result indicates that out of 387 firms, 115 firms are not exporters.

previous export behaviour studies in that the firm's internal environmental measures are more critical to its decision of whether or not to become involved in exporting, (e.g., Cavusgil and Nevin 1981). However, the firm's external environmental measures were found to be more critical to its level of exporting (see Chapter Eight).

9.2.2 An Integration Approach to Identify and to Predict Exporter Status

It was decided to combine the two environmental dimensions (i.e., 26 factors) together in order to gain more insight into export behaviour and to understand better the differentiation between exporters and non - exporters.

Figure (9.1)

All Groups Stacked Histogram: Internal Dimension (Run * 1) G1 = Exporters G2 = Non - Exporters



Such a combination of the two useful dimensions provides a unique and important view of export behaviour and it allows the researcher to identify the whole firm's environment rather than isolated fragments. Accordingly, it was decided to combine the factors of the two dimensions (i.e., 26 factors) included in the analysis so as to identify and predict export behaviour.

The DFA programme was performed in a separate run on the two dimensions (combined together) to predict the group's membership. The results obtained from the third run of DFA show that integration improves the final solution in terms of the degree of association, the discrimination between exporters and non - exporters and the predictive power of group membership (see Table 9.5).

Table (9.5)

Function E	igenvalue	Canonica Correlat	1 Will ion Lamb	da Squar	- DF	Sign.
1 2	.8	.86	.26	493	21	0.0000
			Clas	sification	n Matrix	
Actual Group	No. o	f Cases	Predicted Group G1 G2		Membership Total	
Group (1) exporters	27	0	265	5	270	
Group (2) non - export	ers 11	3	5	108	113	
Total	38	3	270	113	383	

Canonical Discriminant Function and Classification Matrix (Run * 3)



Figure (9.3)

All Groups Stacked Histogram: Combination (Run * 3) G1 = Exporters G2 = Non - Exporters

CANONICAL DISCRIMINANT FUNCTION I



In the third discriminant function, the degree of relationship between all the predictors (i.e., internal and external factors) and export behaviour (i.e., exporters vs. non - exporters) is increased due to the improvement in the eigenvalue (2.8) and the associated Canonical correlation (0.86). The ability of the discriminant function to differentiate between the two groups (i.e., exporters and non exporters) is also improved because of the reduction in the value of Wilk's lambda to 0.26. Equally important is the improvement in the predictive power of the discriminant function which is increased to about 97.39% (see Table 9.5). The classification obtained from the third discriminant function, integrated approach exceeds 19.39% the perfectly accurate level of Hair et al. (i.e., 97.39% - 66.25% =31.14%)

The combination of the internal and external predictors, in one discriminant function (Run 3), indicated that 98.1% of exporters and 95.6% of non - exporters are correctly assigned to their respective groups. Furthermore, this function also showed significant improvement in the discrimination between the two groups, (see the result of McNemar test section 9.1.2, Chapter 9).

9.2.3 Validation of the Prediction of Export Behaviour (Exporter's and Non - Exporter's Group Membership

The question of predictive power is important. As it has been reported earlier, "relative - to - chance" is considered an important way of checking the accuracy of the classification results obtained from two groups DFA. This procedure is commonly used to test whether the proportion of correctly classified cases in the sample is significantly different from the correct proportion that would be expected by chance. However, the "relative - to - chance" measure will be biased if it is applied to the same sample of data used to estimate the discriminant function coefficient. This bias is due to the sampling means of the population (Frank & Massy 1965); "the direction of the bias is to show greater predictive power in classification that actually exists among the true populations. Its magnitude will decrease as the sample size becomes larger".

In order to check the validity of the prediction of power (classification) of export behaviour, two methods of discriminant validity were used, namely, the Split half and the Jacknife method.

(1) Split Half Method

In order to reduce the bias caused by sampling error and then, increase the efficiency of predictive power in DFA, one could split the original sample and use one part for analysis (i.e., analysis sample) and the other for validation (i.e., validation sample). This approach uses the discriminant coefficients derived from the analysis sample to predict group membership for each number of the validation sample (Tabachnick and Fidell 1983).

Therefore, it was decided to check on the validation prediction of exporter's and non - exporter's group membership obtained from each

DFA run. The key steps in conducting the validation procedures are as follows (Frank and Massy 1965);

(1) the original sample (387 cases) was split into two sub samples on the basis of odd and even numbers: one for analysis and the other for validation,

(2) the analysis sample used to determine the discriminant coefficients and to generate a classification table,

(3) using the discriminant coefficients estimated from the analysis sample, predictor of group membership were made for each member,

(4) the differences between the classification of each sample are perfectly acceptable level (i.e., 66.25%, see section 9.2.1).

(5) the above steps were applied to the three set of data: internal, external, and combined respectively.

The results of the validation are shown in Table (9.6), and they confirm our initial findings in that the internal environmental dimension is better than the external dimension in terms of predicting group membership.

Table (9.6)

Validation of the Discriminant Functions: Comparison of the Analysis and validation Sample

Discriminant Analysis sample		amp1e	Validation sample		
function	% of correct classification	> or < H & A's [*]	% of correct classification	> or < H & A's	
- Internal	94.79	+28.54	96.89	+30.64	
- External	80.73	+14.48	84.97	+18.72	
- Both	97.38	+31.13	97.92	+31.77	

* H & A's Approach, i.e., Hair, J., R. Anderson & R. Tatham (1987).

Referring to Table (9.6), the percentage of correct classification analysis sample was less than its counterpart in the validation sample for each discriminant function because of the effects of sampling error. However, the discriminant function of the internal measures is still better than the external dimension measures in predicting group membership.

(2) Jacknife Method

For further confirmation of the earlier results, the Jacknife method was also applied to validate the discriminant function of each of the three DFA runs. There is an evidence that the Jacknife method is superior to other discriminant validation methods including the splitsample approach (Eisenbeis 1977). Since the Jacknife method, unlike the other methods, makes use of all the available data without any serious bias in the estimating error rate (Dillon & Goldstein 1984).

Using the Jacknife method to validate a discriminant function involves leaving out each of the cases in turn, calculating the function based on $n_1 + n_2 - 1$ cases, and then classifying the left out cases. This process is repeated until all the cases are classified (Eisenbeis 1977). Since that the case which is being classified is not included in the calculation of the discriminant function, the method yields almost unbiased estimates of the misclassification probabilities (Dillon and Goldstein 1984).

In order to validate each of our three discriminant functions, the Jacknife method was applied, to the same data, using the BMDP compute package series number P7M. Table (9.7) presents a comparison between the two hit ratio results obtained from the DFA and the Jacknife method with respect to each of the three sets of discriminant functions (i.e., internal, external, and combined).

Table (9.7)

Discriminant Function	The Hit Ratios of the DFA Functions	The Hit Ratios of the Jacknife
Internal	94.5 %	93.8 %
External	83.4 %	82.3 %
Both	97.7 %	96.6%

A Comparison Between the Hit Ratios of the DFA and the Jacknife Method

Table (9.6) shows that the percentage of correct classification of the Jacknife is slightly less higher than the corresponding percentages which were obtained from the DFA function in three runs (i.e., internal, external, and both). On this ground, it could be concluded that each of the three discriminant functions is a valid model in discriminating between the two groups (i.e., exporters and non - exporters).

The findings of the two methods (i.e., split - half and Jacknife) provide concrete evidence on the ability of internal combined with external, to profile and predict export behaviour. Also, the combination of both measures of export behaviour, in one discriminant function, produces a valid classification (see Tables 9.6 and 9.7).

9.2.4 The Relative Important of the Predictor Factors in Terms of Their Contributions

One of the most interesting results of DFA is the relative importance of the predictor factors in terms of their contributions to discriminate between the two groups under investigation. Table (9.8) lists the most important external and internal predictors of export behaviour. It also shows the standardized discriminant coefficients and group means along these predictor factors. <u>Interpreting these results is straight forward</u>, the higher the discriminant coefficient is, the more important the factor is as a discriminator between the two groups (i.e., exporters and non – exporters). The sign of the coefficients associated with the predictor factor indicates the direction of their relationship to the dependent variable. Group means are generally used to identify how the groups differ in pairwise fashion in each of the predictor in the analysis (Perreault *et al.* 1979). The relative importance of the factors of each dimension is presented and discussed here respectively:

(1) The Internal Dimension Factors

Examination of the absolute value of the standardized discriminant coefficients listed in Table (9.8) reveals that the most important internal factors which distinguish the exporter's group from the non - exporter's group in their order of importance are: "managerial aspirations" (0.56), "export benefits" (0.49) "business experience", (0.40), "willingness to change" (0.39), "perceived risk" (0.22), "market competitive strengths" (0.21), "technological strengths" (0.16), "size of the firm" (0.14), "product characteristics" (0.12), "export objectives" (0.12), "demographic" (0.12), "commitment to export related activities" (0.11), "education" (0.08), and "operational obstacles" (0.07).

Looking at Table (9.8), we observe that the first factor (i.e., the "managerial aspirations") is regarded as the most important one related to export behaviour. The group means' results indicate that the exporter's group attached more importance to this factor than the non - exporter's group. This result is inconsistent with export behaviour studies, such as, Bilkey and Tesar (1977) & Cavusgil and Naor (1987).

Table (9.8)

Independent factor	Group Means		Standardized	
	G1	G2	Discriminant Coefficient	
(1) Internal Dimension Factors	a (waxada ti	en tel exe	relation beautifue	
01. Managerial aspirations	0.42	- 0.98	0.56	
02. Export benefits	- 0.07	0.14	- 0.49	
03. Business experience	0.28	- 0.69	0.40	
04. Willingness to change	0.39	- 0.94	0.39	
05. Perceived risk	- 0.11	0.30	- 0.22	
06. Competitive strengths	0.27	- 0.64	0.21	
07. Technological strengths	0.09	- 0.22	0.16	
08. Size of the firm	0.06	- 0.15	- 0.14	
09. Product characteristics	0.06	- 0.11	- 0.12	
10. Exporting objectives	0.28	- 0.67	0.12	
11. Demographic	0.04	- 0.12	- 0.12	
12. Commitment to export	0.17	- 0.43	0.11	
related activities	Supplier out	distantal.	Entering and a	
13. Education	0.28	- 0.71	0.08	
14. Operational obstacles	- 0.19	0.47	- 0.08	
(2) External Dimension Factors	axport a	indering p	restring address i	
15. Awareness of the government export	0.26	- 0.61	0.30	
policies	6 636531 258	Kalilanba 4	3942 C. Steller 1999	
16. Gov. export policies	0.26	- 0.60	0.26	
17. Raw material problems	0.07	- 0.15	0.22	
18. Foreign market conditions	- 0.27	0.64	- 0.17	
19. Gov. export procedures	- 0.08	0.19	- 0.17	
20. Physical distribution	- 0.02	0.05	0.10	
21. Domestic market conditions	- 0.01	0.04	- 0.07	

Group Means and Standardized Discriminant Coefficient

G1 = Exporter's group; G2 = Non - Exporter's group

* These factors are presented in descending order of their standardized discriminant coefficient (separately for each dimension).

* The group means here show the difference between the means of each group from the means of the factor score.

The "export benefits" factor is the second most important one that is associated with export behaviour. The result indicates that the non - exporter's group perceives exporting benefits are higher than the exporter's groups. This might be due to the fact that the exporter's group could be more realistic than its counterpart, the non - exporter's group in terms of the perception of exporting benefits. The non - exporter's group might overestimate the exporting benefits, because of the lack of experience in exporting.

The "business experience" factor is ranked the third among the most discriminating internal factors. The result indicates that the exporter's group has a much wider business experience than the non exporter's group.

The fourth factor of the most important internal factors is the "willingness to change". The result indicates that the exporter's group is more willing to change its export marketing policies than its counterpart, the non - exporter's group. This result appears to agree with previous works in this field, such as Kolhede (1984), Weinrauch and Rao (1974), and Samiee & Walters (1990a)

A further most important factor of the internal dimension related to export behaviour is the "perceived risk" factor. The result indicates that the non - exporter's group perceives a higher risk than the exporter's group. This result is supported by many export studies, such as Czinkota and Johnston (1981) and Tesar (1977).

The "market competitive strengths" factor is the sixth most important one compared with other internal factors. The result indicates that the exporter's group perceived this factor (i.e., market competitive strengths) much higher than the non - exporter's group. This result seems to be in agreement with previous works (e.g., Kirpalani and Macintosh 1980, and Bourantas & Halikias 1991).

The "technological strengths" factor stands out to be as the seventh most important factor that discriminated between the two groups. In comparison between the means result of each group, it may be concluded that the exporter's group perceives this factor much more than the non - exporter's group. This result is supported by many other studies, such as, Hirsch (1971), McGuinness and Little (1981), and Ogram (1982).

The eighth most important factor is the "size of the firm". This factor contributes significantly in discriminating between the two groups. The result indicates that the exporter's group tends to be much larger than the non - exporter's group.

In comparison to the other factors of the internal dimension, the "product characteristics" factor is ranked as the ninth most important factor distinguishing the two groups. The results reveal that the exporter's group perceives this factor much higher than the non exporter's group. This result appears to be in agreement with previous work, for example, Mayer and Flynn (1973), Kolhede (1984), Cavusgil and Naor (1987).

Another most important factor is the "export objectives" factor. The result indicates the exporter's group attaches much more importance to this factor than the non - exporter's group.

The "demographic" factor comes to be ranked as the eleventh most important related to export behaviour. The result indicates that the exporter's group tends to be older than the non - exporter's group in respect of their managers' ages. This appears to be in conflict with previous studies. Previous studies reported that exporters are more likely to be younger than non - exporters. The possible explanation for this conflict might be due to the fact that this measure is tested here at aggregate level (i.e., factor).

This factor "commitment to export related activities" stands out to be ranked as the twelfth most important among the internal factors. The result indicates that the exporter's group tends to be much more committed to export activities than the non - exporter's group. This result is in line with many previous studies, such as Cavusgil (1976), Cavusgil and Nevin (1981), and Kolhede (1984).

The "education" factor comes to be ranked as the thirteenth most important internal factor that associated with export behaviour. The result reveals that the exporter's group in respect to the "education" factor tends to be much higher than the non - exporter's group.

The last most important internal factor comes to be the "operational obstacles" factor. The result indicates that the exporter's group perceives this factor to be much lower than the non exporter's group. This result is in line with other export behaviour studies, (e.g., Bilkey and Tesar (1977), Alexandrides (1971), and Schlegelmilch (1983, 1986), etc.).

(2) The External Dimension Factors

This category of discriminating factors between exporters and non - exporters should represent the primary concern to the government in order to stimulate and encourage non - exporters to become involved in export operations.

As shown in Table (9.8), there are 6 most important factors related to the external dimension. In their order of importance, they are; "awareness of the government's export policies and assistance" (0.30), "government's export policies and assistance" (0.26), "raw material related problems" (0.22), "foreign markets conditions" (0.17), "government export procedures" (0.17), "physical
distribution related problems" (0.10), and the "domestic market conditions" (0.07).

Looking at Table (9.8), we observe that the first factor of the external dimension (i.e., the "awareness of the government's export policies and assistance") is regarded as the most important by the two groups (i.e., exporters and non - exporters). The result indicates that the exporter's group has higher knowledge, or is much more aware of government's export policies and assistance than the non exporter's group. The possible explanation of the importance of this factor might be attributed to two factors; the first one being the wider experience of the exporter's group in comparison with its counterpart the non - exporter's group, and the second factor is the little effort that the government may put in to make these policies known among the non - exporter's group (see Chapter 3). Therefore, it may be concluded that awareness of the government's export policies and assistance are important for those firms which want to become involved in exporting.

The "government's export policies and assistance" factor is the second factor of importance. The result indicates that the exporter's group attached a higher importance to this factor than the non exporter's group. The possible explanation of that might be due to either of two factors; the lack of knowledge or awareness of these export policies among the non - exporter's group, or perhaps the negative attitudes of the non - exporter's group towards the current government's export policies and assistance.

The "raw material related problems" factor is ranked third among the most discriminating external factors. The result indicates that the exporter's group attaches slightly higher importance to this factor than the non - exporter's group. This might be attributed to

the fact that the exporter's group is more realistic, because the exporter's group has a wider experience than the non - exporter's group.

A further important factor related to the external dimension is the "foreign markets conditions" factor. The result indicates that the non - exporter's group perception the problems of the foreign market conditions are much worse than the export's group. The possible explanation of that might be due to the lack of experience of the non - exporter's group in foreign markets.

The "government's export procedures" factor comes to be ranked the fifth most important factor discriminating between the two groups. The result indicates that the exporter's group attaches little important to this factor than the non - exporter's group. Problems associated with government export procedures are considered the main obstacles inhibiting firms to become involved in exporting (see Chapter 3). Therefore, the government has to simplify these procedures as much as possible, if it wants to stimulate and encourage these firms to become involved in exporting.

The "physical distribution related problems" factor is ranked the sixth most important among the external factors in discriminating between the two groups. The result indicates that the non - exporter's group perceives this factor much worse than the exporter's group. This might indicate that the lack of physical distribution facilities in the domestic market might discourage or inhibit firms from becoming involved in exporting.

The last important external factor related to export behaviour is the "domestic market conditions". The result indicates that the non exporter's group perceives this factor to be much worse than the exporter's group.

9.3 The Differences Between Exporters and Non - Exporters: Variables Findings

The t - test is used to check the following null hypothesis: HO(11): There is no significant difference between the two groups

(i.e, exporters and non - exporters) on the basis of

variables which make up each factor, taken separately. This major hypothesis can be further divided into 26 hypotheses according to the number of the factors included in the analysis. A summary of the results of these hypotheses is provided in Table (9.9).

The results indicate that, out of these 26 null hypotheses, only 18 null hypotheses were rejected, one null hypothesis was accepted and 7 null hypotheses were mixed (i.e., some variables were found to be significant).

9.4 The Interpretation of the Variable Findings

The possible interpretation of the results of the variables is presented here in terms of each factor to which they belong, as follows:

(1) The "Managerial Aspirations" Factor: Variables T - Test

This factor consists of 6 independent variables (Table 9.10). Each of them was regarded as important in differentiating between the two groups. The result indicates that the exporter's group attaches a higher importance to all of these attributes than the non - exporter's group. In comparison to other attributes of this factor, the increasing of the firm's overall profit attribute is the most important to differentiate between the two groups.

This result is supported by many studies in previous works, such as Schlegelmilch (1986), Bilkey and Tesar (1977), and Cavusgil and Naor (1987).

Table (9.9)

A Summary of the Results of the T - Test of the Differences Between Exporters and Non - Exporters in Terms of Variables Which Constitute Each Independent Factor, Taken Separately

Hypotheses	Factors	No. of Variables	Signif.
F(HO)01	Managerial aspirations	6	All signif.
F(HO)02	Operational obstacles	8	All signif.
F(HO)03	perceived risk	5	All signif.
F(HO)04	Export objectives	5	All signif.
F(HO)05	Export benefits	4	All insign.
F(HO)06	Export barriers	3	2 signif.
F(HO)07	Competitive situation	2	1 signif.
F(HO)08	Competitive strengths	5	4 signif.
F(HO)09	Size of the firm	3	All signif.
F(HO)10	Technological strengths	3	All signif.
F(HO)11	Business experience	2	All signif.
F(HO)12	Product characteristics	3	All signif.
F(HO)13	Willingness to change	4	All signif.
F(HO)14	Commitment to export related activities	7	All signif.
F(HO)15	Demographic	2	1 signif.
F(HO)16	Education	2	All signif.
F(HO)17	The government's export policies and assistances	9	All signif.
F(HO)18	Awareness of the government export	6	All signif.
F(HO)19	The government's export facilities problems	6	All signif.
F(HO)20	The government export procedures.	3	2 signif.
F(HO)21	The foreign market	5	All signif.
F(HO)22	Commercial facilities	4	All signif.
F(HO)23	Raw material problems	4	2 signif.
F(HO)24	Export Costs	4	All signif.
F(HO)25	Physical distribution	3	All insgnif.
F(HO)26	The domestic conditions	2	All signif.

Table (9.10)

No	Variables	Groups	Means	T -	DF	Sig.	
	a a a a a a a a a a a a a a a a a a a	G1	G2	varue			
V31	Improving the market position	4.94	3.49	7.64	227	.0000	
V32	Increasing the overall profit	5.99	3.56	12.88	151	.0000	
V33	Securing steady growth	4.78	4.37	2.10	192	.0370	
V34	Developing market strategies	5.34	3.52	9.30	174	.0000	
V35	Reducing cost of production	5.84	3.53	12.41	162	.0000	
V36	Increasing the sales volume	5.94	3.63	12.59	164	.0000	
		1		1	1		

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Managerial Aspiration" Factor

Therefore, it may be concluded that these attributes (taken together or separately) are important for those firms which intend to get involved in exporting, in particular, the increasing of the firm's overall profit attribute.

(2) The "Operational Obstacles" Factor: Variables T - Test

Each of the variables comprising the "operational obstacles" factor is found to be important in discriminating between the two groups (Table 9.11). The result indicates that the non - exporter's group attaches relatively higher importance to each of these attributes than the exporter's group. The most common operational obstacle which is considered relatively important by the two groups is the lack of personal expertise. This result appears to agree with other export behaviour studies, such as Schlegelmilch (1983), Alexandrides (1971), Pavord and Bogart (1975), Kolhede (1984) and Keng & Jiuan (1989).

Therefore, it may be concluded that these attributes either hinder or discourage those firms which want to become involved in exporting, in particular, the lack of knowledge in communicating with overseas customers.

No	Variables	Groups means		T -	DF	Sig.
		G1	G2	, and		
V67	Difficulty of obtaining sales representative.	4.42	5.26	- 4.51	383	.0000
V68	Difficulty of pricing our products abroad.	4.23	5.02	- 4.24	281	.0000
V65	Lack of knowledge in comm.	4.38	5.64	- 7.73	345	.0000
v70	Difficulty of locating potential demand	4.42	5.33	- 5.56	300	.0000
V66	Lack of personal expertise	4.69	5.82	- 6.98	332	.0000
V75	Size of production	4.30	5.59	- 5.78	312	.0000
V69	Inability to establish distribution system	4.03	5.07	- 5.77	282	.0000
v71	Difficulty of maintaining control over middleman	4.51	5.19	- 3.74	273	.0000

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Operational Obstacles" Factor

(3) The "Perceived Risk" Factor: Variables T - Test

The "perceived risk" factor consists of five variables, (Table 9. 12). The result indicates that the non - exporter's group attaches higher importance to each of these attributes than the exporter's group. In comparison to other variables of this factor, the political risk attribute is considered as the most important by the two groups.

The attributes of this factor (taken together) are also found to be related to export behaviour. This result is in line with many previous works, such as Bilkey and Tesar (1977), Kolhede (1984), Simpson and Kujawa (1974), and Schlegelmilch (1983, 1986).

Therefore, it may be concluded that the attributes of this factor either hinder or discourage those firms which want to become involved in exporting, in particular, the perception of the political risk attribute.

No	Variables	Groups means		T -	DF	Sig.
depote	13.8.	G1	G2	value		
V77 V78	High risk of collecting High foreign exchange risk	4.80	5.77	- 5.83	303 270	.0000
V88 V93 V98	High risk of transportation High risk of production High political risk	4.52 4.46 4.12	5.43 5.23 6.10	- 4.62 - 3.82 - 12.79	259 257 325	.0000

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Perceived Risk" Factor

(4) The "Export Objectives" Factor: Variables T - Test

The "export objective" factor is composed of five variables (Table 9.13). The results indicate that the exporter's group attaches relatively higher importance to each of these variables than the non - exporter's group. When compared to other variables of this factor, increasing the firm's growth attribute is regarded relatively as the most important by the two groups.

Table (9.13)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Export Objectives" Factor

No	Variables	Groups	means	T - Value	DF	Sig.
		G1	G2			
V27	Making greater profit	5.59	4.62	6.88	205	.0000
V26	Using the excess capacity	5.58	4.61	6.58	219	.0000
V30	Increasing the annual sales	5.50	4.92	5.08	191	.0000
V28	Increasing the security of the firm's investment	5.33	5.06	1.80	206	.0000
V21	Increasing the growth of the firm	5.98	4.45	9.93	229	.0000

This result is supported by many export behaviour studies (e.g., Cavusgil and Nevin (1981) and Bilkey (1978)). Therefore, it may be concluded that these attributes (taken together or separately) are important for potential exporters which want to become involved in exporting.

(5) The "Export Benefits" Factor: Variables T - Test

The results indicates that each of the four attributes comprising this factor (i.e. "Export Benefits") is not important to differentiate between the two groups (Table 9.14). However, the result with regard to these attributes (taken together) is found to be important in differentiating between the two groups.

The possible explanation of this might be attributed to two factors: (1) the use of the aggregate measure may produce more power (a factor) than depending on each attribute separately, and (2) it might be due to the different types of the industrial firms under investigation. Some of these types might consider these attributes taken together as being critical in order to become involved in exporting. This result still needs further investigation.

Table (9.14)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Export Benefits" Factor

No	Variables	Groups	means	T -	DF	Sig.
	d the "Competitive Lines South	G1	G2	varue		
V16	Overcoming instability demand	5.27	5.08	1.26	310	.210
v29	Better opportunities in the foreign markets	5.72	5.64	.75	237	.457
V20	Benefiting from the gove. export assistances	5.13	4.98	.94	331	.347
V22	Preventing dependency in the domestic market	4.86	4.45	1.57	317	.111

(6) The "Export Barriers" Factor: Variables T - Test

The "export barriers" factor consists of three variables (Table 9.15). The results indicate that only difficulty of collecting money on foreign sales and inability to provide follow up services are considered relatively as important by the two groups. The lack of significance for the inability to provide follow up service might indicate that it may be more important to determine the level of exporting.

However, the attributes of this factor (taken together) are regarded as not important by the two groups. This might be due to the different types of the industrial firms under investigation, some of which might only consider some attributes of this factor as being important barriers to their involvement in exporting.

Table (9.15)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Export Barriers" Factor

No	Variables	Groups means		T -	DF	Sig.
		G1	G2	varue	Laure Prov	8 - 5 m - 4 h
V73	Inability to provide	3.65	3.85	- 0.94	196	.3500
V7 4	Inability to provide	4.03	4.67	- 1.17	252	.0002
V72	Diff. of collecting money	4.51	5.48	- 4.99	289	.0000

(7) The "Competitive Situation" Factor: Variables T - Test

The "competitive situation" factor consists of two variables (Table 9.16); keeping pace with local competitors, and overcoming competition in the domestic market. The result indicates that each of these is not shown to be important in discriminating between the two groups.

These attributes of this factor taken together are also found to be not important in discriminating between the exporter's group and the non - exporter's group.

Therefore, it may be concluded that the attributes of the "competitive situation" factor whether taken together or separately are not important for firms who want to become involved in exporting.

Table (9.16)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Competitive Situation" Factor

No	Variables	Groups	means	T - Value	DF	Sig.
		G1	G2			
V23	Overcoming competition in the domestic market	4.68	4.36	1.93	282	.054
V19	Keeping peace with local competitors	4.90	4.88	.09	371	.925

(8) The "Market Competitive Strengths" Factor: Variables T - Test

The "market competitive strengths" factor is composed of five attributes, (Table 9.17). Among these five variables, only the proximity to the foreign market is not regarded as important by the two groups. The results indicate that the exporter's group perceives all of these attributes higher than the non - exporter's group. When compared to other attributes of this factor, the strength of management is considered as the most important in differentiating between the two groups. This result appears to agree with previous works, such as Bilkey and Tesar (1977) and Simpson (1973).

Therefore, it may be concluded that the attributes of the "market competitive strengths" factor are important for firms which intend to become involved in exporting, in particular, the strength of management skills.

No	Variables	Groups	means	T -	DF	Sig.
inep/1	tions to chose is fersioned?	G1	G2	Value		
V54	Ability to hire trained persons	4.53	3.16	7.92	232	.0000
V53	Ability to offer discount price	3.90	2.92	5.36	255	.0000
V24	Strength of management	5.42	3.45	10.47	178	.0000
V18	Proximity to the foreign market	5.43	5.35	.59	313	.5560
V49	Technological standard	4.22	3.30	4.47	292	.0000

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Market Competitive" Factor

(9) The "Size of the Firm" Factor: Variables T - Test

The "size of the firm" factor consists of three variables (Table 9.18). The results indicates that the exporter's group tends to be much larger than the non - exporter's group in terms of each of these variables. In comparison to other attributes of this factor, the total capital assist attribute is shown to be the largest in discriminating between the two groups. This result is supported by many export studies, such as Gronhang & Lorenzen (1982), Withey (1980), Keng & Jiuan (1989) and Culpan (1989). However, it is in conflict with other studies, such as Axinn (1985, 1988), Kirpalani and Macintosh (1980),

Table (9.18)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Size of the Firm" Factor

No	Variables	Groups	means	T -	DF	Sig.
		G1	G2	varue		
V104 V106 V108	Number of employees Total of the capital asset Total sales volume	67.6 492.2 344.94	44.3 169.5 95.4	3.85 5.89 5.37	280 381 382	.0000 .0000 .0000

Bourantas & Halikias (1991), Liouville (1992) and Samiee & Walters (1990a & 1990b). The possible explanation for this conflict might be due to the different size of the firms in the countries under investigation. In fact, the size of Jordanian firms are smaller in comparison to those in developed countries.

(10) The "Technological Strengths" Factor: Variables T - Test

The "technological strengths" factor is composed of three variables (Table 9.19). The results indicate that the exporter's group perceives each of these variables much higher than the non - exporter's groups. In comparison to other variables of this factor, the ability to modify export marketing policies attributes is regarded relatively the most important by the two groups.

Therefore, it may be concluded that these attributes should be considered important for firms which want to become involved in exporting.

Table (9.19)

No	Variables	Groups	means	T -	DF	Sig.
		G1	G2	value		
v50 v51 v55	The superior of product The superior of production High ability to modify product policies	5.61 5.12 4.76	4.95 4.47 3.82	3.81 4.13 5.26	189 220 231	.0000 .0000 .0000

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Technological Strengths" Factor

(11) The "Business Experience" Factor: Variables T - Test

The "business experience" factor consists of two variables (Table 9.20); the number of years in exporting and the number of years in business. The results indicate that the exporter's group has a wider

experience than the non - exporter's group. When compared to other type of experience, the number of years in exporting tends to be more important in differentiating between the two groups.

While this result is in agreement with other studies in respect of the number of years in exporting (e.g., Daniels and Goyburo 1976, Ursic and Czinkota 1983, and Bilkey 1978), it is found to be in conflict with other studies in respect of the firm's age, i.e., the number of years in business, (e.g., Cavusgil and Naor 1987, Keng & Jiuan 1989, Tseng & Yu 1991, and Samiee & Walters 1990a). The possible explanation for this conflict might be due to the difference of the originations of the firms under investigation.

Table (9.20)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Business Experience" Factor

No	Variables	Groups means	T -	DF	Sig.
neco.		G1 G2	varue		
v103 v107	Number of years in business Number of years in exporting	10.4 8.1 8.4 1.00	3.91 24.90	243 271	.000

(12) The "Product Characteristics" Factor: Variables T - Test

Each of the three variables which constitute the "product characteristics" factor is found to be an important by the two groups (Table 9.21). The results indicate that the exporter's group attaches relatively a higher importance to each of these attributes than the non - exporter's group. In comparison to other attributes of this factor, the uniqueness of product attributes is regarded relatively as the most important by the two groups.

This result appears to agree with previous works (e.g., Cavusgil and Naor (1987), and Mayer and Flynn (1973)). Therefore, it may be concluded that the attributes of the "product characteristics" factor are important for firms which intend to become involved in exporting.

Table (9.21)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Product Characteristics" Factor

No	Variables	Groups means		T -	DF	Sig.
		G1	G2	varue		
V25 V52 V17	Product competitive price Product quality Uniqueness of product	5.42 4.69 4.35	5.00 4.36 3.27	3.10 1.93 5.44	240 282 304	.0020 .0570 .0000

(13) The "Willingness to Change" Factor: Variables T - Test

The "willingness to change" factor consists of four variables (Table 9.22). The results indicate that the exporter's group attaches relatively a higher importance to each of these attributes than the non - exporter's group. In comparison to other attributes of this factor the willingness to change pricing attribute is considered relatively the most important by the two groups.

Table (9. 22)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Willingness to Change" Factor

No	Variables	Groups	means	T -	DF	Sig.
-		G1	G2	Value		
V45	Willingness to modify organisational system	4.26	3.65	4.33	354	.0000
V47	Willingness to change pricing policies	4.79	3.39	9.42	190	.0000
V46	Willingness to adapt product policies	5.35	4.78	3.62	230	.0000
V48	Willingness to modify distribution system	4.97	3.87	6.96	205	.0000

This result is supported by many export studies such as Kolhede (1984), Weinrauch and Rao (1974) and Layton & Dunphy (1970).

Therefore, it may be concluded that these attributes (taken together or separately) should be considered important for firms which want to become involved in exporting, in particular, the willingness to change pricing policies to meet foreign markets conditions.

(14) The "Commitment to Export Activities" Factor: Variables T - test

The "commitment to export activities" factor is composed of eight variables (Table 9.23). The result indicates that the exporter's group tends to be much more committed to export activities than the non exporter's group. When compared to other attributes of this factor, having a formal policy for handling export activities attribute is regarded as the most important in discriminating between the two groups.

This result is found to agree with previous works, such as, Malekzadeh & Nahavandi (1985) and Cavusgil *et al.* (1979).

Table (9.23)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Commitment to Export Related Activities" Factor

No	Variables	Groups means		T -	DF	Sig.
	and the second of the second s	G1	G2	value		
741	We promote our product to	4.54	3.19	9.11	296	.0000
/40	We participate in the international trade fairs	4.52	3.48	6.97	358	.0000
142	We plan to get into foreign	5.11	3.12	13.57	281	.0000
143	We send sales rep.	4.45	2.85	10.08	299	.0000
11.1.	We have a formal policy for	4.90	2.08	21.36	251	.0000
v44 v38	We consistently seek export opportunities	5.31	3.03	15.94	202	.0000
139	We carry out a systematic	4.45	2.26	15.08	222	.0000
137	We use export research	3.97	3.59	2.36	244	.0190

Therefore, it may be concluded that the attributes of this factor are important for those firms which wish to get involved in exporting, in particular, having a formal policy for handling export activities.

(15) The "Demographic" Factor: Variables T - Test

The "demographic" factor consists of two variables; age and experience (Table 9.24). The results indicate that each of these is not important to differentiate between the two groups. However, this result is found to be in conflict with previous work in export behaviour. For example, Layton & Dunphy (1970), and Pinney (1968) reported that exporters were more likely to be younger than non exporters. The possible explanation of this might be attributed to the different demographic characteristics of the managers of the firms in the countries under investigation. Jordanian managers at the higher level of management tend to be in the same age group.

Table (9.24)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Demographic" Factor

No	Variables	Groups	means	T - Value	DF	Sig.
	and the second sec	G1	G2			
V115 V114	Experience Age	9.8 41	8.6 42	1.68	225 187	.0750 .7250

(16) The "Education" Factor

The "education" factor is composed of two types of education (Table 9.25); the level of education and proficiency in a foreign language. The result indicates that the exporter's group is much more highly educated than the non exporter's in terms of education and linguistic ability. With regard to the education determinant, the proficiency in foreign language is found to be more important in differentiating between the two groups.

While this result is found to be in agreement with other export studies, such as Reid (1981), Simpson (1973), Langston & Teas (1976), Keng & Jiuan (1989) and Holzmuller & Kasper (1990), it was found to be in conflict with others such as Cavusgil & Naor (1987), Axinn (1988) and Bourantas & Halikias (1991). This might be due to the differences between the key respondent of the firms under investigation.

Therefore, it may be concluded that the attributes of the "education" factor are important for firms which want to become involved in exporting, in particular, the proficiency in a foreign language attribute.

Table (9.25)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Education" Factor

No	Variables	Groups means		T -	DF	Sig.
		G1	G2	varue		
V117 V120	Level of education Proficiency in foreign language	12.6 54.1	9.7 27.9	5.83 9.80	178 174	.0000

(17) The "Government's Export Policies and Assistant" Factor: Variables T - Test

The "government's export policies and assistance" factor consists of nine variables (Table 9.26). The result indicates that the exporter's group attaches higher importance to each of these attributes than the non - exporter's group. The possible explanation of this result might be due to the lack of knowledge or awareness of these export policies and assistance among the non - exporter's group, or it could be related to the negative attitudes of the non - exporter's group towards the government's export procedures. When compared to other attributes of this factor, the devaluation of JD and the income tax rebate policy attribute are considered the most important by the two groups.

Table (9.26)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Comprise the "Government Export Assistance and Policies" Factor

No	Variables	Groups	means	T -	DF	Sig.	
	ion nooje.ed to other date	G1	G2	Varue	d (; Alise	Legh	
V1	The government's trade agreement policies	4.95	4.03	4.61	240	.0000	
V2	Export credit discount	4.52	3.78	2.73	238	.0007	
V3	Duty drawback facilities	4.68	3.39	6.40	243	.0000	
V4	The Chamber of Industry export services	5.04	3.79	7.19	244	.0000	
V 5	The Trade Centre export services	5.22	3.94	8.06	281	.0000	
V6	The Free Trade Zones facilities	4.48	3.71	4.43	265	.0000	
V7	Devaluation of JD	5.55	3.73	10.80	216	.0000	
V8	Temporary entry facilities	5.22	3.54	8.96	244	.0000	
v9	The income tax rebate	5.62	3.95	9.03	232	.0000	

This result emphasises the role of the government's export policies and assistance by enabling firms to become involved in exporting. However, it appears to be in conflict with previous works. For example, Kolhede (1984), Rabino (1980) and Sullivan & Bauerschmidt (1990) reported that government's export incentives were not important either for exporters or non - exporters. This might be due to the different abilities of the firms in the countries under investigation. In fact, it was reported that firms in developing countries are in need of their governments' assistances, because of their lack of financial resources and know - how. Therefore, it may be concluded that the government's export policies and assistance are critical to Jordanian firms which want to become involved in exporting.

(18) The "Awareness of the Government Export Policies and Assistant" Factor: Variables T - Test

The "awareness of the government export policies and assistance" factor is composed of six variables (Table 9.27). The results indicate that the exporter's group is more aware of these attributes than the non - exporter's group. The explanation of this differences might be due to the wider experience of the exporter's group, or it could be due to the little attention that the government might be paying to promote these policies among the non - exporter's group.

when compared to other attributes of this factor, the lack of awareness of the duty - drawback system is the most common attribute considered by the two groups.

Therefore, it may be concluded that the awareness of these policies and assistances are important issues for firms to become involved in exporting, in particular the awareness of the duty - drawback system.

Table (9.27)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Awareness of the Government Export Assistance and Policies" Factor

No	Variables	Groups	means	T -	DF	Sig.
	The bank quaternice req.	G1	G2	Varue	678.	-100%
V10	Knowledge of Free Trade Zones facilities	4.33	3.37	5.61	261	.0000
V11	Awareness of the duty - drawback facilities	4.44	2.73	9.40	263	.0000
V12	Awareness of the export credit programme	4.40	2.66	9.06	253	.0000
V13	Awareness of the manag. export services	4.90	3.29	8.52	227	.0000
V14	knowledge of temporary entry facilities	5.05	3.65	7.18	210	.0000
V15	Awareness of the export services provided by the chamber of Industry	4.71	3.75	5.01	227	.0000

(19) The Government Export Facilities Related Problems" Factor: Variables T - Test

The "government export facilities related problems" factor consists of 6 variables (Table 9.28). The result shows that the non exporter's groups perceives each of these attributes in a much worse way than the exporter's group. When compared to other attributes of this factor, the bank guarantee requirement is regarded relatively as the most important problem by the two groups.

Therefore, it may be concluded that these attributes (taken together or separately) might hinder or discourage firms to become involved in exporting, in particular, the bank guarantee requirement.

Table (9.28)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Comprise the "Government Export Facilities" Factor

No	Variables	Groups means		T -	DF	Sig.
		G1	G2	Varue		1.8300
V59	Inadequate tax incentive	4.60	5.03	- 2.25	267	.0250
V60	Cost of export documents	4.32	5.12	- 4.28	263	.0000
v61	Complexity of the duty - drawback system	4.58	5.19	- 3.62	290	.0000
V62	Lack of commercial	4.57	5.00	- 2.42	281	.0160
V63	High duties imposed on the spare parts	4.88	5.31	- 2.44	276	.0150
V86	The bank guarantee req.	5.68	6.24	- 3.99	327	.0000

(20) The "Government Export Procedures Related Problems" Factor: Variables T - Test

The "government export procedures related problems" factor consists of three attributes (Table 9.29); the customs export clearances, the export licence procedures, and the export inspection process. The results indicate that only the export licence procedures and the export inspection process are important in differentiating between the two groups. When compared to the export licence procedures, the export inspection process is considered as a more important problem by the two groups.

However, these attributes acting as a factor are found to be important in discriminating between the two groups. The possible explanation of this might be attributed either to one of two reasons: (1) the use of the aggregate measure (i.e., factor) might produce much power than being based on each attribute acting separately, or (2) perhaps some export procedures are perceived as much worse than others.

Table (9.29)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Comprise the "Government Export Procedures" Factor

No	Variables	Groups means		T -	DF	Sig.
		G1	G2	Value		and an and a second second
V56	The custom export clearance	4.56	4.60	- 0.22	255	.8300
V57	The export inspection proc.	4.11	4.65	- 5.25	276	.0000

(21) The "Foreign Market Conditions Related Problems" Factor: Variables T - Test

The "foreign market conditions" factor is composed of five variables (Table 9.30). The results indicate that the non - exporter's group perceives each of these variables much worse than the exporter's group.

When compared to other attributes of this factor, the difficulty of business practise attribute is considered as the most important problem by the two groups. Therefore, it may be concluded that these attributes might hinder or discourage firms to become involved in exporting, in particular, the difficulty of business practice in foreign markets.

When compared to other attributes of this factor, the difficulty of business practise attribute is considered as the most important problem by the two groups.

Therefore, it may be concluded that these attributes might hinder or discourage firms to become involved in exporting, in particular, the difficulty of business practice in foreign markets.

Table (9.30)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Foreign Market Conditions" Factor

No	Variables	Groups means		T -	DF	Sig.
		G1	G2	varue		
V94	Diff. of business practise	3.93	5.83	- 13.5	383	.0000
V95	The intensity of competition	4.41	6.18	- 9.2	273	.0000
V96	Foreign government restric.	3.98	5.44	- 8.5	270	.0000
V97	Geographical market distance	4.26	5.16	- 5.1	260	.0000
V98	Political instability	4.11	6.09	- 12.8	325	.0000

(22) The "Commercial Facilities Related Problems" Factor: Variables T - Test

This factor consists of 4 variables (Table 9.31). The results indicate that the two groups regard each of these variables as being important. When compared to the other variables of the "commercial facilities related problems" factor, the lack of reliable information about the foreign markets and the lack of export credit guarantee against non - payment risk are considered as the most important problems by the two groups. Therefore, it may be concluded that these attributes (taken together or separately) will hinder or discourage firms to become involved in exporting, in particular, the lack of export credit guarantee against non - payment risk.

Table (9.31)

T - Te	est Between	Expo	rters and No	on - Exporte	ers in	Terms	of Va	ariables
Which	Constitute	the	"Commercial	Facilities	Relate	d Prob	lems"	Factor

No	Variables	Groups means		T -	DF	Sig.
	and the second se	G1	G2	Varue		
V64	Lack of reliable infor.	5.52	6.06	- 3.37	266	.0001
V76	Lack of export credit	5.51	6.16	- 4.24	297	.0000
V79	Diff. of obtaining banking facilities	4.27	5.10	- 4.15	265	.0000
V86	Lack of trade companies	4.98	5.70	- 3.96	290	.0000

(23) The "Raw Material Related Problems" Factor: Variables T - Test The "raw material related problem" factor consists of four variables (Table 9.32). The results indicate that only shortage of raw materials in the domestic market and the high price of raw materials regarded as being relatively important by the two groups.

However, these attributes acting as a factor are found to be important in discriminating between the two groups. The possible explanation of this might be related either to one of two reasons: (1) the use of the aggregate measure (i.e., factor) may produce more power than depending on each attribute acting alone, or (2) perhaps the different types of the industrial firms under investigation. Some of these types might consider these attributes taken together as being important problems to their involvement in exporting.

No	Variables	Groups	means	T -	DF	Sig.
	and the second second second second	G1	G2	varue		.4310
V85	Diff. of obtaining raw	4.44	4.61	8	230	.4310
V84	Low quality of raw material	4.20	4.11	.4	235	.6590
V83	Shortage of raw material in the domestic market	4.98	4.45	2.6	242	.0100
V92	High price of raw material	4.70	5.47	- 3.7	269	.0000

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Raw Material Related Problems" Factor

(24) The "Export Costs" Factor: Variables T - Test

The "export costs" factor is composed of 4 variables (Table 9.33). The results indicate that each of these variables is relatively considered as important by the two groups. When compared to other variables of this factor, the high interest rate attribute was the most important in differentiating between the exporter's group and the non - exporter's group. Therefore, it may be concluded that these attributes are hinder or discourage firms to become involved in exporting, in particular, the high interest rate.

Table (9.33)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Export Costs" Factor

No	Variables	Groups	means	T -	DF	Sig.
New York	erts greap. When enablighed	G1	G2	value	d an	Militia
V89	High cost of communication	4.62	4.98	- 2.07	274	.0390
V90	High cost of labour skills	4.67	3.30	- 3.63	272	.0000
v91	High cost of obtaining warehouses facilities	4.20	5.83	- 4.47	283	.0000
v101	High interest rate	5.26	5.20	- 5.51	289	.0000

(25) The "Physical Distribution Related Problems" Factor: Variables T - Test

The "physical distribution related problem" consists of three variables (Table 9.34). The results indicate that each of these is not important to in differentiate between the two groups.

Table (9.34)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Physical Distribution Related Problems" Factor

No	Variables	Groups	means T - D Value	DF	Sig.	
	The sales is aspection The back docestic desaid	G1	G2	varue	8812 1812	0000
V81	Diff. of obtaining	4.25	4.60	- 1.77	255	.0780
V82	Diff. of obtaining	3.40	3.75	- 1.78	230	.0760
v87	High freight cost	5.04	5.05	- 0.03	294	.9790

Therefore, it can be concluded that these attributes acting as a factor or separately are not important in differentiating the exporter's group from the non - exporter's group. This might be due to fact the two groups encountered the same physical distributions problems in the domestic market.

(26) The "Domestic Market Conditions" Factor: Variables T - Test

The "domestic market conditions" factor consists of two attributes (Table 9.35). The results indicate that the non - exporter's group attaches a higher importance to each of these attribute than the exporter's group. When compared to the intensity of competition attribute, the high domestic demand is regarded as more important by the two groups. Therefore, it may be concluded that these attributes either discourage or hinder firms to become involved in exporting, in particular, the high domestic demand attribute.

Table (9.35)

T - Test Between Exporters and Non - Exporters in Terms of Variables Which Constitute the "Domestic Market Condition" Factor

No	Variables	Groups means T -	DF	Sig.		
		G1	G2	value		
v99	The intensity competition	3.89	5.76	- 12.08	351	.0000
v100	The high domestic demand	3.93	5.83	- 13.51	382	.0000

internal contransmul accurren bad a higher deper of association with export behavior, (1), the classification of grapp association in the twenthing forward discretion was found to be made sequenced in the twenthing forward the set grapps (1.6., separters and eas exponents) that the vertical discretion, This respire size folder a loss determined factors are more critical thes extract for field the become explicit is cateding. It supports the field of the secondary evolves is cateding. It supports the field of the secondary evolves is cateding. It supports the field of the secondary evolves is cateding. The supports the field of the

3. The action of the ordereal contrological analyses to the pregates environmental anomales improved the productive soder of the control fraction of prove sometricals (mention 9.1.2).

The continuation of the internal and calcings environmental prediction (i.e., % factors) in the discriminant flactions chemical distinguishing between the distinguishing between the two stress.

Summary and Conclusions

In this chapter, the testing of the research hypotheses (HO6) to HO(11) conducted and the main findings of the other objectives of the study (i.e., 5 and 6) were presented and discussed respectively. Drawing together the findings of the various analysis conducted in this chapter, a number of conclusions emerge:

1. It was possible to discriminate between the two groups (i.e., exporters and non - exporters) in terms of their internal environmental measures (i.e, 16 factors) and external environmental measures (i.e., 10 factors) taken separately and together (see section 9.1.1 and 9.1.2).

2. In comparison to the external environmental measures, the internal environmental measures had a higher degree of association with export behaviour (i.e., the classification of group membership). In other words, the internal dimension was found to be more important in distinguishing between the two groups (i.e., exporters and non exporters) than the external dimension. This result might indicate that internal factors are more critical than external factor for firms to become involved in exporting, It supports the findings of the previous studies, such as these reported by Cavusgil and Nevin (1981), Kolhede (1984) and Axinn (1985).

3. The addition of the external environmental measures to the internal environmental measures improved the predictive power of the classification of group membership (section 9.1.2).

4. The combination of the internal and external environmental predictors (i.e., 26 factors) in one discriminant function showed significant improvement in distinguishing between the two groups

Table (9.36)

Profile Analysis For The Two Groups Means in Terms of the Components of the Internal and External Environmental Dimensions

Independent factors	Profile Analysis			
	Exporters	Non - Exporters		
(1) Internal Factors	(1) "Report lossef	itan, (S) Storage		
 Managerial aspirations Export benefits Business experience Willingness to change Perceived risk Competitive strengths Technological strengths Size of the firm Product characteristics Export objectives Demographic Commitment to export Education Operational obstacles 	much higher slightly lower higher slightly lower higher slightly higher slightly higher slightly higher higher higher higher lower	<pre>much lower slightly higher much lower much lower higher much lower lower smaller slightly lower much lower much lower much lower much lower much lower much lower much lower much higher</pre>		
(2) External Factors	A CAL PROPERTY OF	and and the second second second		
 Awareness of the government's export Gov. export policies 	higher higher	much lower much lower		
 03. Raw material problems 04. Foreign Markets 05. Gov. export procedures 06. Physical distribution 07. Domestic market 	slightly higher lower slightly lower slightly lower slightly lower	slightly lower much higher higher slightly higher slightly higher		

* The comparison is made in terms of the difference between each group mean (i.e., exporter's group mean ; non - exporter's group mean) from the overall mean of each factor score. (i.e., exporters and non - exporters), and the prediction was better than on each environmental measure individually (section 9.2.2).

5. The relative importance of internal and external factors were outlined and discussed respectively after the validation of the discriminant function was assessed and presented. The most important internal factors according to their order of importance were found to be: (1) "Managerial aspirations", (2) "Export benefits", (3) "Business experience", (4) "Willingness to change", (5) "Perceived risk", (6) "Market competitive strengths", (7) "Technological strengths", (8) "Size of the firm", (9) "Product characteristics", (10) "Export objectives", (11) "Demographic", (12) "Commitment to export related activities", (13) "Education", and (14) "Operational obstacles".

The most important external factors in terms of their order of importance are: (1) "Awareness of the government's export policies and assistance", (2) "Government export policies and assistance", (3) "Raw material related problems", (4) "Foreign markets conditions", (5) "Government export procedures", (6) "Physical distribution related problems", and the (7) "Domestic market conditions" (section 9.2.4).

6. Two methods were used to check the prediction of the export behaviour (exporter's group membership), namely, The Split - Half and Jackknife methods. The results of the validation confirm our findings that the internal environmental measures are better than the external environmental measures on the basis of prediction group membership of export behaviour. Also, they confirm the result that the combination of the two dimensions was better than taking each dimension acting separately in terms of prediction and classification of the group membership (sections 9.2.2 and 9.2.3).

7. The integration approach of the two environmental dimensions (internal and external) allows us to delineate a feasible profile for

the two groups, and in turn, to answer the question, who are exporters and who are non - exporters?. A summary profile for the differences between the exporter's group and the non - exporter's group is given in Table (9.36).

8. The results of differences between the two groups (i.e., exporters and non - exporters) in terms of their means score on the variables (taken separately) consisting of each independent factor were presented and discussed respectively. The results indicated that the attributes (i.e., variables) of 18 factors were found to be important in differentiating between the two groups and some attributes of 7 factors were found to be related to export behaviour (i.e., the differentiation between exporters and non - exporters). However, only attributes of one factor were not found to be relevant to export behaviour (see Table 9.9).

(9) The results showed some disagreement with previous works in respect of the attributes of the following factors: (1) The "Size of the firm", (2) "Business experience", (3) "Demographic", (4) "Government's export policies and assistances" and "Education". The possible explanations of these conflicts were attributed to either of the two factors; (1) the use of the aggregate measure (i.e., factor), or (2) the export situations of the firms in the countries under investigation. For more details see section 9.4, Factors 10, 11, 15, and 17.

CHAPTER TEN

SUMMARY, CONCLUSIONS AND RECOMMENDATIONS

10.1 Summary

10.2 Main Conclusions of the Research Findings

10.3 Research Contributions

10.3.1 Theoretical Contribution 10.3.2 Practical Contribution

10.4 Recommendations

10.4.1 Recommendations For Current and Potential Exporters 10.4.2 Recommendations For Public Policy - Makers

10.5 Research Limitations

10.6 Areas for Further Research

SUMMARY, CONCLUSIONS AND RECOMMENDATIONS

10.1 Summary

This research has attempted to investigate the influence of the firm's internal and external environmental measures upon its export behaviour and the level of exporting. The specific objectives of this study have been explicitly presented in the first Chapter.

In order to achieve the study objectives, and to conduct the research in a systematic approach, a conceptual framework was developed. The conceptual framework tied together the major factors proposed to influence the firm's export behaviour and the level of exporting. The key factors were presented under two broad dimensions, i.e., internal and external.

The firm's internal environmental dimension was proposed under five major constructs: (1) Managerial aspirations, (2) Management's expectations, (3) Comparative advantages, (4) Commitment to export activity, and the (5) Socio - demographic characteristics of the decision - makers. The firm's external environmental dimension was suggested under two major constructs: (1) The economic and commercial infrastructural environment, and the (2) Political - legal environment (for more detail, see Chapter 4).

The data for this research were collected through structured directed interviews with 387 respondents. The target respondents were the manufacturing firms in Jordan, and the key respondent approach was employed.

These primary data have been analysed by a variety of multivariate statistical techniques including; Stepwise multiple regression statistical technique; discriminant function analysis, the Jacknife

and Split - Half methods for validating the DFA functions; correlation analysis (using the Pearson Correlation Coefficient); the Chi - square test, the Univariate F - ratio test; McNemar test, and the t - test.

The findings of this study have been presented and discussed in detail in Chapters 7, 8, and 9. Chapter (7) outlined the main findings of a factor analysis, i.e., the factors that underlie each construct of the internal and external dimensions. Chapter (8) was devoted to the determinants of the firm's level of exporting, and Chapter (9) dealt with the determinants of the firm's export behaviour (i.e., exporters vs. non - exporters).

10.2 Main Conclusions of the Research Findings

The main conclusions that can be drawn from the findings of this study are as follows:

a. Twenty six factors were extracted from the seven major constructs of the internal and external dimensions. Sixteen of them were derived from the five constructs of the internal dimension; managerial aspirations (one factor), management's expectations (6 factors), comparative advantages (5 factors), commitment to export activity (2 factors), and socio - demographic characteristics (2 factors). The other ten factors were extracted from the two constructs of the external dimension; political - legal environment (4 factors) and economic and commercial infrastructural environment (6 factors). These 26 factors were successfully identified and labelled, and they were subsequently used to answer the research questions by using regression and discriminant analysis.

b. The analysis provides empirical evidence that the integration approach of the firm's internal environmental and its external environmental measures give better explanation not only of the

prediction of export behaviour (i.e., classification of export group membership), but also of the prediction of the level of exporting (i.e., percentage of export sales volume). This result supports the theories of the firm's behaviour which stress that the organisational behaviour is inseparably linked to the environment in which it takes place (e.g., Cyert and March 1963). Therefore, a better understanding of export behaviour (or level of exporting) requires that the firm's environmental measures as a whole to be viewed (i.e., the interaction of internal and external environments) rather than isolated fragments (e.g.,, only a single environmental dimension).

c. The study's findings show that factors determining the firm's level of exporting are quite different from the ones that distinguish exporters from non - exporters. For example, while the "government export procedures related problems" factor is not shown to be importantly related to the level of exporting, it is regarded as one of the most important factors that distinguish the exporters' group from the non - exporters' group. This result is of particular importance to public policy decision - makers who want to encourage firms to become involved in exporting, because any attempt to remove or reduce such problems associated with export procedures will be more helpful for potential exporters. (For more details about other differences, see Appendix [3]).

d. The results indicate that factors existing outside the boundary of the firm (i.e., external environment) are more critical for its level of exporting than those related to its internal environment. However, in order to become involved in exporting, the internal factors are more important than those existing in its external environment. This result might represent a primary concern for public policy makers, in the way that they can develop different promotional

strategies and make the required changes within the Government departments, either to improve the level of exporting, or/ and to encourage firms to become involved in exporting.

e. The Determinants of the Level of Exporting

(1) The application of the Stepwise Multiple Regression analysis indicates that not only the firm's external environmental measures are important determinants of the level of exporting, but its internal environmental measures as well. This result is of a particular importance not only to manufacturers, but also to Government authorities, in the sense that one can identify the types of change required within their organisations in order to enhance the level of exporting.

(2) By the integration of the firm's internal environmental measures (i.e, 16 factors), only three factors are shown to be significantly related to the firm's level of exporting. In their order of importance, they are: 1. "Willingness to change", 2. "Commitment to export related activities", and 3. "Size of the firm".

(3) By the integration of the firm's external environment measures (i.e., 10 factors), only five factors are shown to be significantly related to the level of exporting. According to the order of importance, they are: 1. The "Foreign market conditions", 2. "Export costs", 3. The "Domestic market conditions", 4. "Government's export policies and assistances", and 5. "Awareness of the government's export policies and assistances.

(4) The research results indicate that the external environmental dimension produces better explanation of the variation of the level of exporting than the internal environmental dimension. In other words, factors existing outside the boundary of the organisation (external factors) are more critical to the firm's level of exporting than those

related to its internal environment. This result should be the main concern for Government departments to consider that any change in or action with regard to their current export policies and assistances will be helpful for firms which intend to improve their level of exporting.

(5) Integration of the internal and external dimensions (i.e., 26 factors) produces a better prediction and explanation of the variation of the level of exporting than each dimension acting alone. The result shows that 64% of the variance of the level of exporting could be explained by this combination, while each of internal and external dimension could only explain about 49.5% and 52% of the variance of the level of exporting respectively. This produces an empirical evidence that the level of exporting could be better explained through the combination of factors related to the firm's internal process and factors existing outside its boundary.

(6) Individual analysis of the attributes of the internal and external factors (i.e., 26 factors) show some difference from the results of taking attributes together (i.e., as a factor) in respect of the level of exporting. Four factors are shown to be important to the level of exporting, but some of their attributes are not. These factors are: 1. "Government export facilities related problems" 2. "Government's export policies and assistances", 3. "Demographic", 4. "Export benefits". The possible explanations for this disagreement might be attributed either to the use of the aggregate measure (i.e., factor) which may produce more power than each of their attributes acting alone, or perhaps the different types of industrial firms under investigation. Some of these types might consider these attributes taken together as being critical in order to improve their level of exporting.
(7) The analysis shows that out of 26 factors, all the attributes of 17 factors and some attributes of 6 factors are important to the level of exporting (see Table 8.33, Chapter 8). This potentially allows identification of necessary changes that organisations can make to improve the level of exporting. For example, compared to other attributes in the "size of the firm" factor, the capital of assets are shown to be the most importantly related to a high level of exporting. This implies that manufacturers who want to improve their level of exporting might need to maximise their capital assets as much as possible. For the other important attributes of each factor related to a high level of exporting, see Appendix [3]).

(8) While many of this study's findings support previous research, results of the following two attributes and two factors are found to be in conflict; (1) the decision - maker's age (2) the decisionmaker's foreign language skills, (3) "technological strengths", and (4) "willingness to change" factors.

For example, Simmonds and Smith (1968), Reid (1980) and Moon & Lee (1990) reported little or no relationship between the decision maker's foreign language skills and the level of exporting. The result of this study shows a positive relationship of such skills. However, it agrees with the findings of studies reported by Burton and Schlegelmilch (1987), Schlegelmilch (1986), Khan (1978) & Langston (1976). This differences in findings might be attributed to the difference between the countries under investigation.

This study found no relationship between the decision - maker's age and the level of exporting which is in conflict with previous studies, such as Dichtl *et al.* (1984, 1990) and McConnell (1979). This might be attributed to the difference between the ages of managers of the firms in the countries under investigation. Managers at a higher level of

management in Jordan are nearly about of the same age, i.e., almost old.

The study shows that the "technological strengths" factor is not important to the level of exporting. While this result is supported by Axinn (1985, 1988) and Reid (1986), it is in conflict with results of other studies, such as Cavusgil and Nevin (1981), Schlegelmilch (1983, 1986), Aaby & Slater (1989), and Walters and Samiee (1990). A possible explanation might be attributed either to general low technological level of the firms surveyed here, or perhaps to the use of an aggregate measure here (i.e., factor).

The study indicates that the "willingness to change" factor is importantly related the level of exporting. While this result is in agreement with previous studies such as those reported by Cooper & Kleinschmidt (1985), Samiee & Walters (1990a), and Oritz-Buonafina (1990), it is in conflict with the findings of other studies, such as Seifert & Ford (1989) and Christensen *et al.* (1987). This difference in findings might be attributed to the use of the aggregate measures (i.e., factor), or it might be due to the different types of the firms in the countries under investigation.

f. The Determinants of the Firm's Export Behaviour

(1) The results obtained from applications of the Split Half and Jacknife to the validity of the discriminant functions of the three set of dimensions (i.e., internal, external, and combined) provide strong evidence that each of these discriminant functions is a valid model in discriminating between the two groups (i.e., exporters and non - exporters). For more details, see section 9.2.3, Chapter 9.

(2) The application of Discriminant Function Analysis indicates that there is a strong relationship between the values of 16 factors of the internal dimension and the classification of the export group

membership. The analysis shows that the internal factors are able to correctly classify 95.84% of the population of the study into two groups (exporter's group and non - exporter's group). This result should represent a primary concern for manufacturers who want to become involved in exporting, since by comparing themselves with those already involved in exporting, they can implement the required changes within their organisations.

(3) The Discriminant Function Analysis shows that there is also a relationship between the 10 factors of the external dimension and the classification of the export group membership. The analysis shows that the external factors are able to correctly classify 83.38% of the same population into exporter's group and non - exporter's group. This result is of a particular importance to public policy makers who want to encourage firms to become involved in exporting, in the way that the Government can meet the needs of each of these groups alone.

(4) In comparison between the solutions produced by the two dimensions (internal and external) in respect of the classification of export group membership, it is inferred that factors related to the internal processes of the firms are more critical for them to become involved in exporting than those existing outside of their boundaries. This is in line with other export behaviour studies, such as Cavusgil (1976), Cavusgil and Nevin (1981), Axinn (1985, 1988), Kolhede (1984), Gomez-Mejia (1988) and Gemunden (1991). It indicates that the real barriers to export are more related to the aspects under the control of the firm. This result is of a particular concern for the Government if it intends to encourage firms to move into exporting. The Government action is more likely to be effective where it is designed to make an impact on the firm's internal environmental characteristics.

(5) The integration approach of the internal and external dimension provides empirical evidence that export behaviour is better predicted by the combination (interaction) of these dimensions than by each dimension acting alone. The analysis shows that the addition of the 10 factors of the external dimension to the 16 factors of the internal dimensions produce a significant improvement in the classification of group membership and it is able to correctly classify 97.39% of the same population into two groups (i.e., exporters and non - exporters). This result supports the behaviour theories of the firm which suggest that organisational behaviour is a function of the interaction of the firm's internal with its external environment. It also gives a better understanding of export behaviour by viewing the organisation as a whole rather than being isolated fragments (i.e., previous studies only covered the internal environment).

(6) The analysis of the discriminant analysis function also shows that out of the 26 factors of the combination 21 could be considered as significant discriminators between the exporters' group and the non - exporters' group. These 21 factors belong to both of the two key dimensions of the environments: (1) Internal environmental dimension (14 factors) and (2) External environmental dimension (7 factors).

With regard to the factors related to the internal environmental dimension, the 14 principal discriminators, in their order of importance are: 1. "Managerial aspirations", 2. "Export benefits", 3. "Business experience", 4. "Willingness to change", 5. "perceived risk", 6. Market competitive strengths", 7. "Technological strengths", 8. "Size of the firm", 9. "Product characteristics", 10. "Export objectives", 11. "Demographic", 12. "Commitment to export related activities", 13. "Education", and 14. "Operational obstacles".

With regard to the factors related to the firm's external environmental dimension, the 7 most important factors, in their order of importance, they are: 1. "Awareness of the government's export policies and assistance", 2. "Government's export policies and assistances", 3. "Raw material related problems", 4. "Foreign markets conditions", 5. "Government export procedures related problems", 6. "Physical distribution related problems", and 7. the "Domestic market conditions".

(7) Individual analysis of the attributes of 26 factors show some differences from the result of taking attributes together (i.e., as a factor) in respect of the differentiating between the exporters' group and the non - exporters' group. While 4 factors are shown to be important in discriminating between the two groups, some of their attributes are regarded as not important by the two groups. They are: 1. "Export benefits", 2. Demographic", 3. Government's export procedures related problems", and 4. "Physical distribution related problems". The possible explanations for this disagreement of results might be attributed to two factors; the use of the aggregate measure here (i.e., the factor) may produce more power than on each attribute acting alone, and secondly the different types of the industrial firms under investigation. Some of these types might regard these attributes taken together as being essential in order to become involved in exporting.

(8) The research findings show some disagreement with previous export behaviour studies in respect of the results of the following factors and their associated attributes; 1. the "Size of the firm" 2. "Business experience", 3. "Demographic", and 4. "Government's export policies and assistances", 5. "Education".

The attributes of the "size of the firm" factor were found to be important in discriminating between the two groups (i.e., exporters and non - exporters). While this result is supported by Reid (1980), Withey (1980), Gronhang & Lorenzen (1982), Culpan (1989), Keng & Jiuan (1989), Ortiz-Buonafina (1990), Samiee & Walters (1991) and Liouville (1992), it is in conflict with others, such as Cavusgil and Naor (1987), Axinn (1985, 1988), Czinkota & Johnston (1983), and Kirpalani and Macintosh (1983). This might be attributed either to the use of the aggregate measure (i.e., factor), or it might be due to the different sizes of the firms in the countries under investigation. The size of the Jordanian firms are quite small in comparison with those firms in developed countries, such as the U.K, U.S.A, etc.

In respect of the attributes of the "business experience" factor, the result indicates that there is no relationship between the firm's age and the probability of being an exporter. While this result is supported by Keng and Jiuan (1989) and Samiee & Walters (1991), it is found to be in conflict with Cavusgil and Naor's study (1987) who reported that the number of years in business (i.e., age) is an important attribute in differentiating between exporters and nonexporters. Also, Tseng and Yu (1991) reported that exporters have been in business for a significantly shorter period of time than nonexporters. The possible explanation for this disagreement might be due to the differences of business experience of the firms in the countries under investigation. Industrial development in Jordan is quite recent in comparison to business experience of firms in developed industrial countries, such as the UK.

The age of the decision - maker as one of the attributes of the "demographic" factor is not shown to be important in differentiating between the two groups. However, this result does not support previous

studies, such as those by Layton & Dunphy (1970) and Pinney (1968), which reported that exporters were more likely to be younger than non - exporters. The difference of findings might be attributed to the difference of the countries under investigation. In fact, managers at the higher level of management in Jordan are almost old (i.e., above 40 years). The survey indicates that about 70% of the managers interviewed were above 40 years old.

Previous studies, such as Kolhede (1984), Rabino (1980) and Sullivan & Bauerschmidt (1988, 1990) reported that government export incentives are not important either for exporters or non-exporters. However, the present results show that the government's export assistance and policies are regarded as important by the two groups. This disagreement might be attributed to the different abilities of the firms in the countries under investigation. Jordanian firms might not be in a position or able to export without the government's export assistance, because of the lack of financial resources and know - how.

In respect to the attribute of "education" factor, the study indicated that there is a relationship between the decision-maker's type of education and the firm's export behaviour. While this result is supported by many studies such as Holzmuller & Kasper (1990, 1991) and Simpson and Kujawa (1974), it is found to be in conflict with other studies such as Axinn (1985, 1988), Cavusgil and Naor (1987) and Bourantas & Halikias (1991). The possible explanation of this disagreement might be related to the differences between the respondents of the firms of the countries under investigation.

10.4 Research Contributions

Several contributions to existing knowledge are made in this research. These contributions are theoretical and practical.

10.4.1 Theoretical Contribution

(1) Extending the understanding of export behaviour by testing the phenomenon in a new environment. In the literature review, it was pointed out that most of the research in this area was conducted in developed countries. To the best knowledge of the researcher, export behaviour at the firm level has never been investigated in Jordan or in any other country similar to Jordan.

(2) In comparison with the previous studies conducted in the same field, this study might be considered to be more comprehensive in terms of the number of variables investigated, i.e., the internal and external environmental measures. In other words, the study might be one of a few attempts to investigate one hundred and eleven variables separately and together (i.e., 26 factors).

(3) This study has been conducted in a systematic manner. A conceptual framework was designed to guide it. The framework is based on the integration of the internal and external factors which are thought to influence export behaviour and the level of exporting. For more details, see chapter 4.

(4) The researcher also has developed some measures that relate to the firm's external environment, in particular, the measures of the political - legal environment in Jordan. To the best knowledge of the researcher, these measures have not been previously used in export studies (for more details, see Chapter 3).

(5) The findings of this study reinforce many findings of the previous works in this area. However, some conflicting results were

also reported, for example, the age of the decision - maker was irrelevant either to the level of exporting or export behaviour. These differences raise new explanations when variables are, or are not, relevant to export behaviour.

(6) To the researcher knowledge, this research might be one of the few studies in this area testing the reliability of the scale of measurement of data collection (for more details see Appendix [1]).

(7) To the researcher knowledge, the McNemar test might be employed here for the first time in export behaviour studies, namely, to test the improvement in the classification of group membership after the addition of the external environmental measures to the internal environmental measures.

10.3.2 Practical Contributions

The practical contributions of this study relate to the firm's management, international marketers, and public policy makers.

(1) The decision - makers of manufacturing firms will benefit from this study's findings. A better understanding of the factor determining export behaviour and the level of exporting will help them to implement the required changes within their organisations.

(2) International marketers will find this study relevant in understanding export behaviour of manufacturing firms in one of the developing countries (i.e., Jordan). As indicated in Chapter one, export behaviour has been mostly investigated in developed countries. Little is known about the export behaviour of manufacturing firms in developing countries. In investigating the export behaviour of manufacturing firms in Jordan, this research has added a new dimension to the existing knowledge about export behaviour.

(3) The public policy - maker will benefit from the findings of this study. A better knowledge of the factors that determine the level of exporting or export behaviour will help the Jordanian Government, in particular, to formulate an effective export promotion programme and to make the right changes with regard to the current export policies, procedures, facilities and assistances.

(4) A comparison between the results of the determinants of the level of exporting and the determinants of export behaviour was also reported (see Appendix [3]). This comparison will help both the public policy - makers and manufacturers in Jordan to implement the required changes within their organisations for the purpose(s) of either to improve the level of exporting or / and to encourage other firms to become involved in exporting.

(5) The implications of the research findings are stated under two types of recommendations. These are: (a) recommendations for public policy - makers and (b) recommendations for current and potential exporters (see the next section 10.5)

10.5 Recommendations

The various analysis carried out in this research have provided detailed information about the possible factors that govern the level of exporting and export behaviour in Jordanian firms. Drawing from the key findings obtained from this empirical study, two types of recommendations can be addressed (see Table 10.1 and Table 10.2):

(1) Recommendations for current and potential exporters,

(2) Recommendations for public policy - makers

10.5.1 Recommendations For Current and Potential Exporters

a. The improvement of the level of exporting requires special emphasis on a number of the firm's internal characteristics. The Stepwise Multiple Regression identifies that, out of the 16 factors of the internal dimension, only three factors as being significantly related to a high level of exporting. On this ground, we recommend that if any firm intends to improve its level of exporting, it needs to pay a special attention, at least, to these three most important factors and their associated attributes^{*}: 1. "Willingness to change", 2. "Size of the firm", and 3. "Commitment to export related activities".

<u>Based upon this above conclusion</u>, the following actions are recommended for firms who want to improve their levels of exporting:

(1) The management should try to adapt (change) their marketing mix policies (i.e., pricing, distribution, product and promotion) to meet foreign markets requirements. This can be done by collecting more information about the foreign market environment to where firms want to export.

(2) The management should devote human and material resources to carry out the following export related activities; 1. promoting * see Appendix [3] for detail of the attributes involved

their products to overseas customers 2. participating in international trade fairs, 3. keep sending sales representatives to foreign markets in order to explore and seek marketing opportunities in these markets, 4. using systematic research for the possibilities of increasing export sales, and 5. establishing a separate department for handling several export related activities.

(3) The firm should try to enlarge its business size in general and maximize the capital assets as much as possible in particular (see Appendix [3], factor 4).

b. The adoption of an export strategy also requires much work on a number of the firm's internal characteristics. The discriminant function analysis identifies 14 out of the 16 factors of the internal dimension as being significant discriminators between the two groups (i.e., exporters and non - exporters). Accordingly, we recommend that any firm intending to become involved in exporting needs to pay a special attention to, at least, the following fourteen most important factors and their attributes: 1. "Managerial aspirations", 2. "Export benefits", 3. "Business experience", 4. "Willingness to change", 5. "Perceived risk", 6. Market competitive strengths", 7. "Technological strengths", 8. "Size of the firm", 9. "Product characteristics", 10. "Export objectives", 11. "Demographic", 12. "Commitment to export related activities", 13. "Education", and 14. "Operational obstacles".

<u>In addition to those actions</u>, the following are recommended for potential exporters:

(1) Management should match its firm's profile with the profile of those already involved in exporting. This will help the management to identify and implement the required changes within their organisation. A profile of the exporter's group is presented in Chapter 9.

Table (10.1)

Recommendations and Suggested Actions For Exporters and Potential Exporters

Recommendation s	Suggested Actions & Activities
[A] Develop or improve commitment to export activities.	[O1] Firms should devote and allocate financial and human resources to undertake several important export marketing activities. These activities include:
	 Participation in international trade fairs (e.g., contact Jordan Trade Centre). Hiring or training export sales persons (e.g., contact University of Jordan). Hiring or using export sales agencies. Firms which are not able to establish their own distribution system in foreign markets can hire export agencies to sell on their behalf. Sending sales representative to foreign markets for seeking export opportunities. Establishing a unit or department to handle all of the export related activities. Purchasing or collecting information about foreign market (e.g., contact Chamber of Industry). Adapting or changing the firm's organisational and marketing policies to meet foreign markets requirements. These policies include modifying product, distribution and pricing policies. Attending training courses on how to communicate with overseas markets (e.g., Jordan Training Centre).
[B] Develop or improve the firm's differential advantages. These include; size, experience, technological strengths, innovationsetc.	[02] Enlarging the firm's size as much as possible. This can be done by increasing the number of employees, or increasing the capital investment. Based on the type of ownership, firms can increase their investment capital in many ways. For example, Join Stock companies can issue new shares. Another alternative is to get loans from financial institutions, such as the Industrial Bank Development in Jordan.
	[03] Increasing the rate of expenditure on R & D for the purpose of making or innovating high quality products.
	[04] Increasing or improving the firm's technological standard level. This can be done by modernising the current methods and equipments of production.
	[05] Attending international conference or training course on how to reduce the costs of production. Price terms in foreign markets plays an important role in foreign markets. Therefore, costs of production should be minimized as much as possible.

Continue Table (10.1)

[C] Managers Should develop or improve their international outlook.

[D] Firms should remove or learn how to reduce their export operational obstacles.

- [06] Skills and qualities of management should be improved. This can be done by hiring new managers or attending training courses in this respect (e.g., contact Jordan Training Centre).
- [07] Improving foreign languages skills. (e.g., contact University of Jordan).
- [08] Attending international conferences and /or seminars that are related to the aspects of export marketing.
- [09] Subscribing to international trade periodicals.
- [10] Organising frequent visits to overseas markets.
- [11] Attending training courses on how to communicate with overseas markets (e.g., contact Jordan Training Centre).
- [12] Getting in touch with the Governmental departments for the facilities and assistance they offer in respect to exporting (e.g., Jordan Trade Centre).
- [13] Using or hiring export agencies to sell the firm's product in overseas markets.
- [14] Using or hiring export research agencies to collect the required information about foreign markets or contact the Chamber of Industry in this respect.
- [15] Firms could form themselves into export groups. This unity might strengthen their abilities to overcome their export operational obstacles, (contact or join Jordan Export Trade Association).
- [16] Using assistance and facilities offered by different governmental departments (e.g., facilities offered by the Jordan Trade Centre and the Chamber of Industry..etc.).
- [17] Firms can seek help from similar firms experienced in exporting but not competing with them, and form other counselling centres available in Jordan such as the Royal Scientific Society in Jordan.

(2) Managers should attempt to determine how foreign marketing can help accomplish their firms' goals, such as growth, making greater profit, keeping up with competitors, reducing the cost of production, etc. (see Appendix [3], factors 1 and 4).

(3) Managers should try to improve their foreign language skills. This can be done through travelling or/ and attending training courses in foreign languages.

(4) Managers should try to increase their foreign experience. Attending international business courses and subscribing to international trade periodicals are some ways by which managers can build and retain a realistic outlook in foreign markets, which is an important factor in determining firms' export behaviour.

(5) Firms should try to improve and strengthen their competitive advantages in general and management skills and the level of technology (i.e., equipment and methods of production) in particular (see Appendix [3], factor 8).

(6) Firms should try to develop and design a high quality and competitive price product for overseas customers (see Appendix [3], factor 7).

(7) To reduce or remove export operational obstacles (see Appendix [3], factor 2), it is highly recommended for management to attend international trade conferences in this areas, or training courses on export marketing in an attempt to learn how to overcome such problems.

10.2.3 Recommendations for Public Policy - wakers

(1) A direct implication of the findings of this study for the public policy formulation is that aggregate measures to stimulate export will not be effective unless they are coupled with measures taken at the individual firm's level. In other words, if the

government decides either to enhance the level of exporting, or to induce firms to move into exporting, it would be better to formulate export promotion programmes at the level of individual firms rather than setting them at a national level.

(2) The study's findings show that the factors determining the firm's level of exporting are quite different from the ones determining its export behaviour. This conclusion implies that different Government export promotional programmes are required, depending on whether the objective is to improve existing levels of exporting, or to encourage other firms to become involved in exporting. For more details, see Appendix [3].

(3) To improve the level of exporting, the Government should make a special effort on a number of the firm's external characteristics. The Stepwise Multiple Regression analysis identifies 5 out of the 10 factors of the external dimension as being significantly related to a high level of exporting. Accordingly, we recommend that if the Government intends to enhance the level of exporting, it needs to pay a special attention to, at least, these five most important factors and their associated attributes: 1. The "Foreign markets conditions", 2. "Export costs", 3. The "domestic market conditions", 4. "Government export policies and assistances. For detail about the attributes involved, see Appendix [3].

(4) To encourage firms to become involved in exporting, the Government should pay a special attention to a number of the firm's external characteristics. The DFA identifies that, out of the 10 factors of the external dimension, only seven factors of significant importance in discriminating between the two groups (i.e., exporters and non - exporters). Accordingly, we recommend that if the Government

Recommendations and Suggested Actions For Government to Improve and Encourage Export Involvement

	Recommendations		Suggested Actions and Activities
[A]	Governmental export procedures should be simplified with respect to time, costs and amount of paper work involved. These include: export licensing, inspection process and Customs procedures.	[01]	Establish a one stop department in the Industry of Trade to handle all the governmental export requirements, i.e., have one unit or department where the exporter can conduct and meet the governmental export procedures in a minimum of time and cost.
		[02]	Publicizing information about governmental export requirements and procedures. This might help the potential exporter to better understand and meet the governmental export requirements.
		[03]	Organising training courses or seminars for potential exporters on how to reduce the time and costs associated with conducting export procedures.
CBJ I	Overcoming imperfections in the provision of the government's export assistance. These imperfection include the following:	_	
		15	
	Inadequate tax incentives	[04]	Government should give export oriented firms more allowances for tax purposes.
		[05]	Government should give more compensation for additional costs in export through tax reduction and profit tax exemptions. This would help the manufacturers to sell their products in competitive terms.
- Pay	Payment procedures [C & financial problems [C	[06]	Government should establish export credit guarantee system against non – payment risk.
		[07]	The Central Bank of Jordan should either abolish the bank guarantee requirement, or improve its terms with respect to the time & the value amount of this guarantee.
		[08]	The Central Bank of Jordan should provide assistance in payment procedures for potential exporters by acceptance of the rediscount export bill in a good term.
		C093	Government should offer the best exchange rate for potential exporters and current exporters. This might encourage or help firms to become involved in exporting.
		C103	Publicizing instructions or organising seminars about the payment procedures in overseas markets.

- Customs Duties facilities including duty - drawback system and temporary entry schemes.
- [C] Overcoming imperfection in the domestic market facilities. These include inadequate commercial and infrastructural facilities, such as transportation, warehouses, banking facilities, etc.

- Foreign market representatives

- Foreign market information

& communication

- [11] The government should try to establish more commercial -Attaches and trade centres wherever necessary.
- [12] Establishing a databank unit or department at the Chamber of Industry for providing information about foreign markets environmental characteristics.
- [13] Encouraging the establishment of specialized trade offices for collecting information about foreign markets.
- [14] Organising seminars or training courses on how to get information about foreign markets
- [15] Purchasing up-dated information from foreign markets.
- [16] Publicizing information about foreign markets.
- [17] Producing brochures about potential foreign markets.
- [18] Encouraging participation in international Trade fairs.
- [19] Designing export training courses or / and seminars covering all aspects of export marketing (insurance, shipment, banking. etc.).
- [20] Publicizing instructions or / and organising seminars about about the customs duties facilities. This can be done in in cooperation with academic institutions in Jordan.
- [21] Simplifying the procedures concerning using these facilities.
- [22] Encourage establishment of specialized trade companies or export agents in the domestic market in order to export Jordanian products and give those companies similar incentives which are given to exporters.
- [23] Government should attempt to reduce costs associated with obtaining such facilities by setting new regulations that affect the price of these facilities, in particular, using communication and transportation facilities for export purposes.
- [24] Encouraging the commercial banks to give short and long terms loans to export oriented firms. This can be guaranteed by the Central bank of Jordan.
- [25] Providing training courses or seminars on how to reduce the cost of obtaining transportation, warehouses and communication facilities related to export activity.

[D] Reinforcement and improvement of the current export assistance and policies.

- [26] Organising seminars or designing promotional programmes with the objectives of creating knowledge and awareness regarding the types of export assistance that government offers to the exporters.
- [27] Organising seminars for exporters and potential exporters on how to benefit from governmental export assistances and facilities. This can be done in cooperation with colleges and universities in Jordan.
- [28] Publicizing information about the available types of assistance from which firms could benefit from when they become involved in exporting.
- [29] Establishing a unit or a department in the Ministry of Trade to give the relevant information to the exporter's needs.
- [30] The Government should help potential exporters to overcome problems related to foreign governments trade restrictions. This can be done by making bilateral trade agreements with foreign governments.

wants to induce firms to become involved in exporting, it needs to pay a special attention to, at least, these seven most important factors and their associated attributes: 1. "Awareness of the government's export policies and assistance", 2. "Government's export policies and assistances", 3. "Raw material related problems", 4. "Foreign markets conditions", 5. "Government export procedures related problems", 6. "Physical distribution related problems", and 7. the "Domestic market conditions". For detail about the attributes involved, see Appendix [3].

<u>The following actions are recommended</u>, whether the Government intends to enhance the firms' levels of exporting or/ and encourage other firms to become involved in exporting:

(5) The Government should take corrective action concerning some current export measures, including: export polices, procedures, facilities and assistances. The following are recommended in the light of the study's findings (see Appendix 3, Factors 17, 18, 19 and 20):

- The Government's export procedures should be simplified. In other words, the amount of paper work required, the costs, and time associated with export procedures need to be reduced as much as possible. In particular, the export licence requirement, the inspection process and the custom procedures should be closely examined for simplification.

- The Government should design educational programmes with the purpose of creating awareness of its export facilities and policies. Special educational and promotional programmes should be developed to inform firms about the duty - drawback system facilities, because these were not known by a large number of manufacturing firms.

- The Government should simplify the current payment procedures in general, and the bank guarantee requirement in particular, because they were found to be the most important problems regarded by firms, either those who want to improve their level of exporting or those who want to become exporters.

- The Government should seek to remove export barriers in its aim to stimulate export. For example, the Ministry of Trade should organize counselling sessions that cater for individual firms' enquire. Such measures might help to remove some anxieties about exporting problems.

(6) The Government should establish Commercial - Attaches or trade centres wherever possible. Such measure is expected to be helpful for a firm which wants to enter foreign markets as an exporter. The study's finding in this respect showed that the lack of Commercial -Attaches in foreign market is regarded as an important problem by current and potential exporters. A summary of results on this issue is provided in Appendix (3), Factor 19.

(7) Information about potential foreign markets should be up dated and circulated among manufacturing firms by whatever possible media. Lack of reliable information about foreign markets is considered as one of the most important problems that inhibit or discourage firms either to become involved in exporting or to improve their level of exporting.

(8) The Government should design export training courses or organise seminars on export marketing. This can be done by cooperating with the colleges and universities in Jordan. Special programmes or courses can also be developed to increase management's perceived gains from undertaking exporting activity. These can also include reduction

of obstacles, reduction of risks and costs associated with undertaking export activities.

(9) The Government should establish an export credit guarantee system against non - payment risks. A lack of export credit guarantee against non - payment risk is regarded as an important problem by both the current exporters who wish to increase their level of exporting, and potential exporters who wish to enter foreign markets as exporters.

(10) The Government needs to provide more trade incentives and facilities in general and tax incentives for export oriented industries in particular. Inadequate tax incentive for export oriented industries is regarded as one of the main problems by current and potential exporters (see Appendix [3], factor 3).

(11) The Government policies should foster growth in export development through counselling and training programs including mechanics of exporting, export documents and the use of the Government's export facilities.

(12) The Government should use its advantage with foreign government to support export sales by the Jordanian firms. It can provide such support through bilateral trade agreement with foreign government. This will also help the Jordanin firms to overcome problems that relate to foreign governments trade restriction.

(13) The Government should also take corrective actions in respect of the current infrastructural facilities existing in the domestic market. These facilities are found to hinder rather than stimulate export. In order of their importance, they are (1) transportation facilities and (2) warehouse facilities.

10.4 Research Limitations

(1) The study was tested in one geographic area, i.e., it was only conducted in the Jordanian environment. Therefore, the lack of external validity of this research means that any generalization of the research findings should be taken with caution.

(2) The data analysis was cross - sectional. As with all cross sectional studies, the parameters tended to be static rather than dynamic. This drawback limits the generalization of the study's findings to further situations and beyond the specific population from which the data was gathered.

(3) The study used the single informant approach for data collection, i.e., the data were collected from a key person in each organisation (see Chapter 5, section 5.5). This approach might not provide the best view of the organisation as a whole. However, by using the multiple informant approach (i.e., more than one person in each organisation) in future research, the problem of aggregate responses should be solved.

(5) Because of the time and cost associated with data collection, the study population was confined to six types of industrial sectors. Therefore, the generalisation of the results beyond these industrial sectors should be taken with caution.

10.5 Areas for Further Research

Since this is the first study to address the export behaviour of manufacturing firms in Jordan, there are many issues that could not be covered in this research and deserve further research. The suggested areas for further study are as follows:

(1) The external validation of the current research findings. It is important that part of the future research be directed towards the

replication of the findings of this research. It is suggested that future researches should use the same dimensions of the internal and external environment.

(2) Future studies should be directed to investigate the same phenomena in other types of industrial sectors.

(3) In this research, the level of exporting was measured by one criterion for the reasons explained in Chapter 4. Future research is required to develop multiple measures for the level of exporting. The purpose would be to find out whether there is any difference between the results of using one criterion and the results of using multiple criteria together.

(4) Future research is required to explore and develop more measures relating to the firm's external environment. These could be incorporated with others already investigated. The purpose is to extend the number of aspects of the firm's external environmental measures, and to verify the results which were obtained in this study.

REFERENCES

- Aaby, N.E. & S.F. Slater, "Management Influences on Export Performance: A Review of the Empirical Literature 1978 - 88", International Marketing Review, Vol. 6, No. 4, (1989), PP. 7 - 24.
- Aaker, D.A., <u>Multivariate Analysis in Marketing Theory and</u> <u>Application</u>, Wadsworth, Inc., California, (1973).
- Aaker, D.A. & G.S. Day, <u>Marketing Research</u>, 3rd edition New York: John Wiley and Sons, (1986).
- Abdeljaber, T., "Jordan's Economy: Current Situation, Objectives and Policies, (Unpublished Paper), (1986).
- Abdel-Malek, T., "Import Substitution Vs. Export Orientation", <u>Columbia Journal of World Business</u>, Vol. 4, (Sep. / Oct. 1969), PP. 29 - 38.
- Abdel-Malek, T., <u>Managerial Export Orientation: A Canadian Study</u>, School of Business Administration, University of Western Ontario (1974).
- Abdel-Malek, T., "Export Marketing Orientation in Small Firms", <u>American Journal of Small Business</u>, Vol. 111, No. 1, (July 1978) PP. 25 - 34.
- Abuzir, W.I., <u>Economic Development and Foreign Trade in Jordan</u>, Final Report, Central Bank of Jordan (March 1988).
- Aharoni, Y., <u>The Foreign Investment Decision Process</u>, Cambridge, MA: Harvard University Press, (1966).
- Airaksinen, T., Export Performance of the Firms in the Finnish Engineering Industry, Helsinki School of Economics, (1978).
- Albaum, G., "Effectiveness of Government Export Assistance for US Smaller-Sized Manufacturers: Some Further Evidence", <u>International</u> Marketing Review, Vol. 1, (1983), PP. 68 - 75.
- Albaum, G., J. Strandskav, E. Duerr, & L. Dowd, <u>International</u> <u>Marketing and Export Management</u>, Addison - Wesley Publishing Co., (1989).
- Aldrich, H.E., Organisations and Environments, Englewood Cliffs, New York: Prentice - Hall, (1979).
- Alexandrides, C.G., "How the Major Obstacles to Expansion Can Be Overcome", <u>Atlanta Economic Review</u>, (May 1971), PP. 12 - 15.
- Alexandrides, C.G. & G.P., Moschis, Export Marketing Management, New York: Praeger, (1983).
- Alfred, M., L. Campbell-Boross & P. Rayment, Exports and Economic of Growth of Developing Countries, Cambridge, University Press, Economic and Social Studies, Vol. XXV, (1968).

- Ali, A. & P.M., Swiercz, "Firm Size and Export Behaviour: Lessons From the Midwest", Journal of Small Business Management, Vol. 29, (April 1991), PP. 71 - 78.
- Amerah, M. & M. Al Hajji, "<u>Import Capacity and Economic Growth</u> <u>in Jordan (1973 - 1985)</u>, Royal Scientific Society, Amman - Jordan, (1987).
- Amerah, M. & M. Samdi, <u>Promotion of Manufacturing Exports in Jordan</u>, Royal Scientific Society, Amman - Jordan (1979).
- Amine, L. & T. Cavusgil, "Exploring Strategic Aspects of Export Marketing", <u>International Marketing Review</u>, Vol. 1, (1983), PP. 5 -11.
- Ansoff, H.I., Corporate Strategy, New York: McGraw Hill, (1965).
- Archibald, G.C. (ed.), <u>The Theory of the Firm</u>, Middlessex, England: Penguin Books, Ltd, (1971).
- Argries, C., <u>Understanding Organisational Behaviour</u>, Dorsey: Homewood III, (1960).
- Atkinson, J.W., "Motivational Determinants of Risk-Taking Behaviour", Psychological Review, Vol. 64, (1957), PP. 359 - 372.
- Axinn, C.N., "An Examination of Factors that Influence Export Performance", Ph.D dissertation, Michigan State University, (1985).
- Axinn, C.N., "Export Performance: Do Managerial Perceptions Make a Difference", <u>International Marketing Review</u>, Vol. 5, (Summer 1988), PP. 61 - 71.
- Aya1, I., "International Product Life Cycle: A Reassessment and Product Policy Implications, <u>Journal of Marketing</u>, Vol. 45, (Fall 1981), PP. 91 - 96.
- Ayal, I., "Industry Export Performance: Assessment and Prediction", Journal of Marketing, 46, (Summer 1982), PP. 51 - 61.
- Ayal. I. & S. Hirsch, "Manufacturing Factors in Small Country Manufacturing Exports: Are Market Share and Market Growth Rate Really Important?", Journal of International Business Studies, 2, (Dec. 1977), PP. 327 - 344.
- Baker, M., "Export Myopia", <u>The Quarterly Review of Marketing</u>, (Spring 1979), PP. 1 -10.
- Baldwin, R., "Determinants of the Commodity Structure of US Trade" American Economic Review, Vol. 61, (1971), PP. 126 - 146.
- Barker, A. & E. Kaynak, "An Empirical Investigation of the Differences Between Initiating and Continuing Exporters", <u>European Journal of</u> <u>Marketing</u>, Vol. 26, No. 3 (1992), PP. 27 - 36.

- Barnhart, J., "Export Profitability: An Analysis Among Indiana Firms.", Ph.D Dissertation, Indiana University, Bloomington, Indiana, (1968).
- Barrett, J. & L.F. Wilkinson, "Export Stimulation: A Segmentation Study of the Exporting Problems at Australian Manufacturing Firms", <u>European Journal of Marketing</u>, Vol. 19, No. 2, (1985), PP. 53-72.
- Barrett, J. & L. Wilkinson, "Internationalisation Behaviour, Management Characteristics of Australian Manufacturing Firms by Level of International Development", <u>Research in International</u> <u>Marketing</u>, Turnbull & Pliwoda (ed), London: Croom Helm, (1986).
- Bauerschmidt, A., D. Sullivan, & K. Gillespie, "Common Factors Underlying Barriers to Export: Studies in the U.S Paper Industry", <u>Journal of International Business Studies</u>, (Fall 1985) PP. 11 - 123.
- Beamish, P. & H. Munro, "The Export Performance of Small and Medium -Sized Canadian Manufacturers", <u>Canadian Journal of Administrative</u> <u>Science</u>, Vol. 3, No. 1, (1986), PP. 29 - 40.
- Beamish, P. & H. Munro, "Exporting for Success as a Small Canadian Manufacturers", <u>Journal of Small Business and Enterprenership</u>, Vol. 4, (1987), PP. 38 - 43.
- Bell, J.F., <u>A History of Economic Thought</u>, New York: Ronald Press, (1953).
- Bell, J., M. Murray & K. Madden, "Developing Exportise: An Irish Perspective", <u>International Small Business Journal</u>, Vol. 10. No. 2, (Jan. / March 1992), PP. 37 - 45.
- Belsey, D., E. Kuh, & R. Welsch, <u>Regression Diagnostics: Identifying</u> <u>Influential Data and Sources of Collinearity</u>, New York: John Wiley & Sons, (1980).
- Bendig, W., "The Reliability of Self Rating as a Function of the Amount of Verbal Anchoring and the Number of Categories and the Scale", Journal of Applied Psychology, Vol. 37, (1953), PP. 38-41.
- Bendig, W., "Reliability & the Number of Rating Scale Categories", Journal of Applied Psychology, Vol. 38, (1954), PP. 38 - 40.
- Bennett, S. & D. Bomers, <u>An Introduction to Multivariate Technique</u> for Social and Behavioural Sciences, McMillan, London, (1976).
- Bernard, C.I., <u>Functions of the Executive</u>, Cambridge: Harvard University Press, (1938).
- Berry, W. & S. Feldman, <u>Multiple Regression in Practices</u>, <u>Series</u> <u>Quantitative Application on the Social Science</u>, Sage Publication, Inc, U.S.A, (1985).
- BETRO Trust Committee, Concentration on Key Markets, Royal Society of Arts, London, (1977).

- Bhagwati, J., "The Pure Theory of International Trade", <u>The Economic</u> <u>Journal</u>, Vol. 74, (March 1964), PP. 1- 84.
- Bilkey, W.J., <u>Industrial Simulation</u>, Lexington, MA: Heath Lexington Books, (1970).
- Bilkey, W.J., "An Attempted Integration of the Literature of the Export Behaviour of Firms", Journal of International Business Studies, (Summer 1978), PP. 33 - 46.
- Bilkey, W.J., "Variables Associated with Export Profitability", <u>Journal of International Business Studies</u>, Vol. 13, (Fall 1982), PP. 71 - 80.
- Bilkey, W.J., "Export Marketing Practice and Export Profitability" Research in Marketing, Vol. 7, (1984), PP. 211 - 235.
- Bilkey, W.J., "Development of Export Marketing Guidelines", <u>International Marketing Review</u>, Vol. 2, No. 1, (Spring 1985), PP. 31 - 40.
- Bilkey, W.J. & G. Tesar, "The Export Behaviour of Smaller Sized Wisconsin Manufacturing Firms", <u>Journal of International Business</u> Studies, Vol. 7, (Spring/ Summer 1977), PP. 93 - 98.
- Blough, R., <u>International Business: Environment and Adaptation</u>, McGraw - Hill, New York, (1966).
- Boddewyn, J., "The Analysis of Business Systems & Their Environment", <u>University of Washington Business Review</u>, Vol. 26, (1967), PP. 68-79.
- Bodur, M. & S.T. Cavusgil, "Export Market Research Orientation of Turkish Firms", <u>European Journal of Marketing</u>, Vol. 19, (1985), PP. 5 - 16.
- Boote, A.S., "Reliability Testing of Psychographic Scales", Journal of Advertisement Research, Vol. 21, No. 5, (Oct. 1981), PP. 53 -60.
- Boulding, E., "Implication of General Economic of More Realistic Theories of the Firm", <u>American Economic Review</u>, Vol. XIII, No. 2, (May 1982), P. 35.
- Bourantas, D. & J. Halikias, "Discriminating Variables Between Systematic and Non - Systematic Exporting Manufacturing Firms in Greece", Journal of Global Marketing, Vol. 4, No. 2, (1991), PP. 21 - 62.
- Bradley, F.M., "Attitudes of Export Marketing Growth Among Small Scale Enterprises: A Discriminant Analysis", <u>Proceedings Annual</u> <u>Conference of the Marketing Education Group</u>, St. Patrick's College, Drumcondra, Dublin, (July 1981), PP. 7 - 9.
- Bradley, F.M., & P. Keogh, "Export Management: Motivated, Open -Minded", Journal of Irish Business and Administrative Research, Vol. 3, (Oct. 1981), PP. 29 - 35.

- Brady, D. & W. Bearden, "The Effect of Managerial Attitudes on Alternative Exporting Methods, <u>Journal of International Business</u> <u>Studies</u>, Vol. 10, No. 4, (1979), PP. 79 - 84.
- Brasch, J., "Export Management Companies", <u>Journal of International</u> <u>Business Studies</u>, (Spring / Summer, 1978), PP. 59 - 71.
- Brasch, J., "Deciding on an Organizational Structure for Entry into Export Marketing", <u>Journal of Small Business Management</u>, Vol. 24, (April 1981), PP. 17 - 15
- Brillinger, D., "The Application of the Jacknife to the Analysis of Sample Surveys", <u>Journal of the Market Research</u> <u>Society</u>, Vol. 8, (April 1966), PP. 74 - 80.
- Brooks, M. & P. Rosson, "A Study of Export Behaviour of Small and Medium - Sized Manufacturing Firms in Three Canadian Provinces". In <u>Export Management: An International Context</u>, Edited by Czinkota, M. & G. Tesar, New York, (1982), PP. 39 - 54.
- Brown, F.E., <u>Marketing Research: A Structure for Decision Making</u>, Addision Wesley Publishing Co., Reading Massachusetts, (1980).
- Buatsi, S.N., "The Influence of Organisational Factors on Export Potential & Export Growth of Firms: A Study of Export Involvement by Firms in the UK Plastic Processing Industry", Ph.D. Thesis, MBS, University of Manchester, (1980).
- Buatsi, S.N., "Organisational Adaptation to International Marketing", International Marketing Review, Vol. 3, No. 4, (1986), PP. 17-26.
- Burns, A. & M. Harrison, "A Test of Reliability of Psychographics", Journal of Marketing Research, Vol. 16, (Feb. 1979), PP. 32 - 38.
- Burton, F. & B. Schlegelmilch, "Profile Analysis of Non Exporters Versus Exporters Grouped by Export Involvement", <u>Marketing</u> International Review, Vol. 27, No. 1, (1987), PP. 38 - 47.
- Business America, <u>Jordan</u>: A Country Making the Most of its Limited Resources and Markets Must New Adjust to a Decline in External Income, (Feb. 1987), PP. 30 - 3.
- Business International, "How Three Firms Deal With Production Shortage" <u>Business International</u> (March 22, 1974), P. 92.
- Campbell, T., "The Informant in Quantitative Research", America Journal of Sociology, Vol. 60, (1955), PP. 332 - 342.
- Cannon, T., "International and Export Marketing", <u>Management</u> Bibliographies and Review, Vol. IV, (1979), PP. 91
- Cannon, T., "Managing International and Export Marketing", <u>European</u> Journal of Marketing, Vol. 14, No. 1, (1980), PP. 34 - 49.
- Cannon, T., "The Smaller Firm in the Export Environment", in "Policy <u>Issues in Small Business Research"</u>, edited by Gibb, A. & T. Webb, (1980), PP. 140 - 154.

- Carmines, E. & R. Zeller, <u>Reliability and Validity Assessmen</u>t, Series Quantitative Application in the Social Science, No. 17, London: Sage Publication Int., (1979).
- Cateora, P. & J. Hess, <u>International Marketing</u>, Richard D. Irwin, Inc. Homewood, III., 6th edition, (1987).
- Cattell, R.B., <u>The Scientific Use of factor Analysis in Behaviour</u> and Life Science, Plenum Press, New York, (1978).
- Cavusgil, S.T., "Organizational Determinants of Firms Export Behaviour: An Empirical Analysis", Unpublished Ph.D Dissertation, University of Wisconsin - Madison, (1976).
- Cavusgil, S.T., "On the Internationalisation Process of Firms", European Research, Vol. 8, (Nov. 1980), PP. 273 - 281.
- Cavusgil, S.T., "Some Observations on the Relevance of Critical Variables for Internationalisation Stages". In M. R. Czinkota & G. Tesar (eds) <u>Export Management: An International Context</u>, New York: Praeger (1982), PP. "76 - 86.
- Cavusgil, S.T., "Success Factors in Export Marketing", <u>Journal of</u> <u>International Marketing and Marketing Research</u>, Vol. 8, No. 2, (June 1983), PP. 63 - 73.
- Cavusgil, S.T., "Organizational Characteristics Associated With Export Activity", <u>Journal of Management Studies</u>, Vol. 21, No.1, (1984), PP. 3 - 22.
- Cavusgil, S.T., "Differences Among Exporting Firms Based on Their Degree of Internationalisation", <u>Journal of Business Research</u>, Vol. 12, No. 2, (1984), PP. 195 - 208.
- Cavusgil, S.T., W.J. Bilkey & G. Tesar, "A Note on the Export Behaviour of Firms: Exporter Profiles", <u>Journal of International</u> Business Studies, Vol. 10, (Spring / Summer 1979), PP. 91 - 97.
- Cavusgil, S.T. & Y.M. Godiwalla, "Decision Making for International Marketing: A Comparative Review", <u>Management Decision</u>, Vol. 20, (1980), PP. 47 - 54.
- Cavusgil, S.T. & J. Naor, "Firm and Management Characteristics as Discriminantors of Export Marketing Activity", <u>Journal of Business</u> <u>Research</u>, Vol. 15, (1987), PP. 221 - 235.
- Cavusgil, S.T. & J.R. Nevin, "Internal Determinants of Export Marketing Behaviour: An Empirical Investigation", <u>Journal of</u> <u>Marketing Research</u>, Vol. 18, (Feb. 1981), PP. 114 - 119.
- Central Bank of Jordan, <u>Monthly Statistical Bulletin</u>, Department of Research & Studies, Vol. 25, No. 6, (1989).
- Chenery, H.B. & A.M. Strout, "Foreign Assistance and Economic Development", <u>American Economic Review</u>, Vol. 56, No. 4, (Sep. 1966).

- Cheong, K. & K. Chong, "Export Behaviour of Small Firms in Singapore", <u>International Small Business Journal</u>, Vol. 6, No. 2, (1988), PP. 34 - 41.
- Chipman, J., "A Survey of the Theory of International Trade", <u>Econometrica</u>, Vol. 33, Part 1 and 2, (July and October 1965).
- Christensen, C., A. de Rocha, D. Gerther, & R. Karbel, "An Empirical Investigation of the Factors Influencing Exporting Success of Brazilian Firms", <u>Journal of International Business Studies</u>, (Fall 1987), PP. 61 - 71.
- Churchill, G.A. & J.P. Peter, "Research Design Effect on the Reliability of Rating", <u>Journal of Marketing Research</u>, Vol. 16, (November 1984), PP. 360 - 375.
- Churchill, G.A., <u>Marketing Research: Methodological Foundation</u>, 4th edition, New York: Holt, Rinehart and Wislon, (1987).
- Coase, R.H., "The Nature of the Firm", <u>Economics</u>, New Series, Vol. 4, (Nov. 1937), PP. 388 405.
- Cohen, J. & P. Cohen, <u>Applied Multiple Regression Correlational</u> <u>Analysis for the Behavioural Sciences</u>, New York: John Wiley and Sons, (1975).
- Cohen, L. & M. Holliday, <u>Statistics for Social Scientists</u>: An Introductory Text with Computer Programs in Basic, London: Harper and Row, Publishers (1982).
- Cook, T.D. & D.T. Campbell, "The Design and Conduct of Quasi -Experiments and the True Experiments in Field Settings". In <u>Handbook of Industrial and Organisational Psychology</u>, Edited by M. D., Dunnett, Chicago: Rand McNally, (1976), PP. 223 - 326.
- Cook, T.D. & D.T. Campbell, <u>"Quasi Experimentation: Design and</u> Analysis Issues for Field Setting, Chicago: Rand McNally, (1979).
- Cook, V.J., "Marketing Strategy and Differential Advantage", Journal of Marketing, (Spring 1983), PP. 68 - 75.
- Cooper, W.W., "Revisions of the Theory of the Firm", <u>American Economic</u> <u>Review</u>, (1949), PP. 1204 - 1222.
- Cooper, W.W., "A Proposal for Extending the Theory of the Firm", Quarterly Journal of Economics, (1951), PP. 87 - 109.
- Cooper, R., K. Hartley, & C. Harvey, <u>Export Performance and the</u> <u>Pressure of Demand</u>, New York: Humanities Press, (1970).
- Cooper, R. & E. Kleinschmidt, "The Impact of Export Strategy on Export Sales Performance", <u>Journal of International Business</u> <u>Studies</u>, Vol. 15, (Spring 1985), PP. 37 - 55.
- Cooper, P. & A. Weeks, Data Model and Statistical Analysis, Oxford: Philip Allan Publishers Ltd., (1983).

Corsuch, R.L., Factor Analysis, Dryden Press, Hinsdile III, (1973).

- Cox, E.P., "The Optimal Number of Response Alternatives for a Scale: A Review", <u>Journal of Marketing Research</u>, Vol. 17, (Nov. 1980), PP. 407 - 422.
- Crask, M.R. & D. William and J. Parreault, "Validation of Discriminant Analysis in Marketing Research", <u>Journal of</u> Marketing, Vol. XIV, (Feb. 1977), PP. 60 - 68.
- Craver, T. & V. Crowder, <u>California Manufacturing Firms and Their</u> Export Practices, Security Pacific National Bank, December (1976).
- Crawford, F.M. & R.A. Lomas, "Factor Analysis: A Tool for Data Reduction", <u>European Journal of Marketing</u>, Vol. 14, (1980), P. 414
- Cullwick, T. & P.J. Mellalicu, "Business Attitudes to Government Export Services and Export Marketing Behaviour", <u>New Zealand</u> Journal of Business, Vol. 3, (1981), PP. 33 - 54.
- Culpan, R., "Export Behaviour of Firms: Relevance of Firm Size", Journal of Business Research, Vol. 18, (May 1989), PP. 207 - 218.
- Cunningham, M. & R. Spigel, "A Study in Successful Exporting", British Journal of Marketing, Vol. 5, (Spring 1971), PP. 2 - 12.
- Czinkota, M., "An Analysis of Export Development Strategies in Selected U.S. Industries", Ph.D dissertation, Ohio State University (1979).
- Czinkota, M., <u>Export Development strategy: Its Promotion Policy</u>, New York: Praeger Publishing Co., (1982).
- Czinkota, M. & W. Johnston, "Segmenting U.S. Firms for Export Development, <u>Journal of Business Research</u>, Vol. 9, (Dec. 1981), PP. 353 - 365.
- Czinkota, M. & W. Johnston, "Exporting: Does Sales Volume Make a Difference?", Journal of International Business Studies, Vol. 14, (Spring/ Summer 1983), PP. 147 - 153.
- Czinkota, M. & W. Johnston, "Exporting: Does Sales Volume Make a Difference? Reply", <u>Journal of International Business Studies</u>, Vol. 16, No. 2, (Spring/ Summer 1985), PP. 157 - 165.
- Czinkota, M. & G. Tesar, (ed.), Export Management; An International Context, New York, Praeger, (1983).
- Czinkota, M.R. & D. Ricks, "Export Assistance: Are We Supporting the Best Program?", <u>Columbia Journal of World Business</u>, Vol. 16, (Summer, 1981), PP. 73 - 78.
- Czinkota, M. & I. Ronkainen, <u>International Marketing</u>, London, The Dryden Press, (1988).
- Czinkota, M. & I. Ronkainen, <u>International Marketing</u>, 2nd ed., London, The Dryden Press, (1990).

- Czinkota, M. & M. Ursic, "An Empirical Analysis of the Psychic Distance Argument for Export Development". In <u>Developments in</u> <u>Marketing Science</u>, edited by Rogers, J.C., 111 Academy of Marketing Science, (1983).
- Cyert, R.M. & J.G. March, "<u>A Behavioural Theory of the Firm</u>" Englewood Cliffs: Prentice - Hall, Inc., (1963).
- Daniels, J. & J. Goyburo, "The Exporter Non Exporter Interface A Search For Variables", <u>Foreign Trade Review</u>, Vol. 2, (July 1976) PP. 258 - 282.
- Daniels, J.D. & F. Robles, "The Choice of Technology and Export Commitment: The Peruvian Textile Industry", <u>Journal of</u> <u>International Business Studies</u>, (Spring /Summer 1982), PP. 67 -87.
- Daniels, J.D. & F. Robles, "Perceived Industrial Innovation Advantage and Export Commitment", <u>Journal of Business Research</u>, Vol. 11, No. 3, (July 1985), PP. 363 - 378.
- Darling, J.R., "Keys for Success in Exporting to the U.S Marketing", European Journal of Marketing, (1985), PP. 17 - 30.
- Da Rocha, A., C. Christensen, and C. Da Cunha, "Aggressive and Passive Exporters: A Study in the Brazilian Furniture Industry", International Marketing Review (UK), Vol. 7, (1989), PP. 30 - 34.
- Day, A.J., Exporting for Profit, London: Graham and Trotman, (1976).
- De La Torre, J., "Marketing Factors in Manufactured Exports from Developing Countries". In <u>The Product Life Cycle and International</u> <u>Trade</u>. Edited by L. T. Wells, Jr. Boston: Graduate School of Business Administration, Harvard University, (1972).
- Denis, J.E. & D. Depelteau, "Market Knowledge, Diversification and Export Expansion", <u>Journal of International Business Studies</u>, Vol. 3, (Fall 1985), PP. 77 - 89.
- Dess, G.G. & D.W. Beard, "Dimensions of Organizational task Environment", <u>Administrative Science Quarterly</u>, (1984), PP. 52 -73.
- Diamantopoulos, A. & K. Inglis, "An Empirical Investigation of Differences in Export Characteristics Between High and Low -Involvement Exporters in the Scottish Food and Drink Industry", Working Paper Series No. 85/10, University of Edinburgh, (1985).
- Diamantopoulos, A. & K. Inglis, "Identifying Differences Between High and Low - Involvement Exporters", International Marketing Review, Vol. 5, (Summer 1988), PP. 52 - 60..
- Dichtl, E., G. Koglmayr & S. Muller, "The Export Decision of Small and Medium - Sized Firms: A Review", <u>Management</u> International Review, Vol. 2, (1984), PP. 49 - 60.

- Dichtl, E., G. Koglmayr & S. Muller, "Identifying Export Potential: A Comparative Analysis of German and Japanese", <u>Advance in</u> <u>International Marketing</u>, Vol. 1, (1986), PP. 233 -254.
- Dichtl, E., G. Koglmayr & S. Muller, "International Orientation as a Precondition for Export Success", <u>Journal of International</u> <u>Business Studies</u>, Vol. 21, (1990), PP. 23 - 54.
- Dichtl, E., F. Kondo, H. Koglmayr & S. Muller, "The Foreign Orientation of Managers", <u>Ritsumeikan Business Review</u>, Vol. 23, No. 2, (1984), PP. 1 - 33.
- Dichtl, E., M. Leibold, H. Koglmayr & S. Muller, "The Foreign Orientation of Management as a Central Construct in Export -Centred Decision - Making Processes", <u>Research for Marketing</u>, Vol. 10, (1983), PP. 7 -14.
- Dillon, W.R. & M. Goldstein, <u>Multivariate Analysis: Methods and</u> <u>Applications</u>, John Wiley & Sons, New York, (1984).
- Dill, W.R., "Environment as an Influence on Managerial Autonomy", Administrative Science Quarterly, Vol. 2, (1958), PP. 409 - 443.
- Douglas, M., "Testing a Methodology for Measuring the Interaction Between Organisation and Environment", <u>Management International</u> Review Vol. 1, (1975), PP. 89 - 97.
- Douglas, S., "Patterns & Parallels of Marketing Structure in Several Countries", <u>MSU Business Topic</u>, (Spring 1971), PP. 38 -48.
- Douglas, S.P. & S.C. Craig, "Examining Performance of U.S Multinationals in Foreign Markets", <u>Journal of International</u> Business <u>Studies</u>, Vol. 14, (Winter 1982), PP. 51 - 62.
- Douglas, S.P. & S.C. Craig, <u>International Marketing Research</u>, New Jersy: Englewood Cliffs; Prentice Hall, (1983).
- Douglas, S. & B. Dubois, "Looking at the Cultural Environment for International Marketing Opportunities", <u>Columbia Journal of World</u> <u>Business</u>, (Winter 1977), PP. 102 - 109.
- Douglas, S. & Y. Wind, "Environmental Factors and Marketing Practice" European Journal of Marketing, (Winter 1973), PP. 155 - 165.
- Doyle, R.W. & N.A. Schommer, <u>The Decision to Export: Some</u> <u>Implications</u>, A Motivation Study Commissioned By the Minnesota District Council, (1976).
- Draper, N. & H. Smith, <u>Applied Regression Analysis</u>, New York: John Wiley, Inc, (1966).
- Drucker, P.F., <u>The Practice of Management</u>, New York: Harper & Row Publishers, (1954).
- Drucker, P.F., "Marketing and Economic Development", Journal of Marketing, Vol. 23, No. 3, (1958), P. 255.

- Duncan, R.B., "Characteristics of Organisational Environments and Perceived Environmental Uncertainty", <u>Administrative Science</u> <u>Quarterly</u>, (1972), PP. 313 - 327.
- Dunteman, G.H., <u>Introduction to Mlutivariate Analysis</u>, Sage Publications Inc., Beverly Hills, London, New Delhi, (1984).
- Edmunds, S.E. & S.J. Khoury, "Exports: A Necessary Ingredient in the Growth of Small Business Firms", <u>Journal of Small Business</u> <u>Management</u>, Vol. 24, No. 4, (Oct. 1986), PP. 54 - 65.
- Edwin, D., K. Fumio, K. Hansz George, & S. Muller, "The Foreign Orientation of Managers: A Key Variable for Export Promotion and Export Results", <u>Ritsumeikan Business Review</u>, Vol. 23, Part 2, (1984), PP. 1 -33.
- Eisenbeis, R.A., "Pitfalls in the Application of Discriminant Analysis in Business Finance and Economics", <u>Journal of Finance</u>, Vol. XXX11, No. 3, (1977), PP. 875 - 900.
- Eisenbeis, R.A. & R.B. Avery, <u>Discriminant Analysis</u> and <u>Classific-ation Procedures</u>, Lexington, Massachusetts: Heath, (1972).
- EIU (The Economic Intelligence Unit), Jordan to 1990 Coping with Changes, The Economist Publications Ltd., London, (1986).
- E1-Agraa, A., <u>The Theory of International Trade</u>, London: Cromm Helm Ltd., (1983).
- E1-Ansary, A., "Managerial Gap Analysis: A Frame Reference for Improving International Business Relations With the Middle East". In Kaynak, E. (ed), <u>International Business in the Middle East</u>, Walter de Gruyter Berlin, New York, (1985), PP. 43 - 55.
- Ellsworth, P.T., <u>The International Economy</u>, 3rd edition, New York: Macmillan, (1964).
- Estafen, D., "Methods for Management Research in the 1970's An Ecological System Approach, <u>Academy of Management Journal</u>, (1971).
- Etgar, M. & J. McConnell, "International Marketing as Decision Making Behaviour of Business Organisation", (1976). In Bilkey's W. J., "An Attempted Integration of the Literature of the Export Behaviour of Firms", Journal of International Business Studies, (Summer 1978), PP. 33 - 46.
- Fayerweather, J., <u>International Business Management</u>, New York: McGraw Hill Book Company, (1969).
- Fenwick I. & L. Amine, "Export Performance and Export Policy: Evidence from the U.K. Clothing Industry", <u>Journal of the</u> <u>Operational Research Society</u>, Vol. 30, No. 8, Great Britain, (1979), PP. 747 - 754.
- Ferber, R., <u>Handbook of Marketing Research</u>, MacGraw Hill, New York (1974).
- Fishbein, M. & L. Ajzen, <u>Belief</u>, <u>Attitude</u>, <u>Intention and Behaviour</u>: <u>An</u> <u>Introduction to Theory and Research</u>, Reading, MA: Addison - Wesley Publishing Company (1975).
- Fisher, R.A, "The Use of Multiple Measurement in Taxonomics Problems", Ann Eugenks, Vol. 7, (1936), PP. 179 - 188.
- Forbes, A.M., "Export Targeting", <u>European Journal of Marketing</u>, (1977), PP. 370 - 373.
- Ford, J.L., <u>The Ohlin Heckscher Theory of the Basis and Effects of</u> Commodities Trade, Asia Publishing House, London, (1965).
- Ford, D. & Leonidon, "Research Developments in International Marketing: A European Perspective". In <u>New Perspectives on</u> <u>International Marketing</u>, Edited by Paliwoda, S. J., London: Routledge (1991).
- Frank, D.E. & W.F. Massy, <u>Innovation and Brand Choice</u>, American Marketing Association, Chicago, (1964).
- Frank, D.E., W.F. Massy & D.G. Morrison, "Bias in Multiple Discriminant Analysis", <u>Journal of Marketing Research</u>, Vol. 11 (August 1965), PP. 250 - 258.
- Garnier, G., "Characteristics and Problems of Small and Medium Exporting Firms in Quebec Manufacturing Sector With Special Emphasis on those using Advanced Production Techniques", Ottaawa: <u>Technological Innovation Studies</u>, Program, Dep., of Industry, Trade and Commerce, (August 1974).
- Garnier, G., "Comparative Export behaviour of Small Canadian Firms in the Printing and Electrical Industries in <u>Export Management: An</u> <u>International Context</u> edited by Czinkota, M. & Tesar, G., New York, (1982), PP. 257 - 275.
- Gemunden, H., "Success Factors of Export Marketing". In New <u>Perspectives on International Marketing</u>, Edited by Paliwoda, S. J., London: Routledge (1991).
- Gomez-Mejia, R., "The Role of Human Resources Strategy in Export Performance: A longitudinal Study", <u>Strategic Management Journal</u> Vol. 9, (1988), PP. 38 - 43.
- Goodnow, J.D. & J.E. Hansz, "Environment Determinants of Overseas Marketing Entry Strategies", <u>Journal of International Business</u> Studies, (Spring 1972), PP. 33 - 50.
- Gottko, J. & R. McMahon, <u>Export Marketing Activities of Small Medium</u> <u>Sized Manufacturers</u>, Paper Presented at the Annual Meeting of Western Marketing Association, New York, Beach, California (1988).
- Green, T.R. & W.H. Cunnigham, "The Determinants of U.S Foreign Investment: An Empirical Examination", <u>Management International</u> Review, Vol. 2/3, (1975), PP. 113 - 120.

- Green, R.T. & J.M. Lutz, "U.S. High Technology Import/ Export Performance in Three Industries", <u>Journal of International</u> <u>Business Studies</u>, Vol. 11, (Spring 1980), PP. 112 - 117.
- Green, P. & D. Tull, <u>Research for Marketing Decision</u>, 4 edition, Prentice - Hall, Inc., Englewood Cliffs, New Jersey, (1978).
- Gripsrud, G. "The Determinants of Export Decisions and Attitudes to a Distant Market Norwegian Fishery Exports to Japan", <u>Journal of</u> <u>International Business Studies</u>, Vol, 21, No. 3, (1990), PP. 469 -485.
- Groke, P.O. & J.R. Kreidle, "Export ! Why or Why not ? Managerial Attitudes and Action for Smaller Sized Business Firms", <u>Business</u> and Society, Vol. 8, (Fall 1967), PP. 7 - 12.
- Gronhang, K. & T. Lorenzen, "Exploring Industrial Export Strategies", In <u>Assessment of Marketing Thought and Practice</u>, American Marketing Association Proceedings, Vol. 18 (1982), PP. 294 - 8.
- Gronhang, K. & T. Lorenzen, "Exploring the Impact of Governmental Export Subsides", <u>European Journal of Marketing</u>, Vol. 17, (1985) PP. 5 - 15.
- Haberler, G., <u>Theory of International Trade</u>, New York: Macmillan, (1937).
- Haberler, G., <u>A Survey of International Trade Theory</u>, Revised edition Princeton University, International Finance Section, (July 1961).
- Hackett, D.W., "A Comparison of Small Export and Domestic Oriented Manufacturers' Attitudes Regarding the National and International Marketplace", Ph.D dissertation, University of Oklahoma, (1973).
- Hackett, D.W., "Penetrating International Markets: Key Consideration for Smaller Firms", <u>Journal of Small Business Management</u>, Vol. 15 (January 1977), PP. 10 - 16.
- Hair, J., R. Anderson & J. Bernice, <u>An Introduction to Applied</u> <u>Multivariate Statistics</u>, Virginin Beach, VA: Dynamic Press, (1974).
- Hair, J., R. Anderson & R. Tatham, <u>Multivariate Data Analysis</u>, 2nd edition, U.S. Macmillan Publishing Company, (1984).
- Hair, J., R. Anderson & R. Tatham, <u>Multivariate Data Analysis</u> <u>With</u> <u>Reading</u>, Macmillan - New York, (1987).
- Hair, J., R. Anderson, R. Tatham & B. Grablowsky, <u>Multivariate Data</u> Analysis, Oklahoma: Petroleum Publishing Company, (1979).
- Hecksher, E., <u>The Effects of Foreign Trade on the Distribution of</u> Income in Ellis and Metzlor, (1919).
- Heeler, R.M. & M.L. Ray, Measure Validation in Marketing", <u>Journal of</u> Marketing Research, Vol. 9, (Nov. 1972), PP. 361 - 370.

- Hibbert, E.P., <u>The Principles and Practice of Export Marketing</u>, London: William Heinemann Ltd., (1985).
- Hibbert, E.P., <u>Marketing Strategy in International Business</u>, London: McGraw - Hill, (1989).
- Hirsch, S., "<u>The Export Performance of Six Manufacturing Industries:</u> A Comparative Study of Denmark, Holland, and Israel, New York: Praeger Publisher, (1971).
- Hirsch, S. & Z. Adar, "Firm Size and Export Performance", World Development, Vol. 2, No. 7, (1974), PP. 41 - 46.
- Hirsch, S. & B. Lev, "Sales Stabilization Through Export Diversification", <u>Review of Economic and Statistics</u>, (August 1971), PP. 270 - 77.
- Holzmuller, H. & H. Kasper, "The Decision Maker and Export Activity: Across - National Comparison of the Foreign Orientation of Australian Managers", <u>Management International Review</u>, Vol. 30, No. 3, (1990), PP. 2127 - 230.
- Holzmuller, H. & H. Kasper, "On a Theory of Export Performance: Personal and Organizational Determinants of Export Trade Activities Observed in Small and Medium - Sized Firms", <u>Management</u> International Review, Vol. 31, (1991), PP. 45 - 70.
- Homans, G., The Human Groups, New York: Harcourt, Brace, (1950).
- Hoogstrat, E.E., "Attack on the Value of the Profit Motive Theories of Business Behaviour, in McGuire, J., <u>Theories of Business</u> Behaviour, Chapter 3, (1964), PP. 14 - 35.
- Hook, R. & M. Czinkota, "Export Activities and Prospects of Hawaiian Firms", <u>International Marketing Review</u>, Vol. 5, No. 4 (1988), PP. 51 - 57.
- Hook, R. & M. Czinkota, "Export Activities and Prospects of Hawaiian Firms", <u>European Journal of Marketing (UK)</u>, Vol. 23, No. 11 (1989), PP. 27 - 35.
- Hovell, P.J. "Export Pricing Policies", <u>District Bank Review</u>, Vol. 8, (Sep. 1967), PP. 34 - 55.
- Howard, D. & D. Borgia, "Exporting and Firm Size: Do Small Exporters Have Special Needs?", <u>Journal of Global Marketing</u>, Vol. 4, No. 1, (1990), PP. 79 - 96.
- Hunt, H.G., "Export Management in Medium Sized Engineering Firm", Journal of Management Studies, Vol. 6, (Feb. 1969), PP. 33 - 44.
- Hunt, H.G., J.D. Froggatt & P.J. Hovell, "The Management of Export Marketing in Engineering Industries", <u>British Journal of</u> Marketing, Vol. 1, (Spring 1967), PP. 10 - 24.

- Huu-Phuong, T. & T. Soo-Jiuan, "Export Factoring: A Strategic Alternative for Small Exporters in Singapore", <u>International Small</u> <u>Business Journal</u>, Vol. 8, No. 3 (1989) PP. 49 - 57.
- Jackson, B., <u>Multivariate Data Analysis: An Intermediate</u>, Illinois: Richard D. Irwin, Inc., Homewood, (1983).
- Jacoby, J. & M. Matell, "Three Point Likert Scales are Good Enough", Journal of Marketing Research, (1971), PP. 495 - 500.
- Jain, S.C., <u>International Marketing Management</u>, 2nd edition, Boston: Kent Publishing Co., (1982).
- Jain, S.C., <u>International Marketing Management</u>, 3rd edition, Boston: Kent Publishing Co., (1990).
- Jain, A.K., C. Pinson & B.T. Ratchford, <u>Marketing Research: Applica-</u> tions and <u>Problems</u>, New York: John Wiley and Sons, (1982).
- Jangho, L., "Strategic Choice, Foreign Involvement and Firm Performance", Ph.D dissertation, University of Washington, (1986).
- Jatusripitak, S., "Exporting Behaviour of the Firm: A Study of the Decision Making Process in US Manufacturing Firms", Ph.D dissertation, Northwestern University, (1984).
- Johanson, J. & J. Vahlne, "The Internationalization Process of the Firm: A Model of Knowledge Development and Increasing Foreign Commitments", Journal of International Business Studies, Vol. 8, (Spring/ Summer 1977), PP. 23 - 32.
- Johanson, J. & F. Wiedersheim Paul, "The Internationalization of the Firm - Four Swedish Cases", <u>The Journal of Management Studies</u>, (Oct. 1975), PP. 305 - 322.
- John, D. & H. Lee, <u>International Business Environment and</u> <u>Operations</u>, Fifth edition, Addison - Wesley Publishing Company (1989).
- John, G. & D. Roedder, Reliability Assessment: Coefficient Alpha and Beta". In the Changing Market Environment: <u>New Theories and</u> <u>Application</u>, Edited by Berhardt et al, Chicago: American Marketing Association, (1981).
- Johnston, J., <u>Econometric Methods</u>, 2nd edition, New York: McGraw Hill, (1972).
- Johnston, W. & M. Czinkota, "Managerial Motivations as Determinant of Industrial Export Behaviour". In <u>Export Management: An</u> <u>International Context</u> edited by Czinkota, M. & G. Tesar, Praeger: New York, (1982), PP. 3 - 17.
- Johnston, W. & M. Czinkota, "Export Attitudes of Industrial Manufacturers", <u>Industrial Marketing Management</u>, Vol. 14, (1985), PP. 113 - 32.

- Jordan, Department of Statistics, <u>External Trade Statistics</u>, Vol. 1, (1989).
- Jordan, Department of Statistics, <u>The Jordan Economy in Figure</u>, (Nov. 1987).
- Jordan, Department of Statistics, <u>Statistical Year Book</u>, No. 40, (1989).
- Jordan, Ministry of Planning, <u>Five Year Plan for Economic and social</u> <u>Development</u> (1986 - 1990), (1986).
- Jordan Legislation Encyclopedia, <u>The Function of the Chamber of</u> <u>Industry and Trade</u>, Vol. 18, No. 14, of Year 1949 and Law No. 21 of Year 1961.
- Joynt, P., "An Empirical Study of Norwegian Export Behaviour". In <u>Export Management: An International Context</u>, edited by Czinkota, M. & G. Tesar, New York, (1982), PP. 55 - 59.
- Joynt, P. & C. Welch, "A Strategy for Small Business Internationalization" <u>International Marketing Review</u>, Vol. 2. No. 3, (Autumn 1985), PP. 61 - 73.
- Junt, M. & P. Walters, "The Internationalization of Norwegian Firms: A Study of the U.K Experience", <u>Management International Review</u>, Vol. 27, (1987), PP. 58 - 67.
- Kachigan, S.K., <u>Statistical Analysis: An Interdisciplinary Intro-</u> <u>duction to Univariate and Multivariate Methods</u>, New York: Radius Press, (1982).
- Kacker, M.P., "Export Oriented Product Adaptation It Patterns and Problems", <u>Management International Review</u>, (1975/1976), PP. 111 - 114.
- Kanovsky, E., <u>Jordan's Economy from Prosperity to Crisis</u>, The Moshe Dayan Centre for Middle Eastern and African Studies, The Shillouh Institute, Tel Aviv Univ., Israel, (1989).
- Karafakioglu, M., "Export Activities of Turkish Manufacturers", International Marketing Review, (Winter, 1986), PP. 34 - 43.
- Karafakioglu, M. & T. Harcar, "Internal Determinants Affecting Interest in Exporting: An Empirical Analysis of Turkish Non -Exporting Companies", <u>Journal of Global Marketing</u>, Vol. 3, No. 4, (1990), PP. 99 - 111.
- Kathawala, Y., R. Judd, M. Monipallil and M. Weinrich, "Exporting Practices of Illinois Firms", <u>Journal of Small Business</u> <u>Management</u>, (January 1989), PP. 53 - 59.
- Kaynak, E., "Internationalization of Nova Scotia Manufacturers Export for Survival", <u>Der Market Nr</u>, Vol. 84, (1982), PP. 137 - 146.

Kaynak, E., Marketing in the Third World, New York: Praeger, (1982).

- Kaynak, E., "Correlates of Export Performance in Resource Based Industries". In <u>Global Perspectives in Marketing</u>, Edited by Kaynak, E., New York et al, (1985).
- Kaynak, E., "Across Regional Comparison of Export Performance of Firms in Two Canadian Regions", <u>Management International Review</u>, Vol. 32, No. 2, (1992), PP. 163 - 180.
- Kaynak, E., P. Ghuri & T. Olofsson-Bredenlow, "Export Behaviour of Small Swedish Firms", <u>Journal of Small Business Management</u>, Vol. 25, No. 2, (Winter 1987), PP. 26 - 32.
- Kaynak, E. & M.N. Gurol, "Export Management in Less Developed Countries: A Case Study of Turkey in Light of the Japanese Experience", <u>Management International Review</u> Vol. 22, (1987), PP. 54 - 65.
- Kaynak, E. & B. Hudanah, "Operationalising the Relationship Between Marketing and Economic Development: Some Insight From Less - Developed Countries", <u>European Journal of Marketing</u>, Vol. 21, No. 1, (1987), PP. 48 - 65
- Kaynak, E. & V. Kothari, "Export Behaviour of Small and Medium -Sized Manufacturers: Some Policy Guidelines for International Marketers", <u>Management International Review</u>, Vol. 24, No. 2, (1984), PP. 16 - 19.
- Kaynak, E. & V. Kothari, "Export Behaviour of Small Manufacturers: A Comparative Study of American and Canadian Firms", <u>European</u> <u>Management Journal</u>, Vol. 2, (1984), PP. 41 - 47.
- Kaynak, E. & V. Kothari, "Export Behaviour of Small and Medium Sized Manufacturers", <u>Journal of Small Business Management</u>, (April 1987), PP. 26 - 32.
- Kaynak, E. & A. Samli, "Marketing Practice in Less Developed Countries", <u>Journal of Business Research</u>, Vol. 13, No. 1, (March 1984).
- Kaynak, E. & L. Stevenson, "Export Orientation of Nova Scotia Manufacturers". In M. R. Czinkota and G. Tesar (eds) <u>Export</u> <u>Management: An International Context</u>, New York: Praeger, (1982).
- Keats, W.B. & M.A. Hitt, "A Causal Model of Linkage Among Environmental Dimensions, Macro Organisational Characteristics and Performance", <u>Academy of Management Journal</u>, (Sep. 1986), PP. 572 -593.
- Kedia, B.L. & J. Chhokar, "Managerial Attitudes of Exporters and Non - Exporters and Their Impact on Export Behaviour of Small and Medium - Sized Firms in Two Industries: Paper Presented in the Academy of Management National Convention, San Diego, California, (August 1985).
- Kedia, B.L. & J. Chhokar, "Factors Inhibiting Export Performance of Firms: An Empirical Investigation", <u>Management International</u> <u>Review</u>, Vol. 26, No. 4, (1986), PP. 33 - 43.

- Keegan, W.J. "Five Strategies for International Marketing", <u>European</u> <u>Business</u>, (January 1970), PP. 35 - 40.
- Keegan, W.J. <u>"Multinational Marketing Management</u>, 2nd edition, Englewood Cliffs, New Jersey: Prentice - Hall, (1984).
- Keesing, D.B., <u>Trade policy for Developing Countries</u>, World Bank Staff Working Paper No. 353, (August, 1979).
- Keng, K. & T. Jiuan, "Differences between Small and Medium Sized Exporting and Non - Exporting Firms: Nature or Nurture", <u>International Marketing Review</u>, Vol. 6, No. 4, (1989), PP. 27 -39.
- Kerlinger, F., Foundations of Behavioural Research, CBC Publishing Company: New York, (1986).
- Kerlinger, F. & N. Pehhazur, <u>Multiple Regression in Behavioural</u> <u>Research</u>, New York: Holt, Rinhart and Winston, Inc., (1973).
- Khan, M.S., "A Study of Success and Failure in Exports in the Swedish Export Industry", Research Report, Department of Business Administration, University of Stockholm, (1975).
- Khan, M.S., <u>A Study of Success and Failure in Exports</u>, Department of Business Administration, University of Stockholm, (1978).
- Kim, J. & C. Muller, <u>Introduction to Factor Analysis What it is and</u> <u>How to do it</u>, Series Quantitative Application in Social Science, Sage Publication, Inc. U.S., (1978).
- Kindelberger, C.P., <u>Foreign Trade and National Economy</u>, New Haven, Conn: Yale University Press (1962).
- Kindleberger, C.P., <u>American Business Abroad</u>, Six Lectures on Direct investment, New Haven: Yalf University Press, (1969).
- Kindelberger, C.P., <u>International Economics</u>, Fifth edition, Richard D., Irwin, Homewood, Illinois, (1973).
- Kindra, G.S. (ed.), <u>Marketing in Developing Countries</u>, London: Croom Helm, (1984).
- Kinnear, T. & J. Taylor, <u>Marketing Research: Applied Approach</u>, 4th ed., New York: McGraw-Hill, (1991).
- Kinsey, J., <u>Marketing in Developing Countries</u>, Hong Kong: Macmillian Education Ltd., (1988).
- Kirpalani, V. & N. Macintosh, "International Marketing Effectiveness of Technology Oriented Small Firms", Journal of International <u>Business Studies</u>, (Winter 1980), PP. 81 - 90.
- Kizilbash, A. & C. Maile, "Export Behaviour in a Changing Economic Environment", <u>Journal of Small Business Management</u>, Vol. 15, (1977), No. 1, PP. 1 - 6.

- Klecka, W.R., <u>Discriminant Analysis</u>. In SPSS, McGraw-Hill, 2nd ed., (1975), PP. 434 467.
- Klecka, W.R. <u>Discriminant Analysis</u>, London: Sage Publication, Inc., Beverly Hills, (1980).
- Kleinshmidt, E.J., "Export Strategies, Firm Internal Factors and Export Performance of Industrial Firms: A Canadian Empirical Analysis", Ph.D dissertation, McGill University, (1982).
- Kleinschmidt, E.J., <u>The Impact of Foreign Ownership on the Export</u> <u>Behaviour of Industrial Firms</u>, Canada, McMaster University, (1986).
- Kneith, D., L. Robert & L. Blomstrom, <u>Business and Its Environment</u>, London: McGraw - Hill Book Co., (1966).
- Knight, R., "Export as a Determinant of Success for Canadian High Technology Entrepreneur", Working Paper Series, No. 89 - 07, Research & Publication: The School of Business Administration The University of Western Ontario, (March 1989).
- Kobrin, S. "The Environment Determinants of Foreign Direct Investment An Ex-Post Empirical Analysis", <u>Journal of International Business</u> <u>Studies</u>, (Fall 1976).
- Koh, A., "Relationships Among Organisational Characteristics, Marketing Strategy and Export Performance", <u>International</u> <u>Marketing Review</u>, Vol. 8, (1991), PP. 46 - 60.
- Koglmayr, M. & S. Muller, "The Export Decision of Small and Medium Sized Firms: A Review", <u>Management International Review</u>, Vol. 24, No. 2, (1984), PP. 49 - 60.
- Kolhede, E., "The Underlying Determinants of Export Orientation of Small Manufacturing Firms", Ph.D dissertation, Golden State University, (1984).
- Korth, C.M., "Managerial Barriers to U.S. Exports", <u>Business Horizons</u> (March- April 1991), PP. 18 - 26.
- Kothari, V., "International Marketing Opportunities for Small Business Firms", <u>The SFA Business Review</u>, Vol. 3, No. 1, (1977), PP. 15 -20.
- Kreinin, M.E., <u>International Economics</u>, New York: Harcourt, Brace, Jovanich, (1971).
- Langston, C.M., "An Analysis of the International Orientation of Chief Executives of Exporting and Non - Exporting Manufacturers", Ph.D dissertation, The University of Oklahoma, (1976).
- Langston, C.M. & R.K. Teas, "Export Commitment and the Characteristics of Management", Paper Presented at the Annual Meeting of the Midwest Business Association, St Louis, Missouri, (1976).
- Laurie, J. Mullins, <u>Management and Organisational Behaviour</u>, 2nd ed., London: Pitman Publishing, (1989).

- Lawrence, P. & J. Lorsch, <u>Organisation and Environment</u>, Division of Research, Graduate School of Business Administration, Harvard University, Boston, (1967).
- Layton, R.A. & D.C. Dumphy, "A Summary Report for the Development Council on Export Attitudes", <u>Management Practices and Marketing</u> <u>Skills</u>, University of New South Wales, (1970).
- Lee, C.S., "Comparative Marketing Practices of Foreign and Domestic Firms in Developing Countries: A Case Study of Malaysia", <u>Management International Review</u>, (1973/1974), PP. 91 -102.
- Lee, W.Y. & J.J. Brasch, "The Adoption of Export as an Innovative Strategy", <u>Journal of International Business Studies</u>, (Summer 1978), PP. 62 - 73.
- Lee, C.S. & Y. Yang, "Impact of Export Market Expansion Strategy on export Performance", <u>European Journal of Marketing</u>, Vol. 25 (1991), PP. 29 - 39.
- Lehmann, D.R., <u>Market Research and Analysis</u>, 3rd edition, Richard and Urwin, U.S, (1987).
- Lehmann, R. & J. Hulbert, "Are Three Point Scale Always Good Enough", <u>Journal of Marketing Research</u>, Vol. 9, (1972), PP. 440 -446.
- Leighton, D.S., "The Internationalisation of American Business", Journal of Marketing, Vol. 34, (July, 1970), P.5
- Lenz, R.T., Environment, Strategy, Organisation Structure and Performance: Patterns in One Industry", <u>Strategic Management</u> <u>Journal</u>, Vol. 1, (1980), PP. 209 - 226.
- Linder-Burenstam, S., <u>An Essay on Trade and Transformation</u>, New York John Wiley & Sons, (1961).
- Livtak, A. & P. Banting, "A Conceptual Framework for International Business Arrangements". In <u>Marketing and the New Science of</u> <u>Planning</u>, Robert L. King, (ed.) (Chicago: American Marketing Association (1968), PP. 460 - 467.
- Liouville, J., "Under What Conditions Can Exports Exert a Positive Influence on Profitabilities", <u>Management International Review</u>, Vol. 2, No. 1, (1992), PP. 41 - 54.
- Loeinger, T., "The Technology Factor and the Export Performance of U.S. Manufacturing Industry", <u>Economic Inquiry</u>, Vol. X111, (June 1975), PP. 221 - 236.
- Louter, P., C. Ouwerkerk & B. Bakker, "An Inquiry into Successful Exporting", <u>European Journal of Marketing</u>, Vol. 25, No. 6, (1991), PP. 7 - 23.
- Luck, D. & R. Rubin, <u>Marketing Research</u>, Prentice Hall Inc., 7th ed. Englewood Cliffs, New Jersy, (1987).

- Maclayton, D., M. Smith & J. Hair, "Determinants of Foreign Market Entry: A Multivariate Analysis of Corporate Behaviour", <u>Management</u> <u>International Review</u>, (1980), PP. 40 -53.
- Madsen, T., "Successful Export Marketing Management: Some Empirical Evidence", <u>International Marketing Review (UK)</u>, Vol. 6, Part 4, (1989), PP. 41 - 57.
- Magnas, B. & R.E. Lipsey, "The Export Performance of U.S. and Swedish Multinationals", Research Paper 6361, The University of Warwick, (1989).
- Malekzadeh, A., "An Empirical Study of the Export Perceptions of California Manufacturers", Ph.D dissertation, The University of Utah, (1982).
- Malekzadeh, A. & A. Nahavandi, "Small Business Exporting: Misconceptions Are Abundant", <u>American Journal of Small Business</u>, Vol. 9, No. 4, (Spring 1985), PP. 7 - 14.
- Malekzadeh, A. & S. Rabino, "Manufacturers' Export Strategies", <u>International Marketing Review</u>, Vol. 3, No. 4., (1986), PP. 27 -33
- Malgrem, H., "Information, Expectation and the Theory of the Firm", <u>Quarterly Journal of Economic</u>, Vol. LXXXVI, (August, 1961), PP. 399 - 421.
- March, J. & H. Simon, Organisation, New York: Wiley, (1958).
- Marshall, A., The Pure Theory of Foreign Trade, Published privately reprinted together with: <u>The Pure Theory of Domestic Values</u>. Gordon, Gordon School of Economics and Political Science, (1930).
- Mason, R.H., R.R. Miller, & E.R. Weigel, <u>The Economics of Interna-</u> tional Business, John Wiley and Son Inc, New York, (1975).
- Mayer, C.S. & J.E. Flynn, "Canadian Small Businesses Abroad: Opportunities, Aids and Experiences", <u>Business Quarterly</u>, (Winter 1973), PP. 33 - 47.
- McConnell, J.E., "The Export Decision: An Empirical Study of Firm Behaviour", <u>Economic Geography</u>, Vol. 55, No. 3, (1979), PP. 171.
- McCracken, P.W., "America's Expanding Foreign Trade", <u>The Wall Street</u> Journal, Vol. 20 (September 1973), P. 12.
- McDonald, M., "International Marketing Planning: The Gap Between Theory and Practice", <u>International Marketing Review</u>, Vol. 1, Part 1, (Autumn 1983), PP. 42 - 58.
- McDougall, G.H. & B.W. Stening, "Something to Think About: Identifying the High Performance Exporters", <u>Canada Commerce</u> (Dec. 1975), pp. 12 - 15.
- McFarlane, G., "Scots Queen Awards Winners Don't Excel", <u>Marketing</u>, (April 1978).

- McGuire, J., <u>Theories of Business Behaviour</u>, Englewood Cliffs, New Jersey: Prentice - Hall, (1964).
- McGuinness, N., "The Impact of Technology and Product Characteristics on the International Sales of New Canadian Industrial Products: A Diffusion Analysis", Unpublished Ph.D Dissertation, University of Western Ontario, Canada, (Feb. 1978).
- McGuinness, N. & B. Little, "The Influence of Product Characteristics on the Export Performance of New Industrial Products", <u>Journal of Marketing</u>, Vol. 45, (Spring 1981), PP. 8 - 16.
- Meidan, A., "Nonmetric Multidimensional Analysis of Export Marketing Management", Ph.D Thesis, University of Bradford, (1974).
- Meidan, A., "The Export Marketing Matrix A Tool For Investigating the Marketing Concept in Exporting", <u>Management International</u> <u>Review</u>, (1975/76), PP. 51 - 59.
- Michell, P., "Infrastructure & International Marketing Effectiveness" <u>Columbia Journal of World Business</u>, Vol. 4. (Sep./Oct. 1979), PP. 29 - 38.
- Mintzberg, H., <u>The Nature of Managerial Work</u>, New York: Harper and Row, (1973).
- Ministry of Finance & Customs, <u>Jordan's Customs and Excise</u> Law No. 1, (1962), PP. 1 - 20.
- Ministry of Industry and Trade, <u>Export Regulations</u>, No. 66 of 1979, Trade Publications, Amman (1979).
- Monro, H. & P. Beamish, "Distribution Methods and Export Performance". In Rosson, P. & S. Reid (ed), <u>Managing Export Entry and Expansion:</u> <u>Concepts and Practice</u>, Praeger, New York, (1987), PP. 316 - 34.
- Moon, J. & H. Lee, "On the Internal Correlates of Export Stage Development: An Empirical Investigation in the Korean Electronics Industry", <u>International Marketing Review</u>, Vol. 7, No. 5, (1990), PP. 16 - 26.
- Morrison, D., "On the Interpretation of Discriminant Analysis, Journal of Marketing Research, Vol. 6, (1969), PP. 156 - 163.
- Morrison, D., <u>Multivariate Statistics Methods</u>, McGraw Hill, New York, (1990).
- Morrison, T., <u>Manufactured Export from Developing Countries</u>, New York, London: Praeger Publishers, (1976).
- Muir, A., <u>Pricing Policies on Export Marketing in Successful Export</u> <u>Strategy</u>, Graham and Tratmn, London, (1977).
- NEDO, (National Economic Development Office), <u>International Price</u> <u>Competitiveness Non - Price Factors and Economic Performance</u>, NEDO London, (1977).

- Neidell, L.A., "Comparative Export Practices of Small Firms in the U.S and Scandinavian", <u>Research Report to Federal Reserve Bank of Boston</u>, (1965), No. 29.
- Neidell, L.A., "The Ingredients for Successful Exporting", <u>Marquette</u> <u>Business Review</u>, Vol. 25, (1971), PP. 31 - 44.
- Nunnaly, J.C., <u>Psychometric Theory</u>, New York: McGraw Hill Book Company, (1967), PP. 188 - 226.
- Ohlin, B., Inter Regional and Internationalisation, Oxford: The University Press, (1933).
- Olson, Ch. H. & F. Wiedersheim Paul, "Factors Affecting the Pre - Export Behaviour of Non - Exporting Firms". In <u>European Research</u> <u>in International Business</u>", edited by Ghertman, M. & J. Leontiads, (1978), PP. 283 - 305.
- Ogram, E., "Exporters and Non Exporters: A Profile of Small Manufacturing Firms in Georgia". In Czinkota, M. and G. Tesar (eds), Export Management, Praeger, New York, (1982), PP. 70 - 84.
- Ong, C.H. & A.W. Pearson, "The Impact of Technical Characteristics on Export Activity: A Study of Small and Medium - Sized Electronic Firms", <u>R & D Management</u>, Vol. 12, (1982), 189 - 196.
- O'Rourke, A.D., "Differences in Exporting Practices, Attitudes and Problems by the Size of Firms", <u>American Journal of Small</u> <u>Business</u>, Vol. 9, Part 3, (1985), PP 25 - 29.
- Ortiz-Buonafina, M., "Small Business Exporting", <u>Journal of Global</u> <u>Marketing</u>", Vol. 3, No. 4, (1990), PP. 33 - 57.
- Osborn, R.N. & J.G. Hunt, "Government and Organisational Effectiveness", Administrative Science Quarterly, (1974), PP. 231 - 232.
- Pagonlatos, E. & R. Sorensen, "Domestic Market Structure and International Trade: An Empirical Analysis", <u>Quarterly Review</u> of Economic and Business, Vol. 16, (1976), PP. 45 - 59.
- Papandreou, A., "Some Basic Problems in the Theory of the Firm". In a Survey of Contemporary Economic, edited by B. Haley, Homewood, Richard D Irwin, (1952).
- Parasuraman, A., <u>Marketing Research</u>, Addison Wesley Publishing Co. Inc U.S., (1986).
- Pavord, W.C. & R.G. Bogart, "The Dynamics of the Decision to Export", Akron Business and Economic Review, (Spring 1975), PP.6 -11.
- Pennings, J.M., "Coordination Between Strategically Independent Organisation". In <u>Handbook for Organisational Design</u>, Nystrom, P. & W. Starback (ed), Amsterdam: Elsever, (1979).

- Perkett, W.O., "An Analysis of the Obstacles to Increased Foreign Trade which Confront British Columbia Industrial Machinery Manufacturers" Ph.D Dissertation, University of Washington, Seattle, Washington, (1963).
- Perlmutter, H.V., "The Tortuous Evolution of the Multinational Corporation", Columbia Journal of World Business, (1969), PP. 9 - 15.
- Perreault, W.D., D.N. Behrman & G.M. Armstrong, <u>"Alternative Approaches for Interpretation of Multiple Discriminant Analysis in Marketing Research"</u>, Journal of Business Research, (1979), PP. 151 173.
- Peter, J. Paul, "Reliability: A Review of Psychometric Basics and Recent Marketing Practice "Journal of Marketing Research, (Feb. 1979), PP. 6 - 17.
- Peter, J. Paul, "Construct Validity: A Review of Basic Issues and Marketing Practices, <u>Journal of Marketing Research</u>, Vol. 19, (May 1981), PP. 133 - 45.
- Philip, R., "Industry Review, Jordan", <u>Arab Industry Review</u>, USA Dallas Inter Crescent Publication, (1986), PP. 243 252.
- Philips, L.W., "The Study of Collective Behaviour in Marketing: Methodological Issues in the Use of Key Informant", Ph.D dissertation, Northwestern University, Illinois, (1980).
- Philips, L.W., "Assessing Measurement Error in Key Informants Reports: A Methodological Note on Organizational Analysis in Marketing", Journal of Marketing Research, (November 1981), PP. 395 - 415.
- Philpot, N., "Managing the Export Function", <u>Management Survey</u> <u>Report</u>, No. 6, British Institute of Management, London, (1975).
- Piercy, N., "Company Internationalization: Active and Reactive Exporting", <u>European Journal of Marketing</u>, Vol. 13, No. 3 (1981), PP. 26 - 40.
- Piercy, N., "Export Marketing Management in Medium Sized British Firms", <u>European Journal of Marketing</u>, Vol. 17, No. 1, (1983), PP. 48 - 67.
- Pinney, J.K., "The Process of Commitment to Foreign Trade, Selected Smaller Indiana Manufacturing Firms", Ph.D dissertation, Indiana University, (1968).
- Pinney, J.K., "Process of Commitment to Foreign Trade", <u>Bulletin</u> Published by the Indiana Department of Commerce, Indianapolis, (1970).
- Pinney, J.K., "Obstacles to Foreign Trade of 209 Indiana Manufacturers", <u>Bulletin Published By</u> the Indiana Department of Commerce Iniannpolis, Indiana, (1971).

- Poynter, T.A., "Government Intervention in Less Developed Countries: The Experience of Multinational Companies", <u>Journal of Intern-</u> <u>ational Business Studies</u>, (Spring/ Summer, 1982).
- Rabino, S., "DISC Contribution to Improved Competitiveness of Export Market" A Study of American Manufacturing Exports", Ph.D Dissertation, University of New York, (1979).
- Rabino, S., "An Examination of Barriers to Exporting Encountered by Small Manufacturing Companies", <u>Management International Review</u>, Vol. 20, No. 1, (1980), PP. 67 - 73.
- Rabino, S., "An Attitudinal Evaluation of an Export Incentives Program: The Case of DISC", <u>Columbia Journal of World Business</u>, Vol. 15, (Spring 1980), PP. 61 - 65.
- Rao, C.P., "Export Marketing for Regional Export Expansion", Arkaness Business and Economic Review, (Summer 1974), PP. 20 - 33.
- Rao, C.P., "Impact of the Domestic Economic Environment on International Business Involvement and Practise of Manufacturer - Exporters", Paper Presented at the Academy of International Business Annual Meeting, USA, Orlando Florida, (August 1977), PP. 174 - 7.
- Rao, C.P., M.K. Erramilli & G.K. Ganesh, "Impact of Domestic Recession on Export Marketing Behaviour", <u>International Marketing Review</u>, Vol. 7, (1990), PP. 54 - 68.
- Rao, C.P. & D.D. Weinrauch, "Significance of Changes in the Export Trade Environment as Perceived by Exporters and Potential Exporters", <u>Baylor Business Studies</u>, Vol. 96, (1973), PP. 23 - 40.
- Reid, S.D., "The Contextual Environment for Foreign Entry: Theoretical & Methodological Issues", Proceedings of the Southern Marketing Association, Wichita Atate University, (1980).
- Reid, S.D., " A Behavioural Approach to Export Decision Making", American Marketing Association Proceedings (1980), PP. 265 - 268.
- Reid, S.D., "Export Behaviour in the Small Canadian Owned Manufacturing Enterprise - An Empirical Investigation", Ph.D Thesis, (1981), York University.
- Reid, S.D., "The Decision Maker and Export Entry and Expansion", <u>Journal of International Business Studies</u>, Vol. 12, (Fall 1981), PP. 101 - 112.
- Reid, S.D., "The Impact of Size on Export behaviour in Small Firms". In Czinkota, M. R. and Tesar, G. (eds), <u>Export Management</u>, New York, Praeger, (1982), PP. 18 -38.
- Reid, S.D., "Managerial and Firm Influences on Export Behaviour", <u>Journal of the Academy of Marketing Science</u>, Vol. 2, No. 3, PP. (1983), PP. 323 - 332.

- Reid, S.D., "Information Acquisition and Export Entry Decisions in Small Firms", <u>Journal of Business Research</u>, Vol. 12, (1984), PP. 141 - 157.
- Reid, S.D., "Exporting: Does Sale Volume Make a Difference? -Comment", <u>Journal of International Business Studies</u>, Vol. 16, (1985), PP. 153 - 155.
- Reid, S.D., "Is Technology Linked With Export Performance in Small Firms". In Hubner (ed), <u>The Art and Science of Innovations</u> <u>Management</u>, Elsevier Science Publishers, Amsterdam, (1986), PP. 273 - 83.
- Reid, S.D., Export Structure, Strategy and Performance: An Empirical Study of Small Italian Manufacturing Firms". In P. Rosson and S. Reid (eds) <u>Managing Export Entry and Expansion: Concepts and</u> <u>Practice</u>, New York: Praeger, (1987), PP. 335 - 357.
- Reith, A.R., "A Study of the Perceptions of Selected Small Massacusetts Manufacturers Toward Exporting", Ph.D dissertation, University of Massachusetts, (1976).
- Reith, R. & E.J. Ryan, "A Study of the Perception of Selected Massachusetts Manufacturers Toward Exporting", <u>Development in</u> <u>Marketing Science</u>, Vol. 4, Miami, The Academy of Marketing Science, (1980), PP. 101 - 104.
- Reith, R. & E.J. Rayn, "A Study of the Perceptions of Selected Small Massachusetts Manufacturers Toward Exporting", Developments in Marketing Science, U.S.A, The Academy of Marketing Science, Vol. 14, (1981), PP. 97 - 100.
- Reothlisberger, F.J., <u>Management</u> and <u>Moral</u>, Cambirdly: Harverd University Press, (1941).
- Ricardo, D., <u>On the Principle of Political Economy and Taxation</u>, London: John Murray (1817), In Piero Straffa (ed.), <u>The Works and</u> <u>Correspondence of David Ricardo</u>, Vol. 1, Cambridge, University Press, (1951).
- Rogers, E.M., <u>Diffusion of Innovation</u>, New York: The Free Press, (1961).
- Rojemta, S., "Constraints to Export Trade to Thailand: A Study of Selected California Manufacturing Firms", Ph.D Thesis, State International University, (1986).
- Ronald, F., F. Massy & D. Morrison, "Bias in Multiple Discriminant Analysis", <u>Journal of Marketing Research</u>, Vol. 2, (August 1962), PP. 250 - 8.
- Rosson, P.J. & D. Ford, "Export Marketing Channels", <u>Management</u> <u>International Review</u>, Vol. 20, No. 4, (1980), PP. 31 - 37.
- Rosson, P.J. & D. Ford, "Manufacturer Overseas Distributor Relations and Export Performance", <u>Journal of International</u> <u>Business Studies</u>, Vol. 13, (Fall 1982), PP. 57 - 72.

- Rothchild, K.W., "Price Theory and Oligoply", <u>Economic Journal</u>, Sep. (1947), PP. 289 320.
- Roux, E., "The Export Behaviour of Small and Medium Size French Firms: The Role of the Managers Profile". In Mattsson L. G. and Wiedersheim - Paul, F. (eds), Recent Internationalisation of Business, Uppsala, (1979).
- Roy, D. & C. Simpson, "Attitudes Towards Exporting in the Smaller Manufacturing Firm", <u>Baylor Business Studies</u>, (1980), PP. 35 - 44.
- Roy, D. & C. Simpson, "Export Attitudes of Business Executives in the Smaller Manufacturing Firms", <u>Journal of Small Business Management</u> Vol. 19, Part 2, (April 1981), PP. 16 - 22.
- Rummel, R.J., <u>Applied Factor Analysis</u>, Evanston: Northwestern, University Press, (1970).
- Samiee, S. & P. Walters, "Rectifying Strategic Gaps in Export Management", <u>Journal of Global Marketing</u>, Vol. 4, No. 1, (1990a), PP. 7 - 37.
- Samiee, S. & P. Walters, "Influence of Firm Size on Export Planing and Performance", <u>Journal of Business Research</u>, Vol. 20, (1990b), PP. 230 - 248.
- Samiee, S. & P. Walters, "Segmenting Corporate Exporting Activities: Sporadic Versus Regular Exporters", <u>Journal of the Academy of</u> <u>Marketing Science</u>, Vol. 19, (Spring 1991), PP. 93 - 104.
- Samules, J., S. Greenfield & H. Mpuku, "Exporting and the Smaller Firm", <u>International Small Business Journal</u>, Vol. 10, No. 2, (1992), PP. 24 - 36.
- Schlegelmilch, B., "A Profile Analysis of Divergent Export Behaviour in the U.K. and West German Mechanical Engineering and Food Processing Industries", Ph.D Thesis, University of Manchester (UMIST), (1983).
- Schlegelmilch, B., "Controlling Country Specific and Industry -Specific Influences on Export behaviour", <u>European Journal of</u> <u>Marketing</u>, Vol. 20, No. 2, (1986), PP. 54 - 71.
- Schlegelmilch, B.B. & J.N. Crook, "Firm Level Determinants of Export Intensity", Annual Meeting of the Academy of International Business, London, November, (1986), PP. 21 -23.
- Schlegelmilch, B.B. & A. Diamantopoulas, "Innovation and Export Performance: An Empirical Analysis", Working Paper 86/B, University of Edinburgh, Department of Business Studies, (1986).
- Scott, W.R., Organisation, Englewood Cliffs: Prentice Hall, (1981).
- Schollhammer, H. & D. Nigh, "The Effects of Political Events on Foreign Direct Investment by German Multinational Corporations", <u>Management International Review</u>, No. 2, (1984), PP. 5 - 27.

- Schore, R., <u>The Pure Theory of International Trade</u>, Macmillan, Basingstoke, 1972.
- Seely, R. & H. Iglarsh, "International Marketing in the Context of the Small Business", <u>American Journal of Small Business</u>, Vol. 6, No. 2, (1981), PP. 33 - 77.
- Seidler, J., "On Using Informants: A Technique for Collecting Quantitative Data and Controlling for Measurement Error in Organisation Analysis", <u>America Sociological Review</u>, Vol. 39, (Dec. 1974), PP. 816 - 831.
- Seifert, B. & J. Ford, "Are Exporting Firms Modifying their Product, Pricing and Promotion Policies", <u>International Marketing Review</u>, Vol. 6, Part 6, (1989), PP. 35 - 68.
- Sharkey, W., J. Lim & K. Kim, "Export Development and Perceived Export Barriers: An Empirical Analysis of Small Firms", <u>Management</u> <u>International Review</u>, Vol. 29, No. 2, (1989), PP. 33 - 40.
- Sharman, G.H., <u>Thinking Managerially About Export</u>, London, Institute of Export (1971).
- Sharrard, G.F., "Understanding the Environment of New Business Ventures". In <u>Successful Product and Business Development</u>, Edited by Newman H. Giragosian, New York: Marcel Dekker Inc. (1975), PP. 99 - 114.
- Sheth, J., "The Multivariate Revaluation in Marketing Research", Journal of Marketing, Vol. 35, (January 1971), PP. 13 - 19.
- Siegel, S., Non Parametric Statistics for the Behavioural Sciences, New York: McGraw-Hill, (1956)
- Siegel, S., "Level of Aspirations and Decision Making", Psychological Review, Vol. 64, (1957), PP. 253 - 62.
- Simon, H.A., <u>Administrative Behaviour</u>: A Study of the Decision -Making Process Administrative Organisation, second edition, Free Press, New York, (1947).
- Simmonds, K. & H. Smith, "The First Export Order: A Marketing Innovation", <u>British Journal of Marketing</u>, Vol. 5, (1968), PP. 93 - 100.
- Simpson, C., "The Export Decision: An Interview Study of the Decision Process in Tennessee Manufacturing Firms" Ph.D Dissertation Georgia State University, Atlanta, Georgia, (1973).
- Simpson, C. & D. Kujawa, "The Export Decision Process: An Empirical Inquiry", <u>Journal of International Business Studies</u>, (Spring 1974), PP. 107 - 117.
- Sinai, C.C., "An Investigation of Selected Characteristics of Export Practicing Manufacturing Firms", Ph.D dissertation, University of Washington, (1970).

Smith, A., The Wealth of Nations, New York: Modern Library, (1937).

- Snavely, W.P. & P. Weiner, H. Ulbrich & E. Enright, "Export Survey of the Greater Harford Area", Vols. 1 & 2, The University of Connecticut, Storrs, Connecticut, (1964).
- SPSS, <u>Statistical Package for the Social Sciences</u>, New York: McGraw -Hill, Inc., Second Edition, (1975).
- SPSS, Users Guide SPSS X, New York: McGraw Hill Book Co. (1983).
- SPSS, <u>SPSS X Advanced Statistics Guide</u>, McGraw Hill Chicago, (1990).
- Sood, J. & P. Adams, "A Model of Management Learning Styles as a Predictor of Export Behaviour and Performance", <u>Journal of</u> <u>Business Research</u>, Vol. 12, No. 2, (April 1984), PP. 169 - 182.
- Souza, L.M., A. Schmidt & J.L. Colaiacovo, "Pre Export Behaviour: An Analysis of the Variables Influencing the Decision Process, In M. R. Czinkota, (ed.), <u>Export Promotion:</u> <u>The Public and Private</u> <u>Sector Interaction</u>, New York: Praeger Publishers, (1983).
- Sriram, V., J. Neelankavil & R. Moore, "Export Policy and Strategy Implications for Small - to - Medium Sized Firms", <u>Journal of Global Marketing</u>, Vol. 3, No. 2, (1989), PP. 43 - 75.
- Sriram, V. & H. Sapienza, "An Empirical Investigation of the Role of Marketing for Small Exporters", <u>Journal of Small Business</u> <u>Management</u>, (October 1991), PP. 33 - 43.
- Strauess, J.H., "The Entrepreneur: The Firm", <u>Journal of Political</u> <u>Economy</u>, Vol. 52, (June 1944), PP. 122 - 127.
- Sullivan, D. & A. Bauerschmidt, "A Common Factors Underlying Incentives to Export: Studies in the European Forest Products Industry", <u>European Journal of Marketing</u>, Vol. 22, (1988).
- Sullivan, D. & A. Bauerschmidt, "A Common Factors Underlying barriers to export: A Comparative Study in the European and U.S. Paper Industry", <u>Management International Review</u>, Vol. 2, (1989) PP. -32.
- Sullivan, D. & A. Bauerschmidt, "Incremental Internationalization: A Test of Johanson and Vahlne's Thesis", <u>Marketing International</u> <u>Review</u>, Vol. 30, (1990), PP. 19 - 30.
- Suzman, G.L. & L.H. Wortzel, "Technology Profiles and Export Marketing Strategies", <u>Journal of Business Research</u>, Vol. 12, (1984), PP. 183 - 194.
- Tabachnick, B & L. Fidell, <u>Using Multivariate Statistics</u>, New York: Harper and Row Publishers, (1983).
- Tesar, G., "Empirical Study of Export Operations Among Small and Medium - Sized Manufacturing Firms", Ph.D Thesis, University of Wisconsin, (1975).

- Tesar, G., "Identification of Planning, Attitudinal & Operational Differences Among Types of Exporters", <u>American Journal of Small</u> <u>Business</u>, Vol. 11, No. 2, (1977), PP. 16 - 21.
- Tesar, G. & J. Tarleton, "Comparison of Wisconsin and Virginia Small - and Medium - Sized Exporter: Aggressive and Passive Exporters". In <u>Export Management: An International Contexts</u>, edited by Czinkota, M. & G. Tesar, New York, (1982).
- Tessler, A., "Britain's Over Ambitions Exporters", <u>Marketing U.K.</u>, (Feb. 1986), PP. 67 - 74.
- Terpstra, V., <u>International Marketing</u>, 2nd edition, Illinois: The Dryden Press, (1978).
- Terpstra, V., <u>International Marketing</u>, 4th edition, Chicago: The Dryden Press, (1987).
- Terpstra, V., International Dimensions of Marketing, 2nd edition, Boston: PWS - Kent Publishing Co., (1988).
- Thach, S. & C. Axinn, "Pricing and Financing Practices of Industrial Exporting Firms", <u>International Marketing Review</u>, Vol. 8, No. 1, (1991), PP. 32 - 46.
- Thomas, M.J., <u>International Marketing Management</u>, Scranton, Pennsylvania, International Textbook Co., (1969).
- Thomas, M.H., "Changing Views of Comparative Advantage", <u>Monthly</u> <u>Review of the Federal Reserve Bank of Richmond</u>, Vol. 58, (July 1972), P. 10.
- Thomas, M. & L. Araujo, "Theories of Export behaviour: A Critical Analysis", <u>European Journal of Marketing</u>, Vol. 19, (1985), PP. 42 -52
- Thompson, I.D., Organisation in Action, New York: McGraw Hill, (1967).
- Tookey, D.A., "Factors Associated with Success in Exporting", <u>Journal</u> of <u>Management Studies</u>, Vol. 1, (March 1964), PP. 48 - 64.
- Tookey, D.A., <u>Export Marketing Decisions</u>, Penguin Books, Hamondsworth, (1975).
- Tseng, J.Y. & C.M. Yu, "Export of Industrial Goods to Europe: The Case of Large Taiwanese Firms", <u>European Journal of Marketing</u>, Vol. 25, No. 9, (1991), PP. 51 - 63.
- Tull, D. & D. Hawkins, <u>Marketing Research: Measurement and Methods</u> Fifth edition, Macmillan Publishing Co., New York, (1990).
- Turnbull, W. & P. Stanley (ed.), <u>Research in International Marketing</u>, London: Cromm Helm, (1986).
- United Nations, <u>International Trade Statistics</u>, 1984, Vol. 1, New York, (1987), P. 556.

- UNIDO, (United Nations Industrial Development Organisation), Industrial Development, Review Series, Jordan Stimulating Manufacturing and Exports, Prepared By the Regional and Country Studies Branch, (1987).
- Ursic, M. & M. Czinkota, "Impact of Export Growth Expectations on Smaller Firms", <u>International Marketing Review</u>, Vol. 1, (Winter 1983), PP. 26 - 33.
- Ursic, M. & M. Czinkota, "A Experience Curve Explanation of Export Expansion", <u>Journal of Business Research</u>, Vol. 12, No. 2, (1984), PP. 159 - 168.
- Verhoeven, W., "The Export Performance of Small and Medium Sized Enterprises in the Netherlands", <u>International Small Business</u> Journal, Vol. 6, Part 2, (1988), PP. 20 - 33.
- Vernon, R., "International Investment and International trade: U. S. Exports of Consumer Durable", <u>Quarterly Journal of Economics</u>, Vol. 83, (Feb. 1966), PP. 152 - 162.
- Vernon, R., "The Product Life Cycle Hypothesis in a New International Environment, Oxford, <u>Bulletin of Economic and Statistics</u>, Vol. 41, No. 8, (Nov. 1971), PP. 255 - 267.
- Vernon R. & L. Wells, <u>Manager in the International Economy</u>, Englewood Cliffs, NJ: Prentice - Hall, (1986).
- Walker, J., "Exploring Export Potential", <u>Management Review</u>, (April 1975), PP. 49 - 51.
- Walsh, 1., <u>International Marketing</u>, 2nd edition, Pitman Publishing Ltd, UK, (1981).
- Walters, P.G., "Export Information Sources: A Study of Their Usage and Utility", <u>International Marketing Review</u>, Vol. 1, (Winter 1983), PP. 34 - 43.
- Walters, P.G., "A Study of Planning for Export Operations", <u>International Marketing Review</u>, (Autumn 1985), PP. 74 - 81.
- Walters, P.G. & S. Samiee, "A Model For Assessing Performance in Small U.S. Exporting Firms", <u>Entrepreneurship: Theory and Practice</u>, Vol. 15, (Winter 1990), PP. 33 - 50.
- Weaver, K.M. & J. Pak, "Export Behaviour and Attitudes of Small and Medium - Sized Korean Manufacturing Firms", <u>International Small</u> <u>Business Journal</u>, Vol. 8, No. 4, (1990).
- Weekly, J.K. & E.J. Bardi, "Managerial Perceptions of Exporting Area", <u>Baylor Business Studies</u>, Vol. 6, No. 4, (1975/1976), PP. 17 -26.
- Weinrauch, J.D., "Export Marketing Activities and Problems: An Exploratory Study of Experiences and Perceptions of Business Firms in Arkaness", Ph.D dissertation, University of Arkansas, (1973).

- Weinrauch, J.D., & C.P. Rao, "The Export Marketing Mix: An Examination of Company Experiences and Perceptions.", <u>Journal of Business</u> <u>Research</u>, Vol. 2, (Oct. 1974), PP. 447 - 452.
- Wells, L.T., "A Product Life Cycle for International Trade", <u>Journal</u> of Marketing, Vol. 32, (July 1968), PP. 1 -6.
- Wells, L.T. (ed), <u>The Product Life Cycle and International Trade</u>, Boston: Harvard University Press, (1972).
- Wells, W.D. & J.N. Sheth, <u>Factor Analysis in Marketing Research</u>, McGraw - Hill, U.S., (1971).
- Welch, L. & F. Wiedersheim-Paul, "Domestic Expansion and the Internationalization Process", Working Paper, No. 2, Uppsala, University of Uppsala, (1978).
- Welch, L. & F. Wiedersheim-Paul, "Export Promotion Policy A New Approach", <u>Australian Journal of Management</u>, Vol. 4, Part 2, (1979), PP. 165 - 177.
- Welch, L. & F. Wiedersheim-Paul, "Initial Exports: A Marketing Failure?", <u>The Journal of Management Studies</u>, (October 1980) PP. 333 - 344.
- Wiedersheim-Paul, F. & J. Johanson, "The Internationalization of the Four Swedish Case Studies", <u>Journal of Marketing Studies</u>, Vol. 12, No. 3, (1978), PP. 305 - 322.
- Wiedersheim-Paul, F., H. C. Olson and L. S. Welch, "Pre Export Activity: The First Step in Internationalization", <u>Journal of</u> <u>International Business Studies</u>, Vol. 9, (Spring/ Summer 1978), PP. 47 - 58.
- Wiedersheim-Paul, F. & L. Welch, "Domestic Expansion and the Internationalization Process", Centre for International Business Studies, University of Uppsala, Sweden, (1978).
- William, A.A., <u>International Trade Theory</u>, New York: Random House, (1965).
- Wills, G., Towards an Integration of Export Marketing". In Wills, G. (ed.), <u>Explorations Marketing Thought</u>, Crosby Lockwood, (1971).
- Wind, Y., S. Douglas & H. Perlmutter, "Guidelines for Developing International Marketing Strategies", <u>Journal of Marketing</u>, (April 1973), PP. 14 - 20.
- Withey, J.J., "Differences Between Exporters and Non Exporters: Some - Hypotheses Concerning Small Manufacturing Business" <u>American Journal of Small Business</u>, Vol. 4, No. 3, Winter (1980), PP. 29 - 37.
- Wood, R.V., "Analysis of a Foreign Environment: The Importance of Foreign Environmental Decision for Exporters and Potential Exporters", Ph.D dissertation, University of Oregon, (1982).

World Bank, World Development Report, Washington, (1987).

- Yang, Y.S.," An Export Behaviour Model of Smaller Firms: Identification Potential and Latent Exporters Among Small and Medium Manufacturers", Ph.D Thesis, The University of Texas, (1988).
- Yang, Y.S., R.P. Leone & D.L. Alden, "A Market Expansion Ability Approach to Identify Potential Exporters", <u>Journal of Marketing</u>, (January 1992), PP. 81 - 96.
- Yaprak, A., "A Empirical Study of the Differences between small Exporting and Non - Exporting US Firms", <u>International Marketing</u> <u>Review</u>, (Summer 1985), PP. 72 - 83.
- Yasai-Ardekani, M., "Structural Adaptation to Environments", Academy of Management Review, Vol. 11, No. 1, (1986) PP. 9 -21.
- Young, J.P., <u>The International Economy</u>, 4th edition, New York: Ronald Press Co., (1963).